Cosmological Instabilities

Neil Barnaby

Doctor of Philosophy

Department of Physics

McGill University Montréal, Québec, Canada February, 2007

A thesis submitted to McGill University in partial fulfillment of the requirements of the degree of Doctor of Philosophy

Copyright © Neil Barnaby, 2007



Library and Archives Canada

Published Heritage Branch

395 Wellington Street Ottawa ON K1A 0N4 Canada

Bibliothèque et Archives Canada

Direction du Patrimoine de l'édition

395, rue Wellington Ottawa ON K1A 0N4 Canada

> Your file Votre référence ISBN: 978-0-494-38560-9 Our file Notre référence ISBN: 978-0-494-38560-9

NOTICE:

The author has granted a nonexclusive license allowing Library and Archives Canada to reproduce, publish, archive, preserve, conserve, communicate to the public by telecommunication or on the Internet, loan, distribute and sell theses worldwide, for commercial or noncommercial purposes, in microform, paper, electronic and/or any other formats.

The author retains copyright ownership and moral rights in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

AVIS:

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque et Archives Canada de reproduire, publier, archiver, sauvegarder, conserver, transmettre au public par télécommunication ou par l'Internet, prêter, distribuer et vendre des thèses partout dans le monde, à des fins commerciales ou autres, sur support microforme, papier, électronique et/ou autres formats.

L'auteur conserve la propriété du droit d'auteur et des droits moraux qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

In compliance with the Canadian Privacy Act some supporting forms may have been removed from this thesis.

While these forms may be included in the document page count, their removal does not represent any loss of content from the thesis.



Conformément à la loi canadienne sur la protection de la vie privée, quelques formulaires secondaires ont été enlevés de cette thèse.

Bien que ces formulaires aient inclus dans la pagination, il n'y aura aucun contenu manquant.

Abstract

Though historically the word "tachyon" has been used to describe hypothetical particles which propagate faster than the speed of light, in a more modern context the term has been recycled to refer to certain unstable states in field theory. This thesis explores the role of tachyonic instabilities in cosmology considering tachyons which arise in string theory and also more conventional, field theoretic instabilities. Our study of such instabilities is, in part, motivated by attempts to embed inflation into string theory. We will argue that the study of string theory models of inflation is well-motivated and may provide a rare potential observational window into string physics.

After reviewing the necessary background material concerning inflation, cosmological perturbation theory and tachyonic instabilities we study in detail the dynamics of the tachyonic instability which marks the end of a particular string theory model of inflation, focusing on the processes of reheating and cosmic string production. We show that the peculiar dynamics of the open string tachyon leads to various novelties in these processes and consider also potential observational consequences.

We consider tachyonic preheating at the end of hybrid inflation in a conventional field theory setting and show that the preheating process can leave an observable imprint on the Cosmic Microwave Background, either through n = 4 contamination of the power spectrum or else through large nongaussian signatures. The possibility of large nongaussianity is particularly interesting since it demonstrates that hybrid inflation provides one of the few well-motivated models which can generate an observable nongaussian signature.

Finally, we study a novel string theoretic model of inflation, *p*-adic inflation. This model is nonlocal, however, it is free of the usual problems (such as ghosts) which plague nonlocal theories. Furthermore, the nonlocal structure of the theory leads to a variety of unexpected dynamics including the possibility of a slowly rolling inflaton, despite an extremely steep potential.

Résumé

Bien qu'historiquement le mot "tachyon" fut introduit pour désigner d'hypothétiques particules se déplaçant plus rapidement que la vitesse de la lumière, de nos jours il désigne aussi certaines configurations instables en théorie des champs. Cette thèse explore le rôle des instabilités tachyoniques en cosmologie, particulièrement celui des tachyons apparaissant en théorie des cordes, mais aussi celui de tachyons plus conventionnels de la théorie des champs. Notre étude de ces instabilités est, en partie, motivée par une tentative d'inscrire l'inflation à l'intérieur de la théorie des cordes. Nous démontrerons que l'étude de modèles d'inflation en théorie de cordes est bien justifiée, et pourrait constituer une fenêtre d'accès privilégiée pour l'observation expérimentale de la théorie des cordes.

Après avoir revu les notions nécessaire en matière d'inflation, de théorie cosmologique des perturbations ainsi que d'instabilités tachyoniques, nous examinons en détails la dynamique de l'instabilité tachyonique qui marque la fin d'un modèle particulier d'inflation en théorie des cordes, en mettant l'accent sur les processus de réchauffement de la matière et de production de cordes cosmiques. Nous montrons que les particularités de la dynamique du tachyon de la corde ouverte apportent des éléments nouveaux à ces processus, et discutons des conséquences possibles pour les observations.

Dans le cadre conventionnel de la théorie des champs, nous étudions le préréchauffement par le tachyon qui survient à la fin de l'inflation hybride et montrons que ce processus peut laisser des empreintes sur le fond diffus cosmologique (CMB), soit par une contamination n = 4 du spectre de puissance, soit par une importante signature non-gaussienne. La possibilité d'une importante non-gaussianité est particulièrement intéressante, puisque parmi les modèles d'inflation qui soit bien motivés l'inflation hybride est l'un des rares possédant une telle signature.

Finalement, nous considérons un nouveau modèle d'inflation s'inscrivant à l'intérieur de la théorie des cordes: l'inflation *p*-adique. Il s'agit d'un modèle non-local, mais néanmoins exempt des difficultés affectant habituellement ce genre de théories (tels les fantômes). Cette non-localité confère au modèle des propriétés remarquables, notamment la possibilité d'une inflation en roulement lent malgré un potentiel extrêmement abrupt.

Acknowledgements

It is a pleasure to thank A. Berndsen, T. Biswas, C. Burgess, X. Chen and H. Stoica for collaborations. I also thank R. Brandenberger, H. Lam and G. Moore for instruction as professors. Thanks to J. J. Blanco-Pillado, R. Brandenberger, C. Byrnes, G. Calcagni, K. Dasgupta, H. Firouzjahi, J. Garcia-Bellido, D. Ghoshal, M. Laidlaw, L. Leblond, J. Lidsey, A. Linde, D. Lyth, K. Malik, G. Moore, A. Notari, G. Rigopolous, A. Riotto, M. Sakellariadou, D. Seery, P. Shellard, H. Stoica, B. van Tent, H. Tye and F. Vernizzi for helpful discussions, comments and suggestions during the course of the work which resulted in this thesis. It is also, of course, a great pleasure to thank J. Cline, my thesis advisor, who has also functioned in all of the above roles during the last five years in addition to providing support, encouragement, helpful feedback and advice.

Thanks to P. Kalyniak who inspired me to pursue graduate studies in theory.

Finally, thanks to L. Hoi for generously providing the LaTeX template for this thesis and to S. Caron-Huot for helping with the French translation of the abstract.

Statement of Original Contributions

This thesis contains material previously published in [1]-[7].

Chaper 2 is based on [1] which was written in collaboration with J. Cline. I performed most of the calculation and wrote a large fraction of the body of the text.

Chapter 3 is based on [2] which was a single author paper and hence I have performed all calculations and wrote the entirety of the text.

Chapter 4 is based on [3] which was written in collaboration with C. Burgess and J. Cline. I studied the mode functions and helped to set up the two throat model described in section 2. I independently verified the calculations in the paper to ensure accuracy.

Chapter 5 is based on [4] which was written in collaboration with A. Berndsen, J. Cline and H. Stoica. I performed all the analytical calculations of defect formation in sections 3 and 4, and studied the comological consequences in collaboration with my co-authors in section 6. I also wrote a large fraction of the body of the text and contributed most of the appendix.

Chapter 6 is based on [5] and [6], both written in collaboration with J. Cline. In [5] I performed all of the cosmological perturbation theory calculations, both at first and second order, including the construction of the tachyon curvature perturbation and the construction of the Green function for the master equation. I performed some (though not all) of the calculations involving the tachyon mode functions section III. I independently verified both the numerical and analytical analysis in sections V and VI and performed the reduction of the KKLMMT model in section VII. I contributed appendices A-E and wrote a large fraction of the body of the text. In [6] I independently verified the adiabatic approximation computation in section III and derived the result demonstrating that it is possible to obtain large nongaussianity without saturating the observational limit on the linearity parameter in this case. I performed all of the calculations in section VI (second order cosmological perturbation theory for a multi-component tachyon). I contributed both appendices. I also wrote

a large fraction of the body of the text.

Chapter 7 is based on [7] written in collaboration with T. Biswas and J. Cline. I contributed section 3 and was responsible for the initial observation that the inflaton can roll slowly despite the steepness of the potential. I independently verified all of the calculations in subsections 4.1, 4.2 and contributed subsection 4.3 on the friction-dominated approximation. I constructed the mode function solutions and the curvature perturbation in section 5 and contributed much of the discussion of initial conditions. I contributed all of the appendices and wrote a large fraction of the body of the text.

Contents

1	Intr	Introduction			
	1.1	1 Modern Cosmology			
		1.1.1	Inflation	4	
		1.1.2	Cosmological Perturbation Theory	7	
		1.1.3	Gaussianity of the Perturbations	17	
		1.1.4	Inflation from String Theory	18	
	1.2	Tachy	onic Instabilities	22	
		1.2.1	Tachyons in Field Theory	23	
		1.2.2	Hybrid Inflation and Tachyonic Preheating	25	
		1.2.3	Tachyons in String Theory	28	
2	\mathbf{Cre}	ating	the Universe from Brane-Antibrane Annihilation	33	
	2.1	Introd	luction	34	
	2.2	Dynai	mical Tachyon Kink Formation on Unstable Dp-branes \ldots	37	
		2.2.1	Effective Field Theory and Equations of Motion	37	
		2.2.2	Solutions Near the Core of the Defect	38	
		2.2.3	Vacuum Solutions	39	
		2.2.4	Stress-Energy Tensor	40	
	2.3	Effect	ive Tachyon Field Theory on the Brane-Antibrane Pair	41	
	2.4	Vorte	x Solutions on the Brane-Antibrane Pair	43	
		2.4.1	Solutions Near the Core of the Defect	44	
		2.4.2	Solutions Away From the Core of the Defect	46	

		2.5.1	Normalization of the Potential	49
		2.5.2	Stress-Energy Tensor at $r = 0$	50
		2.5.3	Stress-Energy Tensor at $r > 0$	51
		2.5.4	Conservation of Energy	51
	2.6	Inclus	ion of Massless Gauge Fields	52
	2.7	Simpli	fied Model of Reheating	56
		2.7.1	Gauge Field Solutions	57
		2.7.2	Spectra of Produced Particles	60
		2.7.3	Energy Density	63
		2.7.4	Efficiency of Reheating	66
	2.8	Conclu	usions	69
3	Cau	istic Fo	ormation in Tachyon Effective Field Theories	74
	3.1	Introd	uction	75
	3.2	First 1	Derivative Tachyon Effective Actions and the Eikonal Equation	78
		3.2.1	The Tachyon Dirac-Born-Infeld Action	78
		3.2.2	The Boundary String Field Theory Action	80
		3.2.3	The Lambert and Sachs Action	82
	3.3	Caust	ic Formation	85
	3.4	Vacuu	m Dynamics in a Higher Derivative Action	86
	3.5	Inhom	nogeneous Solutions in <i>p</i> -adic String Theory	88
	3.6	Concl	usions	98
4	Wa	rped F	Reheating in Brane-Antibrane Inflation	100
	4.1	Introd	luction	101
	4.2	Tale o	f Two Throats	104
	4.3	Brane	-Antibrane Annihilation	110
	4.4	Tunne	ling	115
	4.5	Warpe	ed Reheating	117
	4.6	Concl	usions	119

5	Ove	erprod	uction of Cosmic Superstrings	121
	5.1	Introd	luction	122
	5.2	Brane	Inflation	124
	5.3	Tachy	on Kink Formation in Compact Spaces	127
		5.3.1	Action and Equation of Motion	127
		5.3.2	Lattice Simulation of Kink Formation	128
		5.3.3	Analytical Study of Tachyon Kink Formation	130
	5.4	Tachy	on Vortex Formation in Compact Spaces	132
		5.4.1	Action and Equations of Motion	133
		5.4.2	Lattice Simulations of Vortex Formation	134
		5.4.3	Analytical Study of Vortex Formation	136
		5.4.4	Alternative Complex Tachyon Action	137
	5.5	Comp	parison with Ordinary Cosmic String Formation	139
	5.6	Cosm	ological Consequences	141
		5.6.1	String defects	141
		5.6.2	Domain Walls	145
		5.6.3	Full String Network Simulations	147
		5.6.4	Monopoles	150
	5.7	Summ	nary and Conclusions	151
0	ЪŢ			1 50
6	INO1	igauss	ianity from Tachyonic Preheating in Hybrid Inflation	153
	6.1	Introc		154
	6.2	Metri	c and Matter Perturbations	157
	6.3	Hybri	d Inflation	159
		6.3.1	Background Dynamics	159
		6.3.2	Conditions for a slowly varying tachyon mass	161
		6.3.3	Tachyonic Instability	163
	6.4	Secon	d Order Fluctuations in the Long Wavelength Approximation .	169
		6.4.1	The Master Equation	169
		6.4.2	The Green Function	170

		6.4.3	The Tachyon Source	171
		6.4.4	The Gauge Invariant Curvature Perturbation	173
		6.4.5	The Tachyon Curvature Perturbation	175
		6.4.6	Time-integrated tachyon perturbation	177
	6.5	Bispec	etrum and spectrum of second order metric perturbation	178
		6.5.1	The tachyon contribution to the bispectrum	178
		6.5.2	The tachyon contribution to spectrum	181
	6.6	Constr	raints on hybrid inflation parameter space	183
		6.6.1	The issue of scale invariance	184
		6.6.2	Nonscale-invariant case	185
		6.6.3	Scale-invariant case; the adiabatic approximation $\ldots \ldots \ldots$	186
	6.7	Implic	ations for Brane-Antibrane Inflation	188
	6.8	Implic	ations for P-term Inflation	192
		6.8.1	D-term Inflation	193
		6.8.2	F-term Inflation	194
	6.9	The C	ase of a Complex Tachyon	195
		6.9.1	Cosmological Perturbation Theory for an ${\cal O}(M)$ Multiplet $\ .$.	195
		6.9.2	Complex Tachyon Mode Functions	199
		6.9.3	The End of Symmetry Breaking	200
		6.9.4	Tachyon Curvature Perturbation	200
	6.10	Conclu	usions	201
7	<i>p</i> -ad	ic Infl	ation	203
·	7.1	Introd	uction	204
	7.2	Review	v of <i>n</i> -adic string theory	207
	7.3	Absen	ce of Naive Slow Roll Dynamics	210
	7.4	Appro	ximate Solutions: Analytical techniques	211
		7.4.1	<i>p</i> -adic Scalar Field Evolution	212
		7.4.2	The Stress Energy Tensor and the Friedmann Equation	- 214
		7.4.3	The Friction-Dominated Approximation	216
				-10

	7.5	Fluctu	ations and Inflationary Predictions	221
		7.5.1	<i>p</i> -adic Tachyon Fluctuations	222
		7.5.2	Determining Parameters	225
		7.5.3	Curvature Perturbation and COBE normalization	227
		7.5.4	Comments on Slow Roll and the Relation to Local Field Theory	[,] 230
		7.5.5	Tensor Modes	231
		7.5.6	Comments on Initial Conditions	232
	7.6	Conclu	isions	234
8	Con	clusio	ns and Discussion	237
\mathbf{A}	App	oendix	to Chapter 3	2 40
	A-1	The L	egendre Transformation	241
в	App	oendix	to Chapter 4	244
	B-1	The 4	D View	245
\mathbf{C}	App	oendix	to Chapter 5	248
	C-1	Initial	Conditions for Defect Formation	249
D	App	pendice	es to Chapter 6	251
	D-1	The M	fatching Time N_k	252
	D-2	Pertur	bed Einstein Equations and the Master Equation	253
	D-3	The Ir	iflaton Contribution to $\zeta^{(2)}$	257
	D-4	Fourie	r Transforms, Mode Functions and Inverse Laplacians	261
	D-5	Constr	ruction of The Tachyon Curvature Perturbation	263
	D-6	An Ide	entity Concerning γ_{σ}	266
\mathbf{E}	Арр	oendice	es to Chapter 7	268
	E-1	The S	tress Energy Tensor and the Friedmann Equation	269
	E-2	Incom	plete Cylindrical Functions of the Sonine-Schaefli Form	271
	E-3	Comp	arison to Local Field Theory	272

List of Tables

List of Figures

2.1	Numerical solution for $p(t)$ and $q(t)$ of (2.24-2.25) showing the finite-	
	time slope divergence. $1/p$ and $1/q$ are also shown, demonstrating the	
	linearity of these functions near the critical time. \ldots \ldots \ldots \ldots	45
2.2	$N^{mn}_{M>0}(k)$ versus k for increasing values of the radial quantum number	
	n showing decreased production of heavier modes	62
2.3	Same as fig. 2.2, but varying the angular quantum number m	63
2.4	$N_{M=0}(k)$ versus k for different values of ϵ . $R_0 = 10 l_s$ and $\eta = 1$	64
2.5	$N_{M=0}(k)$ versus k for different values of R_0 . $\epsilon = l_s$ and $\eta = 1$	64
2.6	$N_{M=0}(k)$ versus k for different values of η . $R_0 = 10 l_s$ and $\epsilon = l_s$	65
2.7	$ ho$ versus compactification radius R_0 for different values of $\epsilon = \Lambda^{-1}$	
	with $\eta = 1$. Dotted line is the critical density for efficient reheating ρ_c ,	
	(2.67)	67
2.8	Same as figure 2.7 but for different values of vortex collapse rate η	
	and with $\epsilon = \Lambda^{-1} = 0.7 l_s$.	67
2.9	Warped compactification with branes localized in throats on opposite	
	sides of a stack of branes.	72
3.1	Plot of the potential $V(\phi)$ for even n	89
3.2	Plot of the potential $V(\phi)$ for odd n	89
२.२ २.२	The contours of integration C_1 and C_2	03
0.0	The contours of integration C_1 and C_2	50
3.4	The contour of integration C	94

 The initial data is such that the linearized approximation is valit throughout the evolution. 3.6 Plot of the spatial profiles δφ/φ₀ for a series of increasing time steps. The initial data is such that the linearized approximation breaks dow at late times. 4.1 A Type IIB vacuum with a mildly warped inflationary throat and strongly warped Standard Model throat. This diagram suppresses an image throats arising due to any orientifolds which appear in the compactification. 4.2 The warp factor as function of a bulk radial coordinate in a simplified model of two asymmetric throats. As shown in the figure, th part of the internal space outside of the throats can be regarded as regularization of a 'Planck' brane of a Randall-Sundrum geometry. 4.3 Wave functions of KK gravitons on the internal space. 4.4 Unnormalized wave functions for highly excited KK gravitons with KI numbers n = 1, 100, 1000 and 20,000. 4.5 How to place two throats on an S₁ with Z₂ orbifold symmetry. 5.1 Illustration of the Kibble argument in the case of branes spanning extended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, 1/H ≫ ℓ, s there are no causally disconnected regions along this dimension. The 	96 97 106 106 109
 throughout the evolution	96 97 106 106 109
 3.6 Plot of the spatial profiles δφ/φ₀ for a series of increasing time steps. The initial data is such that the linearized approximation breaks dow at late times	97 106 109
 The initial data is such that the linearized approximation breaks dow at late times	97 106 106 109
 at late times	97 106 109
 4.1 A Type IIB vacuum with a mildly warped inflationary throat and strongly warped Standard Model throat. This diagram suppresses an image throats arising due to any orientifolds which appear in the compactification	106 106 109
 strongly warped Standard Model throat. This diagram suppresses an image throats arising due to any orientifolds which appear in the compactification. 4.2 The warp factor as function of a bulk radial coordinate in a simplified model of two asymmetric throats. As shown in the figure, the part of the internal space outside of the throats can be regarded as regularization of a 'Planck' brane of a Randall-Sundrum geometry. 4.3 Wave functions of KK gravitons on the internal space. 4.4 Unnormalized wave functions for highly excited KK gravitons with KI numbers n = 1, 100, 1000 and 20,000. 4.5 How to place two throats on an S₁ with Z₂ orbifold symmetry. 5.1 Illustration of the Kibble argument in the case of branes spanning extended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, 1/H ≫ ℓ, sthere are no causally disconnected regions along this dimension. The 	106 106 109
 image throats arising due to any orientifolds which appear in the compactification. 4.2 The warp factor as function of a bulk radial coordinate in a simple fied model of two asymmetric throats. As shown in the figure, the part of the internal space outside of the throats can be regarded as regularization of a 'Planck' brane of a Randall-Sundrum geometry. 4.3 Wave functions of KK gravitons on the internal space. 4.4 Unnormalized wave functions for highly excited KK gravitons with KI numbers n = 1, 100, 1000 and 20,000. 4.5 How to place two throats on an S₁ with Z₂ orbifold symmetry. 5.1 Illustration of the Kibble argument in the case of branes spanning extended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, 1/H ≫ ℓ, s there are no causally disconnected regions along this dimension. The 	106 106 109
 pactification	106 106 109
 4.2 The warp factor as function of a bulk radial coordinate in a simple fied model of two asymmetric throats. As shown in the figure, the part of the internal space outside of the throats can be regarded as regularization of a 'Planck' brane of a Randall-Sundrum geometry. 4.3 Wave functions of KK gravitons on the internal space	106 109
 fied model of two asymmetric throats. As shown in the figure, the part of the internal space outside of the throats can be regarded as regularization of a 'Planck' brane of a Randall-Sundrum geometry. 4.3 Wave functions of KK gravitons on the internal space	106 109
 part of the internal space outside of the throats can be regarded as regularization of a 'Planck' brane of a Randall-Sundrum geometry. 4.3 Wave functions of KK gravitons on the internal space	106 109
 regularization of a 'Planck' brane of a Randall-Sundrum geometry 4.3 Wave functions of KK gravitons on the internal space 4.4 Unnormalized wave functions for highly excited KK gravitons with KI numbers n = 1, 100, 1000 and 20,000	106 109
 4.3 Wave functions of KK gravitons on the internal space 4.4 Unnormalized wave functions for highly excited KK gravitons with KI numbers n = 1, 100, 1000 and 20,000 4.5 How to place two throats on an S₁ with Z₂ orbifold symmetry 5.1 Illustration of the Kibble argument in the case of branes spanning extended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, 1/H ≫ ℓ, st there are no causally disconnected regions along this dimension. The 	109
 4.4 Unnormalized wave functions for highly excited KK gravitons with KI numbers n = 1, 100, 1000 and 20,000	
 numbers n = 1, 100, 1000 and 20,000	
 4.5 How to place two throats on an S₁ with Z₂ orbifold symmetry 5.1 Illustration of the Kibble argument in the case of branes spanning extended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, 1/H ≫ ℓ, state there are no causally disconnected regions along this dimension. The 	109
5.1 Illustration of the Kibble argument in the case of branes spanning ex- tended dimensions and wrapping compact ones. The compact dimen- sion is much smaller in size than the Hubble distance, $1/H \gg \ell_{\parallel}$, so there are no causally disconnected regions along this dimension. The	109
tended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, $1/H \gg \ell_{\parallel}$, so there are no causally disconnected regions along this dimension. The	
sion is much smaller in size than the Hubble distance, $1/H \gg \ell_{\parallel}$, s there are no causally disconnected regions along this dimension. Th	
there are no causally disconnected regions along this dimension. Th	ı
	:
branes that will form as topological defects will wrap the same compac	
dimensions as the parent branes, and will be localized on the extende	
dimensions	. 125
5.2 Formation of tachyon kinks for $R = 8M_T^{-1}$. The time axis is measure	
in units of M_T^{-1}	. 129
5.3 Formation of tachyon kinks for $R = 15M_T^{-1}$. The time axis is measured	
in units of M_T^{-1}	

5.4	Behavior of the slope at the core of the kink for different initial condi-	
	tions. The time axis is measured in units of b	132
5.5	Typical initial configuration for numerical studies of vortex formation.	135
5.6	The final configuration of the complex tachyon field for $R = 7M_T^{-1}$.	135
5.7	The final configuration of the complex tachyon field for $R = 15 M_T^{-1}$.	135
5.8	The final configuration of the complex tachyon field for $R = 25 M_T^{-1}$.	136
5.9	Comparison of the tachyon and ϕ^4 potentials. In the case of the ϕ^4	
	theory, the finiteness of the slope of the kink, as well as large oscillations	
	of the field, strongly suppress the formation of defects.	140
5.10	The effect of damping on the formation of vortices in the ϕ^4 theory.	
	Small damping results in a large number of oscillations of the field, and	
	effective homogenization.	140
5.11	The effect of damping on the formation of vortices in the ϕ^4 theory. An	
	unphysically large damping is used in order to show that the density	
	of defects which survive is larger.	141
5.12	Fraction of critical density versus time for long strings, loops, and	
	visible radiation starting from initial values $\Omega_s = 0.4$, $\Omega_l = 0.1$ and	
	$\Omega_{rad} = 0.5$. Not shown is the fraction of energy density in gravity	
	waves produced by decay of the loops.	144
5.13	An array of codimension-two D3-branes from D5-anti-D5 annihilation,	
	partially localized in the compact dimensions, which look like strings	
	in the x_1 - y_1 plane, and domain walls in the large x_i dimensions	146
5.14	Fraction of wound string density about the three anisotropic directions	
	(two large and one small) starting from equal energy density in strings	
	wrapping each direction. Top curve is for strings winding in the small	
	direction	148
5.15	Plot of domain wall network in an anisotropic space with two large and	
	one small dimensions with $L = 80 M_s^{-1}$. This snapshot is taken late in	
	the evolution and shows that domain walls which are localized in the	
	large dimensions are preferred by the defect evolution	149

5.16	Fraction of wound string density about the three anisotropic directions	
	(one large and two small) starting from equal energy density in strings	
	wrapping each direction.	151
6.1	Exact solution and our approximation for the function N_k in eq. (6.27).	166
6.2	Ratio of the exact mode functions to the approximation (6.28) , at large	
	times	167
6.3	Contours of $\ln M_p^2/\lambda v^2$ in the plane of N_* and $\ln c. \ldots \ldots$	168
6.4	sign(f) ln(1 + f) versus N, showing behavior of the function f	
	defined in (6.47), for $\ln c = 1$, $N_* = 22.5$	179
6.5	Log of the integrand of the p integral in eq. (6.52), for a case where	
	the maximum occurs near $\ln(p/\sqrt{c}H) = N_i \dots \dots \dots \dots$	180
6.6	Contours of $\ln \tilde{B} $, defined in (6.54), in the plane of N_* and $\ln c$. $N_i =$	
	-30 in this example	181
6.7	Contours of $\ln \tilde{S} $ in the plane of N_* and $\ln c$, at $N_i = -30$	182
6.8	Excluded regions of the hybrid inflation parameter space, for $\log_{10} v/M_p$ =	=
	-1, -3, -5, -7, -9.	185
6.9	Excluded regions of the KKLMMT brane-antibrane inflation parame-	
	ter space, in the plane of $\log_{10} a_i$ versus $\log_{10} g_s$ for $\log_{10} a_i M_s/M_p =$	
	$-13, -11, \ldots, -5$ from nongaussianity. \ldots \ldots \ldots \ldots	191
7.1	The potential of the <i>p</i> -adic tachyon for $p = 19$	208
7.2	Plot of the energy density of the p -adic scalar as a function of t , for	
	$p = 10^5$. This figure shows that ρ_{ϕ} is approximately constant for	
	$u < p^{-1/2}$. The individual contributions from the potential and kinetic	
	energy are also shown.	221
7.3	Log of the string coupling g_s as a function of $\log_{10} p$ for several values	
	of n_s	229
7.4	Log of the ratio of the string mass scale to the Planck mass as a function	
	of $\log_{10} p$ for several values of n_s .	229

•

Chapter 1

Introduction

1.1 Modern Cosmology

Inflation [8] plays a key role in modern cosmology. Inflation has become the dominant paradigm for explaining the high degree of homogeneity and isotropy of the universe. In addition to explaining why the universe is so homogeneous, inflation also provides a mechanism for explaining the existence (and scale invariant nature) of the small perturbations which are present in the Cosmic Microwave Background (CMB). As we will see, these are generated by the quantum fluctuations of the inflaton field during inflation. The literature on this subject contains a number of excellent review papers on the theory of inflation and cosmological perturbations, for example [9, 10, 11].

We begin with a very brief review of some of the salient features of modern cosmology. Modern cosmology is based on Einstein's theory of General Relativity (GR). One of the essential assumptions of modern cosmology (confirmed by observation) is that, to a first approximation, the universe is homogeneous and isotropic. This assumption turns out to be quite restrictive and the only metric compatible with this assumption is the Friedmann-Robertson-Walker (FRW) metric¹

$$ds^{2} \equiv g_{\mu\nu}dx^{\mu}dx^{\nu} = -dt^{2} + a(t)^{2} \left[\frac{dr^{2}}{1 - Kr^{2}} + r^{2} \left(d\theta^{2} + \sin^{2}\theta d\phi^{2}\right)\right]$$
(1.1)

where $K = 0, \pm 1$ defines the geometry on spatial hypersurfaces and a(t) is the scale factor of the universe. We will typically set K = 0 which is in good agreement with observation and is motivated by the assumption of a period of primordial inflation. The stress-energy tensor for matter content which is consistent with the assumption of homogeneity and isotropy is of the form

$$T^{\mu}_{\nu} = \text{diag}\left(-\rho(t), P(t), P(t), P(t)\right)$$
(1.2)

where ρ , P are the energy density and pressure respectively. It is conventional to define the equation of state ω of the matter by

$$P = \omega \rho \tag{1.3}$$

¹We employ the "mostly plus" convention for the metric signature throughout so that $\eta_{\mu\nu} = \text{diag}(-1, +1, \dots, +1)$.

In many cases of interest ω is constant. In particular, for a gas of nonrelativistic particles (pressureless dust) one has $\omega = 0$ while for a gas of relativistic particles (radiation) one has $\omega = 1/3$. A pure cosmological constant $(T_{\mu\nu} \propto g_{\mu\nu})$ corresponds to $\omega = -1$.

The dynamics of the system is described by the Einstein equations

$$G_{\mu\nu} = \frac{1}{M_{p}^{2}} T_{\mu\nu}$$

$$R_{\mu\nu} - \frac{1}{2} R g_{\mu\nu} = \frac{1}{M_{p}^{2}} T_{\mu\nu}$$
(1.4)

as well as the conservation equations

$$\nabla^{\mu}T_{\mu\nu} = 0 \tag{1.5}$$

Throughout Greek indices $\mu, \nu = 0, 1, 2, 3, 4$ run over the full spacetime while Latin indices only run over the spatial directions i, j = 1, 2, 3. In (1.4) $G_{\mu\nu}$ is the Einstein tensor, $R_{\mu\nu}$ is the Riemann tensor, R is the Ricci scalar and $M_p = (8\pi G_N)^{-1/2} \cong$ 2.43×10^{18} GeV is the reduced Planck mass. Equations (1.4, 1.5), along with the ansatz (1.1), imply the following system

$$H^2 = \frac{\rho}{3M_n^2} - \frac{K}{a^2}$$
(1.6)

$$0 = \dot{\rho} + 3H(\rho + P)$$
 (1.7)

$$\frac{\ddot{a}}{a} = -\frac{1}{6M_p^2}(\rho + 3P) \tag{1.8}$$

In the above the dot denotes differentiation with respect to cosmic time $f = \partial_t f$ and we have defined $H \equiv \dot{a}/a$. Only two of these equations are independent, indeed (1.8) may be derived by differentiating (1.6) with respect to time and using (1.7). With (1.3) the two equations (1.6, 1.7) are sufficient to close the system.

For constant $\omega \neq -1$ and assuming K = 0 one has $\rho \propto a^{-3(1+\omega)}$ and $a(t) \propto t^{\frac{2}{3}\frac{1}{(1+\omega)}}$. For $\omega = -1$ we would have had $\rho, H = \text{const}$ and $a \propto e^{Ht}$. Notice that for both radiation and matter dominated cosmologies one has a *decelerating* universe, $\ddot{a} < 0$, while a universe dominated by a cosmological constant will *accelerate*, $\ddot{a} > 0$.

Historically, the development of inflation was motivated by several conceptual problems associated with big bang cosmology (none of which represented an actual conflict with observation). Though the big bang model enjoyed immense phenomenological success in explaining Hubble's law, the existence (and backbody nature) of the Cosmic Microwave Background (CMB) radiation and the abundance of light elements (nucleosynthesis), it nevertheless seemed to require unnaturally fine tuned initial conditions in order to explain the observed flatness and homogeneity of the universe. Perhaps the most serious conceptual problem with the simple big bang model was the difficulty in constructing a causal theory of structure formation. The resolution to these puzzles, which has now become a widely accepted part of the standard cosmological model, is to posit a phase of accelerated expansion known as *inflation* preceeding the big bang. Because the inflation is a *predictive* scenario (and these predictions are in excellent agreement with observation) we will not linger on its historical motivation and will rather focus on the predictions.

1.1.1 Inflation

How can one obtain a prolonged phase of accelerated expansion? It is clear from our previous discussion that a universe dominated by dust or by radiation will not lead to acceleration. A pure cosmological constant would do the trick; however in that case inflation would never end and there could be no smooth transition to the big bang. The most popular (and perhaps simplest) implementation of this idea is inflation driven by a single scalar field, φ , which we refer to as the *inflaton*. The action for the inflaton field is

$$S_{\varphi} = -\int d^4x \sqrt{-g} \left[\frac{1}{2} \partial_{\mu} \varphi \partial^{\mu} \varphi + V(\varphi) \right]$$
(1.9)

where $V(\varphi)$ is the potential energy of the field. For a homogeneous field configuration $\varphi = \varphi(t)$ the stress tensor associated with the inflaton takes the form (1.2) with

$$\rho = \frac{1}{2}\dot{\varphi}^2 + V(\varphi) \tag{1.10}$$

$$P = \frac{1}{2}\dot{\varphi}^2 - V(\varphi) \tag{1.11}$$

To close the system of dynamical equations describing the inflaton coupled to gravity we need only consider the Friedmann equation (1.6) and the Klein-Gordon equation

1.1 Modern Cosmology

for the inflaton field

$$H^{2} = \frac{1}{3M_{p}^{2}} \left[\frac{1}{2} \dot{\varphi}^{2} + V(\varphi) \right]$$
(1.12)

$$0 = \ddot{\varphi} + 3H\dot{\varphi} + V'(\varphi) \tag{1.13}$$

where $V'(\varphi) = \partial V/\partial \varphi$. Equation (1.13) can be derived by minimizing (1.9) or equivalently by inserting (1.10, 1.11) into (1.7). If we would like to use the inflaton field to drive a period of inflation we want $\rho \cong -P$ so that the field (temporarily) mimics a cosmological constant. From (1.10, 1.11) it is clear that this requires $\dot{\varphi}^2/2 \ll V$. This will ensure that $H \cong$ const, however, if $\ddot{\varphi}$ is too large large then this inflationary phase will not last very long. In order to ensure a long period of inflation we must also therefore demand that $\ddot{\varphi} \ll H\dot{\varphi}$. It is conventional to quantify this by defining the *slow roll parameters*. There are two logically distinct definitions of slow roll parameters which appear in the literature. The first definition, the Hubble slow roll parameters, is in terms of the dynamics of the fields

$$\epsilon_H \equiv \frac{1}{2M_p^2} \frac{\dot{\varphi}^2}{H^2} = -\frac{\dot{H}}{H^2}$$
 (1.14)

$$\epsilon_H - \eta_H \equiv \frac{\ddot{\varphi}}{H\dot{\varphi}} \tag{1.15}$$

In the second equality in (1.14) we have used the fact that $\dot{H} = -(\rho + P)/(2M_p^2) = -\dot{\varphi}^2/(2M_p^2)$ and thus $\epsilon = -\dot{H}/H^2$, which follows from (1.6) and (1.8). It is clear that if $\epsilon_H \ll 1$, $|\eta_H| \ll 1$ then the time evolution of $\varphi(t)$ and H(t) will be slow compare to the Hubble scale. An alternative definition is the potential slow roll parameters

$$\epsilon_V \equiv \frac{M_p^2}{2} \left(\frac{V'}{V}\right)^2 \tag{1.16}$$

$$\eta_V \equiv M_p^2 \frac{V''}{V} \tag{1.17}$$

so that for $\epsilon_V \ll 1$, $|\eta_V| \ll 1$ the scalar potential is flat. For a flat potential one can show that $\epsilon_H \cong \epsilon_V$, $\eta_H \cong \eta_V$ so in this case there is no need to distinguish between these two types of slow roll parameter. We will assume this to be the case throughout this chapter and hence we drop the subscipts and refer to the slow roll parameters simply as ϵ , η . In chapter 7 we will consider a model of inflation where the distinction between Hubble and potential slow roll parameters is crucial. The challenge of inflationary model building is to find potentials which are sufficiently flat but are nevertheless well motivated from a particle physics perspective. We will return to this issue in a subsequent subsection. Assuming that we have found a potential for which ϵ , $|\eta| \ll 1$, then the dynamics is described by the slow roll equations

$$0 \cong 3H\dot{\varphi} + V' \tag{1.18}$$

$$H^2 \cong \frac{1}{3M_p^2} V(\varphi) \tag{1.19}$$

Notice that the time derivatives of the slow roll parameters are second order in the slow roll expansion²

$$\begin{aligned} |\dot{\epsilon}| &= 2\epsilon |\eta - 2\epsilon| H \ll \epsilon H \\ |\dot{\eta}| &= 2\epsilon |\eta| H \ll |\eta| H \end{aligned}$$

which means that we are justified in treating ϵ, η as constant if we work only to leading order in a slow roll expansion. This fact will be quite useful when we solve the perturbation equations.

During inflation the Hubble scale H remains approximately constant so that expansion of the universe is approximately that of de Sitter space $a(t) \cong e^{Ht}$. The acceleration of the scale factor can then be written as

$$\frac{\ddot{a}}{a} = \dot{H} + H^2 = (1 - \epsilon)H^2$$

so that inflation will come to an end once $\epsilon(\varphi_f) \cong 1$ where $\varphi_f \equiv \varphi(t_f)$ is the value of the inflaton field at the end of inflation. It is common to define the total number of e-foldings of inflation between some reference time t and the end of inflation t_f as

$$N_e \equiv \int_t^{t_f} H dt \cong -\frac{1}{M_p^2} \int_{\varphi}^{\varphi_f} \frac{V}{V'} d\varphi$$
(1.20)

so that typically $N_e \cong H(t_f - t)$.

²The equality for $\dot{\eta}$ only strictly holds for V''' = 0 though this contribution will in general be small.

1.1.2 Cosmological Perturbation Theory

In addition to explaining why the universe is so homogeneous and isotropic, inflation also provides a mechanism for generating the small inhomogeneities which lead to the temperature fluctuations of the CMB. In this subsection we will show how quantum fluctuations during the inflationary epoch lead to a nearly scale-invariant large scale power spectrum.

Quantum Fields in de Sitter Space

During inflation the expansion of the universe is approximately that of pure de Sitter space. Noting that the η slow roll parameter can be written as

$$\eta = \frac{m^2}{3H^2}$$

the slow roll condition $|\eta| \ll 1$ therefore implies that the inflaton field must be light compared to the Hubble scale $m \ll H$. Hence the inflaton field fluctuation is expected to behave very much like a light quantum field in de Sitter space.

As a warm-up exercise for the full calculation we therefore consider the dynamics of some quantum field, χ , (not necessarily the inflaton) in pure de Sitter space (see, for example, [12]). For the sake of simplicity we assume a potential of the form $V = m_{\chi}^2 \chi^2/2$. The χ -field is expanded in terms of annihilation/creation operators as

$$\chi(t, \mathbf{x}) = \int \frac{d^3k}{(2\pi)^{3/2}} \left[a_{\mathbf{k}} \chi_{\mathbf{k}}(t) e^{i\mathbf{k}\cdot\mathbf{x}} + \text{h.c.} \right]$$
(1.21)

where h.c. denotes the Hermitian conjugate of the preceeding term. The annihilation/creation operators satisfy

$$\left[a_{\mathbf{k}}, a_{\mathbf{k}'}^{\dagger}\right] = \delta^{(3)} \left(\mathbf{k} - \mathbf{k}'\right)$$

The c-number valued mode functions $\chi_{\mathbf{k}}(t)$ satisfy an equation of the form

$$\ddot{\chi}_k + 3H\dot{\chi}_k + \left[\frac{k^2}{a^2} + m_{\chi}^2\right]\chi_k = 0$$
(1.22)

The calculation is simplest in terms of conformal time, τ , which is related to cosmic time as $ad\tau = dt$. For pure de Sitter space $a(\tau) = -1/(H\tau)$ and it is convenient to define the conformal time Hubble scale as

$$\mathcal{H} \equiv \frac{a'}{a} = -\frac{1}{\tau}$$

In terms of conformal time (1.22) becomes

$$\chi_k'' + 2\mathcal{H}\chi_k' + \left[k^2 + a^2 m_\chi^2\right]\chi_k = 0$$

and, defining $V_k = a\chi_k$, this equation simplifies to

$$V_k'' + \left[k^2 + \frac{1}{\tau^2} \left(\frac{m_\chi^2}{H^2} - 2\right)\right] V_k = 0.$$
 (1.23)

This is the variable in terms of which the action is canonically normalized

$$S \cong -\frac{1}{2} \int d\tau d^3x \left[-(V')^2 + \partial_i V \partial^i V + m_{\text{eff}}^2(\tau) V^2 \right]$$

so that the field V behaves like a quantum field in Minkowski space with a time varying mass, $m_{\text{eff}}(\tau)$, which encodes information about the expansion of the universe. The general solutions to (1.23) are

$$V_k(\tau) = \sqrt{-\tau} \left[c_1(k) H_{\nu}^{(1)}(-k\tau) + c_2(k) H_{\nu}^{(2)}(-k\tau) \right]$$
(1.24)

where the order of the Hankel functions is $\nu = \sqrt{9/4 - m_{\chi}^2/H^2}$. This analysis makes no assumptions (yet) about the size of m_{χ}/H and all the formulae we derive are valid for arbitrary complex ν unless otherwise stated.³ Notice that $-\tau \ge 0$ for all t so that the arguments of the Hankel functions are always positive definite.

In the solution (1.24) there are two unknown coefficients $c_i(k)$, i = 1, 2 corresponding to the two initial data required to uniquely specify the solution of a second order differential equation. How should one fix the initial data? Inspection of (1.21) suggests that different choices of normalization for the mode functions χ_k will lead to alternative definitions of the annihilation/creation operators. Since the vacuum is defined by $a_{\mathbf{k}}|0\rangle = 0$ it follows that different initial data correspond to different

³Of course, for χ corresponding to the inflaton perturbation we are interested in $m_{\chi} \ll H$. However, for the time being we leave m_{χ} unspecified. The behaviour of heavy fields in de Sitter space will be relevant for the analysis of chapter 6.

vacuum choices. This is a standard issue is curved space quantum field theory: there is no unique prescription for how to choose the vacuum state.⁴ There are many interesting physical consequences associated with this ambiguity, however, a complete discussion would be out of place here. (Indeed, the above discussion can be put on rigorous footing using the tools of curved space quantum field theory.) For our purposes it is sufficient to note that on small scales $k/(aH) = -k\tau \gg 1$ de Sitter space is indistinguishable from Minkowski space. Thus, a natural presciption for normalizing the modes χ_k will one be which reproduces the well known Minkowski space result $V_k = e^{-ik\tau}/\sqrt{2k}$ on small scales. This choice is referred to as the Bunch-Davies vacuum prescription. Thus, we fix $c_i(k)$ by demanding that

$$V_k(\tau) \approx e^{-ik\tau} / \sqrt{2k} \tag{1.25}$$

for $-k\tau \gg 1$. The solution (1.24) reproduces the desired asymptotics (1.25) with the choice

$$c_1(k) = \frac{\sqrt{\pi}}{2} \exp\left[\frac{i\pi}{2}\left(\nu + \frac{1}{2}\right)\right], \quad c_2(k) = 0.$$

The solution for the Mukhanov variable V_k , for all k, becomes

$$V_k(\tau) = \frac{\sqrt{-\pi\tau}}{2} \exp\left[\frac{i\pi}{2}\left(\nu + \frac{1}{2}\right)\right] H_{\nu}^{(1)}(-k\tau)$$
(1.26)

In terms of cosmic time and the original field variable (1.26) becomes

$$\chi_k(t) = \frac{1}{2} \sqrt{\frac{\pi}{a^3 H}} \exp\left[\frac{i\pi}{2} \left(\nu + \frac{1}{2}\right)\right] H_{\nu}^{(1)}\left(\frac{k}{aH}\right).$$
(1.27)

Ultimately we will be interested in the large scale behaviour of this solution. Taking the limit $-k\tau \rightarrow 0$ of (1.26) gives

$$V_{k}(\tau) \rightarrow \frac{\sqrt{-\pi\tau}}{2} \exp\left[\frac{i\pi}{2}\left(\nu + \frac{1}{2}\right)\right]$$

$$\times \left[\frac{1}{\Gamma(\nu+1)}\left(-\frac{k\tau}{2}\right)^{\nu} - i\frac{\Gamma(\nu)}{\pi}\left(-\frac{k\tau}{2}\right)^{-\nu}\right]$$
(1.28)

⁴Indeed, this ambiguity is already present in flat-space quantum field theory. However, since Minkowski space enjoys a high degree of symmetry, once the zero-particle state has been defined then all inertial observers will agree on this vacuum.

To simplify further we should focus on either $m_{\chi}/H > 3/2$ or $m_{\chi}/H < 3/2$.

We first consider a field which is heavy compared to the Hubble scale $m_{\chi} \gg H$ (not the inflaton) so that $\nu \approx i m_{\chi}/H$ is pure imaginary. How do the two terms in the square braces of (1.28) compare? The functions $(-k\tau)^{\pm i m_{\chi}/H}$ are oscillatory so the relative magnitude of these two terms depends only on the gamma function prefactors. Using the results (for β real and arbitrary complex z)

$$\Gamma(1+z) = z\Gamma(z), \quad |\Gamma(1+i\beta)| = \sqrt{\frac{\pi\beta}{\sinh(\pi\beta)}}$$

from the theory of gamma functions we find that for $m_\chi/H\gg 1$

$$\left| \frac{1}{\Gamma(1 + im_{\chi}/H)} \right| \sim \sqrt{\frac{H}{2\pi m_{\chi}}} \exp\left(\frac{\pi m_{\chi}}{2H}\right),$$
$$|\Gamma(im_{\chi}/H)| \sim \sqrt{\frac{2\pi H}{m_{\chi}}} \exp\left(-\frac{\pi m_{\chi}}{2H}\right)$$

so that the first term in the square braces on the second line of (1.28) dominates at large m_{χ}/H . Going back to the original variable we find

$$|\chi_k| \sim a^{-3/2} \frac{1}{2^{3/2} \sqrt{m_{\chi}}} \quad \text{for } m_{\chi} \gg H$$
 (1.29)

on large scales $-k\tau \ll 1$.

We consider now a field (such as the inflaton) which is light compared to the Hubble scale $m_{\chi} \ll H$. In this case $\nu \approx 3/2$ is pure real and the first term in the square braces in (1.28) goes to zero and we have

$$|\chi_k| \sim \frac{H}{\sqrt{2k^3}} \quad \text{for } m_\chi \ll H$$
 (1.30)

on large scales $-k\tau \ll 1$.

Notice that for any value of m_{χ}/H we have, by construction, the asymptotics

$$|\chi_k| \sim a^{-1} \frac{1}{\sqrt{2k}}$$
 (1.31)

on small scales $-k\tau \gg 1$.

We now define the power spectrum of the field in terms of the two-point function as

$$\langle 0|\chi^2(t,\mathbf{x})|0\rangle = \int \frac{d^3k}{(2\pi)^3} |\chi_k(t)|^2$$

=
$$\int \frac{dk}{k} \mathcal{P}_{\chi}(k)$$
 (1.32)

where

$$\mathcal{P}_{\chi}(k) = \frac{k^3}{2\pi^2} |\chi_k(t)|^2 \cong \frac{H^2}{(2\pi)^2} \left(\frac{k}{aH}\right)^{n-1}$$
(1.33)

The quantity n in (1.33) is the spectral index. The spectrum is called scale-invariant if n = 1, blue-tilted if n > 1 and red-tilted if n < 1.

From equation (1.29) we see that any heavy field in de Sitter space will have a large-scale spectrum with a spectral index n = 4 and which is damped as $a(t)^{-3/2} = e^{-3Ht/2}$. On the other hand, from equation (1.30) we see that any light field in de Sitter space will have a large-scale spectrum which is almost scale invariant

$$n = 1 + 3 - 2\nu \cong 1 + \mathcal{O}\left(\frac{m^2}{H^2}\right) \tag{1.34}$$

and which suffers no exponential damping. We expect, then, that inflation should generate a nearly scale invariant spectrum of perturbations on large scales. In the next subsubsection we will confirm this intuition.

The Full Calculation

The previous calculation shed some light on the dynamics of quantum fields during inflation, however, this calculation neglected departures from pure de Sitter expansion. Furthermore, this calculation neglected the possibility of metric perturbations. Because Einstein's equations (1.4) couple the geometry to the matter, it is clear that quantum fluctuations in the inflaton $\delta\varphi$ will induce fluctuations of the geometry $\delta g_{\mu\nu}$. A consistent calculation must include both.

We expand the inflaton field in perturbation theory as

$$\varphi(\tau, \mathbf{x}) = \varphi_0(\tau) + \delta\varphi(\tau, \mathbf{x}) \tag{1.35}$$

The most general expansion of the metric is

$$g_{00} = -a(\tau)^2 \left[1 + 2\phi\right] \tag{1.36}$$

$$g_{0i} = a(\tau)^2 \left[\partial_i \omega + \omega_i\right] \tag{1.37}$$

$$g_{ij} = a(\tau)^2 \left[(1 - 2\psi) \delta_{ij} + D_{ij} \chi + (\partial_i \chi_j + \partial_j \chi_i + \chi_{ij}) \right]$$
(1.38)

where $D_{ij} = \partial_i \partial_j - \frac{1}{3} \delta_{ij} \partial^k \partial_k$ is a trace-free operator. It is conventional to decompose the metric perturbations according to their behaviour under three-dimensional rotations. The fluctuations are therefore decomposed such that the vector perturbations are transverse $\partial^i \omega_i = \partial^i \chi_i = 0$ while the tensor perturbations are transverse, traceless and symmetric: $\partial^i \chi_{ij} = 0$, $\chi_i^i = 0$, $\chi_{ij} = \chi_{ji}$. This scalar/vector/tensor decomposition is an inherently nonlocal procedure, however, it is both conventional and convenient.⁵ Clearly at linear order the scalar, vector and tensor modes will decouple and we can study each type of perturbation independently. Our calculation will focus on the scalar perturbations. It can be shown that vector perturbations decay in an expanding universe while tensor perturbations are suppressed with respect to the scalar perturbations by slow roll parameters.

There is one further subtlety to this perturbative expansion. As we have mentioned, quantum fluctations in the matter sector $\delta \varphi$ will induce metric perturbations $\delta g_{\mu\nu}$. However, it is easy to convince oneself that metric perturbations can also be induced by performing coordinate transformations of the form

$$x^{\mu} \to x^{\mu} + a^{\mu} \tag{1.39}$$

where we assume that a^{μ} is a small perturbation. Clearly fluctuations associated with (1.39) have no physical meaning. This problem is completely analogous to what occurs in ordinary electromagnetism: the Maxwell equations $\partial_{\mu}F^{\mu\nu} = 0$ are insufficient to completely determine the vector potential A_{μ} . The ambiguity is related

⁵At linear order and focusing on inflationary perturbations this nonlocality is not manifest, however, if one considers second order perturbations or perturbations associated with field configurations having compact support (such as cosmic strings) then this nonlocality becomes quite evident [13].

to the freedom to perform gauge transformations in electromagnetism $A_{\mu} \rightarrow A_{\mu} + \partial_{\mu} a$. Of course, this ambiguity does not appear in the *physical* quantities; for example the field strength tensor $F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}$ is invariant under $A_{\mu} \rightarrow A_{\mu} + \partial_{\mu}a$. Similarly, the freedom to perform coordinate transformations (1.39) represents a gauge freedom in cosmological perturbation theory. Just as in other gauge theories, there are two ways to proceed:

- 1. Rewrite the system in terms of quantities which are gauge invariant and work only with those quantities.
- 2. Fix the gauge by placing some constraints on the metric perturbations. One must still be careful to ensure that whatever physical quantities are computed in the end are independent of the gauge fixing.

We will adopt the gauge-fixed approach which is typically simpler. We employ the longitudinal gauge, defined by $\omega = \chi = 0$, so that the metric, ignoring vector and tensor modes, takes the simple form

$$ds^{2} = a(\tau)^{2} \left[-(1+2\phi)d\tau^{2} + (1-2\psi)\delta_{ij}dx^{i}dx^{j} \right]$$

The perturbed Einstein equations $\delta G^{\mu}_{\nu} = M_p^{-2} \delta T^{\mu}_{\nu}$ can be found in elsewhere (see [9] for example) and we do not reproduce them here. The $\delta G^i_j = M_p^{-2} \delta T^i_j$ Einstein equation for $i \neq j$ implies that $\phi = \psi$ which is a well known result. We apply this simplification in the following. The $\delta G^0_i = M_p^{-2} \delta T^0_i$ Einstein equation is a constraint

$$\phi' + \mathcal{H}\phi = \frac{1}{2M_p^2}\varphi_0'\delta\varphi \tag{1.40}$$

which means that the first order metric perturbation ϕ and the first order inflaton perturbation $\delta\varphi$ are not independent. Once either ϕ or $\delta\varphi$ is known, the other may be computed from (1.40), though it is simplest to solve for ϕ and use (1.40) to compute $\delta\varphi$. One obtains a dynamical equation for the metric perturbation by applying (1.40) to the sum of $\delta G_0^0 = M_p^{-2} \delta T_0^0$ and $\delta G_i^i = M_p^{-2} \delta T_i^i$ Einstein equations. The result is

$$\phi_k'' + 2\left(\mathcal{H} - \frac{\varphi_0''}{\varphi_0'}\right)\phi_k' + \left[2\left(\mathcal{H}' - \mathcal{H}\frac{\varphi_0''}{\varphi_0'}\right) + k^2\right]\phi_k = 0 \tag{1.41}$$

Notice that the perturbed Klein-Gordon equation for the inflaton is not needed to close the system of equations. We now discuss the slow roll solutions of (1.41).

The conformal time slow roll parameters are

$$\epsilon = 1 - \frac{\mathcal{H}'}{\mathcal{H}^2} = \frac{\kappa^2 \, \varphi_0'^2}{2 \, \mathcal{H}^2},\tag{1.42}$$

$$\epsilon - \eta = \frac{\varphi_0''}{\mathcal{H}\varphi_0'} - 1. \tag{1.43}$$

During a pure deSitter phase $\epsilon = \eta = 0$ and the scale factor evolves as $a(\tau) = -1/(H\tau)$ with *H* constant. During inflation, however, the Hubble scale evolves slowly as (1.42) so that for small ϵ one has [9]

$$a(\tau) = -\frac{1}{H\tau} \frac{1}{1-\epsilon}$$

and

$$\mathcal{H} = \frac{a'}{a} = \dot{a} = aH = \frac{-1}{\tau(1-\epsilon)}.$$
(1.44)

The dynamical equation for $\phi^{(1)}$ (1.41) can be rewritten in terms of the slow roll parameters as

$$\phi_k'' - \frac{2}{\tau}(\eta - \epsilon)\phi_k' + \left[\frac{2}{\tau^2}(\eta - 2\epsilon) + k^2\right]\phi_k = 0$$
(1.45)

where we used (1.44) and dropped higher order terms in ϵ , η . Treating the slow roll parameters as constant, the equation (1.45) has an exact solution

$$\phi_k(\tau) = (-\tau)^{1/2+\eta-\epsilon} \left[c_1(k) H_{\nu}^{(1)}(-k\tau) + c_2(k) H_{\nu}^{(2)}(-k\tau) \right]$$
(1.46)

where $\nu \cong 1/2 + (3\epsilon - \eta)$ to lowest order in slow roll parameters.

It remains to fix the coefficients $c_1(k)$, $c_2(k)$ in (1.46). The variable in terms of which the action is canonically normalized is the Mukhanov variable [10]

$$V_k = a \left[\delta \varphi_k + \frac{\varphi_0'}{\mathcal{H}} \phi_k \right]$$

where the inflaton fluctuation is solved for using the constraint equation (1.40). We fix $c_1(k)$, $c_2(k)$ by requiring that $V_k \cong e^{-ik\tau}/\sqrt{2k}$ on small scales $-k\tau \gg 1$ which corresponds to the usual Bunch-Davies vacuum choice. This leads to

$$c_1(k) = i \frac{H}{M_p} \sqrt{\frac{\pi\epsilon}{2}} \frac{\exp\left[\frac{i\pi}{2}(\nu+1/2)\right]}{2k}, \quad c_2(k) = 0.$$

1.1 Modern Cosmology

The solution for the metric perturbation, then, is

$$\phi_k(\tau) = i\sqrt{\frac{\pi\epsilon}{2}} \frac{H}{M_p} \frac{\exp\left[\frac{i\pi}{2}(\nu+1/2)\right]}{2k}$$
$$\times (-\tau)^{1/2+\eta-\epsilon} H_{\nu}^{(1)}(-k\tau).$$
(1.47)

It is straightforward to construct the inflaton fluctuation using (1.40). The quantity relevant for observation is the *curvature perturbation*

$$\zeta = -\frac{\mathcal{H}}{a\varphi_0'}V = -\phi - \frac{\mathcal{H}}{\varphi_0'}\delta\varphi$$

This quantity is gauge invariant and is approximately constant on large scales (in the absence of nonadiabatic pressure). The resulting large scale power spectrum (defined as in (1.32)) is

$$\mathcal{P}_{\zeta} = A_{\zeta}^2 \left(\frac{k}{aH}\right)^{n-1} \tag{1.48}$$

where the amplitude of the spectrum is

$$A_{\zeta}^{2} = \frac{1}{2M_{p}^{2}\epsilon} \frac{H^{2}}{(2\pi)^{2}}$$
(1.49)

and the spectral index is

$$n-1 = 2\eta - 6\epsilon \tag{1.50}$$

so that $|n-1| \ll 1$. The spectrum (1.48) is meant to be evaluated when scales relevant for CMB observations exited the horizon, roughly 60 e-foldings before the end of inflation. It turns out that $N_e \cong 60$ also coincides with the minimal amount of inflation necessary to resolve the horizon problem of big bang cosmology.

Quantum fluctuations which are "born" on small scales $-k\tau \gg 1$ are redshifted by the expansion of the universe and become classical as they cross the horizon $-k\tau = 1$. The quantum to classical transition is most easily seen by writing the hamiltonian in the form $H_k = \omega_k (n_k + 1/2)$ where ω_k is the energy eigenvalue and n_k is the occupation number. It is straightforward to show that

$$n_k \sim \left(-k\tau\right)^{-3}$$

so that once $-k\tau \ll 1$ we have $n_k \gg 1$, the "quantum" factor of 1/2 in H_k can be neglected and the fluctuations may indeed be considered classical.⁶

The CMB

During inflation quantum fluctuations of the inflaton field with comoving wavenumber k are generated by the usual Minkowski space vacuum fluctuations on small scales $k \gg aH$, as in (1.31). The physical wavenumbers $k_{phys} = k/a \sim e^{-Ht}k$ are redshifted by the expansion of the universe (while H(t) remains approximately constant) until horizon crossing k = aH (or $-k\tau = 1$). Once a mode exits the horizon it becomes "frozen in" as a classical perturbation, as in (1.30).

After inflation the energy density in the inflaton field must be transferred to visible radiation in order to make contact with the successes of the big bang model. This process is called *reheating* (see [14] for a review). Immediately after reheating the universe is radiation dominated so that $a(t) \sim t^{1/2}$, $H(t) \sim t^{-1}$. However, because $\rho_{\rm rad} \sim a^{-4}$ while $\rho_{\rm dust} \sim a^{-3}$ it follows that eventually pressureless dust will come to dominate. During the matter dominated The crucial point is that after inflation physical wavenumbers red-shift as $k/a \sim kt^{-\alpha}$ with some $\alpha < 1$ while $H(t) \sim t^{-1}$. Thus modes which crossed the horizon during inflation will subsequently re-enter the horizon at some point after reheating.

As perturbation modes ζ_k re-enter the causal horizon during radiation- or matterdomination they create density perturbations $\delta \rho_k$ by the gravitational attraction of potential wells and ultimately induce the observed temperature fluctuations δT in the CMB. The amplitude of the spectrum A_{ζ} (equation (1.49)) fixes the amplitude of the temperature fluctuations $\delta T/T$ in the CMB. Consistency with observation requires $\delta T/T \sim 10^{-5}$ on large angular scales [15] which translates into

$$\frac{V}{\epsilon M_p^4} = 6 \times 10^{-7} \quad \Rightarrow \quad \left(\frac{V}{\epsilon}\right)^{1/4} = 6.7 \times 10^{16} \text{GeV}$$
(1.51)

We refer to (1.51) as the COBE normalization of the inflaton potential.

⁶Another way to see the quantum-to-classical transition is by considering the commutator $[\delta\varphi, \delta\dot{\varphi}]$. It is straightforward to show that this quantity damps to zero exponentially as $-k\tau \to 0$.

The prediction $|n - 1| \ll 1$ (equation (1.50)) is in excellent agreement with observation. Indeed, recent WMAP analysis [16] suggests that $n \cong 0.95$. Inflation also predicts a nearly scale-invariant spectrum of tensor perturbations (gravitational waves) with an amplitude that is suppressed by a factor of ϵ as compared to A_{ζ} , though these have yet to be observed.

1.1.3 Gaussianity of the Perturbations

In simple single-field models of inflation (which, so far, we have limited our discussion to describing) the perturbations are almost gaussian. Working strictly to linear order in perturbation theory we find that

$$\langle \zeta_{k_1} \zeta_{k_2} \zeta_{k_3} \rangle = 0 \langle \zeta_{k_1} \zeta_{k_2} \zeta_{k_3} \zeta_{k_4} \rangle = \langle \zeta_{k_1} \zeta_{k_2} \rangle \langle \zeta_{k_3} \zeta_{k_4} \rangle + \text{perms} \dots$$

where "perms" denotes various permutations of the momenta k_i . Similarly all higher odd *n*-point functions vanish while all even *n*-point functions can be reduced to products of two-point functions. This result is simply a consequence of Wick's theorem and the fact that ζ contains only one annihilation/creation operator. Thus, working strictly to linear order in perturbation theory, the two-point function is the only independent statistic. This is precisely the definition of *gaussian* perturbations. Since the inflationary perturbations are small (they are indeed suppressed as compared to the homogeneous background by $\delta T/T \sim 10^{-5}$) it follows that linear theory should be applicable and that the perturbations must be approximately gaussian.

However, Einstein's equations are nonlinear and one generically expects *some* nongaussianity to be generated during inflation, though one must work beyond linear order in perturbation theory in order to see this effect (for a review see [17]). From the above discussion it is clear that the three-point function (the bispectrum) is the lowest order statistic which will allow us to discriminate between gaussianity and

1.1 Modern Cosmology

nongaussianity. Defining P, B by

$$\langle \zeta_{k_1} \zeta_{k_2} \rangle = P(k) \,\delta(k_1 + k_2)$$

 $\langle \zeta_{k_1} \zeta_{k_2} \zeta_{k_3} \rangle = (2\pi)^{-3/2} B(k) \,\delta(k_1 + k_2 + k_3)$

it is conventional to define the *nonlinearity* parameter f_{NL} by

$$B(k_i) = -\frac{6}{5} f_{NL} \left[P(k_1) P(k_2) + \text{perms} \right]$$

The nonlinearity parameter provides a dimensionless measure of the nongaussianity in a particular model of inflation.

We will return to the issue of calculating f_{NL} using second order cosmological perturbation theory in chapter 6. For the time being we simply cite the result of a full calculation. In a simple, single field inflationary model (such as we have discussed in the previous subsection) one finds that [18]

$$f_{NL} \sim \frac{5}{12}(n-1)$$

so that $|f_{NL}| \ll 1$. On the other hand, observation provides the rather weak constraint $|f_{NL}| \lesssim 100$ [16]. Even though the simplest models predict negligible nongaussianity there is still considerable interest in computing f_{NL} in various scenarios because this quantity is extremely model dependent and can (at least in principle) be used as a powerful tool to discriminate between various realizations of inflation. Furthermore, an observation of $|f_{NL}| \gtrsim \mathcal{O}(10)$ would be strongly indicative of some novelty in the dynamics during inflation or shortly afterwards.

1.1.4 Inflation from String Theory

Despite the great phenomenological success of inflation, it remains a paradigm in search of a theory. There are a number of unresolved problems concerning this scenario including:

1. What is the inflaton? The above construction gives no information about how the field φ should be interpreted from a particle physics perspective.
1.1 Modern Cosmology

- 2. How does reheating after inflation take place? A complete understanding of reheating after inflation will require that we understand how the inflaton field couples to standard model degrees of freedom. We will discuss this issue in much greater detail in chapters 2 and 4.
- 3. The issue of fine tunings. The requirements ϵ , $|\eta| \ll 1$ and (1.51) seem to require considerable fine tuning of the inflaton potential. For example, the *chaotic* inflation model [19] with $V = m^2 \varphi^2/2$ requires $\varphi \gg M_p$ in order to achieve slow roll. At such large field values one might expect nonrenormalizable terms in the potential to become important and generically such corrections would spoil the flatness of the potential. Furthermore, the COBE normalization in this model demands $m \sim 10^{13}$ GeV which means that we have introduced a hierarchy into the particle physics model building. It remains an outstanding problem to explain convincingly why the inflaton potential should be so flat and how this flatness is protected against quantum corrections.
- 4. The trans-Planckian problem. If inflation lasts much longer than the minimal ~ 60 e-foldings then physical scales of current interest will have started out smaller than the Planck scale. However, on scales comparable to the Planck scale one expects quantum gravity effects to become important and the semi-classical analysis detailed above to be invalidated.

This list could be extended (see, for example, [10]). These, and other, outstanding problems provide a strong motivation to try to embed inflation into some complete theory of particle physics which incorporates quantum gravity. Indeed, it is difficult to imagine a convincing resolution to the above problems which does not involve identifying the inflaton as a physical degree of freedom in some ultra-violet (UV) complete theory of particle physics. To date there is only one viable candidate for such a theory: *string theory*.

The starting point of string theory is the assumption that elementary particles are not point-like but are, in fact, one-dimensional extended objects (or strings). The characteristic string length is taken to be extremely small so that these objects will appear as point-like at low energies. These strings may appear as closed loops or they may be open. In this picture the various species of particles which we observe in nature arise as distinct vibrational modes of the string. Remarkably, one of these excitations describes a massless spin-2 particle - the graviton - which mediates the gravitational interaction in the same way that the photon mediates electromagnetic interactions. Hence string theory is a quantum theory of gravity which has the potential to unify all the observed forces in nature. (In fact, despite considerable effort, string theory seems to be the only such theory.) It is well known that string theory predicts the existence of compact extra spatial dimensions (in addition to the three large spatial dimensions which we observe). The theory also predicts the existence of extended higher dimensional objects known as "branes" (short for "membrane") on which open strings can end. The difficulty of finding direct laboratory probes of string theory is well known. We will discuss below the possibility of gaining an observational window into string theory through cosmology.

Why can string theory help with the aforementioned problems of inflation? String theory provides inflationary model builders with a vast number of scalar fields, some of which may provide candidates for the inflaton. Since string theory is a unified theory of particle physics, it is in principle possible to identity how some candidate inflaton couples to standard model degrees of freedom and hence it should be possible to address the reheating issue quantitatively. Furthermore, it seems to be possible to generate large hierarchies in string theory in a natural way by taking advantage of the properties of warped compactifications [20, 21] so there may be hope to resolve the naturalness issues which plague inflation. Finally, string theory is a complete theory of quantum gravity, hence it should be able to describe dynamics all the way down to the Planck scale and it is at least in principle possible to resolve the trans-Planckian problem.

We have argued that the search for inflation in string theory is of practical interest for a cosmologist because it is probably necessary in order to resolve the "big" open questions associated with inflation. Embedding inflation into string theory is also interesting from the perspective of a string theorist because it might provide a rare observational window into the theory.

For the reasons described above there has been intense effort on the part of both cosmologists and string theorists to develop inflationary models in string theory (there are far too many proposal to cite here, see [22] for a review and see subsequent chapters for more comprehensive referencing of the literature). Though none of the existing proposals is completely satisfactory, there has been considerable progress recently. In particular, there has been progress in developing fully string theoretic brane inflation models for which the moduli (scalar fields describing the size and shape of the compact dimensions) are stabilized⁷ by the addition of fluxes [24]. Furthermore, in this construction it is possible to identify where the standard model degrees of freedom should reside [25]. In such a construction it is possible to address the issue of reheating after inflation in a quantitative manner [3] (see also [26] for subsequent work). This constuction is not without problems [27]; however, we feel that it is encouraging that realistic string theory constructions can begin to address (at least some of) the aforementioned problems of inflation.

In chapter 7 we will describe a novel string inflation model for which slow roll inflation can proceed even with an extremely steep potential. This provides a particularly interesting possibility for circumventing the issue of fine tuning of the inflaton potential.

It is worth spending some time thinking about why string theory and cosmology can usefully interact with one another [28]. It seems surprising that this should be possible since cosmology deals with the properties of the large scale universe while string theory (ostensibly) describes physics on the smallest scales. Shouldn't the principle of decoupling⁸ forbid string theory from playing any interesting role in cosmology? It is

⁷This stabilization is required for consistency with observation. Time variation of the extra dimensions will lead to a time variation in the four-dimensional effective Newton's constant. Such time variation is tightly constrained by observation (see, for example, [23]).

⁸For example, we know that atoms are made up of smaller constituent electrons, protons and neutrons and futhermore that the protons and neutrons are made up of quarks and gluons. However, a detailed understanding of atomic properties (like chemistry) does *not* depend on the complicated

quite remarkable that string theory and cosmology *can* usefully inform one another. Indeed the fact that this is possible was already evident from the trans-Planckian problem (above). The possibility that inflation could be sensitive to quantum gravity effects means that, from the perspective of a string theorist, the trans-Planckian problem is really the trans-Planckian *window of opportunity* [10]. There appear to be two main reasons why inflation and string theory can interact [28]:

- 1. Access to high energy scales. Inflation probably took place at an extremely high energy scale, close to 10^{16} GeV, for which stringy degrees of freedom would be relevant. (From eq. (1.51) it is clear that the scale of inflation must be $V^{1/4} \leq 10^{16}$ GeV because $\epsilon < 1$. A priori there is no reason why we should be close to the upper bound since almost any value $\epsilon \leq 10^{-2}$ is compatible with observation. However, one might argue that because $\epsilon \ll 1$ requires fine tuning it follows that the most natural value of ϵ is the largest value which is compatible with observation. Furthermore, recent reconstructions of the inflaton potential from the CMB data seem to suggest a large value of ϵ [30].)
- 2. Dependence on UV sensitive properties. The phenomenological success of inflationary model building relies on properties of the underlying particle physics model (like the flatness of the potential) which are UV sensitive. Inflation requires light scalar masses m < H (to keep $|\eta| < 1$), but scalar masses are notoriously difficult to keep small when integrating out high energy degrees of freedom.

1.2 Tachyonic Instabilities

Historically the term "tachyon" has been used to refer to hypothetical particles which travel faster than light. Such particles would carry negative mass-squared (m^2) since the relativistic relation $v = p/\sqrt{p^2 + m^2}$ between velocity v and momentum p clearly

details of the interactions between quarks. In retrospect, the decoupling of small scales from large scale physics was a prerequisite for the success of physics as a discipline.

gives $v \leq 1$ unless $m^2 < 0$. From a more modern perspective the idea of fasterthan-light propagation is abandoned and the term "tachyon" is recycled to refer to a quantum field with $m^2 \equiv V'' < 0$. Clearly V'' can be negative about a maximum of the potential. Fluctuations about such a point will be unstable (since the dynamics of the system will tend to minimize the energy) and hence tachyons are associated with the presence of some physical instability. Tachyons play an important role in symmetry breaking and the formation of cosmic defects. We will illustrate this below using an extremely simple example: a single real scalar field with a double well potential. The results below generalize readily to more complicated theories.

1.2.1 Tachyons in Field Theory

Consider the field theory

$$S = -\int d^4x \left[\frac{1}{2} \partial_\mu \sigma \partial^\mu \sigma + \frac{\lambda}{4} \left(\sigma^2 - v^2 \right)^2 \right]$$
(1.52)

The potential is $V(\sigma) = \lambda \left(\sigma^2 - v^2\right)^2 / 4$ and the Klein-Gordon equation

$$\ddot{\sigma} - \partial^i \partial_i \sigma + V'(\sigma) = 0$$

has static solutions at the critical points where V' = 0. This potential has critical points $\sigma = 0$ and $\sigma = \pm v$.

Let us first consider linearizing the theory about the point $\sigma = 0$ which has V''(0) < 0 and hence represents a tachyonic maximum. Writing $\sigma = 0 + \delta\sigma$ for some small fluctuation $\delta\sigma$ the mode functions obey

$$\delta\ddot{\sigma}_k + k^2\delta\sigma_k - \lambda v^2\delta\sigma_k \cong 0 \tag{1.53}$$

so that

$$\delta\sigma_k \sim \frac{1}{\sqrt{2k}} e^{\sqrt{\lambda v^2 - k^2 t}}$$

and the modes with $k < \sqrt{\lambda}v$ grow exponentially. Quickly the fluctuations become large and the linearized approximation breaks down once $\langle (\delta\sigma)^2 \rangle^{1/2} \sim v$. We see that if the field is initially localized near $\sigma = 0^9$ then it will quickly roll towards the critical points $\sigma = \pm v$.

One the other hand, one might imagine linearizing about the critical points $\pm v$ as $\sigma = \pm v + \delta \sigma$. In this case the mode functions obey

$$\delta\ddot{\sigma}_k + k^2\delta\sigma_k + 2\lambda v^2\delta\sigma_k \cong 0$$

so that

$$\delta\sigma_k \sim \frac{1}{\sqrt{2k}} e^{i\sqrt{2\lambda v^2 + k^2}t}$$

which describes small, stable oscillations about the vacuum. The critical points with V'' > 0 therefore correspond to the true vacua of the theory while the point $\sigma = 0$ is the false (unstable) vacuum.

To see the connection to symmetry breaking, notice that the original field theory (1.52) enjoys a reflection symmetry associated with $\sigma \rightarrow -\sigma$. However, writing $\sigma = +v + \chi$ (we could equally well have chosen $-v + \chi$, the point is that we have to choose *some* vacuum) we get an action for the χ field of the form

$$S = -\int d^4x \left[\partial_\mu \chi \partial^\mu \chi + \lambda v^2 \chi^2 + \lambda v \chi^3 + \frac{\lambda}{4} \chi^4 \right]$$

which contains a term proportional to χ^3 . We see that the reflection symmetry of the original action is spontaneously broken by the vacuum.

It is interesting to note that the action (1.52) admits a classical soliton solution depending on one spatial coordinate $x^1 \equiv x$

$$\sigma(x) = v \tanh\left(\sqrt{\lambda} v x\right) \tag{1.54}$$

called the "kink". This solution interpolates between the vacua $\sigma = -v$ at $x = -\infty$ and $\sigma = +v$ at $x = +\infty$ while the field remains trapped in the false vacuum at x = 0 (the core of the kink). Embedded into the full 3 + 1-dimensional spacetime the kink describes a two-dimensional membrane called a domain wall, with thickness

⁹One could imagine trapping the field near the unstable maximum by thermal effects or else by interactions with other fields. We will give an explicit example below.

 $\Delta x \sim \lambda^{-1/2} v^{-1}$, which is most easily seen by constructing the stress-energy tensor associated with this topological defect

$$T^{00} = -T^{22} = -T^{33} = \frac{\lambda v^4}{2} \operatorname{sech}\left(\sqrt{\lambda} v x\right)$$
$$T^{11} = 0$$
(1.55)

These results generalize in a straightforward manner to more complicated tachyon actions. Of particular interest is the generalization of (1.52) to a complex tachyon. In that case the symmetry which is spontaneously broken is U(1). The theory admits a soliton solution called the vortex which appears as a one-dimensional topological defect called the cosmic string, when embedded into the full spacetime.

1.2.2 Hybrid Inflation and Tachyonic Preheating

The discussion of the previous subsection can be made more concrete by considering an explicit cosmological application. We will consider hybrid inflation [31], which is described by the action

$$S = \frac{M_p^2}{2} \int d^4x \sqrt{-g} R$$

-
$$\int d^4x \sqrt{-g} \left[\frac{1}{2} (\partial \varphi)^2 + \frac{1}{2} (\partial \sigma)^2 + V(\sigma, \varphi) \right]$$
(1.56)

with

$$V(\sigma,\varphi) = \frac{\lambda}{4} (\sigma^2 - v^2)^2 + \frac{m^2}{2} \varphi^2 + \frac{g^2}{2} \varphi^2 \sigma^2$$

= $\frac{\lambda v^4}{4} + \frac{1}{2} (-\lambda v^2 + g^2 \varphi^2) \sigma^2 + \frac{\lambda}{4} \sigma^4 + \frac{m^2}{2} \varphi^2$ (1.57)

Initially we take $g^2 \varphi^2 > \lambda v^2$ so that σ has a positive mass-squared and remains trapped in the false vacuum $\sigma = 0$. This is the inflationary phase. The effective potential driving inflation is then

$$V = \frac{\lambda v^4}{4} + \frac{m^2}{2} \varphi^2 \cong \frac{\lambda v^4}{4}$$

where we assume that $\lambda v^4 \ll 2m^2 \varphi^2$ throughout inflation.¹⁰ The slow roll parameters along the inflationary trajectory are

$$\eta \cong 4 \frac{M_p^2 m^2}{\lambda v^4}, \quad \epsilon \cong 8 \left(\frac{M_p m^2 \varphi}{\lambda v^4}\right)^2$$

so that if $\lambda v^4 \ll 2m^2 \varphi^2$ then $\epsilon \ll \eta$ and $n-1 \cong 2\eta$. During the slow roll phase the inflaton evolves as

$$\varphi(t) = \frac{\lambda^{1/2}v}{g} \exp\left(-\frac{m^2(t-t_c)}{3H}\right) = \frac{\lambda^{1/2}v}{g} \left(\frac{a(t)}{a(t_c)}\right)^{-\eta}$$

where t_c is the time at which $\varphi = \varphi_c \equiv \lambda^{1/2} v/g$ and $H \cong \lambda^{1/2} v^2/(2\sqrt{3}M_p)$. One can show that the COBE normalization in this model fixes the inflaton mass as $m^2 \cong 230g\lambda v^5/M_p^3$ while the constraint $|n-1| \lesssim 10^{-1}$ requires $gv/M_p \lesssim 5 \times 10^{-5}$. Notice that $\eta > 0$ so that this model always gives a blue-tilted spectrum which is disfavoured by the recent WMAP data.

Once $t > t_c$ the field σ becomes tachyonic. The subsequent evolution is called tachyonic preheating [32, 33]. If the dimensionless couplings g, λ are not too small then the system experiences a very strong force along the unstable direction in field space and the tachyonic instability will bring inflation to an end very quickly (this corresponds to satisfying the so-called "waterfall conditions" of hybrid inflation). In this regime the tachyonic preheating completes in a time which is very short compared to H^{-1} and we are justified in neglecting the expansion of the universe to describe this phase. It is also common to make the "instantaneous quench" approximation in which one neglects also the time variation of the mass term $m_{\sigma}^2 \equiv -\lambda v^2 + g^2 \varphi^2$. In this approximation one takes the σ field to be massless at $t = t_c$ while quickly at $t > t_c$ a tachyonic mass $m_{\sigma}^2 = -\lambda v^2$ is turned on. The large scale mode functions grow exponentially as in (1.53) and the variance of the fluctuations grows as [34]

$$\langle (\delta\sigma)^2 \rangle = \frac{1}{8\pi} \int_0^{\lambda v^2} dk^2 \frac{\lambda v^2}{\lambda v^2 - k^2} \sinh^2 \left[(t - t_c) \sqrt{\lambda v^2 - k^2} \right]$$

¹⁰It is not necessary to demand that the false vacuum energy $\lambda v^4/4$ dominate to achieve successful inflation, however, if the mass term dominates then this model is not different from chaotic inflation.

where we have neglected the divergent vacuum contribution. This tachyonic growth persists until $\langle (\delta \sigma)^2 \rangle \sim v^2$ at which point the dynamics is dominated by oscillations about the true minima $\pm v$ and the symmetry breaking process has completed.

We will discuss some cosmological implications of tachyonic preheating after hybrid inflation in chapter 6, showing that preheating can lead to observable consequences in the CMB. In particular, we will show that it is possible to generate significant nongaussianity during tachyonic preheating.

We see that once $t > t_c$ the tachyon field rolls down the unstable direction towards the true minima $\pm v$. Causality forbids the tachyon field σ from rolling to the same vacuum in two spatial regions which are causally disconnected. The universe therefore fragments into domains of $\sigma = +v$ and $\sigma = -v$. Continuity of the field σ means that there must be some region in between these domains where the field remains trapped at $\sigma = 0$: this is exactly the kink solution (1.54). This mean that tachyonic preheating will produce a network of domain wall topological defects in the early universe. Since during radiation/dust domination the causal horizon is $\sim H^{-1}$ it follows that we must expect at least one defect per Hubble volume. However, this is merely a lower bound on the produced number of defects and typically the microscopic dynamics will produce a network with characteristic defect separation $\sim \lambda^{-1/2} v^{-1}$ [32, 33, 34]. (The argument above is a caricature of the Kibble mechanism [35] which ensures that topological defects will be formed at cosmological phase transitions. See [36] for a more modern review.) After formation the defect network will undergo some highly nontrivial nonlinear dynamics which is thought to lead to a scaling solution so that at late times there is roughly one defect per Hubble volume, regardless of the initial density.¹¹ We will discuss the formation of topological defects at the end of brane inflation and the cosmological consequences of the defect network in chapter 5.

We should mention that a cosmological network of domain walls is not compatible with late-time cosmology since these defects will overclose the universe unless their

¹¹In the case of a cosmic string network, the existence of such a scaling solution is fairly well understood. For a domain wall network this remains to be demonstrated conclusively [37, 38], however, a scaling solution is expected.

tension is fine tuned to be unreasonably small. The above discussion also holds with only minor modifications for preheating with a complex tachyon which leads to the formation of a network of cosmic strings. Cosmic string networks are not ruled out and lead to a variety of interesting observational consequences.

1.2.3 Tachyons in String Theory

Certain D-brane configurations in string theory carry no conserved charges and hence are unstable to decay (see [39, 40] for reviews). In particular, the following systems are unstable:

1. The brane-antibrane system. The lightest stretched string mode between a parallel brane and antibrane separated by a distance y has a mass-squared given by

$$m^2 = m_s^2 \left[-\frac{1}{2} + \frac{(m_s y)^2}{(2\pi)^2} \right]$$

where m_s is the string mass. This state become tachyonic once the branes are sufficiently close, signaling that the system has become unstable to decay.

- 2. Wrong dimensional branes. In type IIA/IIB string theory odd/even dimensional branes contain a tachyonic excitation in the spectrum of open strings ending on the brane.
- 3. The brane-brane system with fluxes. Certain brane-brane systems with fluxes turned on have tachyonic instabilities. One example in the D3/D7 system [209] in which case the presence of non-self-dual D7 fluxes can generate a negative contribution to the mass-squared of the stretched string modes between the branes so that a tachyon develops at sufficiently small inter-brane separation.

The decay of unstable objects in string theory may lead to interesting cosmological consequences. In order to explore these consequences we must first derive an effective field theoretic decription of the open string tachyon. To do so we recall that the Dp-brane in string theory is described by the Dirac-Born-Infeld (DBI) action (see, for

example, [41])

$$S = -T_p \int d^{p+1} \sigma e^{-\phi} \sqrt{-\det(G_{AB} + B_{AB} + 2\pi\alpha' F_{AB})}$$
(1.58)

The brane tension is

$$T_p = \frac{m_s^{p+1}}{(2\pi)^p}$$

while the dilaton ϕ is related to the string coupling as $e^{-\phi} = g_s^{-1}$ and $\alpha' \equiv m_s^{-2}$. We define indices $A, B = 0, \dots, p$ which run over the directions parallel to the brane while $\mu, \nu = 0, \dots, D-1$ run over the full D-dimensional spacetime and a, b = $p+1, \dots, D-1$ run over the directions perpendicular to the brane. The coordinates σ^A parameterize the embedding of the brane into the full spacetime. In (1.58) G_{AB} is the pull-back of the spacetime metric $g_{\mu\nu}$ to the hypersurface of the brane

$$G^{AB} = \frac{\partial \sigma^A}{\partial x^{\mu}} \frac{\partial \sigma^B}{\partial x^{\nu}} g^{\mu\nu}$$
(1.59)

Similarly B_{AB} and F_{AB} are the pull-backs of the antisymmetric tensor and the open string U(1) gauge field strength respectively. Setting $F_{AB} = B_{AB} = 0$ and assuming a constant dilaton field the action becomes

$$S = -\tau_p \int d^{p+1}\sigma \sqrt{-\det\left(G_{AB}\right)} \tag{1.60}$$

where

$$\tau_p = \frac{m_s^{p+1}}{(2\pi)^p} \frac{1}{g_s}$$

Intuitively this action makes sense because $\delta S = 0$ minimizes the surface area of the D-brane. We can choose the coordinates along the brane such that

$$\sigma^A = x^A$$

 $\sigma^a = y^a(x)$

(this choice is referred to as the static gauge) so that (1.60) becomes

$$S = -\tau_p \int d^{p+1}x \sqrt{-g} \sqrt{1 + \delta^{ab}g^{\mu\nu}\partial_{\mu}y_a\partial_{\nu}y_b}$$
(1.61)

where the fields $y^a(x)$ are the transverse scalars which describe the fluctuations of the brane surface. Notice that in the absence of transverse fluctuations the stress tensor associated with the D*p*-brane is of the form $T_{AB} = -g_{AB} \tau_p \, \delta^{D-p-1}(\mathbf{x} - \mathbf{x_0})$ while $T_{ab} = T_{Ab} = 0$ (here $\mathbf{x_0}$ gives the position of the brane in the space spanned by $\{x^a\}$).

We would like to generalize the Dp-brane action (1.61) to incorporate the tachyon field which appears on the world-volume of a wrong dimensional brane. A natural generalization, usually called the Sen action, is [42, 43, 44]

$$S = -\int d^{p+1}x\sqrt{-g}\,V(T)\,\sqrt{1+g^{\mu\nu}\partial_{\mu}y^{a}\partial_{\nu}y_{a}+g^{\mu\nu}\partial_{\mu}T\partial_{\nu}T}$$
(1.62)

where T is a real scalar field describing the open string tachyon and V(T) is the tachyon potential which satisfies

$$V(0) = \tau_p, \quad V''(0) < 0,$$

 $V(T_{\min}) = 0, \quad V''(T_{\min}) > 0$

and T_{\min} is the value of T which minimizes the potential (we will eventually be interested in the limit $T_{\min} \to \infty$). The configuration T = 0 is an unstable maximum of the potential which describes a system consisting of a single wrong-dimensional D*p*brane. One expects the tachyon to roll to T_{\min} , the true minimum, which describes the vacuum without any D*p*-brane (since V(T) plays the role of a time-varying brane tension). In a full string theory calculation the brane will decay to a gas of closed strings [45].

Though it has not been derived from first principles, the simple action (1.62) does a remarkably good job of describing the dynamics. In particular:

1. This action can be obtained from string theory in some limit [46] with the potential

$$V(T) = \frac{\tau_p}{\cosh\left[\frac{T}{\sqrt{2}l_s}\right]} \tag{1.63}$$

where $l_s \equiv m_s^{-1}$ is the string length. (For the remainder of the chapter we assume this potential.)

2. This action reproduces the correct tachyonic mass in the unstable vacuum [45].

- 3. It yields the correct stress tensor for homogeneous tachyon condensation [45, 47, 49].
- 4. The action admits soliton solutions corresponding to lower dimensional branes. The tension of the defect reproduces the expected D-brane tension [50].
- 5. The action for the fluctuations about the defect solutions reproduces the expected DBI action for a brane [50].
- 6. In studies of dynamical defect formation using this action, the gradient of the tachyon field blows up in a finite time near the core of the defect [4, 51], leading to a delta function divergence in the stress tensor [1]. The same result was observed in a fully string theoretic calculation in [47, 48].
- The late time solutions close to the tachyon vacuum give a coarse-grained description of the gas of closed strings which are produced during brane decay [52].

In chapter 3 we will describe in detail the dynamics of the tachyon field close to the true vacua $T \to \pm \infty$. In particular, we will show that small initial inhomogeneities in the field profile will lead to the formation of *caustics* where second (and higher) spatial derivatives of the field blow up, invalidating the effective description (1.62).

Let us now try to construct a kink solution the action (1.62) neglecting the transverse scalars $y^a = 0$. The following discussion follows closely [50]. We are interested in a profile which depends on only one spatial coordinate $x^1 \equiv x$. The (x, x) component of the stress-tensor is

$$T_{xx} = \frac{V(T)}{\sqrt{1 + (\partial_x T)^2}}$$

and energy-momentum conservation gives $\partial_x T_{xx} = 0$ so that T_{xx} is constant. Far from the core of the kink (which we take to be at x = 0) the tachyon should be in the true vacuum $T = \pm \infty$ (recall eq. (1.54)) so that $T_{xx} = 0$. At the core of the kink T = 0 so that $T_{xx} = 0$ requires $\partial_x T = \infty$. Clearly the solution appears singular, however, the stress tensor is well-defined and is independent of how we regularize the

1.2 Tachyonic Instabilities

kink solution. Consider the profile

$$T(x) = f\left(\frac{x}{\varepsilon}\right) \tag{1.64}$$

where f(u) satisfies

$$f(u) = -f(-u), \quad f'(u) > 0 \ \forall \ u, \quad f(\pm \infty) = \pm \infty$$

but is otherwise arbitrary. This profile is meant to be understood in the limit $\varepsilon \to 0$. Such a field profile satisfies energy-momentum conservation and is a solution of the full equation of motion for T. The resulting stress tensor gives the expected result for an infinitely thin defect (compare to eq. (1.55))

$$T_{00} = -T_{22} = \dots = \tau_{p-1}\delta(x)$$
$$T_{xx} = T_{0x} = \dots = 0$$

(For finite ε one obtains a strongly peaked function which tends to a delta function in the limit $\varepsilon \to 0$.) The fact that the kink solution is singular no longer seems surprising when one recalls that the D-brane should be infinitely thin. Since the D-brane stress tensor has a delta function profile, it is not surprising that tachyon kink solution must also be a distribution. When we consider the dynamical formation of such a tachyon defect in chapters 2 and 5 we will see that the defect forms within a finite time t_c and that the role of the parameter ε which regulates the kink is played by $t_c - t$.

As we have mentioned previously, the action describing small fluctuations about this kink solution exactly reproduces the expected DBI action for a stable D(p-1)brane. We see that the tachyon kink, which would be expected to form during brane decay, *is* a lower dimensional brane. The above discussion also generalizes to the brane-antibrane system in which case the tachyon is a complex field and the resulting vortex-like singular soliton solution describes a codimension-two brane, which results as a decay product from brane-antibrane annihilation.

Chapter 2

Creating the Universe from Brane-Antibrane Annihilation

Abstract

When p-dimensional branes annihilate with antibranes in the early universe, as in brane-antibrane inflation, stable (p-2)-dimensional branes can appear in the final state. We reexamine the possibility that one of these (p-2)-branes could be our universe. In the low energy effective theory, the final state branes are cosmic string defects of the complex tachyon field which describes the instability of the initial state. We quantify the dynamics of formation of these vortices. This information is then used to estimate the production of massless gauge bosons on the final branes, due to their coupling to the time-dependent tachyon background, which would provide a mechanism for reheating after inflation. We improve upon previous estimates indicating that this can be an efficient reheating mechanism for observers on the brane.

2.1 Introduction

In the last few years significant progress has been made in constructing string theoretic cosmological models where inflation is driven by the naturally occuring potentials between D-branes and their antibranes [24, 25], [53]-[55]. The formation of lower-dimensional branes at the end of inflation can lead to interesting signals: cosmic-string-like or higher dimensional defects could be observable remnants [56, 57, 58], possibly providing a rare clue to the stringy origin of inflation. A more radical idea was explored in [59]: perhaps our own observable universe is such a defect in the higher-dimensional spacetime predicted by string theory. Since the stable branes in Type IIB string theory have spatial dimensionalities which are odd, a 3-brane would have descended from annihilation of 5-branes in this picture, and thus they would be codimension-two defects in of the effective 6D theory. Codimension-two braneworlds have attracted interest lately because of their novel features, which might have some bearing on the cosmological constant problem [60, 61].

Our interest in this scenario is motivated by questions about the efficiency of reheating in brane-antibrane inflation [62]. It is possible that the energy density lib-

erated from the brane collisions will be converted mostly into closed string states, ultimately gravitons, and not necessarily into visible radiation [63]. A generic mechanism which could avoid this problem was proposed in [59], wherein the reheating in D-brane driven inflation is due to the coupling of massless gauge fields to a timedependent tachyon condensate, which describes the annihilation process. However, ref. [59] considered only the formation of a tachyon kink instead of the more realistic case of a vortex, and it used a somewhat crude ansatz for the background tachyon field. The problem of finding the actual tachyon background predicted by string theory was studied numerically in [51] but no attempt was made to improve on the reheating computation. In this chapter we aim to analytically determine the dynamics of formation of lower dimensional branes described as tachyon defects—both kinks and vortices—and to improve on the reheating calculation of [59].

Let us begin by describing the scenario we have in mind. In the simplest version of D-brane inflation a parallel brane and antibrane begin with some separation between them in one of the extra dimensions. Although parallel branes are supersymmetric and have no force between them, the brane-antibrane system breaks supersymmetry so that there is an attractive force and hence a nonvanishing potential energy. It is the latter which drives inflation. Once the branes have reached a critical separation one of the stretched string modes between the branes, T, becomes tachyonic and the branes become unstable to annihilation. The tachyon field starts from the unstable maximum T = 0 and rolls towards the vacuum $T \rightarrow \pm \infty$. However, topological defects may form through the Kibble mechanism [56, 57, 58] so that T = 0 stays fixed at the core of the defect. These defects are known to be consistent descriptions of branes whose dimension is lower than that of the original branes [50, 64, 65]. For example, the brane-antibrane system has a complex tachyon field, leading to vortices which represent codimension-two branes. On the other hand, an unstable brane has a real tachyon which leads to kinks representing codimension-one branes.

The formation of tachyon defects at the endpoint of D-brane inflation is a dynamical process where the tachyon couples to gauge fields which will be localised on the descendant brane. It is thus expected that some radiation will be produced by the rolling of the tachyon and the problem of reheating becomes quantitative: can this effect be efficient enough to strongly deplete the energy density of the tachyon fluid so the the universe starts out being dominated by radiation rather than cold dark matter? It is important to stress that though the situation is somewhat analogous to that of hybrid inflation (where the tachyon plays the role of the unstable direction in field space which allows for inflation to end quickly) the mechanism for reheating is qualitatively different. The difference is that in the low energy effective field theory which describes the tachyon T, the potential is minimized at $T = \pm \infty$ and there are no oscillations about the minimum of the potential. In a normal hybrid inflation model, T would have a minimum at some finite value and the oscillations of T around its minimum would give rise to reheating in the usual way. In the present case, the time dependence of the background is monotonic, not oscillatory. Reheating thus might seem to resemble gravitational particle production [66] rather than the standard picture in which the inflaton decays. However, in this work we highlight an important difference between reheating through tachyon condensation and gravitational particle production, which can make the former much more efficient: there is a divergence in the stress-energy tensor of the tachyon field within a finite time, which corresponds to the formation of the lower-dimension D-brane.

In this chapter we study analytically the dynamical formation of the tachyon vortex and improve the reheating calculation, using a slightly simplified model of particle production by tachyon condensation, which captures the essential physics revealed by the analysis of vortex formation. In section 2.2 we review the formation of tachyon kinks which describes the condensation of a brane to a brane of codimension-one. In sections 2.3-2.5 we study analytically the formation of a tachyon vortex on the brane-antibrane pair. Section 2.3 introduces Sen's action for the complex tachyon field describing this situation. Section 2.4 presents new analytic results for the time-dependent, complex tachyon field representing vortex formation, both near to and far from the vortex core. In section 2.5 we show that the stress-energy tensor for the system splits into a localized, singular piece describing the descendant branes, plus a bulk contribution that describes the rolling tachyon condensate. Section 2.6

introduces the effective action for U(1) gauge bosons which become localized on the final-state 3-brane, in the rolling tachyon background. This provides a model for the visible radiation produced during reheating. In section 2.7 we calculate the energy density of this produced radiation on the 3-brane, using some reasonable simplifying assumptions. Section 2.8 gives our conclusions, including speculation about how the final brane-antibrane system could be stabilized.

2.2 Dynamical Tachyon Kink Formation on Unstable Dp-branes

In this section we review the dynamical formation of a D(p-1)-brane through tachyon condensation on an unstable Dp-brane, and derive a few new results. The equations of motion in this case are simpler than in the case of the vortex and we will use the analysis of this section to reinforce our conclusions when we analyze the vortex since many of the results are quite analogous.

2.2.1 Effective Field Theory and Equations of Motion

We will work with the effective action for the tachyon on an unstable Dp-brane $[42, 44]^1$

$$S = -\int V(T)\sqrt{-\det|\eta_{MN} + \partial_M T \partial_N T|} \, d^{p+1}x$$
(2.1)

where we have set the gauge fields and transverse scalars to zero. We use the potential $V(T) = \tau_p \exp(-T^2/a^2)$ where τ_p is the tension of a D*p*-brane and $a = 2\sqrt{\pi \alpha'}$. The value of the constant *a* is chosen so that the potential satisfies the normalization condition

$$\int_{-\infty}^{+\infty} V(y) \, dy = 2\pi \sqrt{\alpha'} \tau_p = \tau_{p-1} \tag{2.2}$$

¹The convention for indices is that upper case roman indices $\{M, N\}$ run over the full space-time coordinates $\{0, 1, \dots, p\}$, greek indices $\{\mu, \nu\}$ run over the defect coordinates $\{0, 1\}$ and "hatted" greek indices $\{\hat{\mu}, \hat{\nu}\}$ run over the remaining spatial coordinates $\{2, 3, \dots, p\}$. We use metric signature diag $(-1, 1, 1, \dots)$.

proposed in [50]. This normalization was used in [50] to fix the tension of the singular static kink solution of the action (2.1) to correspond to the tension of a D(p-1)brane. For a time-dependent kink solution we take T to be a function of $x^{\mu} = (t, x)$ so that the action (2.1) becomes

$$S = -\int V(T)\sqrt{1 + \partial_{\mu}T\partial^{\mu}T} \, d^{p+1}x.$$
(2.3)

Static solutions of the theory (2.3) are well studied in the literature [50, 67]. Inhomogeneous solutions have also been studied in some detail [59, 51, 68, 69, 70]. The energy momentum tensor for (2.3) is

$$T_{\mu\nu} = \frac{V(T)}{\sqrt{1 + \partial_{\rho}T\partial^{\rho}T}}\partial_{\mu}T\partial_{\nu}T - \eta_{\mu\nu}V(T)\sqrt{1 + \partial_{\rho}T\partial^{\rho}T},$$
(2.4)

and the Euler-Lagrange equation of motion is

$$\partial^{\mu}\partial_{\mu}T - \frac{\partial_{\mu}\partial_{\nu}T\partial^{\mu}T\partial^{\nu}T}{1 + \partial^{\rho}T\partial_{\rho}T} - \frac{V'(T)}{V(T)} = 0$$
(2.5)

where $V'(T) = \frac{\partial V(T)}{\partial T}$. It is worth noting, as in [71], that the equation of motion (2.5) is equivalent to conservation of energy $\partial_{\mu}T^{\mu\nu} = 0$ for nonconstant T since $\partial_{\mu}T^{\mu\nu} = \partial^{\nu}T \left[\partial_{\mu}\left(\frac{\partial \mathcal{L}}{\partial(\partial_{\mu}T)}\right) - \frac{\partial \mathcal{L}}{\partial T}\right]$. It will be useful in the ensuing analysis to define

$$\Sigma = \frac{V(T)}{\sqrt{1 + \partial_{\mu} T \partial^{\mu} T}}.$$
(2.6)

2.2.2 Solutions Near the Core of the Defect

At the core of the kink we expect the field to stay pinned at T = 0. Consider initial data $T(t = 0, x) = T_i(x)$ and $\dot{T}(t = 0, x) = \dot{T}_i(x) = 0^2$. One expects the field to start to roll where $T_i(x) \neq 0$ due to the small displacement from the unstable maximum V'(T) = 0. At t = 0 the equation of motion (2.5) is

$$\ddot{T}_i(x)(1+T'_i(x)^2) = T''_i(x) + 2a^{-2}T_i(x)(1+T'_i(x)^2).$$

²For the remainder of this section the dot denotes differentiation with respect to time while the prime denotes differentiation with respect to the x coordinate.

Clearly any point x_0 where $T_i(x_0) = T''_i(x_0) = 0$ will be a fixed point where $\ddot{T}(t, x_0) = 0 = \dot{T}(t, x_0)$ throughout the evolution. We restrict ourselves only to considering initial data such that $\operatorname{sgn}(\ddot{T}_i(x)) = \operatorname{sgn}(T_i(x))$ for all x to ensure that the solutions are increasing.

At the site of the kink (which we take to be $x_0 = 0$) we have T = 0; hence there should always be some neighbourhood of the point x = 0 where we can take $V'(T) \cong 0$ so that (2.5) yields

$$\ddot{T}(1+T'^2) = (1-\dot{T}^2)T'' + 2\dot{T}T'\dot{T}'.$$
(2.7)

This has an increasing solution with T'' = 0

$$T(x,t) = x \tan\left[\frac{\omega}{a}(t-t_c) + \frac{\pi}{2}\right].$$
(2.8)

Near the site of the kink the slope of the tachyon field diverges as $(t_c - t)^{-1}$ as t approaches the critical time t_c , similar to the solutions in [51].

The finite-time slope divergence was observed both numerically and analytically in [51] and leads to the formation of a singularity in the energy density at $t = t_c$. This effect was also found in an exact string theoretic calculation in [48]. As $t \to t_c$ we have $\Sigma(t \to t_c, x \cong 0) \to 0$. The case $\Sigma = const$ arises as a first integral of the motion in the static case and the limit $\Sigma \to 0$ corresponds to the singular soliton solution of Sen [67]. It is natural to expect, then, that as $t \to t_c$, near x = 0, the tachyon field $T(t = t_c, x = 0)$ coincides with the stable kink solution of Sen and the time evolution in this neighbourhood stops. We will argue that at this point a codimension-one brane has formed.

2.2.3 Vacuum Solutions

Away from the site of the kink the field is expected to roll towards the vacuum $T \rightarrow \pm \infty$ so that $V(T) \rightarrow 0$ at late times for $x \neq 0$. To analytically study the dynamics near the vacuum it is easiest to work in the Hamiltonian formalism [72, 73] since the Lagrangian vanishes in the limit $V(T) \rightarrow 0$, whereas the Hamiltonian remains well-defined. Defining the momentum conjugate to T as $\Pi = \delta S / \delta T$ the Hamiltonian is

given by $\mathcal{H} = \sqrt{\Pi^2 + V(T)^2} \sqrt{1 + \partial_i T \partial_i T}$. It is useful to rewrite Hamilton's equations of motion, $\dot{\Pi} = -\frac{\partial \mathcal{H}}{\partial T}$ and $\dot{T} = \frac{\partial \mathcal{H}}{\partial \Pi}$, in a manifestly covariant way as

$$\partial^{\mu}T\partial_{\mu}T + 1 = \frac{V(T)^2}{\Sigma^2}$$
(2.9)

and

$$\Sigma \partial_{\mu} [\Sigma \partial^{\mu} T] = V(T) V'(T), \qquad (2.10)$$

where $\Sigma = \Pi/\dot{T}$ is defined in (2.6). In the limit $V(T) \to 0$ equations (2.9), (2.10) yield

$$\partial^{\mu}T\partial_{\mu}T + 1 = 0 \tag{2.11}$$

and

$$\partial_{\mu}[\Sigma \partial^{\mu} T] = 0. \tag{2.12}$$

The solutions of (2.11) were found for arbitrary Cauchy data in [68] using the method of characteristics. The generic solutions exhibit the formation of caustics where second and higher order derivatives become singular. Caustics are known to form in systems with a pressureless fluid, which is a good description of the tachyon field as it approaches its ground state $T \to \infty$. It is not known whether the caustics are a genuine prediction of string theory or just an artifact of the derivative truncation which leads to the Born-Infeld Lagrangian for the tachyon.

In any case, caustics are not present in the simplest solution of eq. (2.11), $T = \pm t$, which is the asymptotic form for the homogeneously rolling tachyon. For $\dot{T}^2 = 1$, eq. (2.12), which is equivalent to energy conservation, implies that $\Sigma(t, x) = \Sigma(x)$ is an arbitrary function of x. In this regime the energy momentum tensor (2.4) is identical to that of pressureless dust $T_{\mu\nu} = \Sigma(x)u_{\mu}u_{\nu}$ where $u_{\mu} = \partial_{\mu}T$ is interpreted as the local velocity vector and Σ is interpreted as a Lorentz-invariant matter density [44, 72, 73].

2.2.4 Stress-Energy Tensor

We are interested in the behavior of T_{MN} as $t \to t_c$. First we consider the neighbourhood near x = 0 where $T(t, x) \cong \frac{kx}{t_c - t}$ as $t \to t_c$ (see 2.8). Near x = 0, the Hamiltonian is

$$T_{00} \cong \frac{\tau_p k}{t_c - t} \exp\left(-\frac{k^2 x^2}{a^2 (t_c - t)^2}\right)$$

and

$$\lim_{t \to t_c^-} T_{00} = \sqrt{\pi} a \tau_p \delta(x) = \tau_{p-1} \delta(x)$$
(2.13)

using the normalization (2.2) for the potential. Similarly

$$T_{\hat{\mu}\hat{\nu}} \rightarrow -\tau_{p-1}\,\delta(x)\,\delta_{\hat{\mu}\hat{\nu}}$$

and $T_{11} \to 0$ as $x \to 0$.

Consider now the late-time behavior of T_{MN} away from the site of the kink. Using the solutions of section 2.2.3 we find $T_{00} \rightarrow \Sigma(x)$ while T_{11} , T_{01} and $T_{\hat{\mu}\hat{\nu}}$ tend to zero for $x \neq 0$.

To summarize, we find that in the limit of condensation the energy momentum tensor is identical to that of a D(p-1)-brane:

$$T_{00} = \tau_{p-1} \,\delta(x) + \Sigma(x)$$
$$T_{11} = T_{01} = 0$$
$$T_{\hat{\mu}\hat{\nu}} = -\tau_{p-1} \,\delta(x) \,\delta_{\hat{\mu}\hat{\nu}}.$$

The extra bulk energy density $\Sigma(x)$ is similar to the result in [71] and corresponds to what has been dubbed tachyon matter.

2.3 Effective Tachyon Field Theory on the Brane-Antibrane Pair

We would like to generalize the results of the previous section to study the dynamical formation of a tachyon vortex and hence a codimension-two brane. This is the more realistic situation, since the stable D-branes of a given string theory are those whose dimensions differ by multiples of two. We will work with an effective action proposed by Sen [50] for the tachyon on a brane-antibrane pair. The field content for this system is a complex tachyon field T, massless gauge fields $A^{(1)}_{\mu}$, $A^{(2)}_{\mu}$ and scalar fields $Y^{I}_{(1)}$, $Y^{I}_{(2)}$ corresponding to the transverse fluctuations of the branes. The index (i) = (1), (2), which we call the brane index, labels which of the original branes (actually the brane or the antibrane) the field is associated with. The effective action is:

$$S = -\int V(T, Y_{(1)}^{I} - Y_{(2)}^{I}) \left(\sqrt{-\det M^{(1)}} + \sqrt{-\det M^{(2)}}\right) d^{p+1}x$$
(2.14)

where 3

$$M_{MN}^{(i)} = g_{MN} + \alpha' F_{MN}^{(i)} + \partial_M Y_{(i)}^I \partial_N Y_{(i)}^I + \frac{1}{2} D_M T D_N T^* + \frac{1}{2} D_M T^* D_N T, \qquad (2.15)$$

$$F_{MN}^{(i)} = \partial_M A_N^{(i)} - \partial_N A_M^{(i)}, \quad D_M = \partial_M - i A_M^{(1)} + i A_M^{(2)}.$$
(2.16)

For the remainder of this chapter we will ignore the transverse scalars and choose $V(T,0) = V(T) = \tau_p \exp(-|T|^2/a^2)$ where *a* is chosen so that the static singular vortex solutions of the theory (2.14) have the correct tension to be interpreted as codimension 2 D-branes according the the normalization proposed in [50]. We will discuss the normalization of the potential proposed in [50] in more detail when we calculate the energy momentum tensor for the theory (2.14).

Though the action (2.14) was not derived from first principles it obeys several necessary consistency conditions which are discussed in [50]. There have been various other proposals for the tachyon effective action and vortex solutions on the brane-antibrane pair [74, 75, 76]. See [77] for a discussion of various models including (2.14).

³Greek indices $\{\mu, \nu\}$ are now understood to run over the coordinates $\{0, 1, 2\}$ on which the vortex solutions depend, and hatted greek indices $\{\hat{\mu}, \hat{\nu}\}$ run over the spatial coordinates parallel to the vortex $\{2, 3, \dots, p\}$, where p = 6 for a vortex which describes a 3-brane. Upper case roman indices $\{M, N\}$ still run over the full space-time coordinates $\{0, 1, \dots, p\}$. Finally it will be convenient later on to refer to lower case roman indices $\{m, n\}$ which run over only the time and radial coordinates $\{0, 1\}$.

2.4 Vortex Solutions on the Brane-Antibrane Pair

To construct vortex solutions we ignore the transverse scalars and take the remaining fields to depend only on the polar coordinates $x^{\mu} = (x^0, x^1, x^2) = (t, r, \theta)$ with metric $g_{\mu\nu}dx^{\mu}dx^{\nu} = -dt^2 + dr^2 + r^2d\theta^2$. Since the vortex solution should have azimuthal symmetry we make the ansatz:

$$T(t,r,\theta) = e^{i\theta}f(t,r), \quad A_{\theta}^{(1)} = -A_{\theta}^{(2)} = \frac{1}{2}g(t,r)$$
(2.17)

with all other components of $A^{(i)}_{\mu}$ vanishing. This generalizes the ansatz used in [50] to include time-dependence in the fields f and g. For (2.17) one has

$$D_t T = e^{i\theta} \dot{f}(t, r), \quad D_r T = e^{i\theta} f'(t, r),$$
$$D_{\theta} T = e^{i\theta} i(1 - g(t, r)) f(t, r)$$

and

$$F_{t\theta}^{(1)} = \frac{1}{2}\dot{g}(t,r) = -F_{t\theta}^{(2)}, \quad F_{r\theta}^{(1)} = \frac{1}{2}g'(t,r) = -F_{r\theta}^{(2)}$$

where the dot denotes differentiation with respect to time and the prime now denotes differentiation with respect to the radial coordinate. The matrices $M_{MN}^{(i)}$ are

$$\begin{bmatrix} M_{MN}^{(1)} \end{bmatrix} = \begin{bmatrix} -1 + \dot{f}^2 & \dot{f}f' & \alpha'\dot{g}/2 & 0 \\ \dot{f}f' & 1 + f'^2 & \alpha'g'/2 & 0 \\ -\alpha'\dot{g}/2 & -\alpha'g'/2 & r^2 + (1-g)^2f^2 & 0 \\ 0 & 0 & 0 & \delta_{\hat{\mu}\hat{\nu}} \end{bmatrix},$$

$$\begin{bmatrix} M_{MN}^{(2)} \end{bmatrix} = \begin{bmatrix} M_{MN}^{(1)} \end{bmatrix}^T.$$
(2.18)

We also have $\det(M^{(1)}) = \det(M^{(2)})$ since $M_{MN}^{(1)} = M_{NM}^{(2)}$ and so we omit the brane index (i) on $\det(M^{(i)})$ in subsequent calculations.

The action for this ansatz simplifies to:

$$S = -2 \int V(f) \left((1 + \partial_m f \partial^m f) \left[r^2 + f^2 (1 - g)^2 \right] + \frac{\alpha'^2}{4} \partial_m g \partial^m g - \frac{\alpha'^2}{4} \left(\epsilon^{mn} \partial_m f \partial_n g \right)^2 \right)^{1/2} d^{p+1}x$$
(2.19)

where $x^m = (x^0, x^1) = (t, r)$ and $g_{mn}dx^m dx^n = -dt^2 + dr^2$. For notational convenience we define the scalar quantity

$$\Sigma(t,r) = \frac{V(f)}{\sqrt{-\det(M)}}$$
(2.20)

in analogy with (2.6). The equation of motion for the tachyon is

$$\partial_m \left[\Sigma \left[r^2 + (1-g)^2 f^2 \right] \partial^m f - \frac{\alpha'^2 \Sigma}{4} (\epsilon^{ab} \partial_a f \partial_b g) \epsilon^{mn} \partial_n g \right]$$

= $\Sigma (1 + \partial^m f \partial_m f) (1-g)^2 f + \frac{V'(f) V(f)}{\Sigma},$ (2.21)

and the nontrivial component of the equation of motion for the gauge field is

$$\frac{\alpha'^2}{4} \partial_m \left[\Sigma \partial^m g - \Sigma (\epsilon^{ab} \partial_a f \partial_b g) \epsilon^{mn} \partial_n f \right]$$

= $\Sigma (1 + \partial^m f \partial_m f) f^2 (g - 1).$ (2.22)

Alhough these equations are somewhat cumbersome, inspection of (2.22) tells us that there should exist a solution g(t,r) such that at g = 1, the vacuum, we have $\partial_m g = 0$. This is the asymptotic behavior which corresponds to a vortex solution; it is already known from [50] that the static solution g(r) is a monotonically increasing function which varies between 0 and 1. Thus we shall only consider solutions with the asymptotic behavior $\partial_m g(t,r) \to 0$ as $g(t,r) \to 1$. We will take initial data $g(0,r) = g_i(r)$ such that $g_i(0) = 0, 0 \leq g_i(r) \leq 1$ for all r and $g'_i(r) \geq 0$ for all r. In addition we will focus on initial tachyon profiles $f(0,r) = f_i(r)$ such that $f_i(0) = 0$ and $f'_i(r) > 0$ for all r. For these initial conditions the tachyon must start rolling for $r \neq 0$ due to its displacement from the unstable vacuum V'(f) = 0. Since the asymptotic $g_i(r \to \infty) \to 1$ is an exact solution of (2.22) and g(t,r = 0) = 0 by construction, we therefore expect g(r, t) to increase towards unity for finite $r \neq 0$.

2.4.1 Solutions Near the Core of the Defect

As in [51], to analytically study the dynamics near the core of the vortex, r = 0, we make the ansatz:

$$f(t,r) \cong p(t)r, \quad g(t,r) \cong q(t)r$$
 (2.23)



Figure 2.1: Numerical solution for p(t) and q(t) of (2.24-2.25) showing the finite-time slope divergence. 1/p and 1/q are also shown, demonstrating the linearity of these functions near the critical time.

for small r. Dropping terms which are subleading in r yields a set of coupled ODEs for p(t) and q(t)

$$pq^{3}\ddot{q} - \ddot{p}q^{4} + 2\dot{p}q^{3}\dot{q} - 2pq^{2}\dot{q}^{2} + \frac{4}{\alpha'^{2}}p^{3}q^{2} + \frac{4}{\alpha'^{2}}pq^{2} + \frac{2}{a^{2}}pq^{4} = 0$$
(2.24)

from (2.21) and

$$-p\ddot{p}q^{3} + p^{2}q^{2}\ddot{q} + q^{2}\ddot{q} - 2q\dot{q}^{2} + 2p\dot{p}q^{2}\dot{q} - 2p^{2}q\dot{q}^{2} + \frac{4}{\alpha'^{2}}p^{4}q + \frac{8}{\alpha'^{2}}p^{2}q + \frac{4}{\alpha'^{2}}q + \frac{2}{a^{2}}p^{2}q^{3} = 0$$
(2.25)

from (2.22). Although (2.24,2.25) are difficult to solve analytically it is straightforward to verify that in the regime where \dot{p} , \dot{q} and higher derivatives are large compared to p, q there exists an approximate solution to (2.24-2.25) where both p(t) and q(t) are divergent in finite time t_c :

$$p(t) = \frac{p_0}{t_c - t}, \quad q(t) = \frac{q_0}{t_c - t}$$
 (2.26)

in analogy with the kink solution of [51]. Numerical solutions to (2.24-2.25) agree with this prediction, as shown in figure 2.1.

2.4.2 Solutions Away From the Core of the Defect

We are interested in solutions where $\partial_m g \to 0$ as $g \to 1$ and where $f(t,r) \to \infty$ as $t \to \infty$. Since, as we have seen above, g'(t, r = 0) is diverging in finite time, therefore g must be increasing to unity for $r \neq 0$ so that at late times we expect g(t,r) to resemble a step function. Thus to study the dynamics away from the core of the defect we begin with the ansatz:

$$g(t,r) = 1 - \varepsilon \sigma(t,r) \tag{2.27}$$

and work only to leading order in ε . The leading order contribution to (2.21) decouples completely from $\sigma(t, r)$

$$\begin{bmatrix} \ddot{f}(1+f'^2) - f''(1-\dot{f}^2) - 2\dot{f}f'\dot{f}' - \frac{2f}{a^2}(1-\dot{f}^2+f'^2) \end{bmatrix} r - \begin{bmatrix} 1-\dot{f}^2+f'^2 \end{bmatrix} f' = 0.$$
(2.28)

We can consistently find solutions of (2.28) by taking f to be a solution of the eikonal equation $1 - \dot{f}^2 + f'^2 = 0$. Subject to this constraint the second term in the square braces in (2.28) vanishes trivially. The constraint that the first term in the square braces in (2.28) vanishes is exactly the same as the equation of motion one would derive from $\mathcal{L} = -V(f)\sqrt{1 + \partial_m f \partial^m f}$, the Born-Infeld Lagrangian. The eikonal equation yields the Born-Infeld equation as a differential consequence, which is not surprising since this amounts to minimizing the action by setting $\mathcal{L} = 0$. Thus the PDE (2.28) is automatically satisfied when f is a solution of the eikonal equation. We find, then, that when $g(t, r) \cong 1$ the tachyon field must obey

$$\partial_m f \partial^m f + 1 = 0 \tag{2.29}$$

as in section 2.2.

The ansatz (2.27) yields no simplification of (2.22); however, we may solve for $\sigma(t, r)$ given (2.29) based on more fundamental constraints. The argument of the square root in (2.19) must be nonnegative to ensure reality of the Lagrangian. Thus for f(t, r) given by (2.29) the requirement that the Lagrangian be real translates to

$$\partial_m \sigma \partial^m \sigma - (\epsilon^{mn} \partial_m f \partial_n \sigma)^2 \ge 0$$

or, with f given by (2.29),

$$-\left(\dot{\sigma}\sqrt{1+f'^{2}}-\sigma'f'\right)^{2} \ge 0.$$
 (2.30)

For real fields it is clear that (2.30) can only be solved when the equality is taken. It is worth noting that since the Lagrangian vanishes when the equality is taken in (2.30), this constraint ensures that the full equations of motion are satisfied.

As in section 2.2 we can avoid the difficulties of caustic formation in the general solutions of (2.29) found in [68] by taking the Cauchy data to be linear and using the one-parameter family of solutions

$$f(t,r) = \alpha t + \sqrt{\alpha^2 - 1} r. \qquad (2.31)$$

Reality of the Lagrangian requires

$$\alpha \dot{\sigma} - \sqrt{\alpha^2 - 1} \, \sigma' = 0.$$

This PDE is separable and we find the solution

$$g(t,r) = 1 - \varepsilon \exp\left(-\frac{r}{R}\right) \exp\left(-\frac{t}{R}\sqrt{\frac{\alpha^2 - 1}{\alpha^2}}\right)$$
(2.32)

where R is a separation constant. Note that the solution (2.32) becomes static in the limit $\alpha^2 \to 1$, the homogeneous rolling tachyon. In fact, for $\alpha^2 = 1$ any function $\sigma(t,r) = \sigma(r)$ satisfying the necessary boundary condition $\sigma(r \to \infty) \to 0$ will generate a solution. We will ultimately be interested in this limit.⁴

It is noteworthy that taking g(t,r) close to unity (2.27) ultimately translates into the requirement that $\det(M)$ must vanish. The solutions (2.31-2.32) should be thought of as late-time asymptotics where $V(f) \to 0$ since $f \to \infty$. In this limit $\Sigma(t,r)$ defined in (2.20) has the indeterminant form $\frac{0}{0}$ as in section 2.2. The quantity Σ is of some interest for two reasons. First, it parametrizes the manner in which we take the limits $V(f) \to 0$ and $\det(A) \to 0$ as we approach the vacuum state. Second,

⁴The exact functional form of σ at large r and late times turns out to be of little importance to the ensuing analysis.

the form of Σ for $r \neq 0$ will determine the form of the energy-momentum tensor in that regime, as in the case of the kink. Following the discussion in section 2.2 we will use energy-momentum conservation to place constraints on the aymptotic form of Σ .

2.5 Stress-Energy Tensor

In this section we demonstrate that the vortex solutions found above give rise to the formation of a singularity in the stress-energy tensor of the tachyon field, which corresponds exactly with that of a codimension-two D-brane in the final state, whose tension has the value expected from string theory. We also derive the bulk stressenergy tensor for the leftover tachyon matter, which continues to roll even after the formation of the D-brane.

The stress-energy tensor for the action (2.14) is

$$T^{MN} = -\frac{V(T, Y^{I}_{(1)} - Y^{I}_{(2)})}{r} \left[\sqrt{-\det M^{(1)}} (M^{-1}_{(1)})^{MN}_{S} + \sqrt{-\det M^{(2)}} (M^{-1}_{(2)})^{MN}_{S} \right]$$
(2.33)

where the subscript S denotes the symmetric part of the matrix, *i.e.*, $(M_{(i)}^{-1})_S^{MN} = \frac{1}{2}[(M_{(i)}^{-1})^{MN} + (M_{(i)}^{-1})^{MN}]$. The components of T^{MN} parallel to the vortex simplify to

$$T^{\hat{\mu}\hat{\nu}} = -\frac{2}{r}V(f)\sqrt{-\det(M)}\delta^{\hat{\mu}\hat{\nu}}.$$
(2.34)

For the components involving t, r, θ , it is useful to rewrite $T^{\mu\nu}$ in terms of Σ and the symmetrized cofactor matrix of $M^{(i)}_{\mu\nu}$, which we define as $C^{\mu\nu}_{(i)}$.⁵ Since $C^{\mu\nu}_{(1)} = C^{\mu\nu}_{(2)}$ we drop the brane index on the cofactor matrices. In the t, r, θ directions, then, we have

$$T^{\mu\nu} = \frac{2\Sigma}{r} C^{\mu\nu}.$$
 (2.35)

⁵That is to say $C_{(i)}^{\mu\nu} = \det(M)(M_{(i)}^{-1})_S^{\mu\nu}$.

The nonzero components of the cofactor matrix for the Lagrangian (2.19) are

$$\begin{split} C^{tt} &= \left[r^2 + f^2(1-g)^2\right](1+f'^2) + \frac{\alpha'^2}{4}g'^2, \\ C^{tr} &= -\left[r^2 + f^2(1-g)^2\right]\dot{f}f' - \frac{\alpha'^2}{4}\dot{g}g', \\ C^{rr} &= -\left[r^2 + f^2(1-g)^2\right](1-\dot{f}^2) + \frac{\alpha'^2}{4}\dot{g}^2, \\ C^{\theta\theta} &= -(1-\dot{f}^2 + f'^2). \end{split}$$

2.5.1 Normalization of the Potential

In [50] Sen finds that the action (2.14) provides a good effective description of the tachyon on the brane-antibrane system provided the potential V(T) is chosen to satisfy the normalization constraint

$$\tau_{p-2} = 4\pi \int_0^\infty V(z) \sqrt{z^2 (1 - \hat{G}(z))^2 + \frac{\alpha'^2}{4} (\hat{G}'(z))^2} \, dz \tag{2.36}$$

where $\tau_{p-2} = (2\pi)^2 \alpha' \tau_p$ is the tension of a (p-2)-brane, $\hat{G}(z) = G(F^{-1}(z))$, and $\{f(r) = F(br), g(r) = G(br)\}$ are the static soliton solutions to be understood in the limit that $b \to \infty$. This constraint is necessary to ensure that the vortex solution has the correct tension to be interpreted as a D(p-2)-brane. In the time-dependent case there is some ambiguity as to how to interpret (2.36) since this statement appears to depend on the functional form of the solutions, which would make the right-hand-side apparently time-dependent.⁶

But physically, it makes sense to impose (2.36) at $t \ge t_c$ since t_c is the time by which the brane has actually formed, and in the limit $t \to t_c$ the time-dependent solutions should coincide with the soliton solutions in the neighbourhood of r = 0. In the time-dependent case, it is $(t_c - t)^{-1}$ which tends to infinity (as $t \to t_c$) and plays the role of b. Although the exact functional forms of G(z), F(z) are not known for all z, we can infer from (2.26) that $G(z) \cong q_0 z$ and $F(z) \cong p_0 z$ near z = 0.

⁶Arguments are presented in [50] for why (2.36) is in fact only a constraint on V(T) and not on the solutions T(x) themselves.

Furthermore, we know that $G(z) \to 1$ for sufficiently large z by construction. We also have F(z=0) = 0 and $F(z \neq 0) \neq 0$ so that $F^{-1}(0) = 0$ and $F^{-1}(z \neq 0) \neq 0$.

Let us consider the two terms under the square root in (2.36). The first term, $z^2(1-\hat{G}(z))^2$, is small near z = 0 due to the overall multiplicative factor of z^2 . On the other hand, at large z this term is also small since $F^{-1}(z)$ is large and hence $\hat{G}(z) = G(F^{-1}(z)) \cong 1$. We conclude that the derivative term under the root in (2.36) dominates. At small z we have $F^{-1}(z) \cong p_0^{-1}z$ and thus $\hat{G}(z) \cong q_0 F^{-1}(z) \cong \frac{q_0}{p_0}z$ so that $\hat{G}'(z) \cong \frac{q_0}{p_0}$. For simplicity we take $\hat{G}'(z) \cong \frac{q_0}{p_0}$ for all z since this expression is multiplied by V(z) which tends to zero quickly for large z. We find then that

$$\frac{q_0}{p_0} \int_{-\infty}^{\infty} e^{-z^2/a^2} dz = 4\pi.$$
(2.37)

The normalization (2.37) is equivalent to $a = 4\sqrt{\pi \frac{p_0}{q_0}}$. We shall see later on that the relation $a = 4\sqrt{\pi \frac{p_0}{q_0}}$ may be equivalently viewed as a constraint on the arbitrary function $\Sigma(t, r)$.

2.5.2 Stress-Energy Tensor at r = 0

At $r \cong 0$ and $t \to t_c$ the solutions (2.26) are valid and the Hamiltonian is

$$T^{00} = \frac{2\Sigma}{r} \left(\left[r^2 + f^2 (1-g)^2 \right] (1+f'^2) + \frac{\alpha'^2}{4} g'^2 \right)$$
$$\cong \frac{\tau_p}{r} \frac{q_0 \alpha'}{t_c - t} \exp\left(-\frac{p_0^2 r^2}{a^2 (t_c - t)^2} \right)$$

to leading order in r. Then, using $a = 4\sqrt{\pi \frac{p_0}{q_0}}$,

$$\lim_{t \to t_c} T^{00} = 4\pi \tau_{p-2} \frac{\delta(r)}{r}$$
$$= \tau_{p-2} \,\delta(r\cos\theta) \,\delta(r\sin\theta)$$

and the components parallel to the vortex are

$$T^{\hat{\mu}\hat{\nu}} = -\frac{2}{r} \delta^{\hat{\mu}\hat{\nu}} V(f) \sqrt{-\det(M)}$$
$$\rightarrow -\tau_{p-2} \delta^{\hat{\mu}\hat{\nu}} \delta(r\cos\theta) \,\delta(r\sin\theta)$$

The remaining components, T^{11} and T^{01} , are vanishing at r = 0. The angular component $T^{22} = T^{\theta\theta}$ contains a delta function at r = 0; however this is an artifact of θ being a bad coordinate at r = 0; it can be seen that $T_{\theta\theta} = 0$, and going to Cartesian coordinates confirms that the $T_{\mu\nu} = 0$ for the transverse coordinates, as should be the case for a D-brane. This result is the same as in the static case [50].

2.5.3 Stress-Energy Tensor at r > 0

For r > 0 at late times the solutions (2.31,2.32) are valid. For simplicity we take $\alpha^2 = 1$ and work only to leading order in ε . The Hamiltonian is

$$T^{00} \cong 2r\Sigma(t,r)$$

and the remaining components of T^{MN} vanish at late times for r > 0. Notice that conservation of energy $\partial_M T^{MN} = 0$ at large r forces

$$\Sigma(t,r) = \Sigma(r).$$

That is, Σ is an arbitrary function of r as in section 2.2.

To summarize, we find that in the limit of condensation the energy momentum tensor is identical to that of a D(p-2)-brane

$$T^{00} = \tau_{p-2} \,\delta(r\cos\theta) \,\delta(r\sin\theta) + 2r\Sigma(r)$$

$$T^{11} = T^{22} = 0$$

$$T^{\hat{\mu}\hat{\nu}} = -\delta^{\hat{\mu}\hat{\nu}} \tau_{p-2} \delta(r\cos\theta) \,\delta(r\sin\theta)$$
(2.38)

with all off-diagonal components vanishing. The extra bulk energy density $2r\Sigma(r)$ is similar to the result in [71] and section 2.2 and corresponds to tachyon matter rolling toward $T \to \infty$ in the bulk.

2.5.4 Conservation of Energy

We can constrain $\Sigma(r)$ using conservation of energy. Initially the system consists of two D*p*-branes with energy density $2\tau_p V_2$ in the (p-2)-dimensional subspace spanned by $\{x^{\hat{\mu}}\}$, where V_2 is the volume of the 2-dimensional subspace spanned by $\{r, \theta\}$. At late times, after the codimension 2 brane and its antibrane have formed, the energy density in the (p-2)-dimensional space is given by the sum of the D(p-2)brane tensions, $2\tau_{p-2}$, and the energy density due to tachyon matter $4\pi \int r^2 \Sigma(r) dr$. Conservation of energy thus implies

$$2\tau_p V_2 = 2\tau_{p-2} + 4\pi \int dr \, r^2 \Sigma(r)$$

(2V_2 - 2(2\pi)^2 \alpha') \tau_p = 4\pi \int dr \, r^2 \Sigma(r).

Since $\Sigma(r)$ is arbitrary we can take

$$\Sigma(r) = \frac{\tau_p}{r} + \frac{\Sigma(r)}{4\pi r^2} \tag{2.39}$$

where $\Sigma(r)$ satisfies the constraint

$$\int dr \,\tilde{\Sigma}(r) = -8\pi^2 \alpha' \tau_p. \tag{2.40}$$

The conditions (2.39),(2.40) are equivalent to (2.37) and may be thought of as an alternative to the normalization (2.36).

2.6 Inclusion of Massless Gauge Fields

We will now restrict ourselves to a (5+1)-dimensional spacetime with $\{M, N\} = \{0, 1, \dots, 5\}$, $\{\mu, \nu\} = \{0, 1, 2\}$ and $\{\hat{\mu}, \hat{\nu}\} = \{3, 4, 5\}$. There are two gauge fields in the problem: $A_{(1)}^M$ and $A_{(2)}^M$, or equivalently $A_+^M = A_{(1)}^M + A_{(2)}^M$ and $A_-^M = A_{(1)}^M - A_{(2)}^M$, which have different couplings to the tachyon. We have already shown that A_-^μ is the field which condenses in the vortex, hence its associated gauge symmetry is spontaneously broken. For reheating it is thus $A_+^{\hat{\mu}}$ which most closely resembles the Standard Model photon. We will ignore fluctuations of the heavy fields $A_{-}^{\hat{\mu}}$, A_{+}^{μ} , and A_{-}^{μ} , keeping only the background solution for A_-^{μ} (which was given in section 2.4), and the fluctuations of the photon $A_{+}^{\hat{\mu}}$. This leads to considerable simplification since it ensures that $D_{\hat{\mu}}T = 0$. To compute the production of photons in the time-dependent background, we want to expand the action (2.14) to quadratic order in $A_+^{\hat{\mu}}$.

The matrix $M_{MN}^{(i)}$ of eq. (2.15) can be written in block diagonal form as

$$\begin{bmatrix} M_{MN}^{(i)} \end{bmatrix} = \begin{bmatrix} \begin{bmatrix} \mathcal{V}_{\mu\nu}^{(i)} \end{bmatrix} & \begin{bmatrix} S_{\mu\hat{\nu}}^{(i)} \end{bmatrix} \\ -\begin{bmatrix} S_{\mu\hat{\nu}}^{(i)} \end{bmatrix}^T & \begin{bmatrix} \delta_{\hat{\mu}\hat{\nu}} + \alpha' F_{\hat{\mu}\hat{\nu}}^{(i)} \end{bmatrix}$$
(2.41)

where $\mathcal{V}_{\mu\nu}^{(i)} = M_{\mu\nu}^{(i)}$ is the contribution from the vortex background given in (2.18), $S_{\mu\nu}^{(i)} = \alpha' \partial_{\mu} A_{\nu}^{(i)}$ is the contribution from $\{t, r, \theta\}$ derivatives of $A_{\hat{\mu}}^{(i)}$, and $F_{\hat{\mu}\hat{\nu}}^{(i)}$ is the field strength tensor for $A_{\hat{\mu}}^{(i)}$. Using a well-known identity for determinants we can write

$$\det(M^{(i)}) = \det \mathcal{V}^{(i)} \det \left(\mathbf{1} + \alpha' F^{(i)} + S^T_{(i)} \mathcal{V}^{-1}_{(i)} S_{(i)} \right).$$
(2.42)

Expanding det $(1 + \alpha' F^{(i)} + S^T_{(i)} \mathcal{V}^{-1}_{(i)} S_{(i)})$ to quadratic order in $A^{(i)}_{\hat{\mu}}$, the action (2.14) becomes

$$S \cong -\alpha'^{2} \frac{1}{4} \int d^{p+1} x V(f) \sqrt{-\det \mathcal{V}} \left((\mathcal{V}^{-1})^{\mu\nu}_{(S)} \partial_{\mu} A^{+}_{\hat{\mu}} \partial_{\nu} A^{\hat{\mu}}_{+} + \partial_{\hat{\mu}} A^{+}_{\hat{\nu}} \partial^{\hat{\mu}} A^{\hat{\nu}}_{+} \right)$$

$$(2.43)$$

where we have chosen the gauge $\partial_{\hat{\mu}} A^{\hat{\mu}}_{+} = 0$ and disregarded the piece which does not depend on $A^{\hat{\mu}}_{+}$. We omit the brane index on $\mathcal{V}_{(i)}$ since the determinant and the symmetric part of $\mathcal{V}^{-1}_{(i)}$ are equal for both $i = \{1, 2\}$. Defining an effective metric G_{MN} by

$$[G_{MN}] = \begin{bmatrix} \begin{bmatrix} \mathcal{V}_{\mu\nu}^{(S)} \end{bmatrix} & [0] \\ [0] & [\delta_{\hat{\mu}\hat{\nu}}] \end{bmatrix}$$
(2.44)

the action (2.43) may be written as

$$S = -\frac{{\alpha'}^2}{4} \int V(f) \sqrt{-G} G^{MN} \delta^{\hat{\mu}\hat{\nu}} \partial_M A^+_{\hat{\mu}} \partial_N A^+_{\hat{\nu}} d^{p+1} x.$$
(2.45)

From (2.45) one sees that the fluctuations of the photon behave like a collection of massless scalar fields propagating in a nonflat spacetime described by the metric G_{MN} , with a position- and time-dependent gauge coupling given by $g^2 = 1/V(f(t,r))$.

To get an intuitive sense for the behavior of the action (2.45) we note that the stress-energy tensor derived in section 2.5 can be written as

$$T^{MN} = -\frac{2}{r}V(f)\sqrt{-G}G^{MN}.$$

In the limit of condensation, T^{MN} is given by (2.38), so that once the brane has formed the action (2.45) reduces to a description of gauge fields propagating in a (3+1)-dimensional Minkowski space, with an additional component which couples the gauge fields to the tachyon matter density in the bulk. In other words, the effective metric G_{MN} starts off being smooth throughout the bulk, but within the time t_c , its support collapses to become a delta function $\delta^{(2)}(\vec{x})$ in the relevant extra dimensions $\{r, \theta\}$.

The equations of motion resulting from the effective action (2.45) are difficult to solve analytically since the effective metric G_{MN} depends nontrivially on both r and tand is nondiagonal in the subspace of $\{t, r\}$. For this reason we would like to propose a simplified model of the condensation which captures the essential features of the action (2.45). We have derived solutions for the vortex background valid at small $r, r \leq (t_c - t)$, and at large $r, r \geq (t_c - t)$. Similarly, the energy momentum tensor we have derived corresponding to these solutions has very different behavior in the $r \leq (t_c - t)$ and $r > (t_c - t)$ regions of the spacetime.

For $r \leq (t_c - t)$ the energy momentum tensor contracts to a delta function centered at r = 0, with $(t_c - t)$ playing the role of the small parameter which regularizes the delta function. That is to say,

$$rT^{MN} \cong \frac{1}{t_c - t} \exp\left(-\frac{p_0^2 r^2}{a^2 (t_c - t)^2}\right) H^{MN}$$

at small r, where the matrix entries H^{00} , $H^{\hat{\mu}\hat{\nu}}$ are finite as $t \to t_c$ and the remaining components of H^{MN} tend to zero (near r = 0) as $t \to t_c$.

In the $r > (t_c - t)$ region, the energy momentum tensor has quite different behavior. After condensation of the defect has completed at $t = t_c$, the energy density in the bulk (r > 0) is due entirely to tachyon matter, while the part of the stress-energy which is going into the tension of the defect vanishes in this region. Hence

$$rT^{MN} \rightarrow 2\delta_0^M \delta_0^N r^2 \Sigma(r)$$

as $t \to t_c$, for $r > (t_c - t)$.

In our simplified model of the particle production due to the tachyon condensation we therefore split the energy momentum tensor into brane and bulk pieces $T^{MN} = T^{MN}_{\text{brane}} + T^{MN}_{\text{bulk}}$ where T^{MN}_{brane} contracts to a delta function as $t \to t_c$ and $T^{MN}_{\text{bulk}} \to 2\delta_0^M \delta_0^N r \Sigma(r)$ in the same limit. The action (2.45) then splits into two components $S = S_{\text{bulk}} + S_{\text{brane}}$. We expect most of the particle production to occur near
the end of the condensation, when the background tachyon field is becoming singular near the vortex, so the best approximations for the simplified gauge field action are those which describe the exact expression most accurately near $t = t_c$:

$$S_{brane} \propto -\int \frac{1}{t_c - t} \exp\left(-\frac{p_0^2 r^2}{a^2 (t_c - t)^2}\right) \\ H^{MN} \delta^{\hat{\mu}\hat{\nu}} \partial_M A^+_{\hat{\mu}} \partial_N A^+_{\hat{\nu}} dt dr d\theta dx^{\delta}$$

and

$$S_{bulk} \propto -\int r^2 \Sigma(r) \left(-\delta^{\hat{\mu}\hat{\nu}} \dot{A}^+_{\hat{\mu}} \dot{A}^+_{\hat{\nu}} \right) \, dt dr d\theta dx^{\hat{\alpha}}$$

At earlier times the coefficient of the bulk part of the Lagrangian would have time dependence, and the bulk Lagrangian would contain contributions from all the derivative of the gauge field, but this form is valid close to t_c .

Finally we argue that the bulk part of the action can be ignored. To this end, let us change to coordinates which are comoving with the contraction of the vortex core:

$$\tilde{r} = \frac{r}{t_c - t}, \quad \tilde{t} = t_c - t.$$
(2.46)

In terms of these coordinates the "small r" solutions are valid for $\tilde{r} \leq 1$ and the "large r" solutions are valid for $\tilde{r} > 1$. The Jacobian of this transformation is $-\tilde{t}$ so that

$$S_{\text{brane}} \propto -\int \exp\left(-\frac{p_0^2 \tilde{r}^2}{a^2}
ight) H^{MN} \delta^{\hat{\mu}\hat{
u}} \partial_M A^+_{\hat{\mu}} \partial_N A^+_{\hat{
u}} d\tilde{t} d\tilde{r} d heta dx^{\hat{lpha}}.$$

To lowest order in \tilde{r}^2 the matrix entries H^{MN} in terms of these new coordinates are all constant. Consider now the piece of the action which couples to the tachyon matter density in the bulk, written in terms of these new coordinates:

$$S_{\text{bulk}} \propto -\int \tilde{r}^2 \tilde{t}^2 \Sigma(\tilde{r}\tilde{t}) \left(-\delta^{\hat{\mu}\hat{\nu}}\dot{A}^+_{\hat{\mu}}\dot{A}^+_{\hat{\nu}}\right) d\tilde{t}d\tilde{r}d\theta dx^{\hat{\alpha}}.$$

Since $\Sigma(z)$ is not singular at z = 0, the bulk piece of the action become negligible near the end of the contraction $\tilde{t} \to 0$. This is a consequence of the fact that as the condensation proceeds the gauge field is confined to the descendant brane.

2.7 Simplified Model of Reheating

In the previous section we argued that the gauge field couples most strongly to the part of the tachyon background which is collapsing to form the defect. This closely resembles a gauge theory defined on a manifold in which a two-dimensional subspace which is shrinking with time. As a simplified model of the interaction we thus consider a massless spin-1 field

$$S = -\frac{1}{4} \int \sqrt{-g} g^{MA} g^{NB} F_{MN} F_{AB} d^{p+1} x$$

propagating in a FRW-like background

$$g_{MN}dx^{M}dx^{N} = -dt^{2} + R(t)^{2} \left(d\tilde{r}^{2} + \tilde{r}^{2}d\theta^{2}\right) + \delta_{\hat{\mu}\hat{\nu}}dx^{\hat{\mu}}dx^{\hat{\nu}}.$$
 (2.47)

The coordinate \tilde{r} in (2.47) is fixed with the expansion and is thus corresponds to \tilde{r} defined in (2.46). However for simplicity of notation we will drop the tilde and write r instead of \tilde{r} in the remainder of the chapter. We take r to be dimensionless while $t, x^{\hat{\mu}}$ and R have dimensions of length.

If we restrict ourselves to configurations with $A^{\mu} = 0$, $A^{\hat{\mu}} \neq 0^7$ and impose the gauge condition $\partial_{\hat{\mu}}A^{\hat{\mu}} = 0$ then the action simplifies to

$$S = -\frac{1}{2} \int \sqrt{-g} g^{MN} \delta^{\hat{\mu}\hat{\nu}} \partial_M A_{\hat{\mu}} \partial_N A_{\hat{\nu}} d^{p+1} x \qquad (2.48)$$

Notice that for the metric (2.47) and the gauge field configuration $A^{\mu} = 0$ we have chosen, one has $\nabla_M A_{\hat{\mu}} = \partial_M A_{\hat{\mu}}$ and $R_{MN} A^M A^N = R_{\hat{\mu}\hat{\nu}} A^{\hat{\mu}} A^{\hat{\nu}} = 0$.

We will impose homogeneous boundary conditions at r = 1 and take the scale factor in (2.47) to be

$$R(t) = \begin{cases} R_0 & \text{if } t < 0; \\ R_0 - \eta t & \text{if } 0 \le t \le t_c; \\ R_0 - \eta t_c = \epsilon & \text{if } t > t_c. \end{cases}$$
(2.49)

⁷This restriction will only underestimate the reheating. Since we want to show that the reheating can be efficient, this approximation will not weaken the ensuing argument.

where R_0 represents the initial radial size of the extra dimensions. This approximation for the time-dependence of the vortex core is the simplest form which has the same qualitative behavior as the true background, while still allowing us to solve analytically for the gauge field wave functions in the background. A shortcoming of this approximation is that \dot{R} is discontinuous at the interfaces, which leads to an ultraviolet divergence in the production of gauge bosons. The behavior of the actual R(t) is smooth, and must yield a finite amount of particle production [59]. In addition, we will find a separate UV divergence in the particle production in the limit as the vortex core thickness goes to zero. This is presumably an artifact of the effective field theory which is not present in the full string theory, and we deal with it by introducing the cutoff ϵ on the final radius of the defect core, which we will take to be of order the string length l_s .

2.7.1 Gauge Field Solutions

The first step in computing the production of photons in the time-dependent background is to solve their equation of motion following from (2.48):

$$-2\frac{\dot{R}}{R}\dot{A}^{\hat{\rho}} - \ddot{A}^{\hat{\rho}} + \frac{A^{\hat{\rho}''}}{R^2} + \frac{A^{\hat{\rho}'}}{rR^2} + \frac{\partial^2_{\theta}A^{\hat{\rho}}}{r^2R^2} + \partial^{\hat{\mu}}\partial_{\hat{\mu}}A^{\hat{\rho}} = 0$$
(2.50)

where the dot and prime denote differentiation with respect to t and r, respectively.

Equation (2.50) separates as

$$A^{\hat{\rho}}(t,r,\theta,x^{\hat{\mu}}) = \phi(t)\varphi(r)\Theta(\theta)\chi^{\hat{\rho}}(x^{\hat{\mu}})$$
(2.51)

where

$$\partial^{\hat{\mu}}\partial_{\hat{\mu}}\chi^{\hat{\rho}} = -k^2 \chi^{\hat{\rho}}, \ \partial_{\hat{\nu}}\chi^{\hat{\nu}} = 0, \ k^2 = k^{\hat{\mu}}k_{\hat{\mu}} = \vec{k} \cdot \vec{k}$$
(2.52)

$$\partial_{\theta}^2 \Theta = -m^2 \Theta \tag{2.53}$$

$$\varphi'' + \frac{1}{r}\varphi' + \left(c^2 - \frac{m^2}{r^2}\right)\varphi = 0 \qquad (2.54)$$

$$\ddot{\phi} + 2\frac{\dot{R}}{R}\dot{\phi} + \left(\frac{c^2}{R^2} + k^2\right)\phi = 0$$
 (2.55)

and c is a separation constant.

The particular solutions of (2.52) are labeled by the momenta \vec{k} in the 3 large dimensions ($\{x^4, x^5, x^6\}$) and we take them to be normalized according to

$$\int \chi^{\hat{\mu}}_{\vec{k}} \chi^{\hat{\mu}}_{\vec{k}'} dx^4 dx^5 dx^6 = \delta_{\vec{k}\vec{k}'}.$$

The particular solution of (2.53) is a sum of sines and cosines. The odd parity and even parity modes (under $\theta \rightarrow -\theta$) do not mix with the even ones, and for simplicity, we restrict our attention to the even modes, which include the massless one. This can underestimate the effciency of the reheating by a factor of 2 at most. The solution of (2.53) is thus

$$\Theta_m(\theta) = \frac{1}{\sqrt{\pi}}\cos(m\theta).$$

These are orthogonal for different values of m, and requiring that the solution be single-valued restricts m to be an integer.

So far we have not been specific about the geometry of the two extra dimensions around which the original annihilating branes were wrapped. One simple possibility is a 2-sphere, where the descendant brane and antibrane (vortices) form at antipodal points. Since all the singular behavior of the tachyon background is localized near these points, the curvature and topology in the bulk should have little effect on particle production near the defects. To simplify the mathematics, we therefore replace either of the two hemispheres of the sphere with a flat disk, in the coordinate region $r \leq 1$. The correct boundary condition on radial eigenfunctions of the bulk Laplacian is that their derivatives vanish at r = 1, so that they are smooth at the interface where the two halves of the space are glued together.

The solution of (2.54), subject to the boundary condition $\varphi'(r=1) = 0$ and the requirement that $\varphi(r)$ be regular at the origin, is

$$\varphi_{mn}(r) = \frac{\sqrt{2}}{J_{m+1}(c_{mn})} J_m(c_{mn}r)$$
(2.56)

where c_{mn} is the *n*th zero of $J'_m(r)$. The solutions (2.56) are orthogonal for different values of *n*. The zero mode n = 1, m = 0 must be treated separately; it is the constant solution, where $c_{01} = 0$ and $\varphi_{01}(r) = 1$.

The solution of (2.55) depends on the scale factor. For t < 0 and $t > t_c$ the solutions are trivial and are given by

$$\phi_{mn}(t) = \frac{1}{R_0 \sqrt{2\omega_{mn}}} \left(a_{mn} e^{-i\omega_{mn}t} + a_{mn}^{\dagger} e^{i\omega_{mn}t} \right)$$
(2.57)

and

$$\phi_{mn}(t) = \frac{1}{\epsilon \sqrt{2\bar{\omega}_{mn}}} \left(d_{mn} e^{-i\bar{\omega}_{mn}t} + d_{mn}^{\dagger} e^{i\bar{\omega}_{mn}t} \right)$$
(2.58)

respectively. We have defined $\omega_{mn}^2 = \frac{c_{mn}^2}{R_0^2} + k^2$ and $\bar{\omega}_{mn}^2 = \frac{c_{mn}^2}{\epsilon^2} + k^2$. The mutiplicative factors in (2.57) and (2.58) are introduced for later convenience and ensure that a_{mn} and d_{mn} will be properly normalized annihilation operators in the appropriate spacetime region when the gauge field is quantized. It will be convenient for what follows to introduce phase-shifted annihilation and creation operators in the region $t > t_c$: $\bar{d}_{mn} = e^{-i\bar{\omega}_{mn}t_c}d_{mn}$ and $\bar{d}_{mn}^{\dagger} = e^{i\bar{\omega}_{mn}t_c}d_{mn}^{\dagger}$. In terms of these operators (2.58) becomes

$$\phi_{mn}(t) = \frac{1}{\epsilon \sqrt{2\bar{\omega}_{mn}}} \left(\bar{d}_{mn} e^{-i\bar{\omega}_{mn}(t-t_c)} + \bar{d}_{mn}^{\dagger} e^{i\bar{\omega}_{mn}(t-t_c)} \right).$$

We are suppressing the dependence of the annihilation/creation operators on the 3-momenta \vec{k} .

In the region $0 < t < t_c$ where the scale factor depends nontrivially on time, the solution of (2.55) is

$$\phi_{mn}(t) = \frac{1}{\sqrt{R_0 - \eta t}} \left(B_{mn} J_{p_{mn}} \left[\frac{k}{\eta} (R_0 - \eta t) \right] + C_{mn} J_{-p_{mn}} \left[\frac{k}{\eta} (R_0 - \eta t) \right] \right)$$

$$(2.59)$$

where $p_{mn} = \frac{1}{2\eta} \sqrt{\eta^2 - 4c_{mn}^2}$. Some comments are in order concerning this solution, which has different behavior for massive and massless modes.

In the massless case $c_{mn} = 0$, $p_{mn} = 1/2$ and J_p , J_{-p} are linearly independent, since p is noninteger. The constants B_{mn} and C_{mn} in (2.59) should be interpreted as independent, real-valued constants so that $\phi_{mn}(t)$ is real.

In the massive case $c_{mn} \neq 0$, $\eta^2 \leq 1$,⁸ and the order of the Bessel functions in (2.59), p_{mn} , is pure imaginary. Such Bessel functions are complex-valued and there are several options for constructing real solutions. Since $(J_{\nu}(x))^* = J_{\nu^*}(x)$ for real

x, we can impose $C_{mn} = B_{mn}^{\dagger}$. Calculating the Wronskian of J_{is} , J_{-is} verifies that for real $s \neq 0$ these two solutions are linearly independent. Since we do not need to explicitly quantize the field in the region $0 < t < t_c$, the dagger can be thought of simply as complex conjugation. The interpretation of B_{mn} and B_{mn}^{\dagger} as annihilation and creation operators is unnecessary, since there are no asymptotic states in this region.

We summarize this subsection by putting these results together to write the general solution of (2.50) as

$$A^{\hat{\rho}}(x^{M}) = \sum_{m,n} \sum_{\vec{k}} \chi^{\hat{\rho}}_{\vec{k}}(x^{\hat{\mu}}) \Theta_{m}(\theta) \phi_{mn}(t) \varphi_{mn}(r).$$

where

$$\phi_{mn}(t) = \begin{cases} \frac{1}{R_0 \sqrt{2\omega_{mn}}} \left(a_{mn} e^{-i\omega_{mn}t} + a_{mn}^{\dagger} e^{i\omega_{mn}t} \right) & \text{if } t < 0; \\ \frac{1}{\sqrt{R_0 - \eta t}} \left(B_{mn} J_{p_{mn}} \left[\frac{k}{\eta} (R_0 - \eta t) \right] + C_{mn} J_{-p_{mn}} \left[\frac{k}{\eta} (R_0 - \eta t) \right] \right) & \text{if } 0 \le t \le t_c; \\ \frac{1}{\epsilon \sqrt{2\bar{\omega}_{mn}}} \left(\bar{d}_{mn} e^{-i\bar{\omega}_{mn}(t - t_c)} + \bar{d}_{mn}^{\dagger} e^{i\bar{\omega}_{mn}(t - t_c)} \right) & \text{if } t > t_c. \end{cases}$$

2.7.2 Spectra of Produced Particles

The next step is to impose continuity of $A^{\hat{\mu}}$ and $\partial_M A^{\hat{\mu}}$ at t = 0 and $t = t_c$ in order to compute the Bogoliubov coefficients, which relate the annihilation and creation operators for $t > t_c$ (\bar{d}_{mn} and \bar{d}^{\dagger}_{mn}) to those in the region t < 0 (a_{mn} and a^{\dagger}_{mn}).⁹ Smoothness of the solutions at the interfaces is ensured by the continuity of ϕ_{mn} and $\dot{\phi}_{mn}$, since ϕ_{mn} is the only part of $A^{\hat{\rho}}$ which changes between the different spacetime regions.

⁸One might expect $\eta^2 \leq 1$ on physical grounds. In fact, for the ensuing arguments to hold one need only restrict $\eta \leq 3.7$ to ensure that p_{mn} is pure imaginary and nonzero for all massive modes.

⁹We are ultimately interested in calculating the number of d_{mn} quanta in the vacuum annihilated by a_{mn} . For this purpose it is just as good to use \bar{d}_{mn} , \bar{d}^{\dagger}_{mn} as d_{mn} , d^{\dagger}_{mn} since the phase shift cancels out of the number operator. $d^{\dagger}_{mn}d_{mn} = \bar{d}^{\dagger}_{mn}\bar{d}_{mn}$.

Massive Modes

For the massive modes continuity of $\phi_{mn}(t)$ and $\dot{\phi}_{mn}(t)$ at t = 0 implies

$$\begin{pmatrix} a_{mn} \\ a_{mn}^{\dagger} \end{pmatrix} = U \begin{pmatrix} B_{mn} \\ B_{mn}^{\dagger} \end{pmatrix} = \begin{bmatrix} u_1 & u_2 \\ u_2^{\dagger} & u_1^{\dagger} \end{bmatrix} \begin{pmatrix} B_{mn} \\ B_{mn}^{\dagger} \end{pmatrix}.$$
 (2.60)

The entries of the matrix U are given by

$$u_{1} = \left(\sqrt{\frac{R_{0}\omega_{mn}}{2}} + \frac{i\eta}{2\sqrt{2R_{0}\omega_{mn}}}\right) J_{p_{mn}}(kR_{0}/\eta)$$

$$- ik\sqrt{\frac{R_{0}}{2\omega_{mn}}} J'_{p_{mn}}(kR_{0}/\eta)$$

$$u_{2} = \left(\sqrt{\frac{R_{0}\omega_{mn}}{2}} + \frac{i\eta}{2\sqrt{2R_{0}\omega_{mn}}}\right) J_{-p_{mn}}(kR_{0}/\eta)$$

$$- ik\sqrt{\frac{R_{0}}{2\omega_{mn}}} J'_{-p_{mn}}(kR_{0}/\eta).$$

Continuity at $t = t_c$ similarly gives

$$\begin{pmatrix} \bar{d}_{mn} \\ \bar{d}^{\dagger}_{mn} \end{pmatrix} = V \begin{pmatrix} B_{mn} \\ B^{\dagger}_{mn} \end{pmatrix} = \begin{bmatrix} v_1 & v_2 \\ v_2^{\dagger} & v_1^{\dagger} \end{bmatrix} \begin{pmatrix} B_{mn} \\ B^{\dagger}_{mn} \end{pmatrix}.$$
 (2.61)

The matrix entries v_i are obtained from u_i by replacing R_0 with ϵ and ω_{mn} with $\bar{\omega}_{mn}$.

From (2.60), (2.61) we can write

$$\begin{pmatrix} \bar{d}_{mn} \\ \bar{d}_{mn}^{\dagger} \end{pmatrix} = V U^{-1} \begin{pmatrix} a_{mn} \\ a_{mn}^{\dagger} \end{pmatrix} = \begin{bmatrix} \alpha_{mn} & \beta_{mn} \\ \beta_{mn}^{*} & \alpha_{mn}^{*} \end{bmatrix} \begin{pmatrix} a_{mn} \\ a_{mn}^{\dagger} \end{pmatrix}.$$

The last equality defines the Bogoliubov coefficients. Note that there is no summation implied over any of the indices in the above expression. The indices m and n label the modes of the in- and out-states, and the summation which appears in the general definition of the Bogoliubov coefficients is not present here.

We can now determine the spectrum of Kaluza-Klein (KK) excitations of the photon which is produced in the tachyon vortex background. Observers in the future see a spectrum of massive particles in the final state given by

$$N_{M>0}^{mn}(k) = |\beta_{mn}|^2 = \frac{1}{|\det U|^2} |u_1 v_2 - u_2 v_1|^2.$$



Figure 2.2: $N_{M>0}^{mn}(k)$ versus k for increasing values of the radial quantum number n showing decreased production of heavier modes.

The determinant is

$$\det U = \det V = -\frac{2i\eta}{\pi} \sin(\pi p_{mn})$$
$$= \frac{2\eta}{\pi} \sinh\left(\frac{\pi}{2\eta}\sqrt{4c_{mn}^2 - \eta^2}\right)$$
(2.62)

which may be obtained by using the Wronskian of J_p and J_{-p} . The fact that the two determinants are equal ensures the appropriate normalization $|\alpha_{mn}|^2 - |\beta_{mn}|^2 = 1$ of the Bogoliubov coefficients. The explicit expression for $N_{M>0}^{mn}(k)$ can be obtained analytically, but it is complicated and we do not write it out here; instead we will give numerical results.

The mass of a KK mode with quantum numbers m, n is c_{mn}/ϵ , which increases with m and n. Figures 2.2-2.3 illustrate the dependence of $N_{M>0}^{mn}(k)$ on the 3-momentum k for the lightest few massive modes. The parameters of the vortex background are taken to be $R_0 = 10 l_s$, $\epsilon = l_s$ and $\eta = 1$. (The dependence of $N_{M>0}^{mn}(k)$ on R_0 , ϵ and η is essentially the same as that of the spectrum for massless modes, $N_{M=0}(k)$, which we will discuss in the next section.)



Figure 2.3: Same as fig. 2.2, but varying the angular quantum number m.

Massless Modes

The analysis for the massless modes proceeds similarly to the calculations in the preceding section, but more simply since in this case $p_{mn} = 1/2$ and $\omega_{mn} = \bar{\omega}_{mn} = k$. The Bogoliubov coefficients do not depend on the mode indices:

$$\begin{pmatrix} \bar{d}_{mn} \\ \bar{d}_{mn}^{\dagger} \end{pmatrix} = \begin{bmatrix} \alpha & \beta \\ \beta^* & \alpha^* \end{bmatrix} \begin{pmatrix} a_{mn} \\ a_{mn}^{\dagger} \end{pmatrix}.$$

The observed spectrum of particles in the final state

$$N_{M=0}(k) = |\beta|^2$$

$$\cong \frac{1}{\epsilon^2} \frac{\eta \pi R_0}{8k} \left(J_{\frac{1}{2}} \left(k R_0 / \eta \right)^2 + J_{\frac{3}{2}} (k R_0 / \eta)^2 \right)$$
(2.63)

where the second line shows the leading small- ϵ behavior. The exact result can also be obtained in closed form. Using explicit representations of $J_{n/2}$, we will be able to integrate this expression exactly to obtain the energy density of produced radiation. Figures 2.4-2.6 show plots of $N_{M=0}(k)$ as a function of k for various values of the parameters R_0 , ϵ and η .

2.7.3 Energy Density

To find the total energy density of radiation produced by the massive modes we should integrate over all k and sum over the mode indices m and n and the number



Figure 2.4: $N_{M=0}(k)$ versus k for different values of ϵ . $R_0 = 10 l_s$ and $\eta = 1$.



Figure 2.5: $N_{M=0}(k)$ versus k for different values of R_0 . $\epsilon = l_s$ and $\eta = 1$.



Figure 2.6: $N_{M=0}(k)$ versus k for different values of η . $R_0 = 10 l_s$ and $\epsilon = l_s$.

of polarizations (3 for massive vector bosons):

$$\rho_{M>0} = 3\sum_{m,n} \int \frac{d^3k}{(2\pi)^3} \sqrt{k^2 + \frac{c_{mn}^2}{\epsilon^2}} N_{M=0}^{mn}(k).$$
(2.64)

For the massless modes we only integrate over k and sum over the 2 polarizations, since $N_{M>0}(k)$ has no dependence on the mode indices:

$$\rho_{M=0} = 2 \int \frac{d^3k}{(2\pi)^3} k N_{M=0}(k).$$
(2.65)

As anticipated above, the integrals in (2.64), (2.65) are not convergent due to the fact that the simplified tachyon background (2.49) has discontinuous time derivatives at t = 0 and $t = t_c$. This has the consequence that the spectrum N(k) decreases like k^{-2} for large k, which is too slow for convergence. In reality *n*th derivatives of R(t) do not exceed $O(1/l_s^n)$, so the production of modes with $k > l_s^{-1}$ should be exponentially suppressed. We therefore introduce a UV cutoff, $k_{\max} = \Lambda \sim l_s^{-1}$. Moreover N(k) is divergent as the vortex radius $\epsilon \to 0$, so ϵ is also presumably limited by the string scale, $\epsilon \sim l_s$.

In the final state, the heavy KK modes have mass given by c_{mn}/ϵ , where the c_{mn} 's are order unity and larger. These are near the cutoff, so their contributions are of the same order as other UV contributions which we are omitting. For consistency we should thus neglect the massive states' contribution to the total energy density

on the vortex. Again, this underestimates the efficiency of particle production and makes our estimates conservative. Henceforth we will refer to $\rho_{M=0}$ as simply ρ .

Although the integral (2.65) can be performed analytically, the resulting expression for ρ is cumbersome. Rather than write it out explicitly we will discuss some noteworthy features, plot ρ with respect to the parameters of the model and present some useful simplifications of the complete expression in various limits.

Figures 2.7-2.8 show the dependence of ρ on the initial size of the brane. For R_0 greater than a few times l_s , the energy density is relatively insensitive to changes in R_0 . We can therefore reduce the dimensionality of the parameter space by simply assuming that the extra dimensions are somewhat larger than l_s . In fact in the limit of large R_0 , the energy density takes the very simple form

$$\lim_{R_0 \to \infty} \rho = \frac{\eta^2}{8\pi^2} \frac{\Lambda^2}{\epsilon^2}$$
(2.66)

which is the main result of this section. We recall that η , which parametrizes the speed at which the vortex forms, is predicted from eq. (2.46) to be $\eta = 1$.

As a check on our calculations, we have also considered the limit as $R_0 \rightarrow \epsilon$, which corresponds to a static background, with no vortex condensation. As expected, the energy density of produced particles goes to zero,

$$\rho \cong \frac{\Lambda^4}{4\pi^2 \epsilon^2} (R_0 - \epsilon)^2 - \frac{\Lambda^4}{4\pi^2 \epsilon^3} (R_0 - \epsilon)^3 + \dots$$

as $R_0 \to \epsilon$.

2.7.4 Efficiency of Reheating

To quantify the efficiency of the reheating we need to determine how much energy is available to produce the photons on the final-state 3-brane. Initially the system consisted of a D5-brane plus antibrane, whose 3D energy density was given by $2\tau_5V_2$, where τ_5 is the tension of a D5-brane and V_2 is the volume of the compact 2-space $\{r, \theta\}$ wrapped by the branes. The final state consists of a D3-brane/antibrane with total tension $2\tau_3$. Until now we considered just half of the 2-sphere and focused on a single vortex located at r = 0. Conservation of Ramond-Ramond charge requires the



Figure 2.7: ρ versus compactification radius R_0 for different values of $\epsilon = \Lambda^{-1}$ with $\eta = 1$. Dotted line is the critical density for efficient reheating ρ_c , (2.67).



Figure 2.8: Same as figure 2.7 but for different values of vortex collapse rate η and with $\epsilon = \Lambda^{-1} = 0.7 l_s$.

second vortex, which we place at the south pole of the sphere to preserve azimuthal symmetry. These two defects are identical and are matched at the equator of the sphere. The vortex at the south pole represents the D3-antibrane.

Since reheating on each final-state brane should be equally efficient, the 3D energy density available for reheating on one them, which we call the critical energy density ρ_c , is half the difference between the initial and final tensions of the branes and antibranes:

$$\rho_c \equiv \tau_5 V_2 - \tau_3 = \tau_3 \left(\frac{V_2}{4\pi^2 \alpha'} - 1\right)$$

where we have used the recursion relation $2\pi\sqrt{\alpha'}\tau_p = \tau_{p-1}$. The tension of a D3-brane is given by [78]

$$\tau_3 = \frac{1}{g_s} \frac{1}{(2\pi)^3 \alpha'^2}$$

The string coupling g_s , in 5+1 dimensions of which two are compact, is determined by the gauge coupling evaluated at the string scale, $\alpha(M_s)$ [54]:

$$g_s = \frac{V_2}{2\pi^2 \alpha'} \alpha(M_s).$$

Thus we have

$$l_s^4 \rho_c = \frac{1}{16\pi^3 \alpha(M_s)} \left(1 - \frac{4\pi^2 l_s^2}{V_2} \right).$$
 (2.67)

In the regime where $V_2 \sim 2\pi R_0^2 \gg l_s^2$, which was where we could most easily quantify the particle production, the second term in parentheses can be neglected, and in any case it would be unimportant for a rough estimate unless it accidentally canceled the first term (1) to high accuracy. Hence we drop this term and take $l_s^4 \rho_c \cong (16\pi^3 \alpha(M_s))^{-1}$.

The energy density ρ_c is the critical value at which the conversion into radiation would be 100% efficient. In our analysis we take $\alpha(M_s) \cong \frac{1}{25}$ [54] which gives $\rho_c l_s^4 =$ 0.05. The critical energy density is shown as a dashed horizontal line in figures 2.7 and 2.8. We see that the criterion for efficient reheating can be achieved for moderate values of the parameters. We only need for the length-scale cutoffs $1/\Lambda$ and ϵ to be somewhat smaller than the string scale, while the size of the extra dimensions should exceed a few times l_s . Using (2.66), we can write the criterion for efficient reheating as

$$\sqrt{\frac{\epsilon}{\eta\Lambda}} \lesssim \left(2\pi\alpha(M_s)\right)^{1/4} l_s \cong 0.7 \, l_s \tag{2.68}$$

We have not taken into account the back-reaction of the particle production on the tachyon background, which is why our calculation allows for more reheating than is energetically possible. The back reaction will suppress somewhat the actual efficiency of reheating, but we don't expect a dramatic reduction. Given that we have been conservative in our estimates, such as ignoring the contributions from produced KK photons which will decay into massless photons, our result makes it plausible that a large fraction of the original energy can be converted into visible radiation.

2.8 Conclusions

We have argued for the possibility that our visible universe might be a codimensiontwo brane left over from annihilation of a D5-brane/antibrane pair at the end of inflation. In this picture, reheating is due to production of standard model particles (e.g. photons) on the final branes, driven by their couplings to the tachyon field which encodes the instability of the initial state as well as the vortex which represents the final brane. We find that reheating can be efficient, in the sense that a sizable fraction of the energy available from the unstable vacuum can be converted into visible radiation, and not just gravitons.

The efficiency of reheating is greatest if the radius of compactification of the extra dimensions is larger than 2-3 times the string length l_s . The efficiency also depends on phenomenological parameters we had to introduce by hand in order to cut off ultraviolet divergences in the calculated particle production rate: namely ϵ , a nonvanishing radius for the final brane, and Λ , an explicit cutoff on the momentum of the photons produced. The latter must be introduced to correct for discontinous time derivatives in our simplified model of the background tachyon condensate; the actual behavior of the condensate corresponds to a cutoff of order $\Lambda \sim 1/l_s$. It is less obvious why the effective field theory treatment should give divergent results as the

thickness of the final brane (2ϵ) goes to zero, but it seems clear that a fully stringtheoretic computation would give no such divergence, and therefore it is reasonable to cut off the field-theory divergence at $\epsilon \sim l_s$. Given only these mild assumptions, our estimates (2.66,2.67) predict that the fraction of the available energy which is converted into visible radiation is $\rho/\rho_c \cong \pi\alpha(M_s) \cong 0.25$. This simple estimate counts only photons; in a more realistic calculation, it would be enhanced by the number of light degrees of freedom which couple to the tachyon, which could be much greater 1. Moreover it could also be enhanced by the production of massive KK modes, which correspond to string excitations when the vortex has formed [76, 79].

Our analysis makes significant improvements to the previous work of [59]. We considered the formation of a vortex in the tachyon field rather than a kink, in line with the descent relations for stable Dp-branes. We found analytic solutions for the tachyon field which give the time dependence in the vicinity of the defect while it is forming, for both the vortex and the kink solutions. In the latter case we verified that this solution reproduces the known dynamics of kink formation which was determined numerically in [51], giving us more confidence in the vortex solutions, which are quite analogous. Our explicit solution for the tachyon background is nevertheless too complicated for computing the production of particles on the defect. We therefore approximated it by a simpler ansatz with the same qualitative behavior, which allows for analytic solutions of the gauge fields in the background. This ansatz resembles a gauge field in a 6D spacetime with two compact spatial dimensions which are contracting with time, and leads to simple analytic results for the energy density of photons produced during the contraction, in the regime where the extra dimensions are large compared to the string scale.

There are still some outstanding questions to be addressed concerning this scenario. First, we have made reference to the Kibble mechanism for the creation of the final state defects. If we assume the causal bound of one defect per Hubble volume then this would imply that the size of the extra dimensions must exceed the inverse Hubble rate; otherwise there would be enough time for the fields to straighten themselves out and the putative vortex-antivortex pair would immediately annihilate. For example if we take the string scale M_s to be 10^{16} GeV then $H \sim M_s^2/M_p$ and we would need the compactification scale to be of order $R_0 \sim (M_p/M_s)l_s$. Our results indicate that efficient reheating is compatible with a large compactification scale. However, taking the remaining four extra dimensions (which the initial 5-brane/antibrane pair do not wrap) to be string scale is not consistent with getting inflation since the initial state 5-brane and antibrane cannot be sufficiently separated to satisfy the slow roll conditions.

One the other hand, our results indicate that the reheating *can* be efficient for R_0 only a few times the string length, though in this scenario a naive application of the Kibble mechanism does not favor having the final state defects span the three large dimensions. These requirements may not be prohibitive since the question of how these defects form dynamically at the end of inflation is a quantitative one which merits further investigation. In principle the correlation length for the initial fluctuations of the tachyon field could be as small as the string length. We point out also that it is possible that the dynamics of the formation of tachyon defects is qualitatively different from defect formation in a conventional scalar field theory. For example, the numerical investigation of [51], it was found that small kinks in the initial configuration which are in causal contact with each other do not dynamically straighten themselves out as they would in a conventional, nontachyonic field theory. Instead, every place where the field crosses zero in the initial state develops a fullblown kink, so long as there was enough energy in the bulk to produce the required number of kinks. Another indication that the dynamics of the tachyon field may be qualitatively different from an ordinary scalar field theory comes from the [80] in which the causal structure of the tachyon Dirac-Born-Infeld action was studied. The authors of [80] found that small fluctuations of the tachyon field propagate according to an effective metric which depends on the tachyon background. In the case of a homogeneous rolling background it was found that as the condensation proceeds the effective metric contracts to the Carroll limit of the Lorentz group so that the tachyon light cone collapses into a timelike half line and the tachyon fields at different spatial points are decoupled. We feel that quantitatively determining the dynamics of the



Figure 2.9: Warped compactification with branes localized in throats on opposite sides of a stack of branes.

formation of tachyon defects at the endpoint of D-brane inflation is a question which deserves further investigation.

In the present analysis we have not included the gravitational or Ramond-Ramond forces between the vortex and antivortex which would attract them toward each other and lead to their eventual annihilation. How do we insure that the braneworld on which we are supposed to live is safe from annihilation with an antibrane in the bulk? One possibility is to have warping caused by a stack of branes which wraps only the equator of the extra dimensions, as illustrated in fig. 2.9. Such a braneworld scenario using the AdS soliton solution for the bulk has been considered in ref. [81, 82, 61]. The advantage for our scenario is that the warping can provide a barrier to the annihilation of the brane-antibrane pair, since it is energetically favorable for them to remain within their respective throats. In chapter 4 we will consider a model of reheating in brane-antibrane inflation which does not suffer from this difficultly.

In solving for the tachyon background we have also ignored the possibility of caustic formation in the bulk [68] by taking initial profiles without too much curvature. It is possible that caustic formation may be an artifact of the derivative truncation which leads to the Born-Infeld type of Lagranian for the tachyon. We will study this issue in greater detail in chapter 3.

Brane-antibrane inflation and braneworld cosmology are two of the most important applications of string theoretic ideas to cosmology. We find it intriguing that these two ideas might be combined in the way we have described. An outstanding challenge is to find some observable signatures that would be able to test our scenario, for example through the gravitational wave component which is expected to be a major component of the radiation produced during reheating.

Acknowledgements

We thank Horace Stoica for helpful discussions concerning warped compactifications as well as Robert Brandenberger and Henry Tye for helpful comments regarding the Kibble mechanism. Our research is supported by the Natural Sciences and Engineering Research Council (NSERC) of Canada and Fonds québécois de la recherche sur la nature et les technologies (NATEQ).

Chapter 3

Caustic Formation in Tachyon Effective Field Theories

Abstract

Certain configurations of D-branes, for example wrong dimensional branes or the brane-antibrane system, are unstable to decay. This instability is described by the appearance of a tachyonic mode in the spectrum of open strings ending on the brane(s). The decay of these unstable systems is described by the rolling of the tachyon field from the unstable maximum to the minimum of its potential. We analytically study the dynamics of the inhomogeneous tachyon field as it rolls towards the true vacuum of the theory in the context of several different tachyon effective actions. We find that the vacuum dynamics of these theories is remarkably similar and in particular we show that in all cases the tachyon field forms caustics where second and higher derivatives of the field blow up. The formation of caustics signals a pathology in the evolution since each of the effective actions considered is not reliable in the vicinity of a caustic. We speculate that the formation of caustics is an artifact of truncating the tachyon action, which should contain all orders of derivatives acting on the field, to a finite number of derivatives. Finally, we consider inhomogeneous solutions in *p*-adic string theory, a toy model of the bosonic tachyon which contains derivatives of all orders acting on the field. For a large class of initial conditions we conclusively show that the evolution is well behaved in this case. It is unclear if these caustics are a genuine prediction of string theory or not.

3.1 Introduction

Recently effective actions describing the open string tachyon have recieved considerable attention in the literature. These effective actions are interesting to study since they permit a relatively simple formulation of the complicated open string dynamics in terms of classical field theories. In particular, the use of effective actions has triggered significant interest in the possible role of the tachyon in cosmology. For example, there have been numerous attempts to make use of the tachyon as either the inflaton or as quintessence [83] (see [84] for reviews of tachyon cosmology). Constraints and shortcomings of these scenarios have been discussed in [85].

Perhaps the most promising application of the tachyon to cosmology comes from D-brane inflation [24, 25], [53]-[55] In the context of D-brane inflation, the annihilation of a brane-antibrane pair at the endpoint of inflation is described by the rolling of the tachyon, T, from the unstable maximum T = 0 to the degenerate vacua of the theory $T = \pm \infty$. The tachyon field may roll to the same vacuum everywhere in the space or topological defects may form through the Kibble mechanism [56, 57, 58]. The latter possibility corresponds to the formation of lower dimensional branes at points where T = 0. In the regions between the descendant branes the tachyon field asymptotically rolls towards the true vacuum of the theory. The formation of lower dimensional branes is of some phenomenological importance since the resulting cosmic-string-like or higher dimensional defects could be observable remnants of inflation [56, 57, 58]. In [86, 1] a more radical idea was explored: it was proposed that our own observable universe could be such a defect in the higher dimensional spacetime predicted by string theory. In this scenario the coupling of the time dependent tachyon condensate to gauge particles which will be localized on the descendant brane can provide an efficient mechanism for reheating at the end of brane-antibrane inflation.

Applications of the tachyon to cosmology generically involve the tachyon field rolling from the unstable maximum of its potential to the true vacuum of the theory, at least in some region of the space, so it is of some interest to study the dynamics in this regime. Often only homogeneous tachyon profiles are considered, though a consistent study of the tachyon field in cosmology must include spatial inhomogeneities. In the context of the formation of lower dimensional branes at the endpoint of D-brane inflation spatial inhomogeneity is unavoidable since the field must cross T = 0 at one or more points in the space. However, even in the absence of topological defects one expects that spatial inhomogeneities will be generated starting from a homogeneous profile by vacuum fluctuations of the tachyon field.

The full tachyon action contains an infinite number of derivatives acting on the field, though the action is not known explicitly to all orders in derivatives. Rather, numerous effective actions describing the tachyon have been proposed in the literature

on different grounds and have been shown to reproduce various nontrivial aspects of the full dynamics expected from string theory. Most studies of the tachyon dynamics in terms of effective actions have considered actions involving only first derivatives of the field and it is not clear how higher derivative corrections may alter the dynamics. Indeed, studies of p-adic string theory suggest that there may be fundamental differences between dynamical equations with an infinite number of derivatives and those with a large finite number of derivatives. In [87] it was shown that the initial value problem is qualitatively different in these two cases.

In this chapter we study analytically the dynamics of the inhomogeneous tachyon field as it rolls towards the vacuum in the context of several different effective field theories. Our interest is motivated by [68, 69] in which inhomogeneous vacuum solutions were discussed using the popular tachyon Dirac-Born-Infeld (DBI) action. The authors of [68, 69] found that in general the inhomogeneous solutions of the DBI action formed caustics where the second and higher derivatives of the tachyon field blow up at some point. In the vicinity of these caustics the tachyon DBI action is not reliable since the higher derivative corrections which have been ignored will become important. It is thus not known if the formation of caustics is a genuine prediction of string theory or an artifact of the DBI action for the tachyon. To shed some light on this question it is interesting to study inhomogeneous vacuum solutions of different tachyon effective actions and look for caustics or similar singularities.

The finite time formation of caustics found in [68, 69] must be distinguished from the finite time divergences in the first derivatives of the tachyon field at points where T = 0 [1, 51] which corresponds to the formation of topological defects. In both cases the divergences in the derivatives of the tachyon field lead to divergences in the energy density. In the case of topological defect formation, however, this finite time divergence in the energy density has the form of a delta function and hence leads to finite total energy [1]. We point out that similar behaviour was found in [88] where the decaying tachyon on the supergravity background corresponding to space-like branes in string theory was studied. In this case the first derivative of the tachyon field and the energy density of the tachyon matter both divergence in finite time producing a space-like curvature singularity in the metric.

The organization of this chapter is as follows. In section 3.2 we consider three tachyon effective actions, all of which contain only first derivatives of the field and show that all three display surprisingly similar dynamics near the true vacuum of the theory. In section 3.3 we briefly review how this universal vacuum structure leads to the formation of caustics. In section 3.4 we consider the vacuum dynamics of the tachyon in the context of yet another effective action which contains terms with second derivatives of the field and show that the problem of caustic formation is not ameliorated. In section 3.5 we briefly review the bosonic tachyon action of p-adic string theory and study small spatial inhomogeneities about a time dependent solution which rolls towards the minimum of the potential (which is unbounded from below). We find that for a large class of initial data our perturbative expansion is reliable throughout the evolution and the solutions are well behaved. Finally we conclude and discuss the difficulty of interpreting caustic formation physically.

3.2 First Derivative Tachyon Effective Actions and the Eikonal Equation

For simplicity we restrict ourselves to studying real tachyon fields in 1+1-dimensional Minkowski space with metric $\eta_{\mu\nu}dx^{\mu}dx^{\nu} = -dt^2 + dx^2$. In the context of brane annihilation the rolling of the tachyon in 1+1 dimensions from the unstable maximum to the ground state describes the decay of an unstable D*p*-brane to either the vacuum (in the case that the tachyon rolls to the same vacuum everywhere) or to a brane of codimension one (in the case that kinks form). We work in units where $\alpha' = 2$.

3.2.1 The Tachyon Dirac-Born-Infeld Action

We first consider the tachyon Dirac-Born-Infeld action [42, 43, 50]

$$\mathcal{L} = -U(T)\sqrt{1 + \partial_{\mu}T\partial^{\mu}T}$$
(3.1)

with potential

$$U(T) = \frac{\tau_p}{\cosh(T/2)} \tag{3.2}$$

where τ_p is the D*p*-brane tension. Solutions of this theory have been widely studied in the literature [86, 1, 68, 69, 50, 89, 90]. This action can be obtained from string theory in some limit [46] and has been shown to reproduce numerous nontrivial aspects of the full string theory dynamic, as discussed in chapter 1.

The action (3.1) should be thought of as describing tachyon profiles which are "close" ¹ to the homogeneous rolling tachyon solution

$$T(t) = 2\sinh^{-1}\left(T_{+}e^{t/2} + T_{-}e^{-t/2}\right).$$
(3.3)

We are interested in the dynamics of (3.1) close to the vacuum $T \to \pm \infty$, $U(T) \to 0$. The vacuum structure of this theory has been studied analytically in [43, 90] which we briefly review. The vacuum structure is most easily described in the Hamiltonian formalism since the Lagrangian does not survive the limit $U(T) \to 0$, but the Hamiltonian does. Defining the momentum conjugate to T as $\Pi = \partial \mathcal{L} / \partial \dot{T}^2$ the Hamiltonian is given by

$$H = \int d^2x \mathcal{H}, \quad \mathcal{H} = \sqrt{\Pi^2 + U(T)^2} \sqrt{1 + T'^2}.$$

In the limit as $U(T) \to 0$ one finds

$$\mathcal{H} = \Pi \sqrt{1 + T^{\prime 2}}.\tag{3.4}$$

In the vacuum Hamilton's equations of motion are

$$\dot{T} = \sqrt{1 + T^{\prime 2}} \tag{3.5}$$

and

$$\dot{\Pi} = \partial_x \left(\Pi \frac{T'}{1 + T'^2} \right) \tag{3.6}$$

¹That is to say taking T_+ , T_- to be slowly varying functions of x^{μ} .

²Here and throughout this chapter $\dot{T} = \partial_0 T = \partial_t T$ and $T' = \partial_1 T = \partial_x T$.

for $\Pi > 0$. Equation (3.5) is precisely the eikonal equation

$$\partial_{\mu}T\partial^{\mu}T + 1 = 0. \tag{3.7}$$

It was observed numerically in [68, 69] that the general solutions of (3.1) tend towards the first order equation (3.7).

3.2.2 The Boundary String Field Theory Action

The action

$$\mathcal{L} = -V(T)F\left(\partial_{\mu}T\partial^{\mu}T\right) \tag{3.8}$$

where the potential is

$$V(T) = \sqrt{2}\tau_p \exp\left(-\frac{T^2}{4}\right) \tag{3.9}$$

and the kinetic term is

$$F(z) = \frac{4^{z} z \Gamma(z)^{2}}{2 \Gamma(2z)}$$
(3.10)

can be derived from boundary string field theory (BSFT) assuming a linear profile $T = a + u_{\mu}x^{\mu}$, and summed to all orders in u_{μ} [91]. This action should therefore be considered reliable only when describing profiles where second and higher derivatives of the tachyon field are small.

The dynamics of the action (3.8) have been discussed in [71] which we briefly review. Before attempting to describe the vacuum dynamics of the theory (3.8) we discuss some limiting behaviour of the function F(z).

At small z the function F(z) has a Taylor expansion

$$F(z) \cong 1 + \ln(4)z + \cdots$$

and at large positive z the function F(z) has the behaviour

$$F(z) \to \sqrt{\pi z} \quad \text{as } z \to \infty.$$
 (3.11)

It is also noteworthy that F(z) is singular at z = -n $(n = 1, 2, \dots)$. Of particular interest to the ensuing analysis will be the limiting behaviour of F(z) near the singular point z = -1

$$F(z) \cong \frac{-1}{2(z+1)}, \quad F'(z) \cong \frac{1}{2(z+1)^2}, \quad F''(z) \cong \frac{-1}{(z+1)^3}.$$
 (3.12)

We now proceed to study the dynamics of the action (3.8). The equation of motion which follows from (3.8) is

$$\partial^{\mu}T\partial_{\mu}z F''(z) + \partial^{\mu}\partial_{\mu}T F'(z) - \frac{1}{2}T zF'(z) + \frac{1}{4}T F(z) = 0$$
(3.13)

where $z = \partial_{\mu} T \partial^{\mu} T$. We will assume initial conditions such that $z \approx 0$ initially. Furthermore, we assume that z does not cross the singularity at $z = -1^{-3}$.

The homogeneous solutions of (3.13) have the asymptotic behaviour $T \to \pm t$ at late times [92] so we expect that the tachyon will roll towards the vacuum $T \to \pm \infty$ as $t \to \infty$. We consider first the case where z does not approach -1 as $T \to \infty$. In this case F(z), F'(z) and F''(z) are well-behaved and the leading contribution to (3.13) is

$$F(z) - 2zF'(z) = 0.$$

It is easy to show that the quantity F(z) - 2zF'(z) is greater than or equal to 0 for all z > -1 and that $F(z) - 2zF'(z) \to 0$ as $z \to +\infty$ (see equation (3.11)). We conclude then that for solutions where z is increasing the equation of motion requires $z \to +\infty$ as $T \to \pm\infty$.

We now consider the possibility that $z \to -1$ at late times. Near z = -1 the function F(z) and its derivatives are given by (3.12) and hence the leading contribution to (3.13) is

$$\partial_{\mu}T\partial^{\mu}z F''(z) - \frac{1}{2}T zF'(z) = 0.$$
 (3.14)

³This is not a particularly restrictive assumption since if z did cross the singularity it would lead to infinite action. Note that this does not exclude the possibility that $z \to -1$ as $V(T) \to 0$.

This equation can be satisfied for profiles where $z \to -1$. Since $z+1 \to 0$ as $T \to \pm \infty$ it is reasonable to assume that we can write z + 1 = f(T). This is consistent with (3.14) which implies

$$z + 1 \sim V(T)^{1/2}$$

It is important to stress that, barring any asumptions on the form of the solution, the equation of motion (3.13) requires that $\partial_{\mu}T\partial^{\mu}T$ tends to either $+\infty$ or -1 as $T \to \pm \infty$. The former case describes tachyon kinks of the form $T(x) = x/\epsilon$ to be understood in the limit that $\epsilon \to 0$. The latter case describes the rolling tachyon and corresponds to the vacuum structure we are interested in. We conclude then that close to the vacuum the tachyon field obeys the eikonal equation

$$\partial_{\mu}T\partial^{\mu}T + 1 = 0.$$

3.2.3 The Lambert and Sachs Action

The action

$$\mathcal{L} = -V(T)K(\partial_{\mu}T\partial^{\mu}T)$$
(3.15)

where the potential is given by (3.9) and the kinetic term is

$$K(z) = e^{-z} + \sqrt{\pi z} \operatorname{erf}\left(\sqrt{z}\right)$$
(3.16)

was proposed in [93]. This action is derived by requiring that the profile

$$T(x) = \chi \sin\left(\frac{x - x_0}{2}\right)$$

be an exact solution to the equations of motion. The residual freedom is fixed by demanding that $\mathcal{L} = \sqrt{2}\tau_p \exp(-T^2/4)$ for constant profiles for agreement with (3.8).

Before proceeding to study the dynamics of (3.15) we consider the asymptotic behaviour of the kinetic term (3.16). For $z \ge 0$ the function K(z) is remarkably similar to the BSFT function F(z) given by (3.10). Near z = 0 the function K(z)has a Taylor expansion

$$K(z) \cong 1 + z + \cdots$$

For large positive values of z the function K(z) has the aymptotic behaviour

$$K(z) \to \sqrt{\pi z}, \ K'(z) \to \frac{1}{2}\sqrt{\frac{\pi}{z}}, \ K''(z) \to -\frac{\sqrt{\pi}}{4z^{3/2}}$$
 (3.17)

so that K(z) tends to infinity slowly as $z \to +\infty$ while K'(z), K''(z) tend to zero. Finally we consider the behaviour of K(z) for large negative z. Using the formula

$$\sqrt{\pi} \operatorname{erf}(iy) \to i e^{y^2} \left(\frac{1}{y} + \frac{1}{2y^3} + \mathcal{O}(y^{-5}) \right)$$

for $y \to \infty$ it is easy to show that

$$K(z) \to -\frac{e^{|z|}}{2|z|}, \quad K'(z) \to +\frac{e^{|z|}}{2|z|}, \quad K''(z) \to -\frac{e^{|z|}}{2|z|}$$
(3.18)

for large negative z. Note that though K(z) has very similar behaviour to F(z) and $\sqrt{1+z}$ for large positive z, these three kinetic terms have very different behaviour on z < 0.

The equation of motion which follows from (3.15) is

$$\partial^{\mu}T\partial_{\mu}z K''(z) + \partial^{\mu}\partial_{\mu}T K'(z) - \frac{1}{2}T zK'(z) + \frac{1}{4}T K(z) = 0$$
(3.19)

where $z = \partial_{\mu}T\partial^{\mu}T$. By construction equation (3.19) has the homogeneous solution

$$T(t) = T_{+}e^{t/2} + T_{-}e^{-t/2}.$$

Comparing this to the homogeneous solution (3.3) demonstrates that a field redefinition is necessary to make accurate comparisons between the two theories. Notice that in the case of the homogeneous solution the tachyon field rolls towards the vacuum $T \to \pm \infty$ at late times with $z \to -\infty$.

We proceed now to study the vacuum structure of the theory (3.15). First we consider the possibility that z tends to some finite positive value as $T \to \pm \infty$. In this case the leading contribution to (3.19) is

$$TzK'(z) = \frac{1}{2}TK(z)$$
 (3.20)

or

$$0 = e^{-z}.$$

This suggests that we require $z \to +\infty$ if z > 0 and indeed, using the limiting behaviour (3.17) the leading contribution to (3.19) for $T \to \pm \infty$ with $z \to +\infty$ is satisfied.

We consider now possibility that z approaches some finite negative value as $T \to \pm \infty$. Assuming $z \neq -\infty$ then K(z), K'(z) and K''(z) are all well behaved and the leading contribution to (3.19) is still (3.20) or equivalently $0 = e^{|z|}$ which has no solution.

Finally we consider the possibility that $z \to -\infty$ as $T \to \pm \infty$. In this case the limiting behaviour (3.18) is applicable and $K(z) \sim -K'(z) \sim K''(z)$ so that the leading contribution to (3.19) is

$$2\partial_{\mu}T\partial^{\mu}z + Tz = 0. \tag{3.21}$$

Since $z \to -\infty$ as $T \to \pm \infty$ it is reasonable to assume z = f(T) in this regime. This assumption is consistent with (3.21) and we find $f(T) = -T^2$ up to an arbitrary additive constant which is irrelevant as $T^2 \to \infty$. One may verify that given these assumptions the leading behaviour (3.21) is order T^3 while the terms we have disregarded in (3.19) are order T so that this series of approximations is self-consistent. We conclude that for z < 0 the equation of motion requires that the tachyon field obeys

$$\partial_{\mu}T\partial^{\mu}T + T^2 = 0.$$

Defining $T = \exp(\tilde{T})$ we find that \tilde{T} obeys the eikonal equation

$$\partial_{\mu}\tilde{T}\partial^{\mu}\tilde{T} + 1 = 0.$$

We stress that the equation of motion (3.19) requires either $z \to +\infty$ or $z \to -\infty$ as $T \to \pm \infty$. The former case describes tachyon kink solutions while the latter case describes the rolling tachyon and corresponds to the vacuum structure we are interested in.

3.3 Caustic Formation

We have found that all three actions (3.1,3.8,3.15) predict that the tachyon field is described by the eikonal equation (3.7) close to the vacuum. Inhomogeneous solutions of (3.7) were found in [68, 69] for arbitrary Cauchy data by the method of characteristics. The solution is given along a set of characteristic curves which, in 1 + 1dimensions, are defined by

$$x(q,t) = q - \frac{T'_i(q)}{\sqrt{1 + T'_i(q)^2}}t$$
(3.22)

where the parameter q defines the initial position of the curve on the x axis such that x(q, t = 0) = q and $T_i(x)$ is the Cauchy data at time t = 0. The parameter q should be thought of as labelling the curves. The value of the tachyon field along a given characteristic curve is

$$T(q,t) = T_i(q) + \frac{t}{\sqrt{1 + T'_i(q)^2}}.$$
(3.23)

It is also worth noting that the derivatives of the field are constant along the curves $T'(q,t) = T'_i(q)$ and $\dot{T}(q,t) = \sqrt{1 + T'_i(q)^2}$. From (3.22) it is clear that in general for initial data where $T''_i \neq 0$ there will be curves originating from different initial points on the x axis which will cross in some finite time. At points where the characteristic curves intersect the field becomes multi-valued since evolving (3.23) along two different curves leads to two different values of T at the point of intersection. This corresponds to the formation of caustics and signals a pathology in the evolution. To illustrate the problem, consider T'' along a characteristic curve

$$T''(q,t) = \frac{T''_i(q)}{1 - \frac{T''_i(q)}{(1 + T'_i(q)^2)^{3/2}}t}.$$

At some time t_c and point q_c where $T''_i(q_c) > 0$ the denominator vanishes and the second derivative blows up. For a given q_c the caustics form at time

$$t_c = \frac{(1 + T'_i(q_c)^2)^{3/2}}{T''_i(q_c)}.$$

In each of the three effective actions considered above (3.1,3.8,3.15) this caustic formation signals a breakdown of the theory since terms in the action involving second and higher derivatives of the field which have been neglected will become important.

3.4 Vacuum Dynamics in a Higher Derivative Action

In light of the analysis of the preceeding section it is tempting to speculate that the caustic formation described above is an artifact of the derivative truncation which leads to (3.1,3.8,3.15) since these effective descriptions break down in the vicinity of a caustic. We would like to reconsider the effective description of the tachyon near the vacuum in the context of an action which does contain terms with second derivatives of the field. One might consider attempting to generalize the action (3.8) to a profile of the form $T = a + u_{\mu}x^{\mu} + v_{\mu\nu}x^{\mu}x^{\nu}$ to obtain an action which is valid for profiles where $\partial_{\mu}T \neq 0$ and $\partial_{\mu}\partial_{\nu}T \neq 0$ without constraint on the size of the first and second derivatives. However, deriving such an action (following [91]) would involve performing a path integral which is no longer Gaussian. On the other hand, the superstring tachyon effective action has been calculated in an expansion in small momenta (derivatives) around a constant profile up to six orders in derivatives ($\mathcal{O}(\partial^6)$) in [94]. The action, in units where $\alpha' = 2$ and truncated to fourth order in derivatives,

is

$$\mathcal{L} = -V(T) \left(1 + \ln(4)\partial_{\mu}T\partial^{\mu}T + \left(\frac{1}{2}(\ln(4))^{4} - \zeta(2)\right) (\partial_{\mu}T\partial^{\mu}T)^{2} + \zeta(2)T\partial_{\mu}\partial_{\nu}T\partial^{\mu}T\partial^{\nu}T + \left[(2\zeta(2) - 8(\ln(2))^{2}) + \frac{\zeta(2)}{2}T^{2} \right] \partial_{\mu}\partial_{\nu}T\partial^{\mu}\partial^{\nu}T \right)$$
(3.24)

where $\zeta(z)$ is the Riemann zeta function. The action (3.24) is valid to all orders in T but for small derivatives and hence naively does not seem applicable to studying the vacuum dynamics where, the analysis of the preceeding sections would suggest, $\partial_{\mu}T\partial^{\mu}T$ is order unity or larger. However, one might expect the vacuum structure of the theory to be significantly modified due to the presence of terms like $T^2\partial_{\mu}\partial_{\nu}T\partial^{\mu}\partial^{\nu}T$ in the action (3.24) which are vanishing on linear profiles but are large in the vacuum for nonlinear profiles. One might hope that the vacuum structure of the theory (3.24) is such that when one considers Cauchy data where the derivatives of T are small but nonzero, then the derivatives stay small throughout the evolution and the approximations which lead to (3.24) remain self consistent. This turns out not to be the case, as we will show.

The equation of motion which follows from (3.24) is cumbersome and we do not write it out here. In the vacuum, as $T \to \pm \infty$, assuming that the derivatives of the field are well behaved ⁴ the leading contribution to the equation of motion is

$$\partial^{\mu}\partial^{\nu}T\partial_{\mu}T\partial_{\nu}T = 0 \tag{3.25}$$

where the subleading terms are of order T^{-1} and smaller.

We note that (3.25) may be re-written as

$$\partial_{\mu}T\partial^{\mu}\left(\partial_{\nu}T\partial^{\nu}T\right) = 0$$

so that the solution set of (3.25) contains the solutions of the first order equation

$$\partial_{\nu}T\partial^{\nu}T = \kappa$$

where κ is a constant which may be set to +1, -1 or 0 by rescaling T. The case $\kappa = -1$ corresponds to the eikonal equation and we conclude that (3.25) does admit solutions

 $^{{}^{4}}$ If we drop this assumption then (3.24) ceases to be a reasonable description of the dynamics.

with caustic formation in the case of Cauchy data such that $\dot{T}_i(x)^2 = 1 + T'_i(x)^2$. However, equation (3.25) is second order and of course it is not necessary to constrain the Cauchy data in this manner. For completeness we discuss the construction of more general solutions of (3.25) in appendix A.

It is beyond the scope of this chapter to perform a comprehensive analysis of the derivative singularities admitted by (3.25). We have shown an explicit reduction of (3.25) to the eikonal equation, which exhibits caustics. As discussed in the appendix, we believe more general solutions of (3.25) exhibit similar pathologies. For our purposes, this is sufficient to demonstrate that the higher derivative action (3.24)does not ameliorate the problem of caustic formation. In light of these results, it is tempting to speculate that it is necessary to use an action which contains all orders of derivatives acting on the field to obtain a fully consistent description of the dynamics of the tachyon near the vacuum.

As an illustrative example we will discuss inhomogeneous solutions in a p-adic string theory, a toy theory of the bosonic string tachyon which contains an infinite number of derivatives acting on the field which is known explicitly to all orders in derivatives.

3.5 Inhomogeneous Solutions in *p*-adic String Theory

The action of p-adic string theory is [95]

$$S = \frac{p^2}{g^2(p-1)} \int d^D x \left[-\frac{1}{2} \phi \, p^{-\frac{1}{2}\partial_\mu \partial^\mu} \phi + \frac{1}{p+1} \phi^{p+1} \right]$$
(3.26)

where ϕ is the open string tachyon, g is the open string coupling constant and, though the action was derived for p a prime number, it appears that p can be continued to any positive integer (the action makes sense even in the limit $p \to 1$ [96]). The differential operator $p^{-\frac{1}{2}\partial_{\mu}\partial^{\mu}}$ is to be understood as the series expansion

$$p^{-\frac{1}{2}\partial_{\mu}\partial^{\mu}} = \exp\left(-\frac{1}{2}\ln(p)\,\partial_{\mu}\partial^{\mu}\right) = \sum_{n=0}^{\infty} \left(-\frac{1}{2}\ln(p)\right)^{n} \frac{1}{n!} \left(\partial_{\mu}\partial^{\mu}\right)^{n}.$$

The equation of motion which follows from (3.26) is

$$p^{-\frac{1}{2}\partial_{\mu}\partial^{\mu}}\phi = \phi^p \tag{3.27}$$

and the potential is



Figure 3.1: Plot of the potential $V(\phi)$ for even p.



Figure 3.2: Plot of the potential $V(\phi)$ for odd p.

The cases of odd and even p are qualitatively different. For odd p the potential is an even function of ϕ and for even p the potential is an odd function of ϕ . In both cases the potential is unbounded from below. In the case of even p the perturbative vacuum is at $\phi = 1$ and in the case of odd p there is an equivalent false vacuum at $\phi = -1$. In both cases the true vacuum of the theory is at $\phi = 0$. Figures 3.1 and 3.2 show plots of the potential (3.28) for the cases of even and odd p respectively.

We note that the action (3.26) is intended to describe the bosonic string tachyon, rather than the superstring tachyon which we have considered in our previous analysis (3.1, 3.8, 3.15, 3.24). Furthermore, (3.26) is a toy model which is only expected to qualitatively reproduce some aspects of a more realistic theory. These points should be kept in mind when comparing analysis using (3.26) to the previous analysis using (3.1,3.8,3.15,3.24). That being said we note that there are several nontrivial qualitative similarities between p-adic string theory and tachyon matter. For example, near the true vacuum of the theory $\phi = 0$ the field naively has no dynamics since its mass squared goes to infinity ⁵. This is the p-adic version of the statement that there are no open string excitations at the tachyon vacuum. A second similarity between p-adic string theory and tachyon matter is the existence of lump-like soliton solutions representing p-adic D-branes [65]. The theory of small fluctuations about these lump solutions has a spectrum of equally spaced masses squared for the modes [97], as in the case of normal bosonic string theory. On the other hand, there are some important differences between the theory (3.26) and (3.1, 3.8, 3.15, 3.24). In the case of tachyon matter the vacuum is at infinity and the tachyon never reaches this point, whereas in the case of the p-adic string the vacuum is at a finite point in the field configuration space and homogeneous solutions rolling towards the vacuum typically pass this point without difficulty [87]. In fact, the numerical studies of [87] found no homogeneous solutions which appeared to correspond to tachyon matter (vanishing pressure at late times).

Keeping in mind that the connection between (3.26) and (3.1,3.8,3.15,3.24) is not entirely clear we proceed to study inhomogeneous solutions of (3.26) as an example of a theory with an infinite number of derivatives which may have some qualitative similarities to the string theory tachyon. At first glance it is not immediately clear

⁵Reference [87] found anharmonic oscillations around the vacuum by numerically solving the full nonlinear equation of motion. However, these solutions do not correspond to conventional physical states.
how to proceed since (3.27) is difficult to solve for profiles with nontrivial dependence on more than one variable. The simplest solution is to study small fluctuations about some known time-dependent solution of (3.27). We are interested in the dynamics near the vacuum and [87] found anharmonic homogenous oscillations near $\phi = 0$ so one might consider small spatial inhomogeneities about these solutions. However, these anharmonic oscillations cannot be found by solving the linearized equation of motion and hence we should not expect to be able to study small inhomogeneities near the vacuum by linearizing about these oscillators. One might consider the closest analogy to the solutions found in the preceeding sections to be small fluctuations about a time dependent solution which interpolates between the unstable vacuum $\phi = 1$ (or also $\phi = -1$ in the case of odd p) and the stable vacuum $\phi = 0$. However, it was shown in [87, 98] that no such time dependent solution to (3.27) exists ⁶. We therefore will consider inhomogeneous fluctuations about the rapidly increasing solution

$$\phi_0(t) = p^{\frac{1}{2(p-1)}} \exp\left(\frac{1}{2} \frac{p-1}{p \ln p} t^2\right).$$
(3.29)

We stress that this solution does not roll from the unstable maximum of the potential to the true vacuum of the theory (indeed no such solution appears to exist) and thus the ensuing analysis may be of limited relevance since the connection between *p*-adic string theory and tachyon matter is unclear. The homogeneous solution (3.29) does bear some qualitative similarity to the solutions of (3.1,3.8,3.15,3.24) in the sense that this solution represents the tachyon rolling down the potential though in the case of (3.29) the tachyon rolls towards $V \to -\infty$ and not V = 0.

Writing $\phi(t, x) = \phi_0(t) + \delta \phi(t, x)$ with $\phi_0(t)$ given by (3.29) the equation of motion (3.27) is

$$p^{-\frac{1}{2}\partial^{\mu}\partial_{\mu}}\delta\phi(t,x) = p\phi_0(t)^{p-1}\delta\phi(t,x)$$
(3.30)

to linear order in $\delta \phi / \phi_0$. The particular solutions of (3.30) may be written as

$$\delta\phi_{\lambda}(t,x) = \phi_0(t)K_{\lambda}(i\alpha t)\chi_{\lambda}(x) \tag{3.31}$$

⁶More precisely, it was shown that there exists no homogeneous nonnegative bounded continuous solution of (3.27) with $\phi(t \to -\infty) = 1$ and $\phi(t \to +\infty) = 0$.

where

$$\alpha = \sqrt{\frac{1}{2} \frac{p^2 - 1}{p \ln p}}$$

and $K_{\lambda}(z)$ is any solution of the Hermite equation

$$\frac{\partial^2 K_{\lambda}(z)}{\partial z^2} - 2z \frac{\partial K_{\lambda}(z)}{\partial z} + 2\lambda K_{\lambda}(z) = 0.$$
(3.32)

Plugging (3.31) into (3.30) one finds that the spatial modes $\chi_{\lambda}(x)$ are determined by the equation

$$p^{-\frac{1}{2}\partial_x^2}\chi_\lambda(x) = p^{1-\lambda}\chi_\lambda(x).$$
(3.33)

Before attempting to solve (3.33) for the spatial dependence of the particular solutions of (3.30) some comments are in order concerning the separation of variables (3.31). It is straightforward to show that the solutions of (3.30) separate as (3.31) using the identity [97]

$$e^{a\partial_t^2} \left[K_{\lambda}(i\alpha t)e^{bt^2} \right] = (1-4ab)^{-1/2} \left(1 + \frac{4a\alpha^2}{1-4ab} \right)^{\lambda/2} \\ \times K_{\lambda} \left(\frac{i\alpha t}{\sqrt{(1-4ab)(1-4ab+4a\alpha^2)}} \right) \exp\left(\frac{bt^2}{1-4ab}\right) (3.34)$$

This identity was considered in [97] for $\lambda = n = 0, 1, 2, \cdots$ and $K_n(z) = H_n(z)$, the Hermite polynomials, though it holds for arbitrary λ which is most easily shown by noting that the two solutions of (3.32), $K_{\lambda}(z) = P_{\lambda}(z)$ and $K_{\lambda}(z) = Q_{\lambda}(z)$, have contour integral representations [99]

$$P_{\lambda}(z) = \int_{C_1} \frac{d\xi}{2\pi i} \,\xi^{-\lambda - 1} e^{-\xi^2 + 2\xi z} \tag{3.35}$$

and

$$Q_{\lambda}(z) = \int_{C_2} \frac{d\xi}{2\pi i} \xi^{-\lambda - 1} e^{-\xi^2 + 2\xi z}$$
(3.36)

where the curves C_1 and C_2 are given in figure 3.3. The standard definition of the Hermite polynomials for $\lambda = n = 0, 1, 2, \cdots$ is given by

$$H_n(z) = n! \left(P_n(z) + Q_n(z) \right) = n! \int_C \frac{d\xi}{2\pi i} \xi^{-n-1} e^{-\xi^2 + 2\xi z}$$
(3.37)

where C is the contour given in figure 3.4. For subsequent calculations we will be particularly interested in defining the Hermite functions $H_{\lambda}(z)$ on $\lambda \leq 1$. For $\operatorname{Re}(\lambda) < 0$ we can contract the path of integration C to the origin and write

$$H_{\lambda}(z) = \frac{1}{\Gamma(-\lambda)} \int_0^\infty d\xi \, \xi^{-\lambda - 1} e^{-\xi^2 - 2\xi z} \tag{3.38}$$

where the normalization has been chosen to agree with [100]. Note that when $\operatorname{Re}(\lambda) > -1$ we can also represent the Hermite functions by a real integral [100]

$$H_{\lambda}(z) = \frac{2^{\lambda} e^{z^2}}{\sqrt{\pi}} \int_0^\infty e^{-\xi^2} \xi^{\lambda} \cos\left(2z\xi - \frac{\lambda\pi}{2}\right) d\xi$$
(3.39)

which coincides with the standard definition of the Hermite polynomials when $\lambda = n = 0, 1, 2, \cdots$. With these conventions the Hermite functions are normalized so that





Figure 3.3: The contours of integration C_1 and C_2 .

We now proceed to determine the spatial dependence of the modes (3.31). Equation (3.33) has solutions

$$\chi_{\lambda}(x) = a_{\lambda} \cos(\omega_{\lambda} x) + b_{\lambda} \sin(\omega_{\lambda} x)$$
(3.40)

where $\omega_{\lambda} = \sqrt{2(1-\lambda)}$. For initial data which are periodic on some interval [-L, +L]one takes $\omega_{\lambda} = \pi m/L$ where $m = 0, 1, 2, \cdots$. With this choice the degree of the Hermite function is

$$\lambda = 1 - \frac{\pi^2 m^2}{2L^2}$$



Figure 3.4: The contour of integration C.

For the zero mode m = 0 we have $\lambda = 1$ and for all other m we have $\lambda \neq 0, 1, 2, \cdots$ unless L is chosen so that $\sqrt{2} L/\pi$ is an integer.

The Hermite equation (3.32) is second order and we expect to be able to find two linearly independent solutions for each λ . It is most convenient to choose one of these to be the Hermite functions as defined by (3.37,3.38,3.39). To obtain a second solution we note the Wronskian formula [100]

$$W[H_{\lambda}(iz), H_{\lambda}(-iz)] = \frac{2^{\lambda+1}\sqrt{\pi}}{\Gamma(-\lambda)}e^{-z^2}.$$
(3.41)

so that

$$\begin{split} \delta\phi_{\lambda}^{(+)}(t,x) &= \phi_0(t)H_{\lambda}(+i\alpha t)\,\chi_{\lambda}^{(+)}(x), \\ \delta\phi_{\lambda}^{(-)}(t,x) &= \phi_0(t)H_{\lambda}(-i\alpha t)\,\chi_{\lambda}^{(-)}(x). \end{split}$$

are linearly independent for $\lambda \neq 0, 1, 2, \cdots$ (the spatial eigenmodes $\chi_{\lambda}^{(\pm)}$ are any linearly independent combinations of $\cos(\omega_{\lambda} x)$ and $\sin(\omega_{\lambda} x)$). The linear independence of these two solutions fails in the exceptional case $\lambda = 1$ for all L. The linear independence of these two solutions may also fail on other values of m if $\sqrt{2} L/\pi$ is an integer. For simplicity we exclude the case of integer $\sqrt{2} L/\pi$ from the present analysis and take $\lambda = 1$ to be the only special case. (We note, however, that is is straighforward to extend our analysis to include integer values of $\sqrt{2} L/\pi$.) In the case $\lambda = 1$ (m = 0) we can take, as one solution of (3.32), the Hermite polynomial (3.37)

$$H_1(z) = 2z$$

The second solution may be written formally as

$$P_1(z) - Q_1(z)$$

where the functions P_{λ} and Q_{λ} are given by (3.35,3.36). The function obtained in this manner does not have a simple closed form expression as does $H_1(z)$.

Putting the results of this section together we write the general solutions of (3.30) as

$$\delta\phi(t,x) = \phi_0(t) \operatorname{Re}\left[P_1(i\alpha t) - Q_1(i\alpha t)\right] \frac{1}{2} a_0^{(1)} + \phi_0(t) \frac{t}{2} a_0^{(2)}$$

$$\phi_0(t) \sum_{m=1}^{\infty} \tilde{H}_{1-\pi^2 m^2/2L^2}^{(1)}(t) \chi_m^{(1)}(x) + \phi_0(t) \sum_{m=1}^{\infty} \tilde{H}_{1-\pi^2 m^2/2L^2}^{(2)}(t) \chi_m^{(2)}(x)$$
(3.42)

where we have defined the functions

$$\tilde{H}_{\lambda}^{(1)}(t) = \frac{\Gamma(\frac{1-\lambda}{2})}{2^{\lambda}\Gamma(1/2)} \frac{1}{2} \left(H_{\lambda}(i\alpha t) + H_{\lambda}(-i\alpha t) \right)$$

and

$$\tilde{H}_{\lambda}^{(2)}(t) = \frac{\Gamma(-\lambda/2)}{2^{\lambda}\Gamma(-1/2)} \frac{1}{2\alpha i} \left(H_{\lambda}(i\alpha t) - H_{\lambda}(-i\alpha t) \right).$$

The normalizations have been chosen so that $\tilde{H}_{\lambda}^{(1)}(0) = 1$ and $\partial_t \tilde{H}_{\lambda}^{(2)}(t)|_{t=0} = 1$. The spatial modes $\{\chi_m^{(i)}\}$ are

$$\chi_m^{(i)}(x) = a_m^{(i)} \cos\left(\frac{\pi m x}{L}\right) + b_m^{(i)} \sin\left(\frac{\pi m x}{L}\right)$$

for i = 1, 2. The coefficients $\{a_m^{(1)}, b_m^{(1)}\}$ determine the fourier expansion of $\delta \phi / \phi_0$ at t = 0 and the coefficients $\{a_m^{(2)}, b_m^{(2)}\}$ determine the fourier expansion of $\partial_t (\delta \phi / \phi_0)$ at t = 0.

For practical numerical computations it is simplest to consider initial data where $a_0^{(1)} = 0$ to eliminate the function $P_1(i\alpha t) - Q_1(i\alpha t)$ from the expression (3.42). For example, this will be true for any initial data such that $\delta \phi / \phi_0$ at t = 0 is an odd function of x.

Recall that the linearized equation (3.30) is valid for $|\delta\phi(t,x)/\phi_0(t)| \ll 1$. The stability of the solution (3.42) therefore depends on the asymptotic behaviour of the functions $\tilde{H}_{\lambda}^{(1)}(t)$ and $\tilde{H}_{\lambda}^{(2)}(t)$ at late times. It is straightforward to show that $\tilde{H}_{\lambda}^{(1)}(t)$ and $\tilde{H}_{\lambda}^{(2)}(t)$ have the asymptotic behaviour

$$\tilde{H}^{(1)}_{\lambda}(t) \sim \tilde{H}^{(2)}_{\lambda}(t) \sim t^{\lambda}$$

for large t. That these two solutions become linearly dependent as $t \to \infty$ was to be anticipated from (3.41). We find, then, that for modes with $\lambda = 1 - \frac{\pi^2 m^2}{2L^2} < 0$ one has $\delta \phi_m^{(i)}(t,x)/\phi_0(t) \to 0$ and the linearized approximation is stable. On the other hand, modes with $\lambda = 1 - \frac{\pi^2 m^2}{2L^2} > 0$ ⁷ are increasing functions of time and hence the linearized approximation will eventually break down. In the case that $L < \pi/\sqrt{2}$ and $a_0^{(1)} = a_0^{(2)} = 0$ then all of the modes are decreasing and the perturbative expansion is reliable throughout the evolution. In this case we can make definitive statements about the absence of caustics or similar singularities in the theory (3.26). In fact, for initial data which satisfy these restrictions, the profile $\phi(t, x)$ becomes homogeneous at late times. One might argue that (3.42) implies the absence of caustic formation even for more general initial data (for example when L is large and many of the modes $\delta \phi_m^{(i)}(t, x)/\phi_0(t)$ are increasing) since in this case the homogeneous zero mode of $\delta \phi/\phi_0$ increases faster than all other modes.



Figure 3.5: Plot of the spatial profiles $\delta \phi / \phi_0$ for a series of increasing time steps. The initial data is such that the linearized approximation is valid throughout the evolution.

Figure 3.5 shows a plot of the spatial profiles $\delta\phi/\phi_0$ given by (3.42) for a series of increasing time steps for the case p = 2, L = 1. The initial data are $\delta\phi(0, x)/\phi_0(0) = 0.01 x e^{-(x/0.4)^2}$ and $\partial_t (\delta\phi(t, x)/\phi_0(t))|_{t=0} = 0.001 \sin(\pi x/L)$ on $-L \leq x \leq +L$. For

⁷Recall that we are excluding the case $\lambda = 0$.



Figure 3.6: Plot of the spatial profiles $\delta \phi / \phi_0$ for a series of increasing time steps. The initial data is such that the linearized approximation breaks down at late times.

this choice of initial data $a_0^{(1)} = a_0^{(2)} = 0$ and $\lambda < 0$ for all m. The perturbative expansion is stable for this example and the field tends towards a homogeneous profile at late times. Figure 3.6 shows the same plot for the initial data $\delta\phi(0, x)/\phi_0(0) = 0.01 x e^{-(x/0.4)^2}$ and $\partial_t (\delta\phi(t, x)/\phi_0(t))|_{t=0} = 0.01$ on $-L \leq x \leq +L$. With this choice of initial data $\delta\phi/\phi_0 \sim t$ at late times since $a_0^{(2)} \neq 0$ and the linearized approximation will eventually break down.

Finally we comment on the interpretation of the Cauchy problem for the linearized differential equation (3.30). We have found a general solution of (3.30) for which we are free to specify the initial field $\delta\phi/\phi_0$ at t = 0 and the time derivative $\partial_t(\delta\phi/\phi_0)$ at t = 0. This may seem surprising since (3.30) contains an infinite number of time derivatives and one might naively expect to have the freedom to specify an infinite number of initial data $\partial_t^{(n)}(\delta\phi/\phi_0)|_{t=0}$ for $n = 0, 1, 2, \cdots$. However, the initial value problem for homogeneous solutions of (3.27) was studied in [87] and it was found that the equation of motion itself imposes an infinite number of consistency conditions on the initial data which one can consider. In fact, it was speculated in [87] that the space of allowable initial conditions for (3.27) may be finite. This conjecture seems consistent with our results. It is interesting to note that (3.30) seems to be an example of an equation containing an infinite number of derivatives whose solution space is surprisingly similar to that of equations containing only two time derivatives.

3.6 Conclusions

We have studied several different effective actions describing the open superstring tachyon. In the case of actions containing only first derivatives of the tachyon field we have studied three different theories proposed on different grounds. These three actions are expected to be valid in different limits of string theory, however, we find that that vacuum structure of all three theories is remarkably similar. In particular, we have shown that all three theories lead to the formation of caustics where the field becomes multi-valued and where second and higher derivatives of the field blow up. Each of these three actions cannot be trusted in the vicinity of a caustic since higher derivative corrections which have been neglected become important. We considered also an effective action containing second order derivatives of the field and found a similar structure of derivative singularities. Finally, in the context of p-adic string theory, we studied small inhomogeneities about a time dependent solution which rolls from the unstable vacuum to infinity in field configuration space. In this case we found that for a broad class of initial data the linearized approximation is reliable throughout the evolution and we could conclusively show the absence of caustics or similar pathologies. This result seems suggestive that the formation of caustics is an artifact of truncating a theory with an infinite number of derivatives, however, the connection between *p*-adic string theory and tachyon matter is still unclear.

Since we have restricted our analysis to the study of effective actions we cannot conclude if the phenomenon of caustic formation is a genuine prediction of string theory or not. If this prediction is borne out by string theory it will be necessary to find some physical interpretation of the caustics, though none is obvious to us. As pointed out in [68, 69] such an interpretation could depend on the dimensionality of the theory. One avenue for future study would be to try to interpret caustic formation in terms of the gas of massive closed strings which is described by the tachyon fluid [52].

If these caustics are in fact a real prediction of string theory it will also be necessary to find some way to predict the field value in the vicinity of a caustic. The situation is qualitatively similar to the formation of shock waves in nonlinear gas dynamics. In that case the multi-valued continuous solution in the vicinity of a shock is replaced with a single valued discontinuous solution using the Rankine-Hugoniot jump condition to quantify the discontinuity in the field. It is possible that some similar auxiliary condition could be used to predict the value of the tachyon field in the vicinity of a caustic.

Acknowledgments

The author would like to thank M. Laidlaw for useful correspondence and J. Cline for many helpful discussions and comments on the manuscript. This research is supported by the Natural Sciences and Engineering Research Council (NSERC).

Chapter 4

Warped Reheating in Brane-Antibrane Inflation

Abstract

We examine how reheating occurs after brane-antibrane inflation in warped geometries, such as those which have recently been considered for Type IIB string vacua. We adopt the standard picture that the energy released by brane annihilation is dominantly dumped into massive bulk (closed-string) modes which eventually cascade down into massless particles, but argue that the this need not mean that the result is mostly gravitons with negligible visible radiation on the Standard Model brane. We show that if the inflationary throat is not too strongly warped, and if the string coupling is sufficiently weak, then a significant fraction of the energy density from annihilation will be deposited on the Standard Model brane, even if it is separated from the inflationary throat by being in some more deeply warped throat. This is due to the exponential growth of the massive Kaluza-Klein wave functions toward the infrared ends of the throats. We argue that the possibility of this process removes a conceptual obstacle to the construction of multi-throat models, wherein inflation occurs in a different throat than the one in which the Standard Model brane resides. Such multi-throat models are desirable because they can help to reconcile the scale of inflation with the supersymmetry breaking scale on the Standard Model brane, and because they can allow cosmic strings to be sufficiently long-lived to be observable during the present epoch.

4.1 Introduction

There has been significant progress over the past years towards the construction of *bona fide* string-theoretic models of inflation. The main progress over early string-inspired supergravity [102] and BPS-brane based [103] models has come due to the recognition that brane-antibrane [104, 105] and related [106, 107] systems can provide *calculable* mechanisms for identifying potentially inflationary potentials. Even better, they can suggest new observable signatures, such as the natural generation of cosmic strings by the brane-antibrane mechanism [104, 56, 57, 58]. The central problem

to emerge from these early studies was to understand how the many string moduli get fixed, since such an understanding is a prerequisite for a complete inflationary scenario.

Recent developments are based on current progress in modulus stabilization within warped geometries with background fluxes for Type IIB vacua [20, 108, 21]. Both brane-antibrane inflation [24] and modulus inflation [109] have been embedded into this context, with an important role being played in each case by branes living in strongly-warped 'throat-like' regions within the extra dimensions. These inflationary scenarios have generated considerable activity [110, 111] because they open up the possibility of asking in a more focused way how string theory might address the many issues which arise when building inflationary models. For instance, one can more fully compute the abundance and properties of any residual cosmic strings which might survive into the present epoch [58]. Similarly, the possibility of having quasirealistic massless particle spectra in warped, fluxed Type IIB vacua [113] opens up the possibility of locating where the known elementary particles fit into the postinflationary world [25], a prerequisite for any understanding of reheating and the subsequent emergence of the Hot Big Bang.

Even at the present preliminary level of understanding, a consistent phenomenological picture seems to require more complicated models involving more than a single throat (in addition to the orientifold images).¹ This is mainly because for the singlethroat models the success of inflation and particle-physics phenomenology place contradictory demands on the throat's warping. They do so because the energy scale in the throat is typically required to be of order $M_i \sim 10^{15}$ GeV to obtain acceptably large temperature fluctuations in the CMB. But as was found in ref. [25], this scale tends to give too large a supersymmetry breaking scale for ordinary particles if the Standard Model (SM) brane resides in the same throat. This problem appears to be reasonably generic to the KKLT-type models discussed to date, because these models tend to have supersymmetric anti-de Sitter vacua until some sort of supersymmetry-

 $^{^{1}}$ Two-throat models are also considered for reasons different than those given here in ref. [112].

breaking physics is added to lift the vacuum energy to zero. The problem is that the amount of supersymmetry-breaking required to zero the vacuum energy also implies so large a gravitino mass that it threatens to ruin the supersymmetric understanding of the low-energy electroweak hierarchy problem.

No general no-go theorem exists, however, and there does appear to be considerable room to try to address this issue through more clever model-building. Ref. [114] provides a first step in this direction within the the framework of 'racetrack' inflation [109]. Another possibility is a picture having two (or more) throats, with inflation arising because of brane-antibrane motion in one throat but with the Standard Model situated in the other (more about this proposal below). By separating the scales associated with the SM and inflationary branes in this way, it may be possible to reconcile the inflationary and supersymmetry-breaking scales with one another.

Besides possibly helping to resolve this problem of scales, multi-throat models could also help ensure that string defects formed at the end of inflation in the inflationary throat have a chance of surviving into the present epoch and giving rise to new observable effects [58]. They are able to do so because if the Standard Model were on a brane within the same throat as the inflationary branes, these defects typically break up and disappear by intersecting with the SM brane.

At first sight, however, any multi-throat scenario seems likely to immediately founder on the rock of reheating.² Given the absence of direct couplings between the SM and inflationary branes, and the energy barrier produced by the warping of the bulk separating the two throats, one might expect the likely endpoint of braneantibrane annihilation to be dump energy only into closed-string, bulk modes, such as gravitons, rather than visible degrees of freedom on our brane. In such a universe the energy which drove inflation could be converted almost entirely into gravitons, leaving our observable universe out in the cold.

It is the purpose of the present work to argue that this picture is too pessimistic, because strongly-warped geometries provide a generic mechanism for channelling

²See ref. [115] for a discussion of issues concerning brane-related reheating within other contexts.

the post-inflationary energy into massless modes localized on the throat having the strongest warping. They can do so because the massive bulk Kaluza-Klein (KK) modes produced by brane-antibrane annihilation prefer to decay into massless particles which are localized on branes within strongly-warped throats rather than to decay to massless bulk modes. As such, they open a window for obtaining acceptable reheating from brane-antibrane inflation, even if the inflationary and SM branes are well separated on different throats within the extra dimensions.

The remainder of the chapter is organized as follows. In §2 we introduce a simple generalization of the Randall-Sundrum (RS) model [116] containing two AdS₅ throats with different warp factors, as a tractable model for the IKLMT inflationary scenario [24] with two throats. Here we recall the form of the KK graviton wave functions in the extra dimension. This is followed in §3 by an account of how the tachyonic fluid describing the unstable brane-antibrane decays into excited closedstring states, which quickly decay into KK gravitons. §4 Discusses the tunneling of the KK modes through the energy barrier which exists between the two throats because of the warped geometry. §5 Gives an estimate of the reheating temperature on the SM brane which results from the preferential decay of the KK gravitons into SM particles. Our conclusions are given in §6.

4.2 Tale of Two Throats

We wish to describe reheating in a situation where brane-antibrane inflation occurs within an inflationary throat having an energy scale of M_i , due to the warp factor $a_i = M_i/M_p$, where M_p is the 4D Planck mass. This throat is assumed to be separated from other, more strongly warped, throats by a weakly warped Giddings-Kachru-Polchinski (GKP) manifold [20] whose volume is only moderately larger than the string scale, so $M_s \lesssim M_p$. In the simplest situation there are only two throats (plus their orientifold images), with the non-inflationary (Standard Model) throat having warp factor $a_{sm} \ll a_i$.

There are two natural choices for the SM warp factor, depending on whether

or not the SM brane strongly breaks 4D supersymmetry. For instance, if the SM resides on an anti-D3 brane then supersymmetry is badly broken and the SM warp factor must describe the electroweak hierarchy à la Randall and Sundrum [116], with $a_{sm} \sim M_W/M_p \sim 10^{-16}$. Alternatively, if the SM resides on a D3 or D7 brane which preserves the bulk's N = 1 supersymmetry in 4D, then SUSY breaking on the SM brane is naturally suppressed by powers of $1/M_p$ because it is only mediated by virtual effects involving other SUSY-breaking anti-D3 branes. In this case the electroweak hierarchy might instead be described by an intermediate-scale scenario [117], where $a_{sm} \sim M_{int}/M_p \sim (M_W/M_p)^{1/2} \sim 10^{-8}$.

A potential problem arises with the low-energy field theory approximation if $a_{sm} < a_i^2$, because in this case the string scale in the SM throat, $M_{sm} \sim a_{sm}M_p$, is smaller than the inflationary Hubble scale $H_i \sim M_i^2/M_p \sim a_i^2M_p$ [118]. In this case string physics is expected to become important in the SM throat, and stringy corrections may change the low-energy description. The intermediate scale, where $a_{sm} \gtrsim a_i^2$, is more attractive from this point of view, since for it the field-theory approximation may be justified.

To proceed we use the fact that within the GKP compactification the geometry within the throat is well approximated by

$$ds^{2} = a^{2}(y)(-dt^{2} + dx^{2}) + dy^{2} + y^{2} d\Omega_{5}^{2}, \qquad (4.1)$$

where y represents the proper distance along the throat, $a(y) = e^{-k|y|}$ is the throat's warp factor and $d\Omega_5^2$ is the metric on the base space of the corresponding conifold singularity of the underlying Calabi-Yau space [119]. Of most interest is the 5D metric built from the observable 4 dimensions and y, which is well approximated by the metric of 5-dimensional anti-de Sitter space.

A simple model of the two-throat situation then consists of placing inflationary brane-antibranes in a throat at $y = -y_i$ and putting the Standard Model brane at $y = +y_{sm}$, as is illustrated in Fig. 4.1. Our analysis of this geometry follows the spirit of ref. [120]. Since most of the interest is in the throats, we simplify the description of the intervening bulk geometry by replacing them with a Planck brane at y = 0, with the resulting discontinuity in the derivative of a(y) chosen to reproduce the smoother (but otherwise similar) change due to the weakly-warped bulk. This approximation is illustrated in Fig. 4.2, with the smooth dashed curve representing the warp factor in the real bulk geometry and the solid spiked curve representing the result using an intervening Planck brane instead.



Figure 4.1: A Type IIB vacuum with a mildly warped inflationary throat and a strongly warped Standard Model throat. This diagram suppresses any image throats arising due to any orientifolds which appear in the compactification.



Figure 4.2: The warp factor as function of a bulk radial coordinate in a simplified model of two asymmetric throats. As shown in the figure, the part of the internal space outside of the throats can be regarded as a regularization of a 'Planck' brane of a Randall-Sundrum geometry.

Of particular interest in what follows are the massive Kaluza-Klein modes in the bulk, since these are arguably the most abundantly-produced modes after braneantibrane annihilation. For instance, focussing on the 5 dimensions which resemble AdS space in the throat, a representative set of metric fluctuations can be parameterized as h(x, y) in the line element,

$$ds^{2} = a^{2}(y)(-dt^{2} + dx^{2} + h_{\mu\nu}dx^{\mu}dx^{\nu}) + dy^{2}.$$
(4.2)

In the static AdS background, the KK modes have spatial wavefunctions of the form

$$h(x,y) = \sum_{n} \phi_n(y) \ e^{ip \cdot x} \tag{4.3}$$

with $p \cdot x = -E_n t + \mathbf{p} \cdot \mathbf{x}$, and $\phi_n(y)$ satisfying the equation of motion

$$-\frac{d}{dy}\left(e^{-4k|y|}\frac{d\phi_n}{dy}\right) = m_n^2 e^{-2k|y|}\phi_n.$$
(4.4)

Here $m_n^2 = p \cdot p$ is the mode's 4D mass as viewed by brane-bound observers.

Exact solutions for $\phi_n(y)$ are possible in the Planck-brane approximation [116, 120, 121], and are linear combinations of Bessel functions times an exponential

$$\phi_n(y) = N_n \, e^{2k|y|} \left[J_2\left(\frac{m_n}{k} e^{k|y|}\right) + b_n \, Y_2\left(\frac{m_n}{k} e^{k|y|}\right) \right] \tag{4.5}$$

where, for low lying KK modes $(m_n \ll k)$ one has

$$b_n \cong \frac{\pi m_n^2}{4k^2} \tag{4.6}$$

while for heavy KK modes $(m_n/k \cong 1)$ one has

$$b_n \simeq -0.47 + 1.04 \left(\frac{m_n}{k}\right).$$
 (4.7)

 N_n is determined by the orthonormality condition, which ensures that the kinetic terms of the KK modes are independent of a_{sm} :

$$\int_{-y_i}^{y_{sm}} dy \, e^{-2k|y|} \phi_n \phi_m = \delta_{nm} \,. \tag{4.8}$$

These wavefunctions are graphed schematically in Fig. 4.3.

For strongly-warped throats it is the exponential dependence which is most important for the KK modes. Because of the exponential arguments of the Bessel functions in eq. (4.5), the presence of the Bessel functions modifies the large-y behaviour slightly. Due to the asymptotic forms $J_2(z) \propto z^{-1/2}$ for large |z|, and similarly for $Y_2(z)$, we see that $\phi_n(y) \sim e^{3k|y|/2}$ for $m_n e^{k|y|} \gg k$. It is only this behaviour which we follow from here on. Taking the most warping to occur in the SM throat we find that m_n is approximately quantized in units of $M_{sm} \equiv a_{sm}M_p$, which is either of order $M_W \sim 10^3$ GeV or $M_{int} \sim 10^{10}$ GeV depending on whether or not supersymmetry breaks on the SM brane. Keeping only the exponentials we find that orthonormality requires $N_n^{-2} \sim \int^{y_{sm}} dy \ e^{-2ky} \ \left(e^{3k|y|/2}\right)^2 \sim (k a_{sm})^{-1}$, and so

$$\phi_n(y) \sim (a_{sm} k)^{1/2} e^{3k|y|/2},$$
(4.9)

showing that these modes are strongly peaked deep within the throat. This is intuitively easy to understand, since being localized near the most highly-warped region allows them to minimize their energy most effectively.

Thus, even the most energetic KK modes still have exponentially larger wave functions on the TeV brane, with the more energetic modes reaching the asymptotic region for smaller y. This is illustrated in Figure 4.4, which shows $\ln |\phi_n(y)|$ versus yin the representative case of a throat having warp factor $a = e^{-10}$, for a series of KK states with masses going as high as M_p (n = 20,000). As the figure shows, the wave functions grow exponentially toward the TeV brane, with the onset of the asymptotic exponential form setting in earlier for larger mode number.³ This behaviour is central to the estimates which follow.

Among the KK modes it is the zero modes which are the exceptional case because their wavefunction is constant, $\phi_0 \sim \sqrt{k}$, and so they are not exponentially peaked inside the throat. It is the strong exponential peaking of the lightest massive KK modes relative to the massless modes which is central to the reheating arguments which follow.

In our simplified model, the presence of two throats is not much more complicated than the original RS model. Mathematically it is the same, except that RS identified the two sides $y \leftrightarrow -y$ through orbifolding. Instead we interpret them as two separate

³For the lowest-lying modes having the smallest nonzero masses it can happen that the asymptotic form of the Bessel functions is not yet reached even when $y = y_{sm}$, in which case the exponential peaking is slightly stronger than discussed above.



Figure 4.3: Wave functions of KK gravitons on the internal space.



Figure 4.4: Unnormalized wave functions for highly excited KK gravitons with KK numbers n = 1, 100, 1000 and 20,000.



Figure 4.5: How to place two throats on an S_1 with Z_2 orbifold symmetry.

throats with different depths defined by the brane locations $-y_i$ and $+y_{sm}$. One can imagine doubling this entire system on S_1 and orbifolding as shown in Fig. 4.5, to define the boundary conditions on the metric and its perturbations at the infrared branes. In this figure, the orbifold identification acts horizontally so that the inflation and SM branes are distinct fixed points.

4.3 Brane-Antibrane Annihilation

In brane-antibrane inflation the energy released during reheating is provided by the tensions of the annihilating branes. Although this annihilation process is not yet completely understood, present understanding indicates that the energy released passes through an intermediate stage involving very highly-excited string states, before generically being transferred into massless closed-string modes. The time frame for this process is expected to be the local string scale.

For instance many of the features of brane-antibrane annihilation are believed to be captured by the dynamics of the open-string tachyon which emerges for small separations for those strings that stretch between the annihilating branes.⁴ In flat space and at zero string coupling ($g_s = 0$), the annihilation instability has been argued to be described by the following tachyon Lagrangian [43]

$$\mathcal{L}_T = -2\tau_0 \, e^{-|T|^2/l_s^2} \sqrt{1 + |\partial_\mu T|^2} \tag{4.10}$$

where T is the complex tachyon field, τ_0 is the tension of either of the branes, and l_s is the string length scale. During inflation, when the brane and anti-brane are well separated, $\dot{T} = 0$ and the pressure of the system p_i is simply the negative of the tension of the two branes, $p_i = -\rho_i$, while afterward $\dot{T} \to 1$ and $p_i \to 0$. In this description the pressureless tachyonic fluid would dominate the energy density of the universe and lead to no reheating whatsoever.

⁴See, however, ref. [122] for a discussion of an alternative mechanism for which the relevant highly-excited strings are open strings, but for which the annihilation energy nonetheless eventually ends up in massless closed string modes.

However, for nonvanishing g_s , the time evolution of the tachyon fluid instead very quickly generates highly excited closed-string states [45, 40]. For Dp systems with p > 2 the rate of closed string production in this process is formally finite, whereas it diverges for $p \leq 2$ (and so passes beyond the domain of validity of the calculation). This divergence is interpreted to mean that for branes with $p \leq 2$ all of the energy liberated from the initial brane tensions goes very efficiently into closed string modes. For spatially homogeneous branes with p > 2 the conversion is less efficient and so can be dominated by other, faster processes. In particular, it is believed that these higher-dimensional branes will decay more efficiently inhomogeneously, since they can then take advantage of the more efficient channels which are available to the lowerdimensional branes. For example a D3 brane could be regarded as a collection of densely packed but smeared-out D0 branes, each of which decays very efficiently into closed strings. Since the decay time is of order the local string scale, $l_s = 1/M_s$, the causally-connected regions in this kind of decay are only of order l_s in size, and so have a total energy of order the brane tension times the string volume, $\tau_0 l_s^3 \sim M_s/g_s$.

These flat-space calculations also provide the distribution of closed-string states as a function of their energy. The energy density deposited by annihilating D3-branes into any given string level is of order M_s^4 , and so due to the exponentially large density of excited string states the total energy density produced is dominated by the most highly-excited states into which decays are possible. Since the available energy density goes like $1/g_s$ the typical closed-string state produced turns out to have a mass of order M_s/g_s , corresponding to string mode numbers of order $N \sim 1/g_s$. On the other hand, the momentum transverse to the decaying branes for these states turns out to be relatively small, $p_T \sim M_s/\sqrt{g_s}$ [45, 40], and so the most abundantly produced closed-string states are nonrelativistic.

How do these flat-space conclusions generalize to the warped Type IIB geometries which arise in string inflationary models? If the annihilating 3-branes are localized in the inflationary throat, then the tension of the annihilating branes is of order $\tau_0 \sim (a_i M_s)^4 / e^{\phi_i}$, where ϕ_i denotes the value taken by the dilaton field at the throat's tip. The highly-excited closed-string states that are produced in this way live in the bulk, with the energy density produced being dominated by those whose masses are of order $a_i M_s / e^{\phi_i}$. Once produced, these closed-string bulk modes decay down to lower energies and, as might be expected from phase space arguments, most of them typically drop down to massless string states very quickly. An important exception to this would arise for those states carrying the most angular momentum at any given string mass level, since these must cascade more slowly down to lower energies in order to lose their angular momentum [123]. However these seem unlikely to be produced in appreciable numbers by brane-antibrane annihilation.

We are led in this way to expect that the annihilation energy is distributed relatively quickly amongst massless string states, or equivalently to KK modes of the higher-dimensional supergravity which describes these states. Although the initial massive string modes would be nonrelativistic, with $M \sim a_i M_s / e^{\phi_i}$ and $p_T \sim a_i M_s / e^{\phi_i/2}$, the same need not be true for the secondary string states produced by their decay, whose masses are now of order the KK mass scales. Consequently these states may be expected not to remain localized in the inflationary throat, and so if the extra dimensions are not too large compared with the string scale these modes would have time to move to the vicinity of the SM throat before decaying further. Once there, they would be free to fall into the potential wells formed by the throats as their energy is lost by subsequent decays into lower-energy levels.

This physical picture is supported by the exponential peaking of the KK-mode wave-functions in the most deeply-warped throats. In order to estimate the efficiency with which energy can be transferred amongst KK modes, we can use the approximate behavior of the wave functions given in the previous section to keep track of powers of the throat's warp factor, a_{sm} . For instance, consider the trilinear vertex among 3 KK states having mode numbers n_1 , n_2 and n_3 which is obtained by dimensionally reducing the higher-dimensional Einstein-Hilbert action, \sqrt{gR} . Keeping in mind that $\sqrt{g}g^{\mu\nu} \propto a^2$ and that $\psi_n \propto a_{sm}^{1/2}/a^{3/2}$ for the nonzero modes (eq. (4.9)), we find that the trilinear vertex involving $0 \leq r \leq 3$ massive KK modes (and 3 - r massless KK modes) has the following representative estimate

$$\mathcal{L}_{int} \sim \int_{-y_{i}}^{y_{sm}} dy \sqrt{|g|} g^{\mu\nu} g^{\alpha\beta} g^{\kappa\sigma} g^{\rho\delta} h_{\alpha\kappa} \partial_{\mu} h_{\sigma\rho} \partial_{\nu} h_{\beta\delta} \sim \int_{-y_{i}}^{y_{sm}} dy \, e^{-2k|y|} \eta^{\mu\nu} \psi_{n_{1}}(x, y) \, \partial_{\mu} \psi_{n_{2}}(x, y) \, \partial_{\nu} \psi_{n_{3}}(x, y) \sim \psi_{n_{1}}(x) \, \psi_{n_{2}}(x) \, \psi_{n_{3}}(x) \, p_{2} \cdot p_{3} \, a_{sm}^{r/2} \int_{-y_{i}}^{y_{sm}} dy \, e^{-2k|y|} (e^{3k|y|/2})^{r} \sim \psi_{n_{1}}(x) \, \psi_{n_{2}}(x) \, \psi_{n_{3}}(x) \, \left(\frac{p_{2} \cdot p_{3}}{k}\right) \, a_{sm}^{\eta}$$
(4.11)

where

$$\eta = 2 - r$$
 if $r \ge 2$, and $\eta = \frac{r}{2}$ if $r = 0, 1$. (4.12)

Notice for this estimate that since derivatives in the compactified directions are proportional to g^{mn} rather than $g^{\mu\nu}$, they suffer from additional suppression by powers of $a = e^{-ky}$ within the throat. Here m, n label the internal directions perpendicular to the large 3 + 1-dimensional Minkowski space.

Thus a trilinear interaction amongst generic KK modes (r = 3), even those with very large n, is proportional to $1/(a_{sm}k) \sim 1/M_{sm}$, and so is only suppressed by inverse powers of the low scale. Similarly, r = 2 processes involving two massive KK modes B and B', and one massless bulk mode ZM — such as the reaction $B \rightarrow B' + ZM$ — are $\propto 1/k \sim 1/M_p$ and so have the strength of 4D gravity inasmuch as they are Planck suppressed. The same is also true of the r = 0 couplings which purely couple the zero modes amongst themselves.⁵ Finally, those couplings involving only a single low-lying massive mode and two zero modes (r = 1) — such as for $B \rightarrow ZM + ZM'$ — are proportional to $a_{sm}^{1/2}/k \sim (M_{sm}/M_p^3)^{1/2}$ and so are even weaker than Planck-suppressed.

Similar estimates may also be made for the couplings of the generic and the massless KK modes to degrees of freedom on a brane sitting deep within the most strongly-warped throat. Using the expressions $\phi_0(y_{sm}) \sim 1$ and $\phi_n(y_{sm}) \sim 1/a_{sm}$ for

⁵The appendix shows that this agrees with the size of the couplings found in the effective 4D supergravity lagrangian which describes the zero-mode and brane couplings.

massless and massive KK modes respectively, this gives:

$$\mathcal{L}_{i} = M_{p}^{-1} \left(h_{\mu\nu}^{(0)} \phi_{0}(y_{sm}) + \sum_{n} h_{\mu\nu}^{(n)} \phi_{n}(y_{sm}) \right) T_{sm}^{\mu\nu} \\ \sim \left(\frac{h_{\mu\nu}^{(0)}}{M_{p}} + \sum_{n} \frac{h_{\mu\nu}^{(n)}}{M_{sm}} \right) T_{sm}^{\mu\nu}.$$
(4.13)

We see here the standard Planck-suppressed couplings of the massless modes (such as the graviton) as compared with the $O(1/M_{sm})$ couplings of the massive KK modes.

The picture which emerges is one for which the energy released by brane-antibrane annihilation ends up distributed among the massive KK modes of the massless string states. Because the wavefunctions of these modes tend to pile up at the tip of the most warped (SM) throat, their couplings amongst themselves — and their couplings with states localized on branes in this throat — are set by the low scale M_{sm} rather than by M_p . Furthermore, because the $\mathcal{O}(1/M_{sm})$ couplings to the massless modes on the SM branes are much stronger than the Planck-suppressed couplings to the massless bulk modes, we see that the ultimate decay of these massive KK modes is likely to be into brane states. If it were not for the issue of tunneling, which we consider below, the final production of massless KK zero modes would be highly suppressed. Although we make the argument here for gravitons, the same warp-counting applies equally well to the other fields describing the massless closed-string sector.

In summary, we see that strong warping can provide a mechanism for dumping much of the energy released by the decay of the unstable brane-antibrane system into massless modes localized on branes localized at the most strongly-warped throat, regardless of whether the initial brane-antibrane annihilation is located in this throat. It does so because the primary daughter states produced by the decaying braneantibrane system are expected to be very energetic closed strings, which in turn rapidly decay into massive KK modes of the massless string levels. The strong warping then generically channels the decay energy into massless modes which are localized within the most strongly-warped throats, rather than into massless bulk modes.

4.4 Tunneling

However the above arguments are too naive, since they ignore the fact that there is an energy barrier which the initial KK gravitons must tunnel through in order to reach the Standard Model throat. The efficiency of reheating on the SM brane will be suppressed by the tunneling probability.

The tunneling amplitude for a KK mode with energy E_n , in a Randall-Sundrumlike two-throat model just like ours, has been computed exactly in [120]:

$$\mathcal{A} \sim a_{\inf}^2 \left(\frac{E_n}{M_i}\right)^2 \tag{4.14}$$

where M_i is the characteristic energy scale at the bottom of the inflation throat, out of which the particle is tunneling. Intuitively, this can be understood in the following way. For a mode with minimum (but nonzero) energy, the tunneling amplitude is given by the ratio of its wave function at the bottom of the throat to that at the top:

$$\mathcal{A} \sim \frac{\phi_n(y_{\text{inf}})}{\phi_n(0)} \sim a_{\text{inf}}^2 \tag{4.15}$$

Since energies in the throat scale linearly with the warp factor, a high-energy mode, with energy E_n , should have the larger tunneling amplitude given by (4.14). In the present case, the highest KK modes have energies determined by the tension of a D0-brane (as argued above); but we must remember that it is the warped tension which counts, so the maximum energy scale is given by $E_n \sim a_{\inf} M_s/g_s$ whereas the characteristic energy scale in the throat is $M_i \sim a_{\inf} M_s$. The tunneling probability is therefore

$$P = \mathcal{A}^2 = \left(\frac{a_{\inf}}{g_s}\right)^4 \tag{4.16}$$

To maximize this, we need a high scale of inflation (so that the inflationary warp factor is not too small) and a small string coupling. Optimistically, we could imagine that inflation is taking place near the GUT scale, 10^{16} GeV, which saturates the bound on the inflation scale coming from gravitational waves contributing to the CMB anisotropy, and $g_s = 0.01$. Then $a_{inf} = 10^{-3}$ and the tunneling probability is $P = 10^{-4}$.

With a small tunneling probability P, the universe immediately after reheating would be dominated by massless gravitons, the final decay product of KK gravitons confined to the inflation throat. Only the small fraction P of the original false vacuum energy density which tunneled into the SM throat would efficiently decay into ultimately visible matter on the SM brane. Such a distribution of energy density would be strongly ruled out by big bang nucleosynthesis were it to persist down to low temperatures. There are several natural ways in which this outcome can be avoided however. Since reheating occurs at a high scale (given that P is not too small, as we shall quantify in the next section), the number of effectively massless degrees of freedom $N(T_{\rm rh})$ could be quite large at the temperature of reheating. As the heavier of these species go out of equilibrium, they transfer their entropy into the lighter visible sector particles, resulting in a relative enhancement factor $N(T_{\rm rh})/N(T_{\rm nuc})$ of the entropy in visible radiation at the nucleosynthesis temperature T_{nuc} . On the other hand the entropy density in gravitons remains fixed because they were already thermally decoupled from the moment they were produced. If this is the only mechanism for diluting gravitons, we would require $P N(T_{\rm rh})/N(T_{\rm nuc}) \gtrsim 10$, so that gravitons make up no more than 10% of the total energy density at BBN.

Additionally, gravitons can be efficiently diluted if any heavy particles decay out of equilibrium at a temperature T_{dec} before BBN, so that they come to dominate the energy density during a significant interval.⁶ In this case the gravitons are diluted by an additional factor of T_{dec}/M by decaying particles of mass M. Similarly, a period of domination by coherent oscillations of a scalar field (for example a flat direction with a large initial VEV, that gets lifted during a phase transition) will behave as though matter-dominated, and give the same kind of dilution.

There is one more criterion which must be satisfied in order for tunneling to be significant: the lifetime of the heaviest KK states should be of the same order as or longer than the typical tunneling time. The typical momentum of the KK modes is of order $M_i/\sqrt{g_s}$, hence the velocity transverse to the decaying brane is of order $\sqrt{g_s}$

⁶We thank Andrei Linde for pointing out this possibility.

[45], and the length of the throat is of order $R = cM_s^{-1}$ where $c \gtrsim 10$ in order to have a reliable low-energy description of the inflationary dynamics. Thus the tunneling time is

$$\tau_t \gtrsim 10 \frac{M_s^{-1}}{\sqrt{g_s}} P^{-1} \tag{4.17}$$

while the lifetime of a KK mode is estimated to be

$$\tau_l \sim \left(g_s^2 \frac{M_i}{g_s}\right)^{-1} \tag{4.18}$$

where the factor of g_s^2 comes from the squared amplitude for the two-body decay and M_i/g_s is the phase space. This translates into the requirement $10 g_s^{9/2} < a_{inf}^3$. To satisfy this, we need to take a string coupling which is somewhat smaller than 0.01, say $g_s = 0.006$. The bound is then saturated for an inflationary warp factor of $a_{inf} = 10^{-3}$. In this case the tunneling probability is 8×10^{-4} .

It would be interesting if there exist warped compactifications in which the background dilaton field is varying between the two throats. In this case it may be possible to have a smaller string coupling in the inflation throat, as would be desirable for the tunneling problem, while keeping the string coupling in the SM throat at a phenomenologically preferred value.

It is worth emphasizing that even when the tunneling probability for energetic KK modes is large enough for reheating, the lifetime of cosmic strings in the inflation throat is still cosmologically large. Copeland *et al.* [58] estimate the barrier penetration amplitude for a string to be

$$e^{-1/a_{\inf}^2} \sim e^{-10^6}$$
 (4.19)

which makes the strings stable on cosmological time scales.

4.5 Warped Reheating

From the previous sections we see that the endpoint of brane-antibrane inflation can be considered as a gas of nonrelativistic closed strings with mass M_i/g_s , density M_i^3 and decay rate $\Gamma \sim g_s M_i$, localized in the inflationary throat. These heavy states cascade down to massless gravitons through a sequence of KK gravitons, a fraction P (eq. 4.16) of which tunnel to the SM brane and decay into visible sector particle.

Initially there will be two relevant reheat temperatures: one for the massless gravitons, T_{grav} and one for the visible sector, T_{vis} . By the standard reheating estimate [125] we see that

$$T_{\rm grav} \sim 0.1 \, (\Gamma M_p)^{1/2} \sim 0.1 \, a_i^{1/2} \, \sqrt{M_s \, M_p} \sim 10^{-3} \, M_p \,,$$
 (4.20)

where the last estimate uses $a_i \sim 10^{-3}$. and $M_s \sim M_p/10$. On the other hand, since a fraction P of the false vacuum energy was converted to visible sector particles, we deduce that

$$T_{\rm vis} = P^{1/4} T_{\rm grav} \sim T_{\rm grav} / 10$$
 (4.21)

using the optimistic estimate of the previous section for P. This estimate is high enough to avoid potential problems to which a low reheat temperature can give rise. One should take this result with a grain of salt since it is marginally larger than both M_i and M_{sm} , and because it is larger than the string scale in the throats it invalidates the 4D field-theoretic calculation on which it is based. A more careful calculation must instead be based on a higher-dimensional, string-theoretic estimate of the energy loss, which goes beyond the scope of this article.

In conventional inflation models, such a high reheating temperature would be in conflict with the gravitino bound (overproduction of gravitinos, whose late decays disrupt big bang nucleosythesis). It is interesting in this regard that the KKLT scenario gives a very large gravitino mass, around $m_{3/2} = 6 \times 10^{10}$ GeV [114], which is so large that there is effectively no upper limit on the reheat temperature (see for instance ref. [133]). The disadvantage of such a large gravitino mass is that supersymmetry is broken at too high a scale to explain the weak scale of the SM. If SUSY is this badly broken, one possibility for explaining the weak hierarchy is that the large landscape of string vacua provides a finely-tuned Higgs mass, as well as cosmological constant, as has been suggested in ref. [134]. If this is the case, then the degree of warping in the SM model brane would not be crucial for determining the TeV scale, and the existence of an extra throat to contain the SM model brane would be unnecessary. However, given the large number of 3-cycles in a typical Calabi-Yau manifold, each of which can carry nontrivial fluxes, the existence of many throats should be quite generic, and it would not be surprising to find the SM brane in a different throat from the inflationary one.

4.6 Conclusions

We have argued that for brane-antibrane inflation in strongly-warped extra-dimensional vacua — such as have been considered in detail for Type IIB string models — there is a natural mechanism which channels a fraction of the released energy into reheating the Standard Model degrees of freedom. This is because a nonnegligible fraction of the false vacuum energy of the brane-antibrane system naturally ends up being deposited into massless modes on branes which are localized inside the most strongly-warped throats, rather than being dumped completely into massless bulk-state modes.

This process relies on what is known about brane-antibrane annihilation in flat space, where it is believed that the annihilation energy dominantly produces very massive closed-string states, which then quickly themselves decay to produce massive KK modes for massless string states. What is important for our purposes is that the wave functions for all of the massive KK modes of this type are typically exponentially enhanced at the bottom of warped throats, while those for the massless KK bulk modes are not. This enhancement arises because the energies of these states are minimized if their probabilities are greatest in the most highly warped regions. This peaking is crucial because it acts to suppress the couplings of the massive KK modes to the massless bulk states, while enhancing their couplings to brane modes in the most warped throats.

Although the couplings of the KK modes to SM degrees of freedom are enhanced, the KK modes must first tunnel from the inflation throat to the SM throat. This results in most of the energy density of the brane-antibrane system ending up as massless gravitons, and only a small fraction P going into visible matter. Nevertheless, for reasonable values of the string coupling and the warp factor of the inflationary throat, P can be as large as $10^{-3} - 10^{-4}$. Since the initial reheat temperature is high, there are many decades of evolution in temperature during which the decoupled gravitons can be diluted by events which increase the entropy of the thermalized visible sector particles relative to the gravitons. In this way it is quite plausible that big bang nucleosynthesis bounds on the energy density of gravitons can be satisfied.

From this point of view, it is possible to efficiently reheat the SM brane after braneantibrane inflation, so long as there are no other hidden branes lying in even deeper throats than the SM, which would have a larger branching ratio for visible sector decay than the SM. This observation is all the more interesting given the attention which multiple-throat inflationary models are now receiving, both due to the better understanding which they permit for the relation between the inflationary scale and those of low-energy particle physics, and to the prospects they raise for producing long-lived cosmic string networks with potentially observable consequences. We will discuss the formation and evolution of such cosmic string networks in chapter 5.

Acknowledgements

It is a pleasure to thank Shamit Kachru, Renata Kallosh, Andrei Linde, Juan Maldacena, Anupam Mazumdar, Liam McAllister, Rob Myers, Joe Polchinski, Fernando Quevedo, Raul Rabadán, Eva Silverstein, Horace Stoica and Henry Tye for fruitful discussions. This research is supported in part by funds from NSERC of Canada, FQRNT of Québec and McGill University.

Chapter 5

Overproduction of Cosmic Superstrings

Abstract

We show that the naive application of the Kibble mechanism seriously underestimates the initial density of cosmic superstrings that can be formed during the annihilation of D-branes in the early universe, as in models of brane-antibrane inflation. We study the formation of defects in effective field theories of the string theory tachyon both analytically, by solving the equation of motion of the tachyon field near the core of the defect, and numerically, by evolving the tachyon field on a lattice. We find that defects generically form with correlation lengths of order M_s^{-1} rather than H^{-1} . Hence, defects localized in extra dimensions may be formed at the end of inflation. This implies that brane-antibrane inflation models where inflation is driven by branes which wrap the compact manifold may have problems with overclosure by cosmological relics, such as domain walls and monopoles.

5.1 Introduction

Although it is notoriously difficult to test string theory in the laboratory, exciting progress has been made on the cosmological front through the realization that superstrings could appear as cosmic strings within recent popular scenarios for braneantibrane inflation [56, 58]. Gravity wave detectors and pulsar timing measurements could thus give the first positive experimental signals coming from superstrings [135] (see, however, [136]). If seen, it will be challenging to distinguish cosmic superstrings from conventional cosmic strings. One distinguishing feature is the possibility that superstrings have a smaller intercommutation probability than ordinary cosmic strings [137]. In this chapter we consider whether the mechanism of formation of cosmic superstrings might provide another source of distinction, by studying in detail the formation of string and brane defects, emphasizing differences with the standard picture of defect formation.

To set the stage, let us briefly recall how defect formation works in a standard theory [35] (see chapter 1 for a review). Consider a scalar field theory with potential $\frac{\lambda}{4}(|\phi|^2 - \eta^2)^2$. In the standard picture as the universe cools below some critical temperature T_c the U(1) symmetry is broken and ϕ rolls to the degenerate vacua of the theory $|\phi| = \eta$. Causality implies that the field cannot be correlated throughout the space and hence the field rolls to different vacua in different spatial regions leading to the formation of topological defects. The defect separation is set by the correlation length, ξ . For a universe expanding with Hubble rate H one expects $\xi < H^{-1}$ and hence a minimum defect density of about one per Hubble volume. However, this is only an upper bound on ξ . A more careful estimate can be made using condensed matter physics methods: equating the free energy gained by symmetry breaking with the gradient energy lost. Very close to T_c thermal fluctuations which can restore the symmetry are probable; however, once the universe cools below the Ginsburg temperature, T_G , there is insufficient thermal energy to excite a correlation volume into the state $\phi = 0$ and the defects "freeze out." For the scalar field theory described above this estimate yields a correlation length of order the microscopic scale: $\xi \sim \lambda^{-1}\eta^{-1}$.

The physics of tachyon condensation on the unstable Dp brane-antibrane system is quite different however, due to the peculiar form of the action [43]

$$S = -2\tau_p \int d^{p+1}x \, e^{-|T|^2/b^2} \sqrt{1+|\partial T|^2}.$$
(5.1)

The tachyon potential has a "runaway" form and there are no oscillations of the field near the true vacuum which can restore the symmetry. The decaying exponential potential of the complex tachyon field multiplies the kinetic term. Once condensation starts, T rolls quickly to large value, and damps the gradient energy exponentially. This essentially eliminates the restoring force which would tend to erase gradients within a causal volume in an ordinary field theory.

In this chapter we perform a quantitative analysis of the formation of string defects starting from the unstable tachyonic condensate that describes unstable braneantibrane systems. Having established that defects form with an energy density comparable to that which is available from the condensate, we then examine the possible cosmological consequences of this larger-than-expected initial density.

The organization of this chapter is as follows. In section 5.2 we briefly dis-

cuss brane-antibrane inflation and the formation of defects at the end of braneantibrane inflation by tachyon condensation. In section 5.3 we study the formation of codimension-one branes in the decay of a nonBPS brane both analytically as well as through lattice simulations. In section 5.4 we repeat the study, this time for the formation of codimension-two branes in the decay of the brane-antibrane system. We contrast these results with the formation of conventional cosmic strings in section 5.5. The reader who is interested in the cosmological implications may wish to skip directly to section 5.6 where we consider new constraints on brane inflation models which may arise due to the large initial density of defects. Conclusions are given in section 5.7. In an appendix we justify the assumed initial conditions for the tachyon fluctuations which lead to defects.

5.2 Brane Inflation

Relic cosmic superstrings can form at the end of inflation driven by the attractive interaction of branes and antibranes [24, 25], [53]. In this picture, inflation ends when the brane and antibrane (or a pair of branes oriented at angles) become sufficiently close that one of the open string modes stretching between the branes becomes tachyonic. The subsequent rolling of this tachyon field describes the decay of the brane-antibrane pair. Quantum fluctuations produce small inhomogeneities in the tachyon field, which will cause it to roll toward different vacua in different spatial regions, leading to the formation of topological defects which are known to be consistent descriptions of lower dimensional branes.

The branes which drive inflation must span the three noncompact dimensions and may wrap some of the compact dimensions. The defects which form are lower dimensional branes whose world-volume is within the world-volume of the parent branes. The argument was made in [56] (illustrated in figure 5.1) that by applying the reasoning of the Kibble mechanism to the formation of the lower-dimensional branes one concludes that the branes which are copiously produced as topological defects will always wrap the same compact dimensions that the parent branes wrap and hence these branes appear as cosmic strings to the 3-dimensional observer.¹ This argument is based on the fact that the size of the compact dimensions is orders of magnitude smaller than the Hubble distance during (and at the end of) inflation and hence there are no causally disconnected regions along the compact dimensions (a reasonable estimate is $H^{-1} \sim 6 \times 10^3 M_s^{-1}$ while typically the extra dimensions are compactified at a scale R closer to M_s^{-1}). Therefore along these directions the tachyon field would always roll toward the same vacuum and no topological defects would form. The causally disconnected regions occur only along the extended directions, so the defects are localized along the extended directions. An identical argument implies that cosmic strings should form with a density of about one string per Hubble volume.



Figure 5.1: Illustration of the Kibble argument in the case of branes spanning extended dimensions and wrapping compact ones. The compact dimension is much smaller in size than the Hubble distance, $1/H \gg \ell_{\parallel}$, so there are no causally disconnected regions along this dimension. The branes that will form as topological defects will wrap the same compact dimensions as the parent branes, and will be localized on the extended dimensions.

Reference [56] uses the analogue of the condensed matter physics argument paraphrased in section 5.1 to estimate the correlation length of the tachyon field at the end of inflation. Since the universe has zero temperature at the end of inflation the thermal fluctuations are replaced by the quantum fluctuations of de Sitter space: $H/(2\pi)$. In that analysis the potential depends on the brane separation and as a result of the relative motion of the brane and the antibrane, the curvature of the potential at T = 0 changes sign. It is also assumed that the tachyon potential, V(T),

¹The production of phenomenologically dangerous monopole-like and domain wall-like defects is suppressed.

has a minimum for some finite value of T. In this case there is an interplay between the correlation length, ξ_C given by the curvature of the potential at T = 0, and the Ginsburg length, ξ_G , which is the largest scale on which quantum fluctuations of the field can restore the symmetry. As the brane separation decreases and the shape of the potential changes, the correlation length and the Ginsburg length change as well. The density of defects freezes when $\xi_C = \xi_G$. This analysis yields a correlation length which is not much different from the Hubble scale: $\xi \sim 1.6 \times 10^4 M_s^{-1} \sim 2.7 H^{-1}$.

There are several reasons to consider a more quantitative study of defect formation at the end of brane inflation. Firstly, the estimate of one defect per Hubble volume provides only a lower bound on the defect density via causality. The actual network of defects produced is determined by the complicated tachyon dynamics as it rolls from the unstable maximum of its potential to the degenerate vacua of the theory. Second, the effective field theory which describes the dynamics of the tachyon [43] has a rather unusual causal structure [138] and the usual reasoning of the Kibble mechanism may not be applicable. In [138] it was found that in the case of the homogeneous rolling tachyon the small fluctuations of the field propagate according to a "tachyon effective metric" which depends on the rolling tachyon background. As the tachyon rolls towards the vacuum the tachyon effective metric degenerates, the tachyon light cone collapses onto a time-like half line and the tachyon fields at different spatial points are causally decoupled.

Finally we comment on the validity of the effective field theory description. At some point the effective field theory description of the decaying brane will become inappropriate and the correct degrees of freedom will be the decay products of the brane annihilation. Since the topological defects of the tachyon field form in a short time of order M_s^{-1} , [48, 1], the effective field theory description should be applicable during the period of defect formation. Furthermore, there are reasons to believe that the effective field theory is valid even at late times when the tachyon is close to the vacuum, since in this regime the tachyon effective field theory may give a dual description of the closed strings which are produced during brane decay [52].
5.3 Tachyon Kink Formation in Compact Spaces

In this section we consider the formation of codimension-one branes from tachyon condensation on a nonBPS brane. We study defect formation both by solving the full nonlinear equations of motion on a lattice, as well as providing analytical approximations to the behavior of the tachyon field in different regimes of interest. We study the evolution of the tachyon field starting from a profile which is close to the unstable vacuum T = 0 (in this respect our analysis is similar to [32, 33, 34]). In our analysis there is no parameter which causes continuous variation of the potential, we consider the brane-antibrane system to be coincident at t = 0 when the initial conditions are imposed (this potentially important difference should be kept in mind when comparing our results to [56]).

5.3.1 Action and Equation of Motion

We will work with the tachyon Dirac-Born-Infeld effective action [43, 42, 50]

$$S = -\int V(T)\sqrt{1 + \partial_{\mu}T\partial^{\mu}T}\sqrt{-g}\,d^{3+1+1}x.$$
(5.2)

The action (5.2), with $V(T) = V_0 / \cosh(T/\sqrt{2\alpha'})$, can be derived from string theory in some limit [46] and has been shown to reproduce various nontrivial aspects of the full string theory dynamics [50, 45]. Here we take the potential to be

$$V(T) = \tau_p \exp\left(-T^2/b^2\right) \tag{5.3}$$

where τ_p is the D*p*-brane tension. The constant *b* determines the tachyon mass in the perturbative vacuum as $M_T = \sqrt{2}b^{-1} \sim 1/\sqrt{\alpha'}$. Qualitatively the results will depend very little on the specific functional form of V(T). In fact, for much of the analytical analysis which follows we will not even need to make reference to the specific functional form of V(T).² We consider real tachyon fields in a spacetime with three expanding, noncompact dimensions $\{x^i\}$ (where i = 2, 3, 4) and one compact

²Provided of course that V'(T=0) = 0 and $V(T \to \pm \infty) \to 0$.

dimension $x^1 = x$ which we take to be static. The metric is

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -dt^2 + dx^2 + a(t)^2\delta_{ij}\,dx^i\,dx^j.$$
(5.4)

For simplicity we take the tachyon field to depend only on time, $x^0 = t$, and the compact spatial coordinate $x^1 = x$. The equation of motion which follows from (5.2) with the metric (5.4) is

$$(1+T'^2)\ddot{T} = T''\left(1-\dot{T}^2\right) - \left(3H\dot{T} + \frac{V'(T)}{V(T)}\right)\left(1-\dot{T}^2 + T'^2\right) + 2\dot{T}T'\dot{T}' \quad (5.5)$$

where $\dot{T} = \partial_0 T = \partial_t T$, $T' = \partial_1 T = \partial_x T$, $V'(T) = \partial V / \partial T$ and $H = \dot{a}/a$.

5.3.2 Lattice Simulation of Kink Formation

We have solved (5.5) on a lattice for different values of the compactification radius and the Hubble parameter H (which we take to be constant for simplicity). We consider vanishing initial velocity $\dot{T}(t = 0, x) = \dot{T}_i(x) = 0$. The initial profile T(t = 0, x) = $T_i(x)$ is a Fourier series truncated such that the minimum wavelength is comparable to the lattice spacing.³ The amplitudes of the Fourier coefficients are given by a random Gaussian distribution and the overall amplitude of the initial profile is chosen to be small compared to b.

As in [51] we find that the gradient of the tachyon field near the core of the kink becomes infinite in a finite time, forcing us to halt our evolution. After this point the codimension-one branes have formed and if one wants to follow the evolution beyond this time it is necessary to consider the dynamics between branes and antibranes in a compact space; the field theory description is no longer adequate.

Typically the initial profile crosses T = 0 at many points. In the early stages of the evolution the field begins to grow due to the small displacement from the unstable vacuum. During this phase of the evolution many of the small fluctuations of the field will straighten themselves out. Large Hubble damping tends to kill off the high

³See the appendix for a discussion of the validity of these initial conditions.

frequency fluctuations faster. At the end of this initial stage there may be several locations where the field stays pinned at T = 0, depending most crucially on the radius of compactification. The evolution very quickly enters a nonlinear regime in which the field begins to roll quickly towards $T \to +\infty$ where T > 0 and $T \to -\infty$ where T < 0.

The most important parameter for determining the density of the defect network once the daughter branes have formed is the radius of compactification, R. Perhaps surprisingly, we find that for a compactification radius as small as a few times b, tachyon kinks will form. Once the field enters the nonlinear regime the defect formation depends only on the local physics near the core of the kink. In figures 5.2, 5.3 we plot the tachyon field versus t and x, showing the formation of codimension-one branes from the decay of a nonBPS brane for radii of compactification $R = 8M_T^{-1}$ and $R = 15M_T^{-1}$. The Hubble constant is taken to be vanishing, H = 0, in these figures. We have also considered $H \neq 0$ and find that the Hubble damping has little effect on the final kink/anti-kink network. Hence we find that tachyon kinks do form in the compact directions even when the field is initially in causal contact throughout the extra dimension.



Figure 5.2: Formation of tachyon kinks for $R = 8M_T^{-1}$. The time axis is measured in units of M_T^{-1} .



Figure 5.3: Formation of tachyon kinks for $R = 15M_T^{-1}$. The time axis is measured in units of M_T^{-1} .

5.3.3 Analytical Study of Tachyon Kink Formation

Here we describe analytically the formation of tachyon kinks. Since the full equation of motion (5.5) is difficult to solve analytically for arbitrary initial data, we study the dynamics of (5.5) in different regimes: near the core of the defect, where the field stays pinned at T = 0, and away from the core, where T rolls towards the vacuum. To simplify our analysis we neglect the compact topology of the extra dimension x, which should be a good approximation since the defect solutions are infinitely thin and therefore highly localized.

Solutions Near the Core of the Defect

Here we briefly review the analytical studies of kink formation near the core of the defect presented in [1]. Consider initial data $T(t = 0, x) = T_i(x)$ and $\dot{T}(t = 0, x) = \dot{T}_i(x) = 0$. The field should start to roll where $T_i(x) \neq 0$ due to the small displacement from the unstable maximum V'(T) = 0, and furthermore it must stay pinned at T = 0 at the core of the kink. At t = 0 the equation of motion (5.5) is

$$\ddot{T}_i(x)\left(1+T'_i(x)^2\right) = T''_i(x) + \frac{2}{b^2}T_i(x)\left(1+T'_i(x)^2\right).$$

Clearly any point x_0 where $T_i(x_0) = T''_i(x_0) = 0$ will be a fixed point where $\ddot{T}(t, x_0) = 0 = \dot{T}(t, x_0)$ throughout the evolution. We restrict ourselves to considering only

initial data such that $sgn(\ddot{T}_i(x)) = sgn(T_i(x))$ for all x to ensure that the solutions are increasing. To analytically study the dynamics near x_0 it is therefore reasonable to make the ansatz

$$T(t,x) \cong u(t)(x-x_0).$$
 (5.6)

For $u(t) = \text{const} \to \infty$ this solution corresponds to the singular static soliton solution of Sen [50]. Inserting the ansatz (5.6) into (5.5) and working only to linear order in $(x - x_0)$ one obtains an ordinary differential equation for the slope

$$\ddot{u} = \frac{2}{b^2}u + 2\frac{u\dot{u}^2}{1+u^2} - 3H\dot{u}.$$
(5.7)

We have solved (5.7) numerically for both constant H and $H \sim 1/(t + t_0)$. We find that generically the solutions of (5.7) become singular in a finite time $t = t_c$. Larger H tends to delay the onset of the singularity and soften the singular behaviour. To understand this finite time slope divergence analytically it is simplest to consider H = 0. In this case if we assume that initially \dot{u} is close to zero then the second term on the right hand side of (5.7) can be neglected and therefore

$$u(t) \cong u_{+}e^{\sqrt{2}t/b} + u_{-}e^{-\sqrt{2}t/b}$$

at early times. Clearly \dot{u} grows quickly and the second term on the right hand side of (5.7) very quickly becomes important. In the regime where u and its derivatives become large the solution has the behaviour

$$u(t) \sim \frac{k}{t_c - t}$$

where the critical time t_c depends on the initial conditions. We find that within finite time the slope becomes singular and the time dependent tachyon field near the core of the kink coincides with the singular soliton solution of Sen [50]. Hence we conclude that the codimension-one brane forms in a finite time.

This finite-time slope divergence was observed both numerically and analytically in [51] and leads to a finite-time divergence in the stress-energy tensor. This divergence



Figure 5.4: Behavior of the slope at the core of the kink for different initial conditions. The time axis is measured in units of b.

was also found in an exact string theory calculation in [48]. For the potential (5.3) the stress-energy tensor near $x = x_0$ and $t = t_c$, with H = 0, is [1]

$$T_{00} \cong \frac{\tau_p k}{t_c - t} \exp\left(-\frac{k^2 (x - x_0)^2}{b^2 (t_c - t)^2}\right).$$

Hence, with the normalization $b = 2\sqrt{\pi \alpha'}$ proposed in [50], as $t \to t_c$ we have $T_{00} \to \tau_{p-1}\delta(x-x_0)$. Similarly one can show that in this regime $T_{01} = T_{11} = 0$ and that $T_{ij} = -\tau_{p-1}\delta_{ij}\delta(x-x_0)$. In the limit of condensation, then, the stress-energy near $x = x_0$ is identical to that of a D(p-1)-brane. If we take into account the rolling of the tachyon for $x \neq x_0$ then there will be an additional component to $T_{\mu\nu}$ corresponding to tachyon matter, as in [71].

Solutions Near the True Vacuum

Away from the site of the kink the field will roll towards the true vacuum $T \to \pm \infty$ so that $V(T) \to 0$ at late times for $x \neq x_0$. Solutions in this regime were discussed in detail in chapter 3.

5.4 Tachyon Vortex Formation in Compact Spaces

In this section we generalize the results of section 5.3 to consider the formation of codimension-two branes from tachyon condensation on the brane-antibrane pair. This

situation is of more direct interest to brane-antibrane inflation since inflation ends when a brane-antibrane pair annihilate. This situation is also more realistic since the stable branes in a given string theory are those whose dimension differs by a multiple of two.

5.4.1 Action and Equations of Motion

We wish to generalize the results of the previous section to consider complex tachyon fields which depend on time and two spatial coordinates which we compactify on a square torus. We generalize the action (5.2) to

$$S = -2\tau_p \int e^{-|T|^2/b^2} \sqrt{1 + \partial_\mu T \partial^\mu \bar{T}} \sqrt{-g} \, d^{3+2+1}x$$
(5.8)

with metric

$$g_{\mu\nu}dx^{\mu}dx^{\nu} = -dt^2 + dx^2 + dy^2 + a(t)^2\delta_{ij}dx^i dx^j$$

where $x^1 = x$ and $x^2 = y$ are Cartesian coordinates on the torus and x^i denotes the three noncompact dimensions, as in section 5.3. We consider tachyon fields which depend on $x^0 = t$, $x^1 = x$ and $x^3 = y$. If we separate the tachyon field into into real and imaginary components as

$$T(t, x, y) = T_1(t, x, y) + i T_2(t, x, y)$$

then (5.8) may be rewritten in terms of components as

$$S = -2\tau_p \int \exp\left(-T_J T_J / b^2\right) \sqrt{1 + \partial_\mu T_I \partial^\mu T_I} \sqrt{-g} \, d^{3+2+1} x$$

where the upper-case Roman indices label the real and imaginary components of the tachyon field (I = 1, 2). The equation of motion is (no sum over I, sum over J and K):

$$V\left(|T|\right)\partial_{\alpha}\left[\frac{\sqrt{-g}g^{\alpha\beta}\partial_{\beta}T_{I}}{\sqrt{1+\partial_{\alpha}T_{J}\partial^{\alpha}T_{J}}}\right] + \frac{\partial V\left(|T|\right)}{\partial T_{K}}\left[\frac{\sqrt{-g}\partial_{\alpha}T_{K}\partial^{\alpha}T_{I}}{\sqrt{1+\partial_{\alpha}T_{J}\partial^{\alpha}T_{J}}}\right] - \frac{\partial V\left(|T|\right)}{\partial T_{I}}\sqrt{-g}\sqrt{1+\partial_{\alpha}T_{J}\partial^{\alpha}T_{J}} = 0.$$
(5.9)

We note that the action (5.8) has not been derived from first principles and has several drawbacks from a theoretical perspective. For other proposals of the effective field theory on the brane-antibrane pair see [50],[139]-[141]. One theoretical difficulty is that the action (5.8) does not evade Derrick's theorem [67]. This is of no practical concern to us since we study time dependent solutions and since our interest is in defect formation in a compact space where charge conservation precludes the possibility of isolated defect solutions. More seriously, for the action (5.8) a static profile of the form T = u(x + iy) with $u \to \infty$ does not correctly reproduce the stress tensor for a codimension-two brane. We will attempt to address this difficulty by considering an alternative description of the complex tachyon dynamics in a subsequent subsection.

5.4.2 Lattice Simulations of Vortex Formation

We solve the system of two coupled partial differential equations (5.9). As in subsection 5.3.2 we find that the gradient of tachyon field becomes singular near the core of the defect in a finite time, forcing us to halt our lattice evolution. As in the case of the kink we choose as initial data $\dot{T}(t = 0, x, y) = \dot{T}_i(x, y) = 0$ and $T(t = 0, x, y) = T_i(x, y)$ given by a truncated Fourier series with random Gaussian coefficients with the overall amplitude small compared to b. We take H = 0 for our examples, since the Hubble damping plays little qualitative role in the dynamics. In figures 5.5, 5.6, 5.7 and 5.8 we plot $-|T| = -\sqrt{T_1^2 + T_2^2}$ against $\{x, y\}$ with the Taxis measured in units of b. Figure 5.5 shows typical initial conditions used for our numerical analysis. In figures 5.6, 5.7 and 5.8 we plot the final configurations close to $t = t_c$ when the gradients become infinite for various radii of compactification. (Note that because we are plotting -|T| rather than +|T| the vortices appear as spikes in the final configuration.) Again we find that vortices do form for radii of compactification as small as a few times M_T^{-1} , even though the field is initially in causal contact throughout the compact space.



Figure 5.5: Typical initial configuration for numerical studies of vortex formation.



Figure 5.6: The final configuration of the complex tachyon field for $R = 7M_T^{-1}$.



Figure 5.7: The final configuration of the complex tachyon field for $R = 15M_T^{-1}$.



Figure 5.8: The final configuration of the complex tachyon field for $R = 25M_T^{-1}$.

5.4.3 Analytical Study of Vortex Formation

As in subsection 5.3.3 we expect that in the case of the vortex there will exist fixed points where the field stays pinned at T = 0 throughout the evolution. To see this we consider the equation of motion for T_1 at t = 0 for initial data such that $\dot{T}(t = 0, x, y) = 0$. To simplify the expression we write the equation evaluated at a point (x_0, y_0) such that $T(t = 0, x_0, y_0) = 0$:

$$\begin{bmatrix} -\ddot{T}_1 + \partial_x^2 T_1 + \partial_y^2 T_1 \end{bmatrix} \begin{bmatrix} 1 + (\partial_x T_1)^2 + (\partial_y T_1)^2 + (\partial_x T_2)^2 + (\partial_y T_2)^2 \end{bmatrix}$$
$$- \begin{bmatrix} (\partial_x T_1)^2 \partial_x^2 T_1 + (\partial_y T_1)^2 \partial_y^2 T_1 + 2 \partial_x T_1 \partial_y T_1 \partial_x \partial_y T_1 \\ \partial_x T_1 \partial_x T_2 \partial_x^2 T_2 + \partial_y T_1 \partial_y T_2 \partial_y^2 T_2 + (\partial_x T_1 \partial_y T_2 + \partial_y T_1 \partial_x T_2) \partial_x \partial_y T_2 \end{bmatrix} = 0.$$

The equation for T_2 is identical with $1 \leftrightarrow 2$. It is clear, then, that any point (x_0, y_0) where $T_I(0, x_0, y_0) = \partial_x^2 T_I|_{(0,x_0,y_0)} = \partial_y^2 T_I|_{(0,x_0,y_0)} = \partial_x \partial_y T_I|_{(0,x_0,y_0)} = 0$ (for I = 1, 2) will be a fixed point of the evolution where $\ddot{T}_I(t, x_0, y_0) = \dot{T}_I(t, x_0, y_0) = 0$ for all t and hence the field T stays pinned at zero throughout the evolution. In the neighborhood of the point (x_0, y_0) we thus should be able to write the field in the form $T \cong u(t)(x - x_0) + v(t)(y - y_0)$ with u and v complex. Therefore to study analytically the dynamics of the field near the core of the vortex we make an ansatz of the type:

$$T_1(t, x, y) = u(t)(x - x_0), \qquad T_2(t, x, y) = u(t)(y - y_0).$$
(5.10)

We have chosen v(t) = i u(t) and u(t) real since the vortex solution is expected to take the form

$$T(t, z, \overline{z}) = u(t) \prod_{i} (z - z_i) \prod_{j} (\overline{z} - \overline{z}_i)$$
(5.11)

where we have defined complex coordinates z = x + iy, $\bar{z} = x - iy$. The profile (5.11), with $u(t) = \text{const} \to \infty$ was used for the multi-vortex solutions of [139] where it was shown that each holomorphic zero of (5.11) corresponds to a brane and each antiholomorphic zero of (5.11) corresponds to an antibrane. We insert now the ansatz (5.10) into the equations of motion (5.9) which corresponds to studying the dynamics close to the core of a single vortex located at $z_0 = x_0 + iy_0$. In this regime the equations of motion for the real and imaginary parts of the field, T_1 and T_2 , give the same equation for the slope near $z = z_0$:

$$\ddot{u} = \frac{2}{b^2} u \left(1 + u^2\right) + \frac{3 u \dot{u}^2}{1 + 2u^2}$$
(5.12)

where we have taken H = 0 for simplicity. We have verified numerically that this equation yields solutions which diverge in a finite time for generic initial data. In the regime where u(t) and its derivatives are large the dominant contribution to (5.12) is

$$\ddot{u} \cong \frac{2u^3}{b^2} + \frac{3}{2}\frac{\dot{u}^2}{u}$$
(5.13)

which has the solution

$$u(t) = \frac{b}{2(t_c - t)}.$$
(5.14)

where the critical time when the slope diverges, t_c , is fixed by the initial data. This singular behavior corresponds to the finite-time formation of a codimension-two brane in the annihilation of a brane-antibrane pair.

5.4.4 Alternative Complex Tachyon Action

The action (5.8) used above has the advantage of making the analysis tractable, and the resulting dynamics is analogous to kink formation. However, as discussed in subsection 5.4.1 this action has theoretical drawbacks. Here we consider an alternative description of the complex tachyon dynamics. The tachyon action has been calculation in boundary string field theory (BSFT) in [140] by assuming a linear tachyon profile. For a linear profile, gauge and space-time transformations allow one to write $T = u_1 x + i u_2 y$, and the action one derives is

$$S = -2\tau_p \int d^{p+1}x \, \exp\left[-2\pi\alpha' \left((u_1 x)^2 + (u_2 y)^2\right)\right] F(4\pi\alpha'^2 u_1^2) F(4\pi\alpha'^2 u_2^2) \quad (5.15)$$

where the function F(z) is given by

$$F(z) = \frac{4^z z \Gamma(z)^2}{2\Gamma(2z)}$$

To make our analysis tractable we generalize the action (5.15) to nonlinear profiles using two simplifications. The first is to consider a generalization which reduces to (5.15) for linear tachyon profiles only when $u_1 = u_2 = u$. We feel this is a reasonable simplification since for the profile $T = u_1x + i u_2y$ to minimize the action one requires both u_1 and u_2 to be infinite.

Our next simplification is to replace the complicated function F(z) by $\sqrt{1 + \pi z}$. As justification we note that these two functions have identical behavior at large z since $F(z) \to \sqrt{\pi z}$ for $z \to \infty$.

We consider, then, the action

$$S = -2\tau_p \int \sqrt{-g} \, d^{p+1} x \exp\left(-|T|^2/b^2\right) \left(1 + \partial_\mu T \partial^\mu \bar{T}\right) \tag{5.16}$$

where we have performed a rescaling of T and, for consistency with (5.15), $b = \sqrt{\pi \alpha'}$. This action has also been studied in connection with tachyon condensation in [64]. Writing the tachyon field in real and imaginary components as $T(t, x, y) = T_1(t, x, y) + iT_2(t, x, y)$ the equation of motion one derives from (5.16) is

$$g^{\mu\nu}\nabla_{\mu}\nabla_{\nu}T_{I} + \frac{1}{b^{2}}\left[T_{I}\left(1 + g^{\mu\nu}\nabla_{\mu}T_{K}\nabla_{\nu}T_{K}\right) - 2T_{K}g^{\mu\nu}\nabla_{\mu}T_{K}\nabla_{\nu}T_{I}\right] = 0.$$
(5.17)

We have solved the system of two coupled partial differential equations (5.17) on a lattice, as in subsection 5.4.2. The results are similar to those following from the action (5.8), though in this case the nonlinear effects are less dramatic. One still has defect formation, though on a longer time scale and defects begin to form for somewhat larger radius of compactification ($R \sim 30M_T^{-1}$ is sufficient). The qualitative

result that cosmic strings can form even for compactification scales well below the Hubble scale is unchanged. We consider now the dynamics near the core of the defect by plugging the ansatz

$$T_1(t, x, y) = u(t)(x - x_0), \quad T_2(t, x, y) = u(t)(y - y_0)$$

into (5.17) and working only to leading order in $(x - x_0)$, $(y - y_0)$. For H = 0 the equation for the slope of the field at the core of the kink is

$$\ddot{u} - \frac{1}{b^2}u = 0 \tag{5.18}$$

leading to an exponentially increasing slope. We see that in this case the slope does not become singular in finite time, which can result in a different final density of defects.

5.5 Comparison with Ordinary Cosmic String Formation

If the potential for the tachyon has the usual runaway form, $\exp(-T^2/b^2)$, then once the field starts rolling, it continues rolling towards $T = \pm \infty$. The gradient force is insufficient to stop or reverse the rolling. In this section we remind the reader how this differs from the mechanism for formation of defects in conventional field theories, where their production is much more strongly suppressed. The conventional case corresponds to a theory with a global U(1) symmetry, the potential $\frac{\lambda}{4} (|\phi|^2 - \sigma^2)^2$ and a standard kinetic term.

In the case of ϕ^4 theory, the evolution is well-behaved and one can follow the formation and the annihilation of kinks and anti-kinks indefinitely into the future. Comparing the two potentials $\exp(-T^2/b^2)$ and $\frac{\lambda}{4} (|\phi|^2 - \sigma^2)^2$ (see figure 5.9), we see that in the ϕ^4 theory one expects oscillations of the field which can restore symmetry and wipe out the defects.

In that case the final density of defects formed depends strongly on how fast the oscillations are damped, either through Hubble expansion, or through the coupling of



Figure 5.9: Comparison of the tachyon and ϕ^4 potentials. In the case of the ϕ^4 theory, the finiteness of the slope of the kink, as well as large oscillations of the field, strongly suppress the formation of defects.



Figure 5.10: The effect of damping on the formation of vortices in the ϕ^4 theory. Small damping results in a large number of oscillations of the field, and effective homogenization.



Figure 5.11: The effect of damping on the formation of vortices in the ϕ^4 theory. An unphysically large damping is used in order to show that the density of defects which survive is larger.

the field ϕ to other fields. In figures 5.10 and 5.11 we show the effect of the damping in the ϕ^4 theory for radius of compactification $R = 60 M_s^{-1}$. A large value of Hresults not only in a higher density of defects, but also slows down the motion of the defects that have formed, effectively reducing the rate at which they annihilate. This is in contrast to the tachyon field theory, in which case the Hubble damping plays essentially no role in determining the final density of defects.

5.6 Cosmological Consequences

5.6.1 String defects

Our studies of defect formation in compact space can easily be extrapolated to imply a density of roughly one defect per string volume in the non-compact directions aswell. We now turn to the cosmological implications of having a very large initial density of strings. In normal cosmic string networks, the details of the initial conditions are not important because the network quickly reaches a scaling solution. This can be demonstrated quite simply, using the "one-scale" model for the energy density ρ in long strings whose characteristic length is L [142]:

$$\dot{\rho} \cong -2H\rho - f(P)\frac{\rho}{L}.$$
(5.19)

The terms on the r.h.s. represent respectively the effects of dilution through expansion and the loss due to breaking off small loops, which eventually disappear by shrinking and emitting gravity waves. f(P) is a function of the intercommutation probability, which is believed to go like $f \sim \sqrt{P}$ [143, 144]. Taking $\rho = \mu/L^2$ (where μ is the string tension) and $H = \beta/t$, one can verify that this has a stable attractor solution

$$L = \gamma(t) t \equiv \frac{f(P) t}{2(1-\beta)}$$
(5.20)

known as the scaling solution, since the string length becomes a constant fraction of the horizon size H^{-1} , and the energy density in long strings also tracks that of the dominant component,

$$\rho = \frac{\mu}{\gamma^2 t^2}.\tag{5.21}$$

If the initial energy density was much greater than the scaling value, we can find the time scale for reaching scaling by solving (5.19) in the approximation that the Hubble expansion term is negligible compared to the loop-emission term, giving the solution $\gamma(t) = \gamma_0 + \frac{1}{2} f(P) \ln(t/t_0)$. Inverting, we find that the time required to reach a value $\gamma = f/(2-2\beta)$, starting from high densities where $\gamma_0 \ll 1$, is

$$t \cong t_0 \, e^{1/(1-\beta)} \tag{5.22}$$

which is not much greater than t_0 . For a radiation dominated universe, with $\beta = 1/2$, the scaling solution is reached in $e^2/2$ Hubble times in the usual case, and subsequent evolution is quite insensitive to the initial conditions. This conclusion is unchanged even for very small intercommutation probabilities.

We have investigated the approach to the scaling solution using a more detailed model of network evolution, which takes into account the possibility that loops may reconnect to long strings when the initial density is very high, and thereby possibly delay the onset of scaling. Suppose that the density of small loops with characteristic size l is $1/x^3$, defining the average distance between loops at a given time. One can estimate that the rate per unit volume for loops to recombine with long strings is

$$\frac{dn_s}{dt} = \tilde{f} \, x^{-3} \, L^{-3} \, l \, L^2 \, \frac{v_{\rm rel}}{\min(x, L)}.$$
(5.23)

Here \tilde{f} is the probability of a reconnection, x^{-3} and L^{-3} are the number densities of loops and long strings, respectively, $v_{\rm rel}$ is the relative velocity between loops and strings, which we take to be O(1), and $\min(x, L)$ is the distance a loop typically travels before reaching a string. The probability of a collision must be proportional to l, not l^2 , since the size of the loop in the direction parallel to the long string does not affect the cross section.

In this model, long strings and small loops are treated as two separate components, $\rho_s = \mu/L^2$ and $\rho_l = \mu l/x^3$, whose energy densities are governed by

$$\dot{\rho}_{s} = -2 H \rho_{s} - f \frac{\rho}{L} + \mu l \frac{dn_{r}}{dt}, \dot{\rho}_{l} = -3 H \rho_{l} + f \frac{\rho}{L} - \mu l \frac{dn_{r}}{dt} - \Gamma G \mu^{2} \frac{1}{x^{3}}.$$
(5.24)

The last term represents the power emitted by loops due to gravitational radiation, $\Gamma G \mu^2$ [146], where $\Gamma \cong 50$ and $10^{-11} \lesssim G \mu \lesssim 10^{-6}$ [56, 58]. The loop size is taken to always be a fixed fraction of the long string correlation length: $l/L = \Gamma G \mu$, so long as this is not smaller than the fundamental string length scale l_s . We have integrated these equations numerically, together with the Friedmann equation and the evolution equations for energy density in visible and gravitational radiation, keeping the short distance cutoff l_s on the size of the loops (in fact the results do not change noticeably if we assume the loops remain as small as this cutoff). This more detailed study confirms that the scaling solution is attained in only a few Hubble times, as can be seen from the time evolution of the fractions of the critical density for each component, shown in figure 5.12. We note, however, that in the above analysis we assumed that string velocities remain of order unity; if there is significant freezing out of the relative string motions in the large in the large dimensions this could have significant impact on the approach to scaling [142]. We note also that the effect of friction due to particle scattering may also significantly alter this picture [145].⁴

The initial density of strings is many orders of magnitude greater than the Kibble estimate, which gave the initial correlation length L as $\sim H^{-1}$ at the moment of formation of the network; instead, the initial energy density of the network is comparable

⁴We thank J.J. Blanco-Pillado and Carlos Martins for bringing this to our attention.



Figure 5.12: Fraction of critical density versus time for long strings, loops, and visible radiation starting from initial values $\Omega_s = 0.4$, $\Omega_l = 0.1$ and $\Omega_{rad} = 0.5$. Not shown is the fraction of energy density in gravity waves produced by decay of the loops.

to the total energy density available, $\sim \mu^2$, so that $L \sim \mu^{-1/2}$, smaller by a factor of $\mu^{-1/2}/M_p$ than the Kibble value. Since $\rho = \mu/L^2$, the initial density is greater by a factor of M_p^2/μ than the Kibble value. However, this huge excess is so shortlived that it has no observable effects. For instance, the contribution from the early nonscaling regime to the gravity wave background is negligible due to the small size of the loops which are formed and subsequently radiate during this era. Following ref. [146], one can estimate the amplitude of these gravity waves at frequency ω as being of order

$$h \sim \frac{(G\mu)^{7/6} \rho_0^{7/12}}{M_n^2 \omega^{1/3}} \sim 10^{-66}$$
(5.25)

where ρ_0 is the present energy density of the universe. The estimate (5.25) is some 40 orders of magnitude below the sensitivity of LIGO at the frequencies of interest, $\omega \sim 100$ Hz. As for the cosmic microwave background, the wavelength of density perturbations created during the nonscaling regime of the network is too short to be relevant: initially $\lambda \sim \mu^{-1/2}$, which gets stretched to the scale of the present energy density $\lambda \sim \rho_0^{-1/4} \sim 0.1$ mm. This length scale is also too small to be relevant for the formation of primordial black holes (PBH's) since the mass contained in volume λ^3 is far below that needed for cosmologically long-lived PBH's.

5.6.2 Domain Walls

Thus the fast approach to the scaling solution for three-dimensional string networks erases all sensitivity to the initial conditions, even though the initial density is orders of magnitude greater than for conventional cosmic strings, and this is true regardless of the reduced intercommutation probability. However, there are situations where our modified understanding of the network's initial conditions may make a dramatic difference: namely, in the case of higher dimensional defects. Let us illustrate with the example of D5- $\overline{\text{D5}}$ annihilation, where two of the dimensions are wrapped on an internal manifold with coordinates (y_1, y_2) and the remaining three dimensions span the euclidean space (x^1, x^2, x^3) . The codimension-two defects which form from the annihilation are D3 branes, and these can have various orientations with respect to the world-volume of the parent branes. The choices are exemplified by the three situations:

- 1. extended in x_3, y_1, y_2 directions, localized in x_1, x_2 : looks like D1 in 3D.
- 2. extended in x_2, x_3, y_2 directions, localized in x_1, y_1 : looks like D2 in 3D.
- 3. extended in x_1, x_2, x_3 directions, localized in y_1, y_2 : looks like D3 in 3D.

The first case looks like ordinary cosmic strings to the 3D observer since the defects have only one long direction among the three large ones. Their effects have already been discussed. Case 3 is a network of 3-branes, all of whose dimensions are large. Their tension will contribute to the effective 3D cosmological constant.⁵ Case 2, illustrated in fig. 5.13, is the interesting one because these appear as domain walls to the 3D observer, and their energy density redshifts too slowly: $\rho \sim 1/a^2$ in terms of the scale factor of the large dimensions. A single domain wall of tension $\tau_2 = \eta^3$ within our horizon would dominate the present energy density unless $\eta \lesssim 1$ MeV. The effective tension of a 3-brane wrapping one compact dimension of size R is $\tau_2 = R\tau_3 \sim \mu^{3/2}$; hence the string scale would have to be $\lesssim 1$ MeV, absurdly small.

⁵These can be safely assumed to annihilate quickly since they are not separated in the expanding space (x^1, x^2, x^3) .



Figure 5.13: An array of codimension-two D3-branes from D5-anti-D5 annihilation, partially localized in the compact dimensions, which look like strings in the x_1-y_1 plane, and domain walls in the large x_i dimensions.

Our conclusion differs from that of ref. [57], which speculated that no defects partially localized in compact directions can form. The argument was based on the Kibble mechanism, and thus assumed the correlation length could not be smaller than $H^{-1} \sim M_p/\mu$, whereas the size of the compact dimensions must be much smaller, of order $\mu^{-1/2}$. We have seen that in fact the initial correlation length is of the same order as the string scale, so that this argument is invalidated.

Still, one might be skeptical as to whether defects partially localized in the compact directions can survive until today, since intuitively they might be able to find each other and annihilate very quickly in the compact space. Ref. [147] attempts to address this question, and concludes that domain wall defects like those in case 2 cannot efficiently annihilate, since they are not completely localized in the compact (nonexpanding) space. The analysis of [147] assumes that the number density of D*p*-branes satisfies a rate equation which is nearly the same as that governing monopoles:

$$\dot{n} = -(3 - p_{||})Hn - \frac{\Gamma}{T^{D-2}}n^2$$
(5.26)

where $p_{||}$ is the number of dimensions of the brane spanning the large dimensions, and D is the total number of spacetime dimensions. This ignores the effect of selfintersections for reducing the density of long defects, which is known to be the dominant means for string networks to reach scaling (cf. eq. (5.19)). Further, it unrealistically assumes that the defects are parallel, so that they will annihilate rather than intercommute when they meet. It is therefore not immediately clear how far we can trust their conclusions [147].

On the other hand, numerical evolution of domain wall networks shows that selfintersections are not generic, and it is suggested that the dominant energy-loss mechanism is direct gravitational radiation rather than through collisions [148]. Furthermore it is observed that the approach to the scaling solution is slower for domain walls than for cosmic strings [149].

5.6.3 Full String Network Simulations

To attempt to address the issue of whether domain walls disappear or not, we have considered the dynamics of D3-branes in (5 + 1) dimensions in the approximation of projecting out the dynamics in one compact direction, y^2 , and one noncompact direction, x^3 , to give an effective 3 + 1-dimensional system. In this case the D3branes appear as one-dimensional objects (see figure 5.13) and the dynamics can be modeled by considering string evolution in an anisotropic space with two large and one small dimension. In this setup those "strings"—string-like from the (x^1, x^2, y^1) point of view—which span the large dimensions x^1 and x^2 appear as domain walls to the 3D observer while the "strings" which span the compact dimension y^1 appear as cosmic strings in 3D.

Following the setup of Smith and Vilenkin [150] we performed numerical simulations of the defect evolution in this approximation, keeping track of the extent to which defects preferentially spanned the compact direction y^1 —thus appearing as strings in the x^1, x^2, x^3 subspace, relative to spanning the large directions, which appear as domain walls. Figure 5.14 shows the fractional energy density in wound strings wrapped about the three directions x^1, x^2, y^1 as a function of time. This particular run started with equal energy in each direction so that the correlation lengths, and hence the interaction rates, were roughly the same. We observe that the branes wrapping the large directions lose their winding energy more quickly than those wrapping the compact direction: this can be attributed to several factors, including the smaller cross-sectional area of the long strings, and the greater energy radiated away when two long strings annihilate. The implication is that defects which look like strings to the 3D observer tend to survive preferentially over those that appear as domain walls. However since this is a toy model for the actual higher-dimensional defects, such a conclusion awaits validation from actual domain wall simulations.



Figure 5.14: Fraction of wound string density about the three anisotropic directions (two large and one small) starting from equal energy density in strings wrapping each direction. Top curve is for strings winding in the small direction.

As further evidence that defect evolution after formation tends to favor the defects which are localized in the large dimensions (and hence tends to favor the survival of cosmic string-like defects) we consider a field theory simulation of domain wall evolution for a real scalar field with potential $\frac{\lambda}{4} (\phi^2 - \sigma^2)^2$ and standard kinetic term ⁶ We studied the formation and evolution of domain walls in this theory in an anisotropic (3 + 1)-dimensional space with two large and one compact dimension.

⁶Though it is of more direct interest to consider the dynamics of the tachyon field theory with action (5.2) the finite time slope divergence prevents us from following the dynamics after the defects have formed.

Starting from a random initial profile close to the false vacuum (as in subsection 5.3.2) we find that the evolution of the defects after formation tends to annihilate domain walls which are localized in the compact direction and to favor the survival of domain walls which are localized in the large directions. Figure 5.15 shows a plot of the domain wall network late in the evolution after the domain walls localized in the compact direction have disappeared.

We also explored the effect of increasing the anisotropy, varying the initial distribution of winding modes, and varying the intercommutation probability; however, the system consistently evolved to favor winding about the compact direction. These results seem to corroborate the claim that domain wall-like defects are suppressed.



Figure 5.15: Plot of domain wall network in an anisotropic space with two large and one small dimensions with $L = 80 M_s^{-1}$. This snapshot is taken late in the evolution and shows that domain walls which are localized in the large dimensions are preferred by the defect evolution.

We note that there are several factors which we have not taken into account which may radically alter the picture. For example, in figure 5.14 we have unrealistically neglected the dynamics in two dimensions to facilitate numerical studies. Furthermore, in both of the examples above we have neglected the expansion of the large dimensions—an effect which could play an important role in the dynamics. It is therefore unclear whether domain walls will pose a cosmological problem in models of brane-antibrane inflation where the branes driving inflation wrap the compact space. We feel that this is a problem which merits further investigation and it is likely that a complete resolution of this issue will require full simulations of 3-brane dynamics in an anisotropic (5 + 1)-dimensional spacetime with 3 expanding dimensions. While such simulations would be of interest, they are beyond the scope of the present work.

5.6.4 Monopoles

Finally consider an example of how monopole-like defects may be formed through a cascade of annihilations in $D5-\overline{D5}$ inflation. The initial state D5 and $\overline{D5}$ span the three large dimensions and wrap two compact dimensions. These may produce D3 and $\overline{D3}$ which wrap the two compact dimensions and are extended in one large dimension; hence these defects appear string-like to the 3-dimensional observer. A parellel D3- $\overline{D3}$ pair may then annihilate to produce D1 branes and antibranes which can span the same large dimension as the parent $D3-\overline{D3}$, or alternatively could wrap the compact dimensions. Those D1-branes which span the large dimension appear as cosmic string defects to the 3-dimensional observer while those which wrap the compact dimensions will appear as point-like (monopole) defects. If the compact dimensions admit nontrivial 1-cycles (like T^2 for example) then these monopoles will be stable. Our results indicate that in general both string-like and monopole-like defects should be produced in this cascade. To understand the subsequent evolution of such defects we consider numerically the dynamics of D1-branes in a 3D space with one large dimension and two small dimensions. Figure 5.16 shows the fractional energy density in wound strings wrapped about the three directions x^1, y^1, y^2 as a function of time. Strings wrapping the large direction x^1 appear as genuine cosmic strings to the 3D observer while strings wrapping small directions y^1, y^2 appear as monopole-like defects in 3D. As in subsection 5.6.2 we start with equal energy in each direction so that the correlation lengths are roughly equal. Again we find that strings wrapping the large direction lose their winding energy quicker than those wrapping the compact directions. Physically, this suggests that monopoles are preferentially produced over cosmic strings in this particular cascade of annihilations.

The question of whether such monopole-like defects will pose a cosmological problem depends crucially on the long-range forces between these defects and is, we feel, an issue which merits further investigation (see [151] for a solution of the monopole problem which is independent of inflation).



Figure 5.16: Fraction of wound string density about the three anisotropic directions (one large and two small) starting from equal energy density in strings wrapping each direction.

5.7 Summary and Conclusions

We have studied the formation of topological defects during the decay of a nonBPS brane or a coincident brane-antibrane pair. The problem was treated both analytically, by solving the equation of motion for the tachyon field at the core of the defect, as well as numerically, by evolving the tachyon field on a lattice. We showed that defects form with a correlation length proportional to the string length rather than the horizon. Defects localized within compact dimensions can form even if the compactification radius is as small as $7M_S^{-1}$ and the tachyon dynamics is insentitive the the Hubble damping. Depending on the exact form of the action (e.g., Sen's version,or the boundary string field theory version) the slope of the field at the defect could either increase exponentially in time, or else diverge within a finite time, potentially changing the initial density of defects. We compared the evolution of the tachyon field to that of a scalar field in the ϕ^4 theory and noted that the most efficient way to suppress the formation of defects is through symmetry restoration, caused by large oscillations of the scalar field. This is not possible if the potential has a runaway form, as for the string tachyon, which inevitably leads to the formation of a higher density of defects in the string case. Once the defects form the field theory description is no longer adequate, so in order to analyze the annihilation of the defects formed one has to use a description in terms of branes and antibranes interacting in a compact space.

As a result, the initial density of string defects is much greater than previous estimates. For strings which are genuine 1D objects, we showed that the string network nevertheless attains scaling behavior within just a few Hubble times, so that there are no observable consequences of the initial high string density. Of course this assumes that the network is not frustrated [152], so that scaling can indeed be achieved. Whether this is the case for cosmic superstrings which are bound states of fundamental and D-strings is an interesting open question [153].

On the other hand, we argue that in models where inflation is driven by branes which wrap the compact manifold (for example $D5-\overline{D5}$), domain wall-like and monopolelike defects are inevitably produced. The stability and subsequent evolution of such defects is complicated and may depend crucially on the details of the compactification, for example, on whether or not the compact manifold which the parent branes wrap admits nontrivial 1-cycles. We leave detailed studies of whether such models are phenomenologically viable to future investigations; however, we have shown that the formation of defects at the end of brane-antibrane inflation is much more complicated and model-dependent than one might have expected.

Acknowledgments

We would like to thank Robert Brandenberger, J.J. Blanco-Pillado, Guy Moore and Henry Tye for discussions. This work was supported in part by NSERC and FQRNT.

Chapter 6

Nongaussianity from Tachyonic Preheating in Hybrid Inflation

Abstract

We show that in hybrid inflation it is possible to generate large second-order perturbations in the cosmic microwave background due to the instability of the tachyonic field during preheating. We carefully calculate this effect from the tachyon contribution to the gauge-invariant curvature perturbation. We clarify some confusion in the literature concerning nonlocal terms in the tachyon curvature perturbation; we show explicitly that such terms are absent. We find that large nongaussianity occurs only when the tachyon remains light throughout inflation, whereas n = 4 contamination to the spectrum is the dominant effect when the tachyon is heavy. We compute quantitatively the amount of nongaussianity or spectral distortion generated during preheating and use this calculation to place constraints on the parameter space of hybrid inflation. We apply our results also to popular brane-antibrane, F-term and D-term models of inflation. In the case of warped-throat brane-antibrane inflation we find interesting constraints from nongaussianity. For F-term and D-term inflation models from supergravity, we obtain nontrivial constraints from the spectral distortion effect. Finally, we also establish that our analysis applies to complex tachyon fields.

6.1 Introduction

In the simplest models of inflation, the primordial density perturbations have a negligible degree of nongaussianity. The parameter f_{NL} which characterizes nongaussianity is of the order of $|n-1| \ll 1$ (where *n* is the spectral index) in conventional inflation models [156]-[158], [18] whereas the current experimental limit is $|f_{NL}| \leq 100$ [16]; one can additionally characterize the nongaussianity using the trispectrum, which is also small in conventional models [159]. Nevertheless, there have been intense theoretical efforts to find models which predict observably large levels [160] (see chapter 1 for a review). These efforts are motivated, in part, by the fact that f_{NL} can potentially be used as a powerful tool to discriminate between various theoretical realizations of inflation.

It has been difficult to find examples which can give large f_{NL} . In single field inflation models a small inflaton sound speed is necessary to achieve large nongaussianity [161], as in the models of [162, 163] (unless the inflaton potential has a sharp feature [164]). It is also possible to generate significant nongaussianity using the curvaton mechanism [165] for the generation of cosmological perturbations. Indeed, the curvaton scenario may be ruled out by future non-observation of nongaussianity [166] (see, however, [167]). The simplest multi-field models do not seem to give large nongaussianity [168], though it is not clear if this is true also of more complicated models.

We will show that one of the most prevalent classes of models, hybrid inflation [31, 169], is able to yield large nongaussianity for certain ranges of parameters (see [170] for constraints on the parameter space of hybrid inflation coming from the WMAP data). The effect is due to the growth of the waterfall field—tachyonic preheating—which contributes to the curvature perturbation, and hence the temperature anisotropy, only starting at second order in cosmological perturbation theory; see also [171]-[177].

The scenario is summarized as follows. During hybrid inflation the inflaton field φ is displaced from its preferred value, and inflation is driven by the false vacuum energy of the "waterfall" or tachyon field σ which is trapped in its false vacuum by interactions with the inflaton. When the inflaton reaches some critical value the tachyon effective mass squared becomes negative and its fluctuations grow exponentially due to the spinodal instability. The exponential growth continues until the fluctuations of the tachyon start to oscillate about the true vaccuum. At this stage the tachyon fluctuations become nonperturbative and their back-reaction brings inflation to an end. Here we study the evolution of the second-order curvature perturbation due to the tachyonic instability and find that it can lead to a large level of nongaussianity, if the tachyon was lighter than the Hubble scale $m_{\sigma} < 3H/2$, during a sufficient part of the inflation (see section 1.1.1), however, in this case it is still possible for preheating to impact the Cosmic Microwave Background (CMB) in a nontrivial way (if the

exponential growth of the tachyonic fluctuations during preheating is so large that it overcomes the exponential damping during inflation). In the case of a heavy tachyon the preheating phase generates a nonscale-invariant (n = 4) contribution to the power spectrum. For certain parameter values this nonscale-invariant contribution can be made so large that it dominates over the scale invariant contribution coming from the inflaton.

The possibility that preheating may impact the large scale curvature perturbation has also been considered in [178]-[181] (though these studies were restricted to first order in perturbation theory). In particular, [179, 181] showed that, contrary to previous claims, there is no violation of causality necessary for preheating to amplify the large scale curvature perturbation. This is so because inflation has already set up correlations on large scales, the preheating dynamic simply amplifies those existing large scale fluctuations.

We apply our results for nongaussianity and spectral distortion to popular models of hybrid inflation including brane inflation [24] and P-term inflation [182], which is a synthesis of supergravity inflationary models interpolating between F-term and D-term inflation. In the case of brane inflation we find that large nongaussianity is possible, while for P-term inflation spectral distortion is the dominant effect.

Finally we demonstrate that our analysis applies not only to real tachyon fields but also to the case of complex tachyon fields. The latter situation is more realistic since in that case cosmic strings, rather than domain walls, are formed at the end of inflation.

We begin in section 6.2 by defining the perturbations up to second order in the metric and matter fields. In section 6.3 the hybrid inflations model is reviewed, starting with the dynamics of the fields at zeroth order, and then their first order perturbations, with emphasis on the dynamics of the tachyon field fluctuations toward the end of inflation. In section 6.4 we solve for the second order curvature perturbation which is induced by the tachyonic fluctuations. This is used in section 6.5 to compute the bispectrum (three-point function) and the nonlinearity parameter f_{NL} as well as contributions to the spectrum itself. In section 6.6 we incorporate experimental

constraints on nongaussianity, as well as on the spectrum, to derive excluded regions in the parameter space of the hybrid inflation model. We then adapt these constraints to the cases of brane-antibrane inflation in section 6.7 and P-term inflation in section 6.8. We further extend our analysis to the more realistic case of a complex tachyon field in section 6.9, showing that the extra components of the tachyon add in a simple way and modify the real-field results by factors of order unity. Conclusions are given in section 6.10. Appendix D-1 gives details about the matching between early- and late-time WKB solutions of the tachyon fluctuation mode functions. Appendix D-2 gives details about the second order perturbed Einstein equations. Appendix D-3 deals with the construction of the inflaton curvature perturbation. Appendix D-4 discusses the Fourier transforms of convolutions. Appendix D-5 provides technical details concerning the constructions of the tachyon curvature perturbation. Finally, appendix D-6 gives details about the source term of the curvature perturbation for complex tachyons.

6.2 Metric and Matter Perturbations

In this section we write down the perturbations of the metric and matter fields about a spatially flat Robertson-Walker background following [157]. Greek indices run over the full spacetime $\mu, \nu = 0, 1, 2, 3$ while latin indices run only over the spatial directions i, j = 1, 2, 3. We will frequently employ conformal time τ , related to cosmic time t by $dt = ad\tau$. Differentiation with respect to conformal time will be denoted by $f' = \partial_{\tau} f$ and with respect to cosmic time by $\dot{f} = \partial_t f$.

The metric is expanded up to second order in fluctuations as

$$g_{00} = -a(\tau)^2 \left[1 + 2\phi^{(1)} + \phi^{(2)} \right]$$
(6.1)

$$g_{0i} = a(\tau)^2 \left[\partial_i \omega^{(1)} + \frac{1}{2} \partial_i \omega^{(2)} + \omega_i^{(2)} \right]$$
(6.2)

$$g_{ij} = a(\tau)^{2} \left[(1 - 2\psi^{(1)} - \psi^{(2)})\delta_{ij} + D_{ij}(\chi^{(1)} + \frac{1}{2}\chi^{(2)}) + \frac{1}{2}(\partial_{i}\chi_{j}^{(2)} + \partial_{j}\chi_{i}^{(2)} + \chi_{ij}^{(2)}) \right]$$

$$(6.3)$$

where $D_{ij} = \partial_i \partial_j - \frac{1}{3} \delta_{ij} \partial^k \partial_k$ is a trace-free operator. The fluctuations are decomposed such that the vector perturbations are transverse $\partial^i \omega_i^{(2)} = \partial^i \chi_i^{(2)} = 0$ while the tensor perturbations are transverse, traceless and symmetric: $\partial^i \chi_{ij}^{(2)} = 0$, $\chi_i^{i(2)} = 0$, $\chi_{ij}^{(2)} = \chi_{ji}^{(2)}$. In the above we have neglected the vector and tensor perturbations at first order, which are small, since vector perturbations decay with time, while tensors are suppressed by the slow roll parameter ϵ . The same is not true at second order, however, since the second order tensors and vectors are sourced by the first order scalar perturbations. We adopt the generalized longitudinal gauge defined by $\omega^{(1)} = \omega^{(2)} = \omega_i^{(2)} = 0$ and $\chi^{(1)} = \chi^{(2)} = 0$. The metric in this gauge becomes

$$g_{00} = -a(\tau)^2 \left[1 + 2\phi^{(1)} + \phi^{(2)} \right]$$
(6.4)

$$g_{0i} = 0$$
 (6.5)

$$g_{ij} = a(\tau)^{2} \left[(1 - 2\psi^{(1)} - \psi^{(2)}) \,\delta_{ij} + \frac{1}{2} (\partial_{i} \chi_{j}^{(2)} + \partial_{j} \chi_{i}^{(2)} + \chi_{ij}^{(2)}) \right].$$
(6.6)

In hybrid inflation the matter content consists of two scalar fields which are expanded in perturbation theory as

$$\varphi(\tau, \vec{x}) = \varphi_0(\tau) + \delta^{(1)}\varphi(\tau, \vec{x}) + \frac{1}{2}\delta^{(2)}\varphi(\tau, \vec{x})$$
(6.7)

$$\sigma(\tau, \vec{x}) = \sigma_0(\tau) + \delta^{(1)}\sigma(\tau, \vec{x}) + \frac{1}{2}\delta^{(2)}\sigma(\tau, \vec{x}).$$
(6.8)

where φ is the inflaton and σ the tachyon (or "waterfall" field). In hybrid inflation the time-dependent vacuum expectation value (VEV) of the tachyon field is set to zero $\sigma_0(\tau) = 0$, about which we will say more later.

The perturbations are defined so that $\langle \delta^{(i)}\varphi \rangle = 0$, hence $\langle \varphi(\tau, \vec{x}) \rangle = \varphi_0(\tau)$. At first order in perturbation theory this is automatic since $\delta^{(1)}\varphi$ contains only one annihilation/creation operator. However, at higher order in perturbation theory the homogeneous k = 0 mode of the fluctuation must be subtracted by hand in order to ensure that all of the zero mode of the field is described by the nonperturbative background.

The Einstein tensor and stress-energy tensor expanded up to second order in perturbation theory can be found in [183]. We do not reproduce these results here,

but we have carefully checked all the results from [183] which are relevant for our analysis.

6.3 Hybrid Inflation

We consider hybrid inflation in which both the inflaton and the tachyon are real fields with the potential

$$V(\varphi,\sigma) = \frac{\lambda}{4} \left(\sigma^2 - v^2\right)^2 + \frac{m_{\varphi}^2}{2}\varphi^2 + \frac{g^2}{2}\varphi^2\sigma^2.$$
(6.9)

This potential will give rise to topological defects at the end of inflation—domain walls in the σ field—which could produce large nongaussianities apart from the ones which we consider. To avoid overclosure of the universe by domain walls we must either render the domain walls unstable (for example through the addition of a small term like $\mu\sigma^3$ to the potential) or else consider the generalization to complex tachyon field. In the case of a complex tachyon the symmetry breaking process will lead to the formation of cosmic strings, which are much more phenomenologically viable than domain walls. We will return to this possibility in a subsequent section, showing that our analysis generalizes to complex fields in a straightforward manner.

6.3.1 Background Dynamics

At the homogeneous level, the usual Friedmann and Klein-Gordon equations for the scale factor and the matter fields are

$$3H^2 = \frac{\kappa^2}{2} \left(\dot{\varphi}_0^2 + \dot{\sigma}_0^2 \right) + \kappa^2 V, \qquad (6.10)$$

$$0 = \ddot{\varphi}_0 + 3H\dot{\varphi}_0 + \frac{\partial V}{\partial \varphi} \tag{6.11}$$

where $\kappa^2 = M_p^{-2} = 8\pi G_N$. Here and elsewhere the potential and its derivatives are understood to be evaluated on the background values of the fields so that $V = V(\varphi_0, \sigma_0)$ and $\partial V/\partial \varphi = \partial V/\partial \varphi|_{\{\varphi=\varphi_0,\sigma=\sigma_0\}}$, for example. For the potential (6.9) we have $\partial V/\partial \sigma = \partial^2 V/\partial \sigma \partial \varphi = 0$, provided that $\sigma_0 = 0$. We will apply this simplification to all subsequent results. To see why one should set $\sigma_0(\tau) = 0$, notice that initially the tachyon effective mass is $m_{\sigma}^2 = g^2 \varphi_0^2 - \lambda v^2 > 0$. That is, the tachyon mass squared starts out being positive during inflation. Provided that there was a long enough prior period of inflation, any initial departure of σ from zero would be exponentially damped. At some point, m_{σ}^2 becomes negative, and the tachyonic instability begins. However, it is still true that $\sigma_0(\tau)$ remains zero even then, since the universe will consist of equal numbers of domains with $\sigma > 0$ and $\sigma < 0$. On average, these give zero, which is the definition of the zeroth order field σ_0 . The departures of σ from zero between domain walls which form are taken account in the fluctuations of the field. Thus it is consistent to set $\sigma_0(\tau) = 0$ for all times in our analysis.

Notice that we do not need to replace the average of the fluctuations $\langle (\delta^{(1)}\sigma)^2 \rangle^{1/2}$ with an effective homogeneous background $\sigma_0(\tau)$ because we work to second order in perturbation theory and the effect of these fluctuations enters into the calculation through the second order perturbed energy momentum tensor $\langle \delta^{(2)}T^{\mu}_{\nu} \rangle$. When $\langle \delta^{(2)}T^{\mu}_{\nu} \rangle$ becomes sufficiently large the backreaction will stop inflation. We take $\langle (\delta^{(1)}\sigma)^2 \rangle^{1/2} = v/2$ (below) as our criterion for the end of inflation. We have checked numerically that this is a somewhat more stringent constraint than demanding that the energy density in the fluctuations $\delta^{(1)}\sigma$ does not dominate over the false vacuum energy which drives inflation $\lambda v^4/4$.

We will make extensive use of the slow roll parameters, defined by

$$\epsilon = \frac{\kappa^2}{2} \frac{\dot{\varphi}_0^2}{H^2} = -\frac{\dot{H}}{H^2} \cong \frac{M_p^2}{2V^2} \left(\frac{\partial V}{\partial \varphi}\right)^2$$
$$\epsilon - \eta = \frac{\ddot{\varphi}_0}{H\dot{\varphi}_0} \cong \epsilon - \frac{M_p^2}{V} \left(\frac{\partial^2 V}{\partial \varphi^2}\right)$$

so that, during inflation

$$\eta \cong 4 \frac{M_p^2 m_{\varphi}^2}{\lambda v^4}, \qquad \epsilon \cong 8 \left(\frac{M_p m_{\varphi}^2 \varphi_0}{\lambda v^4}\right)^2 \tag{6.12}$$

Notice that if $m_{\varphi}^2 \varphi_0^2 \ll \lambda v^4$, then $\epsilon \ll \eta$. This is equivalent to demanding that the false vacuum energy of the tachyon dominates during inflation, which is the assumption usually made for hybrid inflation:

$$V(\varphi_0, \sigma_0 = 0) = \frac{\lambda v^4}{4} + \frac{m_{\varphi}^2}{2}\varphi_0^2 \cong \frac{\lambda v^4}{4}.$$

During the slow roll phase the inflaton equation of motion $3H\dot{\varphi}_0 + m_{\varphi}^2\varphi_0 \cong 0$ has solution

$$\varphi_0(t) = \varphi_s \exp\left(-\frac{m_\varphi^2(t-t_s)}{3H}\right) = \varphi_s \left(\frac{a(t)}{a_s}\right)^{-\eta} \tag{6.13}$$

where we used $a(t) = a_s e^{H(t-t_s)}$, with t_s an arbitrary time. The Hubble scale remains approximately constant, $3H^2 \cong \lambda v^4/(4M_p^2)$. Since φ_0 is decreasing and $\sigma_0 = 0$, the slow roll parameter ϵ actually decreases slowly during inflation while η remains constant.

For $\sigma_0 = 0$ the first order metric and inflaton perturbations obey exactly the same equations as in single field inflation. Hence, the analysis presented in chapter 1 all applies without modification. Indeed, the results of section 1.1.1 will come in handy since similar equations will arise when we study the second order metric perturbations.¹

6.3.2 Conditions for a slowly varying tachyon mass

Now we come to an important point for this chapter, that if η is sufficiently small, then the tachyon is a light field during some part of the observable period of inflation. The tachyon mass is given by $m_{\sigma}^2 = -\lambda v^2 + g^2 \varphi_0^2(t)$. If we choose the arbitrary time t_s in (6.13) to be when $m_{\sigma}^2 = 0$, then $g^2 \varphi_s^2 = \lambda v^2$, and

$$m_{\sigma}^{2} = -\lambda v^{2} \left(1 - \left(\frac{a(t)}{a_{s}} \right)^{-2\eta} \right) \cong -2\eta \lambda v^{2} H(t - t_{s})$$
$$= -2\eta \lambda v^{2} N$$
(6.14)

where N is the number of e-foldings of inflation occurring after the tachyonic instability begins. At some maximum value $N = N_*$, inflation will end. If the inflaton rolls slowly enough, then the tachyon mass remains close to zero for a significant number

¹As will be shown, the second order metric fluctuation $\phi^{(2)}$ obeys an inhomogeneous equation where the differential operator is identical to the one which determines the dynamics of $\phi^{(1)}$; thus an understanding of the first order solutions simplifies the construction of the Green function for the second order fluctuations.

of e-foldings. In general, we will have $-N_i \equiv N_e - N_*$ e-foldings of inflation before the spinodal time, followed by N_* e-foldings during the preheating phase.

The approximation in (6.14), that the tachyonic mass changes slowly enough for its time dependence to be approximated as linear, is true so long as $|2\eta N| \ll 1$. This can be rephrased using the definition of η in (6.12), and eliminating m_{φ}^2 using the COBE normalization of the inflationary power spectrum: $V/(M_p^4\epsilon) \approx 150\pi^2(2 \times 10^{-5})^2 =$ 6×10^{-7} (see chapter 1). Using $V = \frac{1}{4}\lambda v^4$ and eq. (6.12) for ϵ , the COBE normalization gives

$$m_{\varphi}^2 \cong 230 \, g\lambda \, \frac{v^5}{M_p^3} \tag{6.15}$$

Then with $N \sim 60$, the requirement $|2\eta N| < 1$ becomes

$$g < 10^{-5} \, \frac{M_p}{v} \tag{6.16}$$

Interestingly, the bound (6.16) turns out to be a requirement that must often be satisfied for different reasons, namely the experimental limit on the spectral index of the first-order inflaton fluctuations. In terms of the slow-roll parameters, the deviation of the spectral index from unity is given by

$$n-1 = 2\eta - 6\epsilon \cong 2\eta$$

where we have used the fact that $\epsilon \ll \eta$ in hybrid inflation (this is equivalent to the requirement that the energy density which drives inflation is dominated by $\lambda v^4/4$). The experimental constraint on the spectral index is roughly $|n-1| \lesssim 10^{-1}$. Writing η in terms of model parameters this translates into the constraint

$$g \frac{v}{M_p} \lesssim 5 \times 10^{-5} \tag{6.17}$$

This is just five times weaker than the technical assumption (6.16).

In passing, we note that there is also a lower bound on g from the assumption that the false vacuum energy density is dominated by $\lambda v^4/4 > m_{\varphi}^2 \varphi_0^2/2$. Using $g^2 \varphi_s^2 = \lambda v^2$ and (6.15), one finds

$$g > 460 \,\lambda \, \frac{v^3}{M_p^3}$$
 (6.18)
6.3.3 Tachyonic Instability

To quantify the evolution of the tachyonic instability at the end of inflation, we consider the equation of motion for the tachyon field fluctuation in Fourier space,

$$\delta^{(1)}\ddot{\sigma}_k + 3H\delta^{(1)}\dot{\sigma}_k + \left[\frac{k^2}{a^2} + \left(g^2\varphi_0^2 - \lambda v^2\right)\right]\delta^{(1)}\sigma_k = 0.$$
(6.19)

Once $\varphi_0 < \lambda^{1/2} v/g$ the tachyon effective mass parameter $\partial^2 V/\partial \sigma^2$ becomes negative and the fluctuations $\delta^{(1)}\sigma$ are amplified due to the spinodal instability. The efficient transfer of energy from the false vacuum energy $\lambda v^4/4$ to the fluctuations $\delta^{(1)}\sigma$ is referred to as *tachyonic preheating* in the literature [32]-[34].

Tachyonic Preheating in the Instantaneous Quench Approximation

The initial studies of tachyonic preheating focused on the flat space dynamics of $\delta^{(1)}\sigma_k$ in the instantaneous quench approximation. In this approximation the field $\delta^{(1)}\sigma$ is initially assumed to have zero mass and at t = 0 a negative mass squared term $-|m_{\sigma}^2|(\delta^{(1)}\sigma)^2/2$ is turned on. We briefly review these dynamics here following closely [32]. Initially the tachyon has the usual Minkowski space mode functions $e^{-ikt+i\vec{k}\cdot\vec{x}}/\sqrt{2k}$ ². Once the negative mass squared term is turned on the modes with $k = |\vec{k}| < |m_{\sigma}|$ grow exponentially with a dispersion

$$\langle (\delta^{(1)}\sigma)^2 \rangle = \frac{1}{4\pi^2} \int_0^{|m_{\sigma}|} dk \, k \, e^{2t\sqrt{|m_{\sigma}^2|-k^2}}$$

$$= \frac{|m_{\sigma}^2|}{4\pi^2 \alpha^2} \Big(e^{\alpha}(\alpha-1)+1 \Big); \quad \alpha \equiv 2|m_{\sigma}|t$$
(6.20)

which produces a spectrum with an effective cutoff $k_{\text{max}} = |m_{\sigma}|$. The tachyonic growth persists until the dispersion saturates at the value

$$\left\langle (\delta^{(1)}\sigma)^2 \right\rangle^{1/2} \cong \frac{v}{2}$$
 (6.21)

at which point the curvature of the effective potential vanishes and the tachyonic growth is replaced by oscillations about the true vacuum. This process completes

²Even in deSitter space the Bunch-Davies vacuum choice will ensure that this behaviour is respected on small scales $k \gg aH$ regardless of the tachyon mass during inflation. See chapter 1 for a review.

6.3 Hybrid Inflation

within a time

$$t_s \sim \frac{1}{2|m_\sigma|} \ln\left(\frac{\pi^2}{\lambda}\right) \tag{6.22}$$

which we call the spinodal time. At this point a large fraction of the vacuum energy $\lambda v^4/4$ has been converted into gradient energy of the field $\delta^{(1)}\sigma$ so that the universe is divided into domains with $\sigma \pm v$ of typical size $l \sim |m_{\sigma}|^{-1}$ and on average one still has $\langle \sigma \rangle = 0$ so that $\sigma_0(t) = 0$. These analytical argument are backed up by semi-classical lattice field theory simulations in [32, 33].

Dynamics of Tachyonic Preheating Beyond the Instantaneous Quench Approximation

The discussion of the dynamics of tachyonic preheating above apply strictly only in Minkowski space. The dynamics of tachyonic preheating including the dynamics of the inflaton but neglecting the expansion of the universe were considered in [34]. The dynamics of tachyonic preheating including both the dynamics of the inflaton and the expansion of the universe were considered in [184] wherein the authors reach conclusions identical to those discussed above. The authors of [184] also find that the spinodal time is somewhat modified from (6.22) due to the background dynamics (see also [185, 186]).

In the present work, we are interested in a situation which is different from the instantaneous quench, where the instability may turn on slowly compared to the Hubble expansion, rather than suddenly. Of course, all of our analysis will also apply to the case where the instability turns on quickly. We are approximating the time dependence of the tachyon mass as being linear around the time when it vanishes, eq. (6.14), so the mode equation can be written in the form

$$\frac{d^2}{dN^2}\delta^{(1)}\sigma_k + 3\frac{d}{dN}\delta^{(1)}\sigma_k + \left[\hat{k}^2 e^{-2N} - cN\right]\delta^{(1)}\sigma_k = 0.$$
(6.23)

where $N = H(t - t_s)$, $\hat{k} \equiv k/H$ and, incorporating the COBE normalization as in (6.15),

$$c \cong 22000 \, g \, M_p / v \,.$$
 (6.24)

From eq. (6.16), c is limited to values

$$c \ll \left(\frac{M_p}{v}\right)^2 \tag{6.25}$$

for the validity of the approximation that the tachyon mass squared varies linearly with time. The quantum mechanical solution in terms of annihilation and creation operators a_k , a_k^{\dagger} has the usual form ³

$$\delta^{(1)}\sigma(x) = \int \frac{d^3k}{(2\pi)^{3/2}} a_k \xi_k(N) e^{ikx} + \text{h.c.}$$
(6.26)

but the mode functions ξ_k will be complicated by the time-dependence of the tachyon mass. We normalize the mode functions ξ_k according to the usual Bunch-Davies prescription which is discussed in 1.1.1.

Since (6.23) has no closed-form solution, we approximate it in two regions. First, when $\hat{k}^2 e^{-2N} > c|N|$, we ignore the mass term and use the massless solutions, $\xi_k \sim a^{-3/2} H_{3/2}^{(1)}(\hat{k}e^{-N})$. We match this onto the solution in the region where $\hat{k}^2 e^{-2N} < c|N|$, where the term $\hat{k}^2 e^{-2N}$ is ignored in the equation of motion. The transition between the two regions occurs at different times N_k for different wavelengths, given implicitly by

$$N_k = \ln \frac{\hat{k}}{\sqrt{c}} - \ln \sqrt{|N_k|} \tag{6.27}$$

This is a multivalued function of $x \equiv \hat{k}/\sqrt{c}$, because for $x < (2e)^{-1/2}$, the \hat{k}^2 term in the differential equation comes to dominate again for a short period around N = 0, the moment when the tachyon is massless. To deal with this, we are going to assume that the solution is still well-approximated by the massive one during this short period. This amounts to replacing the multivalued function with the single-valued one shown in figure 6.1. We checked this approximation in conjunction with other approximations we will make for the mode functions, as described below eq. (6.30). Appendix D-1 provides some technical details about the matching time N_k .

In the second region, with $N > N_k$, the solutions are approximated by Airy functions, but it is more convenient to use the WKB approximation to obtain an

³Our conventions for fourier transforms and mode functions are discussed in detail in appendix D-4.



Figure 6.1: Exact solution and our approximation for the function N_k in eq. (6.27).

expression in terms of elementary functions. Ignoring overall phases (which will cancel out of the final result), in this way we obtain

$$\xi_k \cong \begin{cases} \left(2H\hat{k}^3\right)^{-1/2} \left(1+i\hat{k}e^{-N}\right), & N < N_k \\ b_k e^{-\frac{3}{2}N+\frac{9}{4c}z^{3/2}} (1+|z|)^{-1/4}, & N > N_k \end{cases}$$
(6.28)

with

$$b_k = \frac{1 - i\sqrt{c|N_k|}}{\sqrt{2H}(c|N_k|)^{3/4}} \frac{(1 + |z_k|)^{1/4}}{\exp\left(\frac{9}{4c}z_k^{3/2}\right)}$$
(6.29)

 and

$$z \equiv \left(1 + \frac{4}{9}cN\right); \quad z_k \equiv \left(1 + \frac{4}{9}cN_k\right) \tag{6.30}$$

In the above expressions, we have for simplicity matched the amplitudes but not derivatives of the solutions at $N = N_k$. This will not affect the estimates we make below. We used (6.27) to reexpress exponential dependence on N_k as power law dependence. Notice that exponent in (6.29) becomes purely imaginary when $\frac{4}{9}cN_k < -1$. We also replaced $|z|^{1/4} \rightarrow (1+|z|)^{1/4}$ to correct the spurious singularity at z = 0where the WKB approximation breaks down. We numerically verified that this gives a good approximation to the exact Airy function solutions.

Moreover, we have checked the approximate solution by numerically integrating the mode equations, starting from the small-N region $k^2 e^{-2N} \gg c|N|$, where the massless solutions with known amplitude tell us the initial conditions, and integrating into the large-N region where the exponential growth due to the tachyonic instability becomes important. We did this for two orthogonal solutions to the mode equations, $\xi_{1,2}$, whose behavior in the small-N region is

$$\xi_1 = (2Hk^3)^{-1}e^{-N} \left(k\cos(ke^{-N}) - \sin(ke^{-N}) \right)$$

$$\xi_2 = (2Hk^3)^{-1}e^{-N} \left(k\sin(ke^{-N}) + \cos(ke^{-N}) \right)$$

Evolving these initial conditions to large N, the envelope of these functions, which is also the modulus of the complex solutions, is $\xi = \sqrt{\xi_1^2 + \xi_2^2}$. We compared this numerical solution to the modulus $|\xi_k|$ of our approximation (6.28) for a large range of c and k values. In the large N-region, $|\xi_k|$ agrees with ξ up to a numerical factor of order unity. This factor, the ratio of the actual solution to the approximation, is shown in figure 6.2. Because the exponential growth of the mode function is a very steep function of N, these small errors have an imperceptible effect on the exclusion plots we will present in section 6.6. Furthermore, we have checked which values of c and k actually give constraints in the parameter space of the hybrid inflation model below, and found that in the regions where c is large, k is exponentially small. Extrapolating the results of figure 6.2 indicates that the error becomes quite small as $k \to 0$. Therefore our approximations for the mode functions are quite good for the purposes of this chapter.



Figure 6.2: Ratio of the exact mode functions to the approximation (6.28), at large times.

With the above approximate solution, we are in a position to recompute the dispersion of the tachyon fluctuations taking into account both the dynamics of the tachyon mass and also the background evolution of the scale factor, $\langle (\delta^{(1)}\sigma)^2 \rangle = (2\pi)^{-3} \int d^3k |\xi_k|^2$. Following the discussion of subsection 6.3.3, we set this equal to $v^2/4$ at the end of inflation, $N = N_*$:

$$\int \left. \frac{d^3k}{(2\pi)^3} |\xi_k|^2 \right|_{N=N_*} = \frac{v^2}{4} \tag{6.31}$$

which implicitly determines N_* in terms of parameters of the hybrid inflation model,

$$N_* = N_* \left(g, \lambda, v/M_p \right) \tag{6.32}$$

The main contribution to the integral at $N = N_*$ comes from wave numbers for which the exponentially growing solution in (6.28) applies. These modes satisfy $k < k_{\text{max}} \equiv$ $He^{N_*}\sqrt{cN_*}$. We have numerically performed the integral for a wide range of values of c and N_* . The result is displayed in figure 6.3, where contours of $\ln M_p^2/\lambda v^2$ are shown in the plane of N_* and $\ln c$. Recall that $c = 22000 g (M_p/v)$, eq. (6.24).



Figure 6.3: Contours of $\ln M_p^2/\lambda v^2$ in the plane of N_* and $\ln c$.

6.4 Second Order Fluctuations in the Long Wavelength Approximation

6.4.1 The Master Equation

The authors of [183] have derived a "master equation" for the second order potential $\phi^{(2)}$ which can be written as

$$\phi^{\prime\prime(2)} + 2\mathcal{H}(\eta - \epsilon)\phi^{\prime(2)} + \left[2\mathcal{H}^2(\eta - 2\epsilon) - \partial^k \partial_k\right]\phi^{(2)}$$

= $J(\tau, \vec{x})$ (6.33)

where the source terms are constructed entirely from first order quantities and can be split into inflaton and tachyon contributions

$$J(\tau, \vec{x}) = J^{\sigma}(\tau, \vec{x}) + J^{\varphi}(\tau, \vec{x}).$$

$$(6.34)$$

Although we have explicitly inserted the slow roll parameters, equation (6.33) is quite general and we have not yet assumed that η , ϵ are small (although recall that we have set $\sigma_0 = 0$). We have verified both the second order Einstein equations and the master equation presented in [183] and these results are discussed in appendix D-2.

Because the equation (6.33) is linear we can split the solutions $\phi^{(2)}$ into three parts: the solution to the homogeneous equation, the particular solution due to the inflaton souce and the particular solution due to the tachyon source. The solution to the homogeneous equation will be proportional to $\phi^{(1)}$ since the differential operator on the left hand side of (6.33) is identical to the operator which determines $\phi^{(1)}$ (see section 1.1.1). In cosmological perturbation theory the split between background quantities and fluctuations is unambiguous, since background quantities depend only on time while fluctuations depend on both position and time. However, the split between first order and second order fluctuations is arbitrary and the freedom to include in the solution for $\phi^{(2)}$ a contribution which is proportional to $\phi^{(1)}$ reflects this. We fix this ambiguity by including only the particular solutions for $\phi^{(2)}$ which is due to the source, $J(\tau, \vec{x})$. During the preheating phase the tachyon fluctuations are amplified by a factor v/H and so J^{σ} will come to dominate. This implies that the particular solution for $\phi^{(2)}$ which is due to J^{σ} will come to dominate over the particular solution which is due to J^{φ} . Although our analysis will focus on this part of the solution, we will also consider the inflaton source in appendix D-3 in order to verify that our formalism reproduces previous results.

6.4.2 The Green Function

In this subsection we construct the Green function for the master equation so that $\phi^{(2)}$ may be determined in terms of first order quantities. As discussed previously we consider only the particular solution for $\phi^{(2)}$ due to the source J and neglect the solution to the homogeneous equation, which is equivalent to assuming that the second order fluctuations are zero before the source is turned on. During a quasi-deSitter phase the master equation can be written as

$$\partial_{\tau} \left[(-\tau)^{2(\epsilon-\eta)} \partial_{\tau} \phi_k^{(2)} \right] + (-\tau)^{2(\epsilon-\eta)} \left[\frac{2}{\tau^2} (\eta - 2\epsilon) + k^2 \right] \phi_k^{(2)}$$
$$= (-\tau)^{2(\epsilon-\eta)} J_k(\tau)$$
(6.35)

where it is assumed that $\epsilon, \eta \ll 1$, and the differential operator on the left hand side of the master equation is written in a manifestly self-adjoint form. In deriving (6.35) we used (1.44) and treated the slow roll parameter as constant, which is consistent at first order in the slow roll expansion. The causal Green function for this operator is

$$G_{k}(\tau,\tau') = \frac{\pi}{2} \Theta(\tau-\tau') (\tau\tau')^{1/2+\eta-\epsilon}$$

$$\times [J_{\nu}(-k\tau)Y_{\nu}(-k\tau') - J_{\nu}(-k\tau')Y_{\nu}(-k\tau)]$$
(6.36)

where the order of the Bessel functions is $\nu \approx 1/2 + 3\epsilon - \eta$ as in section 1.1.1.

The solution for the metric perturbation $\phi^{(2)}$ can then be written as

$$\phi_k^{(2)}(\tau) = \int_{-(1+\epsilon)/a_i H}^0 d\tau' G_k(\tau, \tau') \left(-\tau'\right)^{2(\epsilon-\eta)} J_k(\tau') \tag{6.37}$$

where $a_i = a(t_i)$ is the scale factor at the some initial time, well before the tachyonic instability has set in. This solution is quite general; it applies during any slow roll phase, including during the tachyonic instability. There are several interesting limiting cases of (6.36). In the long wavelength limit $k \rightarrow 0$ the Green function reduces to

$$G_k(\tau, \tau') = \Theta(\tau - \tau')(1 + 2\eta - 6\epsilon) \left[-(-\tau)^{1+2\epsilon}(-\tau')^{2(\eta - 2\epsilon)} + (-\tau')^{1+2\epsilon}(-\tau)^{2(\eta - 2\epsilon)} \right].$$

In the case of pure deSitter expansion $\epsilon = \eta = 0$, but for all k, the Green function reduces to

$$G_k(\tau, \tau') = \Theta(\tau - \tau') \frac{1}{k} \sin \left[k \left(\tau - \tau'\right)\right].$$

Finally, the form of the Green function in the case of $\epsilon = \eta = 0$ and $k \to 0$ may be of some interest

$$G_k(\tau, \tau') = \Theta(\tau - \tau')(\tau - \tau'). \tag{6.38}$$

6.4.3 The Tachyon Source

We now study the tachyon contribution to the source (6.34). In position space J^{σ} takes the form (see equation (D-21))

$$J^{\sigma}(\tau, \vec{x}) = a^{2} \kappa^{2} m_{\sigma}^{2} \left(\delta^{(1)} \sigma\right)^{2} - 2\kappa^{2} \left(\delta^{(1)} \sigma'\right)^{2} + 2\kappa^{2} \mathcal{H}(1 + \eta - \epsilon) \Delta^{-1} \partial_{i} \left(\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma\right) + 4\kappa^{2} \Delta^{-1} \partial_{\tau} \partial_{i} \left(\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma\right) - \mathcal{H}(1 + 2\epsilon - 2\eta) \Delta^{-1} \gamma_{\sigma}' + \Delta^{-1} \gamma_{\sigma}''.$$
(6.39)

The quantity γ_{σ} can be written in the form (see equation (66) of [183], or equivalently (D-20))

$$\begin{split} \gamma_{\sigma} &= -\kappa^{2} \Delta^{-1} \left[3\partial_{i} \left(\partial^{k} \partial_{k} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) \right. \\ &+ \left. \frac{1}{2} \partial^{k} \partial_{k} \left(\partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) \right] \\ &= -3\kappa^{2} \Delta^{-1} \partial_{i} \left(\partial^{k} \partial_{k} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) \\ &- \left. \frac{\kappa^{2}}{2} \left(\partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) . \end{split}$$

Notice that the terms in the first line of the (6.39) are local, the terms in the second and third line are non-local (containing an inverse laplacian Δ^{-1}) and the fourth line contains terms which are both local and doubly non-local (containing Δ^{-2}). The Fourier transforms of the source terms are computed in appendix D-4 wherein we also discuss our conventions for the inverse laplacian operators.

In the following, we will need the Fourier transform of terms like $\Delta^{-1}\gamma_{\sigma}$,

$$\mathcal{F} \left[\Delta^{-1} \gamma_{\sigma} \right] = -3 \frac{\kappa^2}{k^4} \int \frac{d^3 k'}{(2\pi)^{3/2}} k'^2 k \cdot (k-k') \delta^{(1)} \tilde{\sigma}_{k'} \delta^{(1)} \tilde{\sigma}_{k-k'} - \frac{\kappa^2}{2k^2} \int \frac{d^3 k'}{(2\pi)^{3/2}} k' \cdot (k-k') \delta^{(1)} \tilde{\sigma}_{k'} \delta^{(1)} \tilde{\sigma}_{k-k'}$$

This expression is operator-valued and can be written in terms of annihilation/creation operators and mode functions as $\delta^{(1)}\tilde{\sigma}_k = a_k\xi_k(t) + a^{\dagger}_{-k}\xi_{-k}(t)$ (see appendix D-4 for more details). In the Fourier transformed expression for $\Delta^{-1}\gamma_{\sigma}$, the scale dependence of the mode functions $\delta^{(1)}\tilde{\sigma}_k$ is integrated over so that $\Delta^{-1}\gamma_{\sigma}$ gets contributions from the tachyon fluctuations on all scales. The large scale limit of terms like $\Delta^{-1}\gamma_{\sigma}$ is not transparent and therefore we do not neglect any terms in the tachyon source which contain inverse Laplacians.

In order to consistently compute $\phi_k^{(2)}$ (6.37) in the large scale limit it is necessarily to keep the next-to-leading order terms in the small $(k/\mathcal{H})^2$ expansion of the Green function (6.36). To see this, notice that powers of k^2 cancel inverse Laplacians in the source:

$$k^2 J_k^{\sigma} = -\gamma_{\sigma,k}^{\prime\prime} + \mathcal{H} \gamma_{\sigma,k}^{\prime} + \cdots$$
(6.40)

where \cdots denotes gradient terms which are small on large scales. In appendix D-2 it is shown that γ_{σ} can be written on large scales as (D-19)

$$\gamma_{\sigma} \cong \frac{3\kappa^2}{2} \left[\left(\delta^{(1)} \sigma' \right)^2 - a^2 m_{\sigma}^2 \left(\delta^{(1)} \sigma \right)^2 \right] - 3\kappa^2 \Delta^{-1} \partial_{\tau} \partial_i \left(\delta^{(1)} \sigma' \partial^i \delta^{(1)} \sigma \right) - 6 \mathcal{H} \kappa^2 \Delta^{-1} \partial_i \left(\delta^{(1)} \sigma' \partial^i \delta^{(1)} \sigma \right).$$

Plugging this result into (6.40) and comparing to (6.39) we see that $k^2 J_k^{\sigma}/\mathcal{H}^2$ contains terms which are of the same form as those which appear in J_k^{σ} (6.39). Hence these terms must be included to consistently study $\phi_k^{(2)}$ (and hence the curvature perturbation) on large scales. Will we see shortly that the inclusion of such terms is crucial for the cancellation of nonlocal terms in the curvature perturbation.

6.4.4 The Gauge Invariant Curvature Perturbation

The curvature perturbation is expanded to second order as

$$\zeta = \zeta^{(1)} + \frac{1}{2}\zeta^{(2)}$$

In hybrid inflation, where $\sigma_0 = 0$, the first order contribution comes entirely from the inflaton sector

$$\zeta^{(1)} = -\phi^{(1)} - \mathcal{H} \frac{\delta^{(1)}\rho}{\rho_0'}$$

where $\rho_0 = -(T_0^0)_{(0)}$, $\delta^{(1)}\rho = -\delta^{(1)}T_0^0$ are the unperturbed and first order stress tensor respectively. It is also conventional to define the comoving curvature perturbation at first order

$$\mathcal{R}^{(1)} = \phi^{(1)} + \mathcal{H} \frac{\delta^{(1)}\varphi}{\varphi'_0}$$

which, on large scales, is related to $\zeta^{(1)}$ as

$$\mathcal{R}^{(1)} + \zeta^{(1)} \cong 0.$$

Because $\sigma_0 = 0$ the inflationary trajectory is straight the case of hybrid inflation and $\zeta^{(1)}$ is conserved on large scales [190]-[193]. However, this is not the case at second order [187] and one expects that $\zeta^{(2)}$ will be amplified due to the tachyonic instability. This amplification of $\zeta^{(2)}$ will continue until $N = N_{\star}$ at which point the backreaction sets in and stops inflation. For $N > N_{\star}$ the inflaton is no longer dynamical since all of its energy has been converted into tachyon fluctuations. Thus for $N > N_{\star}$ the large scale curvature perturbation is conserved at all orders in perturbation theory since only one field (the tachyon) is dynamical and there are no non-adiabatic pressures.

The definition of the first order curvature perturbation $\zeta^{(1)}$ is generally agreed upon in the literature (up to a sign). At second order, however, there are several definitions of the curvature perturbation in the literature (see [187] for a comprehensive discussion). The definition we adopt follows [188] and generalizes the definition of Malik and Wands [189] (valid on large scales) to multiple scalar fields. (Our definition differs from [183]. The definition of [183] generalizes [157] to two scalar fields and applies only during inflatoin [193, 187] since it is conserved only in the slow roll limit.) In [188] the second order large scale curvature perturbation is written in terms of the first and second order Sasaki-Mukhanov variables as

$$\begin{aligned} \zeta^{(2)} &= \frac{1}{3 - \epsilon} \frac{1}{(\varphi_0')^2} \left[\varphi_0' Q_{\varphi}'^{(2)} + a^2 \frac{\partial V}{\partial \varphi} Q_{\varphi}^{(2)} \right] \\ &+ \frac{1}{3 - \epsilon} \frac{1}{(\varphi_0')^2} \left[(Q_{\sigma}'^{(1)})^2 + a^2 m_{\sigma}^2 \left(Q_{\sigma}^{(1)} \right)^2 \right] \\ &+ \frac{1}{3 - \epsilon} \frac{1}{(\varphi_0')^2} \left[(Q_{\varphi}'^{(1)})^2 + a^2 m_{\varphi}^2 \left(Q_{\varphi}^{(1)} \right)^2 \right] \\ &+ 4(3 + \epsilon - \eta) \left(\frac{3 - 2\epsilon}{3 - \epsilon} \right) \left(\frac{\mathcal{H}}{\varphi_0'} Q_{\varphi}^{(1)} \right)^2 \\ &+ (-10 + 2\epsilon + 2\eta) \left(\frac{\mathcal{H}}{\varphi_0'} Q_{\varphi}^{(1)} \right)^2 \end{aligned}$$
(6.41)

where the first order Sasaki-Mukhanov variables 4 are

$$Q_{\varphi}^{(1)} = \frac{\varphi_0'}{\mathcal{H}} \mathcal{R}^{(1)} \tag{6.42}$$

$$Q_{\sigma}^{(1)} = \delta^{(1)}\sigma \tag{6.43}$$

and the second order Sasaki-Mukhanov variable is

$$Q_{\varphi}^{(2)} = \delta^{(2)}\varphi + \frac{\varphi_{0}'}{\mathcal{H}}\psi^{(2)} + (2 + 2\epsilon - \eta)\frac{\varphi_{0}'}{\mathcal{H}}(\phi^{(1)})^{2} + 2\frac{\varphi_{0}'}{\mathcal{H}^{2}}\phi^{(1)}\phi'^{(1)} + \frac{2}{\mathcal{H}}\phi^{(1)}\delta^{(1)}\varphi'.$$
(6.44)

In writing (6.41-6.44) we have restricted ourselves to hybrid inflation, made use of the background equations and inserted ϵ, η (but not assuming slow roll).

We split $\zeta^{(2)}$ into contributions coming from the inflaton and the tachyon as

$$\zeta^{(2)} = \zeta^{(2)}_{\varphi} + \zeta^{(2)}_{\sigma}$$

and study each piece separately. This splitting is different from the one discussed in [188], where the curvature perturbation is defined for each fluid in such a way that the total curvature perturbation is a weighted sum of the individual contributions.

⁴Notice that $Q_{\varphi}^{(1)}$ is related to the variable $V^{(1)}$ discussed chapter 1 by $V^{(1)} = a Q_{\varphi}^{(1)}$.

Instead, we simply divide $\zeta^{(2)}$ into terms which depend respectively on the tachyon and inflaton fluctuations, $\delta^{(1)}\sigma$ and $\delta^{(1)}\varphi$, $\phi^{(1)}$, which is only possible because $\sigma_0 = 0$.

The tachyon part of the curvature perturbation $\zeta_{\sigma}^{(2)}$ gets contributions from the first and second line of (6.41), both explicitly through $Q_{\sigma}^{(1)}$ and implicitly through $\delta^{(2)}\varphi, \psi^{(2)}$. The inflaton part of the curvature perturbation $\zeta_{\varphi}^{(2)}$ contains contributions from the last three lines of (6.41) as well as from the first line of (6.41), both implicitly through $\delta^{(2)}\varphi, \psi^{(2)}$ and explicitly through the definition of $Q_{\varphi}^{(2)}$.

The inflaton part of the curvature perturbation, $\zeta_{\varphi}^{(2)}$ coincides with the $\zeta^{(2)}$ of single field inflation and has been derived previously [18, 158]. We have considered the construction of $\zeta_{\varphi}^{(2)}$ using our formalism and these results are presented in appendix D-3.

6.4.5 The Tachyon Curvature Perturbation

We now compute $\zeta_{\sigma}^{(2)}$. Our focus is on the leading order contribution to $\zeta^{(2)}$ in the slow roll and large scale limit. If we work only to leading order in the slow roll parameters it is sufficient to use the Green function (6.36) in the limit $\epsilon = \eta = 0$ and keep only the terms in the tachyon source (6.39) which are not slow-roll suppressed. However, to consistently compute $\zeta_{\sigma}^{(2)}$ we must keep the next-to-leading order terms in the small $(k/\mathcal{H})^2$ expansion of the Green function, as we have discussed in subsection 6.4.3.

In appendix D-5 we use the second order Einstein equations and the previously derived expressions for the Green function and tachyon source to write $\zeta_{\sigma}^{(2)}$ (equation (6.41)) in terms of first order fluctuations. The result is

$$\begin{aligned} \zeta_{\sigma}^{(2)} &\cong \frac{\kappa^2}{\epsilon} \int_{-1/a_i H}^{\tau} d\tau' \left[\frac{\left(\delta^{(1)} \sigma' \right)^2}{\mathcal{H}(\tau')} \right. \\ &- \frac{\mathcal{H}(\tau')^2}{\mathcal{H}(\tau)^3} \left(\left(\delta^{(1)} \sigma' \right)^2 - a^2 m_{\sigma}^2 \left(\delta^{(1)} \sigma \right)^2 \right) \right] \end{aligned} \tag{6.45}$$

where the tachyon fluctuations $\delta^{(1)}\sigma$ are functions of the integration variable τ' . The corrections to (6.45) are either subleading in the slow roll expansion or are total gradients which can be neglected on large scales. Equation (6.45) is the main result of this section.

Several comments are in order concerning (6.45). First, note that the integration by parts which we have performed gives rise to terms which are evaluated between the initial and final time of the form $[\cdots]_{\tau'=-1/a_iH}^{\tau}$. Since we are interested in the preheating phase during which the fluctuations are amplified exponentially we can safely drop the contribution at $\tau' = -1/a_iH$ relative to the contribution at $\tau' = \tau$. Consistency of this approximation requires that the tachyonic growth of the perturbations overcomes any damping which may have occured during inflation. (If this is not the case then we can safely assume that no significant nongaussianity is produced.) We will return to this issue.

The expression (6.45) satisfies an important consistency check, namely that it is a local expression. Nonlocal operators Δ^{-n} appeared in many of the intermediate steps in the calculation. These nonlocal terms arise due to the decomposition of the metric perturbations into scalar, vector and tensor components. One expects that the nonlocal terms should cancel out of physical quantities, similarly to electrodynamics in Coulomb gauge. The second order curvature perturbation is related to the observable CMB temperature fluctuations in a nontrivial way, so this by itself does not prove that $\zeta^{(2)}$ must be local. However [194] has recently shown that under the conditions present in our model, $\zeta^{(2)}$ should indeed by local.

To clarify, we have shown that at leading order in slow roll parameters the nonlocal contributions to $\zeta_{\sigma}^{(2)}$ cancel. We have not checked that such terms cancel at higher order in the slow roll expansion, though we believe that they do. This cancellation was not observed in previous studies because the subleading (in k^2) corrections to the Green function were not included and thus the large scale expansion was inconsistent.

A comment is in order concerning the long wavelength approximation which we have used in deriving (6.45). In writing the expression for $k^2 J_k^{\sigma}$ we have dropped terms which are total gradients even though such terms will be integrated over time in computing $\zeta_{\sigma}^{(2)}$, due to the time integral in (6.37). Strictly speaking, one should only apply the large scale limit $k\tau \ll 1$ after the time integral has been performed. The reason for this is that the time integral extends from when the modes are well within the horizon, where they oscillate, to when the modes are outside the horizon, including horizon crossing. Mode-mode coupling near the epoch of horizon crossing can contribute momentum-dependent terms which are not suppressed on large scales [18, 195]. However in our application, we are justified in dropping the contributions which are generated near horizon crossing since they are exponentially suppressed compared those which are produced during the preheating phase.

6.4.6 Time-integrated tachyon perturbation

We can make our main result (6.45) more explicit by substituting in the solutions (6.26,6.28). The result is simplified by taking the Fourier transform of $\zeta^{(2)}$ and evaluating it at vanishing external wave number, and at the final time corresponding to the end of inflation, $N = N_*$:

$$\zeta_{k=0}^{(2)}(N_{*}) = \frac{\kappa^{2}}{\epsilon} \int \frac{d^{3}p}{(2\pi)^{3/2}} \left(a_{p} b_{p} a_{p}^{\dagger} b_{-p} + \text{perms} \right) \\
\times \int_{\max(N_{p},N_{i})}^{N_{*}} f(c,N,N_{*}) dN$$
(6.46)

where N_i denotes the value of N at the beginning of inflation, $f(c, N, N_*)$ is given by

$$f(c, N, N_{*}) = e^{-3N + \frac{9}{2c}z^{3/2}} (1 + |z|)^{-1/2} \times \left[\frac{9}{4} \left(1 - e^{3(N-N_{*})} \right) \left| \sqrt{z} - 1 - \frac{2c \operatorname{sign}(z)}{27(1 + |z|)} \right|^{2} - cNe^{3(N-N_{*})} \right]$$

$$(6.47)$$

where $z \equiv (1 + \frac{4}{9}cN)$, and "perms" in (6.46) indicates the three other combinations of $a_p b_p$ and $a_p^{\dagger} b_{-p}$.

For illustration, we show the behavior of the function $f(c, N, N_*)$ for sample parameter values $N_* = 22.5$ and $\ln c = 1$. Figure 6.4 plots $(\text{sign}(f) \ln(1 + |f|))$ as a function of N. The function is exponentially peaked at the initial value $N = N_i$, and at the final value $N = N_*$. Moreover it always becomes negative at N_* because the negative mass squared term in (6.47) comes to dominate. Although it is not obvious in the figure, the negative value is orders of magnitude larger than the positive maximum just preceding it, so the negative extremum dominates in the integral in (6.47).

Whether the extremum of f at N_* or at N_i dominates overall depends on how $|3N_i|$ compares to $9/(2c)z^{3/2}$ evaluated at N_* . To see this, note that the dominant time-dependence of f is determined by the combination $-3N + \frac{9}{2c}z^{3/2}$ in the exponent. The e^{-3N} decay factor is typical of massive modes, which redshift as $\delta\sigma_k \sim a^{-3/2}$. The $e^{\frac{9}{2c}z^{3/2}}$ growth factor is a result of the tachyonic instability. As we noted in the discussion following eq. (6.45), (6.46) is valid only when the late-time behavior dominates (since otherwise terms which we have discarded are no longer subdominant to the terms which appear in (6.45)). Therefore an important consistency condition for all of our analysis is that

$$\frac{9}{2c} z_*^{3/2} \equiv \frac{9}{2c} \left(1 + \frac{4}{9} c N_* \right)^{3/2} > 3|N_i| \,. \tag{6.48}$$

If this condition is violated then the preheating is not playing any significant role in the dynamics and we can safely assume that no significant nongaussianity is produced.

Because of the exponential growth of f at its extrema, it is a very good approximation to the integral to write $f = e^g$ and expand $g = g_m + g'_m(N - N_m)$ in the vicinity of the maximum value, whether it is at N_i or N_* . Since the integral is so strongly peaked near the extremum, there is an exponentially small error in extending the range of integration to the half-line. In this way one obtains

$$\int dN f \cong \frac{e^{g_m}}{|g'_m|} \tag{6.49}$$

We will use this approximation below to numerically evaluate the integral.

6.5 Bispectrum and spectrum of second order metric perturbation

6.5.1 The tachyon contribution to the bispectrum

Here we calculate the leading contribution to the three-point function (bispectrum) of the second-order curvature perturbation due to the tachyon,

$$\left\langle \zeta_{k_1}^{(2)} \zeta_{k_2}^{(2)} \zeta_{k_3}^{(2)} \right\rangle \equiv (2\pi)^{-3/2} \,\delta^{(3)}(\vec{k}_1 + \vec{k}_2 + \vec{k}_3) \,B(\vec{k}_1, \vec{k}_2, \vec{k}_3) \tag{6.50}$$



Figure 6.4: $\operatorname{sign}(f) \ln(1+|f|)$ versus N, showing behavior of the function f defined in (6.47), for $\ln c = 1$, $N_* = 22.5$

It is understood that only the connected part of the correlation function is computed, which is equivalent to subtracting the expectation value of $\zeta^{(2)}$ from the quantum operator. This three-point function is straightforward to compute, using the freefield two-point functions

$$\left\langle \delta^{(1)} \tilde{\sigma}_{p_i} \delta^{(1)} \tilde{\sigma}_{q_i} \right\rangle = \xi_{p_i} \xi_{q_i} \,\delta^{(3)}(p_i + q_i) \tag{6.51}$$

Carrying out the contractions of pairs of fields which contribute to the connected part of the bispectrum, one finds eight terms, which in the limit of vanishing external wave-numbers, are all equal. The result is

$$B = 8\frac{\kappa^{6}}{\epsilon^{3}} \int \frac{d^{3}p}{(2\pi)^{3}} |b_{p}|^{6} \left[\int_{\max(N_{p}, N_{i})}^{N_{*}} dN f(c, N, N_{*}) \right]^{3}$$
(6.52)

The integrand of the *p*-integral is exponentially strongly peaked, either near $\ln(p/\sqrt{cH}) \cong N_i$ if f has its global maximum near $N = N_i$, or else near $\ln(p/\sqrt{cH}) \cong 0$ if f is dominated by its behavior near N_* . The logarithm of the integrand for a typical case is shown in figure 6.5. If the cusp-like peak is dominating, we can use the same approximation as in (6.49) to evaluate the p integral, on both sides of the maximum. If the other local maximum at larger values of p is the global maximum, as sometimes happens, then we should treat the integral as a gaussian, since the derivative of the integrand vanishes at the maximum. In this case we similarly write the integrand of

the p integral as $f = e^g$ and expand $g = g_m + \frac{1}{2}g''_m(p-p_m)^2$, where $g''_m < 0$; then (6.49) is replaced by

$$\int dN f \cong \sqrt{\frac{2\pi}{|g_m''|}} e^{g_m} \tag{6.53}$$

We have carried out the evaluations numerically over a range of values of $\ln c$ and N_* .

 $e^{-6N_i} \left(\kappa^2\right)^3 \tilde{p}$

In figure 6.6 we plot contours of $\ln |\tilde{B}|$, where \tilde{B} is defined by

$$B = \frac{e^{-6N_i}}{2\pi^2} \left(\frac{\kappa^2}{c\epsilon}\right)^3 \tilde{B}$$
(6.54)



Figure 6.5: Log of the integrand of the p integral in eq. (6.52), for a case where the maximum occurs near $\ln(p/\sqrt{c}H) = N_i$.

The boundary between the two regions in fig. 6.6 is described analytically by the relation

$$3|N_i| = \frac{9}{2c} z_*^{3/2} \tag{6.55}$$

with $z_* = 1 + \frac{4}{9}cN_*$. Only the region to the right, where $\frac{9}{2c}z_*^{3/2} > 3|N_i|$, is relevant for our analysis (see eq. (6.48)).

To make contact with experimental constraints, we want to compare the predicted bispectrum with that of single-field inflation, where nongaussianity is conventionally expressed via a nonlinearity parameter f_{NL} , defined through

$$B(k_i) = -\frac{6}{5} f_{NL} \left(P_{\phi}(k_1) P_{\phi}(k_2) + \text{ permutations} \right)$$
(6.56)



Figure 6.6: Contours of $\ln |\tilde{B}|$, defined in (6.54), in the plane of N_* and $\ln c$. $N_i = -30$ in this example.

where $P_{\phi}(k)$ is the usual inflationary power spectrum, $P_{\phi}(k)^{1/2} \sim 10^{-5} (2\pi)/k^{3/2}$. If we assume that all k_i are the same, $k_i \cong k$, then

$$f_{NL} = -\frac{5}{18} (2\pi)^{-4} 10^{20} B(k) k^6$$
(6.57)

We can thus convert our predicted bispectrum into an effective f_{NL} , for which the present experimental constraint is roughly $|f_{NL}| < 100$.

Using (6.54) in (6.57) and recalling that $N_e = |N_i| + N_*$ we find that

$$f_{NL} = -e^{17.5 - 6N_*} c^{-3} \tilde{B} \tag{6.58}$$

where we used $H^2 = V/(3M_p^2)$ as well as the COBE normalization $V/(\epsilon M_p^4) = 6 \times 10^{-7}$. The contours of \tilde{B} can thereby be converted into contours of f_{NL} , and demanding that $|f_{NL}| < 100$ gives a constraint in the parameter space c, N_*, N_i .

6.5.2 The tachyon contribution to spectrum

The analogous calculation can also be carried out for the tachyon contribution to the two-point function of the curvature perturbation.

Closely following the preceding calculation, it is straightforward to show that, in

the limit of vanishing external wave numbers,

$$\left\langle \zeta_{k_{1}}^{(2)} \zeta_{k_{2}}^{(2)} \right\rangle_{\text{con}} = \delta(\vec{k}_{1} + \vec{k}_{2}) \times 2 \frac{\kappa^{4}}{\epsilon^{2}} \int \frac{d^{3}p}{(2\pi)^{3}} |b_{p}|^{4} \left[\int_{N_{p}}^{N_{*}} dN f(c, N, N_{*}) \right]^{2} \\ \equiv \delta(\vec{k}_{1} + \vec{k}_{2}) S(k_{i})$$

$$(6.59)$$

In analogy to (6.54), we define⁵

$$S = \frac{e^{-3N_i}}{4\pi^2} \frac{H}{c^{3/2}} \left(\frac{\kappa^2}{\epsilon}\right)^2 \tilde{S}$$
(6.60)

and we display contours of \tilde{S} in fig. 6.7. Analogously, only the right-hand region is relevent for our analysis (since this is the region where 6.48 is satisfied).



Figure 6.7: Contours of $\ln |\tilde{S}|$ in the plane of N_* and $\ln c$, at $N_i = -30$.

Demanding that the tachyon contribution to the spectrum is sub-dominant to the inflaton contribution gives

$$\left\langle \zeta_{k_1}^{(2)} \zeta_{k_2}^{(2)} \right\rangle_{\text{con}} \le \frac{(2\pi)^2}{k^3} \,\delta(\vec{k}_1 + \vec{k}_2) \,\mathcal{P}_{\zeta}$$
 (6.61)

with $\mathcal{P}_{\zeta}^{1/2} \cong 10^{-5}$. Combining this with (6.59) and (6.60), we define the *linearity* parameter

$$f_L \equiv \frac{S}{P_{\zeta}} \equiv \frac{10^{10}}{(2\pi)^4} \frac{H}{c^{3/2}} \left(\frac{\kappa^2}{\epsilon}\right)^2 \tilde{S} e^{-3N_i} k^3$$
(6.62)

⁵The powers of H and c can be understood as follows: $b_p \sim H^{-1/2}c^{-3/4}$, whereas $p \sim \sqrt{c}H$.

The linearity parameter quantifies the tachyon contribution to the power spectrum. Since we would like the spectrum to be generated by the inflaton, we will demand that $|f_L| < 1$. One might consider being more conservative and imposing, say, $|f_L| < 0.01$ rather than $|f_L| < 1$, as we have done. Our exclusion plots are actually quite insensitive to the value assumed for f_L (this is also true for f_{NL}) since the effect turns on exponentially fast.

6.6 Constraints on hybrid inflation parameter space

We have shown that contours of \tilde{B} , \tilde{S} can be converted into contours of f_{NL} , f_L . Demanding that $|f_{NL}| < 100$, $|f_L| < 1$ gives constraints in the parameter space of c, N_* , N_i . The only obstacle to working directly in the model parameter space g, λ , v is the implicit dependence of N_* on (g, λ, v) . We have numerically inverted the relation depicted in fig. 6.3 to determine $N_*(g, \lambda, v)$. It then becomes straightforward to scan the model parameters, recomputing f_{NL} and f_L . In the next section we will display our constraints in the space of $\log_{10} \lambda$, $\log_{10} g$ for a range of values of $\log_{10} (v/M_p)$.

Note that once a set of values for λ , g and v/M_p are chosen and N_* has been calculated, one must still determine N_i in order to calculate the parameters f_L and f_{NL} . We do so by first computing the total number of e-foldings using the standard result

$$N_e = 62 - \ln\left(\frac{10^{16} \text{ GeV}}{V^{1/4}}\right) - \frac{1}{3}\ln\left(\frac{V^{1/4}}{\rho_{\text{r.h.}}^{1/4}}\right)$$
(6.63)

where $V \cong \lambda v^4/4$ is the energy density during inflation, and $\rho_{\rm r.h.}$ is the energy density at reheating. In the following, we will ignore the gravitino bound $\rho_{\rm r.h.}^{1/4} \lesssim 10^{10}$ GeV and assume instant reheating, $\rho_{\rm r.h.} = V$. The value of N_i then follows from $N_i = N_* - N_e$. We have checked that incorporation of the gravitino bound does not create a noticeable change in the excluded regions.

6.6.1 The issue of scale invariance

In evaluating the tachyonic contributions to the spectrum or bispectrum we find that these are either nearly scale invariant $(S \sim 1/k^3, B \sim 1/k^6)$, or else to scale invariance is badly violated $(S, B \sim k^0)$. In the latter case, the spectral index for the tachyon contribution to the two-point function is n = 4 (this is characteristic of massive fields and respects the Traschen integral constraints on causality in general relativity [196]).

The two regimes, scale-invariant and nonscale-invariant, can be understood in reference to the condition (6.48) which must be satisfied in order for tachyonic preheating to play any significant role. There are two ways to satisfy (6.48). One is to take $cN_{\star} \gg 1$, which usually requires c > 1. This is the regime in which the tachyon mass is not small compared to H during most of inflation, and so it corresponds to nonscale-invariant fluctuations of $\delta\sigma$. The tachyon fluctuations are Hubble damped as $\delta\sigma \sim a(t)^{-3/2}$ prior to inflation, but this suppression can be overcome on large scales if the amplification during the preheating phase is sufficiently large, which typically requires very small values of the self-coupling $\lambda \ll 1$. This nonscale-invariant regime corresponds to a region of the parameter space where the waterfall condition of hybrid inflation is satisfied (and one typically has $N_{\star} \ll 1$).

The second way to satisfy (6.48) is to take c|N| < 1 for all $N \in [N_i, N_\star]$. This gives a scale-invariant spectrum for the tachyon and also for $\zeta_{\sigma}^{(2)}$, which is most easily seen by writing the tachyon mass-squared as $|m_{\sigma}^2|/H^2 = c|N|$. It is clear that if c|N| < 1for all $N \in [N_i, N_\star]$ then the tachyon field will have been light compared to the Hubble scale for all ~ 60 e-foldings of inflation which ensures a nearly scale-invariant spectrum for the tachyon. Also, in this case the instability will typically take several e-foldings to complete so that $N_\star > 1$. This scale-invariant regime corresponds to a region of the parameter space where the usual waterfall condition of hybrid inflation is violated (which is equivalent to saying $N_\star \gtrsim 1$).

6.6.2 Nonscale-invariant case

In the nonscale-invariant case f_L and f_{NL} depend on k. Because the tachyon spectrum is blue in this case the strongest constraint comes from evaluating f_L , f_{NL} at the largest values of k which are measured by the CMB. In deriving our constraints we conservatively take this to be $k = e^6 H e^{-N_e}$ where N_e is the total number of e-foldings of inflation (6.63). The resulting constraints in the plane of $\log_{10} g$ and $\log_{10} \lambda$ are shown in the right-hand region of figure 6.8, for several values of $\log_{10} v/M_p$. We find that in this region the most stringent constraints come from f_L rather than f_{NL} , so we expect to see distortions of the spectrum rather than nongaussianity at the left-hand boundaries of the excluded regions. (The other boundaries are excluded for different reasons described in the next paragraph.)



Figure 6.8: Excluded regions of the hybrid inflation parameter space, for $\log_{10} v/M_p = -1, -3, -5, -7, -9$.

In computing f_L , f_{NL} over a wide range of g, λ , v/M_p we also checked that several additional assumptions were respected: the tachyon mass-squared m_{σ}^2 varies linearly with the number of e-foldings, which was shown previously to require $gv/M_p < 10^{-5}$; the false vacuum energy density $\lambda v^4/4$ dominates during inflation, leading to the bound $g > 460\lambda (v/M_p)^3$; the reheat temperature exceeds 100 GeV, so that baryogenesis can occur at least during the electroweak phase transition, leading to the lower bounds on λ .

6.6.3 Scale-invariant case; the adiabatic approximation

On the left-hand side of fig. 6.8, we display new constraints for which the spectrum and bispectrum are scale-invariant. In contrast to the right-hand side, f_{NL} provides the dominant constraint here, so that nongaussianity is playing the important role. To obtain these results, we employ a different approximation for the tachyon mode functions, namely the adiabatic approximation (described in some detail in appendix F of [5]). Because the tachyon has a small mass during the entirety of inflation (subsequent to horizon crossing), its mass is changing slowly, and we can use the standard mode functions for light fields, but with a time-dependent mass:

$$\delta\sigma(x) = \int \frac{d^3k}{(2\pi)^{3/2}} \frac{H}{\sqrt{2k^3}} (-k\tau)^{\eta_{\sigma}(\tau)} e^{ikx} a_k + \text{h.c.}$$
(6.64)

Here $\eta_{\sigma} = M_p^2 V_{\sigma\sigma} / V$ is the slow-roll parameter for the tachyon, given by

$$\eta_{\sigma}(\tau) = 8\eta \frac{M_p^2}{v^2} \ln |H\tau| \tag{6.65}$$

where $\eta = M_p^2 V_{,\varphi\varphi}/V$. We have also verified that the solution (6.64) can be reproduced by (6.28) in the appropriate limit.

Since the mode functions have a simple form in the adiabatic approximation, it is possible to go farther analytically in this case. Notably, we can find an implicit equation for N_* after evaluating the integral (6.31):

$$N_* \cong \frac{v/M_p}{15000 N_e g} \ln \left[1 + 2 \times 10^6 N_* g \left(\frac{M_p}{v}\right)^3 \right]$$
(6.66)

This expression for N_* is much easier to evaluate than the one which arises in the WKB approximation since the latter leads to a numerical integral $F(N; \lambda, g, v)$ which must be inverted to find the N_* which satisfies $F(N_*; \lambda, g, v) = v^2/4$.

Moreover, the time (N) integral in (6.45) can be evaluated explicitly using the saddle point method, since it is dominated by the exponential growth near $N = N_*$.

⁶ Defining η_f to be $|\eta_\sigma|$ at $N = N_*$, this results in the expression

$$B(\vec{k}_{i}) = 4H^{6\eta_{f}}d_{*}^{3}\int \frac{d^{3}p}{(2\pi)^{3}}|p|^{-3-2\eta_{f}}|p-k_{3}|^{-3-2\eta_{f}}$$

$$(|p+k_{2}|^{-3-2\eta_{f}}+|p+k_{1}|^{-3-2\eta_{f}})$$
(6.67)

for the bispectrum,⁷ where $d_* = H^2 \kappa^2 \eta_f e^{2\eta_f N_*}/(2\epsilon)$. In the limit of small $c \sim \eta_f$, this is manifestly nearly scale invariant, $B \sim 1/k_i^{6(1+\eta_f)}$, by power-counting the divergent behavior of the integral in the infrared. The divergence must be cut off in the usual way, by ignoring modes with p smaller than the horizon. Numerically evaluating the remaining p integral for $k_i \sim k$, and using $k\tau_* \sim e^{N_i}$ for modes near the horizon,⁸ we find

$$B(k) \cong 45 \, k_i^{-6(1+\eta_f)} \left(\frac{\kappa^2 H^2 \eta_f e^{2\eta_f N_e}}{2\pi\epsilon}\right)^3 \tag{6.68}$$

Further using the COBE normalization to write $\kappa^2 H^2/\epsilon = 2 \times 10^{-7}$, we find that the nonlinearity parameter is

$$f_{NL} = -2.6 \times 10^{-5} \left(\eta_f e^{2\eta_f N_e}\right)^3.$$
(6.69)

Moreover the COBE normalization implies $\eta_f = 7360N_*gM_p/v$. Demanding that $|f_{NL}| < 100$ gives the new excluded regions on the left-hand side of fig. 6.8.

We have claimed that in the scale-invariant regime the dominant constraint is coming from f_{NL} and not from f_L , contrarily to the nonscale-invariant regime. We now justify this claim. Repeating the steps above for the tachyon spectrum, S, one obtains

$$|f_L| \sim 10^{-6} \left(\eta_f e^{2\eta_f N_e}\right)^2$$
 (6.70)

Thus, in the scale-invariant regime, the linearity and nonlinearity parameters are related as

$$|f_{NL}| \sim 10^6 |f_L|^{3/2} \tag{6.71}$$

⁶As discussed previously, an integral of the form $\int dNe^g$ is approximated by $e^{g_*}/\sqrt{|g'_*|}$ where g_* is the maximum value (at $N = N_*$) and g'_* is the derivative evaluated at the same point.

⁷In [5], the conformal time when the instability starts is (perhaps confusingly named) $\tau_* = -1/H$, due to our choice of N = 0 for the beginning of the instability.

⁸The horizon-crossing condition is $k\tau_i = 1$, and $\tau_*/\tau_i = e^{N_i}/e^0$.

so that $|f_{NL}| > |f_L|$ except when $|f_L|$ is extremely small. This demonstrates that it is indeed possible to obtain significant nongaussianity in this region of the parameter space. We have verified that the result (6.71) can also be derived using the mode function solutions (6.28).

6.7 Implications for Brane-Antibrane Inflation

We now apply our results to a popular model of inflation from string theory, braneantibrane inflation [24]. This can be done by mapping the low-energy effective action for the brane-antibrane system onto the hybrid inflation model. We focus on the popular KKLMMT scenario [24, 25] which reconciles brane inflation with modulus stabilization using warped geometries with background fluxes for type IIB string theory vacua [21]. In this model, the antibrane is at the bottom of a Klebanov-Strassler (KS) throat [197], with warp factor $a_i \ll 1$, and the brane moves down the throat.⁹ Within the KS throat the geometry is well approximated by

$$ds^{2} = a(y)^{2}g_{\mu\nu}dx^{\mu}dx^{\nu} + dy^{2} + y^{2}d\Omega_{5}^{2}$$

where y is the distance along the throat, $a(y) \cong e^{ky}$ is the warp factor and $d\Omega_5^2$ is the metric on the base space of the corresponding conifold sigularity of the underlying Calabi-Yau space. In the subsequent analysis we ignore the base space and treat the geometry as AdS_5 .

In the following we compute only the nongaussianity which is due to the preheating dynamics and ignore the possible effects of the inflaton sound speed [162, 163, 199]; hence our results may be thought of as a lower bound on the nongaussianity from brane inflation.

In string theory the open string tachyon T between a D3-brane and antibrane,¹⁰

⁹See [198] for other discussions on nongaussianity in string theory models of inflation.

¹⁰We restrict ourselves to inflation models driven by D3-branes since inflation driven by higher dimensional branes have problems with overclosure of the universe by defect formation [4].

separated by a distance y, is described by the action [43]

$$S_{\text{tac}} = -\int d^4x \sqrt{-g} \mathcal{L}$$

$$\mathcal{L} = V(T, y) \sqrt{1 + (a_i M_s)^{-2} g^{\mu\nu} \partial_\mu T^* \partial_\nu T}$$
(6.72)

Here the small-|T| expansion of the potential is

$$V(T,y) = 2 a_i^4 \tau_3 \left[1 + \frac{1}{2} \left(\left(\frac{M_s y}{2\pi} \right)^2 - \frac{1}{2} \right) |T|^2 + \mathcal{O}(|T|^4) + \cdots \right].$$
(6.73)

where M_s is the string mass scale, $\tau_3 = g_s^{-1} M_s^4 / (2\pi)^3$ is the D3-brane tension, and g_s is the string coupling. Notice that in the warped throat scenario the instability does not set in until the branes are separated by the unwarped string length,¹¹ M_s^{-1} . An interesting difference between the string tachyon and that of ordinary hybrid inflation is that (at y = 0) the tachyon potential V(|T|) in the string case does not have a local minimum; rather

$$V(T,0) \sim \tau_3 e^{-|T|^2/4}$$
 (6.74)

The potential is minimized as $T \to \infty$. Therefore T does not have a VEV. Nevertheless, the unstable brane-antibrane system decays into closed strings soon after the instability begins, and the large-T part of the potential is not meaningful for determining the actual evolution of the tachyon. In hybrid inflation, it is also true that the end of inflation occurs somewhat before the fluctuations of the tachyon become as large as the VEV. We will see that even in the absence of a T^4 coupling, we can still define the equivalent of λ and v for the brane-antibrane system, by equating $\frac{1}{4}\lambda v^4$ to the false vacuum energy, and λv^2 to the tachyon mass scale. This amounts to replacing the condition for the end of inflation (6.31) by

$$\int \frac{d^3k}{(2\pi)^3} |\xi_k|^2 \Big|_{N=N_\star} = \frac{\text{false vacuum energy}}{|\text{tachyon mass}|^2}$$
(6.75)

¹¹There is some confusion on the literature on this point, with some papers having stated that the instability is determined by the warped string length, but this is not the case [163]. We thank L. Leblond for pointing this out to us.

Despite that fact that the tachyon potential is only minimized at $|T| \to \infty$ the condition (6.75) is quite reasonable. Detailed numerical simulations of the symmetry breaking in the theory (6.72) were performed in [4]. Comparing to the analysis of [4] one finds that N_{\star} as defined in (6.75) roughly corresponds to the time at which singularities in the spatial gradients of the tachyon field form [51]. The appearance of singularities within a finite time corresponds to the formation of lower dimensional branes [1] and hence by $N = N_{\star}$ the inflaton field ceases to exist as a physical degree of freedom. This means that, as in our previous analysis, for $N > N_{\star}$ there no longer exists any nonadiabatic pressure (since only one field, the tachyon, is dynamical) and the large scale curvature perturbation becomes conserved to all orders in perturbation theory [192, 194].

The effective values of the couplings can be found by rewriting the action in terms of the canonically normalized fields $\sigma = a_i \sqrt{\tau_3} T/M_s$ and $\varphi = \sqrt{\tau_3} y$ (see equations 3.6, 3.10 or C.1 in [24]), and then matching to the hybrid inflation potential (6.9). This gives the correspondence

$$v = \sqrt{\frac{2}{\pi^3}} \frac{a_i M_s}{\sqrt{g_s}} \tag{6.76}$$

$$\lambda = \frac{\pi^3}{4} g_s \tag{6.77}$$

$$g = \sqrt{2\pi g_s} a_i \tag{6.78}$$

For the analysis of the preceding sections to be valid, the inflaton potential must be well-described by $V_0 + \frac{1}{2}m_{\varphi}^2\varphi^2$ during the relevant e-foldings of inflation. The full potential can be written as

$$V_{\rm inf} = \frac{m_{\varphi}^2}{2}\varphi^2 + V_0 \left(1 - \frac{\nu}{4\pi^2} \frac{V_0}{\varphi^4}\right)$$
(6.79)

where $V_0 = 2a_i^4 \tau_3$ and ν is a geometrical factor which is given $\nu = 27/16$ for the KS throat. It is typical to parameterize the inflaton mass in terms of the dimensionless quantity β as $m_{\varphi}^2 = \beta H_0^2$ where $H_0^2 = V_0/(3M_p^2)$. Using the COBE normalization, we find that

$$\beta = 10^{7/2} a_i^3 \left(\frac{M_s}{M_p}\right)^3.$$
(6.80)

Demanding that the mass term in (6.79) dominate over the Coulomb term even when the branes are separated by the local string length yields a lower bound on β :

$$\beta > 324\pi^4 g_s^2 a_i^{10} \left(\frac{M_p}{M_s}\right)^2$$

The parameter β is also bounded from above by the requirement that $|n-1| \lesssim 10^{-1}$ which corresponds to $g_s^2 a_i^{10} (M_p/M_s)^2 \ll 5 \times 10^{-6}$.

Our results apply only in the case that $\beta > 0$; moreover the case where $\beta < 0$ does not make sense from the string theory point of view, since $\varphi = 0$ denotes the bottom of the throat, and the brane must roll toward that point, not away from it [27].

As in hybrid inflation, we still have three parameters even after normalizing the spectrum, which we can take to be g_s , a_i and the ratio of the warped string scale to the Planck scale, $a_i M_s/M_p$. Taking into account the additional restrictions on β , we find that the scale-noninvariant exclusions (right-hand side of fig. 6.8) do not survive at all in the KKLMMT model; however all the scale-invariant ones do. Therefore this model has the potential for producing large nongaussianity, and is even constrained by producing too high levels of nongaussianity.



Figure 6.9: Excluded regions of the KKLMMT brane-antibrane inflation parameter space, in the plane of $\log_{10} a_i$ versus $\log_{10} g_s$ for $\log_{10} a_i M_s/M_p = -13, -11, \ldots, -5$ from nongaussianity.

The constraints in the string parameter space are shown in figure 6.9. The excluded regions shown here correspond to very small values of $g_s \lesssim 10^{-10}$. In the

simplest way of connecting type IIB string theory to low-energy phenomenology, the gauge couplings of the Standard Model are related to g_s by running down from the string scale, which would render such small values of g_s incompatible with observations. However, type IIB strings are dual to themselves under SL(2,Z) transformations which take $g_s \rightarrow 1/g_s$. In the dual picture, the string coupling is very large, and the gauge dynamics at the string scale would be confining. It is conceivable that the Standard Model arises as a remnant of a strongly coupled gauge theory at the high scale, similar to technicolor models. In this case, the small values of g_s which give rise to large nongaussianity could still be compatible with particle physics constraints.

6.8 Implications for P-term Inflation

In realistic models of hybrid inflation from supergravity, the potential is generated by F- or D-terms. P-term inflation is a class of models which combines the two kinds of terms and can interpolate between them [182]. The potential for P-term inflation, along the inflationary trajectory, is

$$V_{\rm inf} = \frac{g^2 \xi^2}{2} \left(1 + \frac{g^2}{8\pi^2} \ln \frac{\varphi^2}{\varphi_c^2} + \frac{f}{8} \frac{\varphi^4}{M_p^4} + \cdots \right)$$
(6.81)

where \cdots denotes terms of order φ^6/M_p^6 and higher. Here $\varphi_c = \xi = \sqrt{|\xi_+|^2 + \xi_3^2}$ is defined in terms of two Fayet-Iliopoulos parameters ξ_+ and ξ_3 , and in (6.81) f must lie in the interval $0 \le f \le 1$, since it is defined as $f = |\xi_+|^2/\xi^2$. The limits f = 0 and 1 correspond to D-term [200] and F-term [201] models, respectively. We will consider each of these limits separately. We do not consider the complications which arise when these models are coupled to moduli fields [202].

As in the models previously considered the false vacuum energy dominates during inflation and the Hubble scale is given by

$$H \cong \frac{g\,\xi}{\sqrt{6}M_p} \tag{6.82}$$

The inflaton couples to two scalar fields Φ_{\pm} , of which one linear combination σ is tachyonic. Its mass-squared is given by

$$m_{\sigma}^2 = g^2 \left(\varphi^2 - \xi\right) \tag{6.83}$$

By comparing (6.82) and (6.83) to the hybrid inflation potential (6.9) we can determine the hybrid inflation model parameters as

$$\lambda = \frac{g^2}{2} \tag{6.84}$$

$$v = \sqrt{2\xi} \tag{6.85}$$

The coupling g retains its original meaning in P-term inflation.

6.8.1 D-term Inflation

D-term inflation corresponds to taking f = 0 in (6.81). During a slow roll phase the inflaton field evolves as

$$\varphi_0(t)^2 = \varphi_c^2 - \frac{g^3\xi}{2\sqrt{6}\pi^2}(t - t_c)$$

which implies that m_{σ}^2 varies linearly with the number of e-foldings. Scales relevant for the CMB left the horizon when $\varphi = \varphi_N$ where

$$\varphi_N^2 = \varphi_c^2 + \frac{g^2 N}{2\pi^2} M_p^2 = \xi + \frac{g^2 N}{2\pi^2} M_p^2$$

Two distinct regimes are possible depending on the value of the coupling g. It is often assumed that g is relatively large so that $g^2 N/(2\pi^2) \gg \xi/M_p^2$ [203] which gives the correct amplitude of density perturbations with $\xi \cong 10^{-5}M_p^2$ and requires $g \gtrsim 2 \times 10^{-3}$ for consistency. In this regime $\varphi_N \gg \varphi_c$ so that slow roll at the onset of the instability is not guaranteed and our previous analysis of the tachyon mode functions does not apply. However, in this regime it is also difficult to satisfy the constraints coming from the cosmic string tension, to avoid overproduction of cosmic strings, $\xi \lesssim 4 \times 10^{-7} M_p^2$ [204]. (Ref. [205] has pointed out that the constraints on the cosmic string tension can be weakened by incorporating the effect which strings have on the observed spectral index.)

We are therefore driven to consider D-term inflation in the regime of small coupling g^2 so that $g^2 N/(2\pi^2) \ll \xi/M_p^2$ and $\varphi_N \cong \varphi_c$. In this case we are guaranteed that the universe will still be in a slow roll phase at the onset of the instability and our previous analysis holds without modification. This corresponds to very small

couplings $g \ll 2 \times 10^{-3}$, however, there is no obstruction to taking such a small coupling since g^2 is not necessarily related to the gauge coupling constant in a GUT [182]. In this regime the COBE normalization fixes $\xi \cong 7 \times 10^{-4} g^{2/3} M_p^2$ so that g is the only independent model parameter. The cosmic string constraint $\xi \lesssim 4 \times 10^{-7} M_p^2$ then restricts the coupling g to be smaller than $g \lesssim 1.3 \times 10^{-5}$.

Applying our previous analysis of hybrid inflation to D-term inflation,¹² including the additional constraints mentioned above, we find that there is a range of couplings,

$$-10.0 < \log_{10} g \lesssim -8.7 \tag{6.86}$$

which is ruled out because of the spectral distortion constraint, in the scale-noninvariant region of figure 6.8. On the other hand there is no constraint coming from nongaussianity for this model.

6.8.2 F-term Inflation

F-term inflation [201, 207] corresponds to taking f = 1 in (6.81). In this case the dynamics are somewhat more complicated than the D-term model. Again there are two possible regimes: a large coupling regime where $\varphi_N \gg \varphi_c$ and our previous analysis does not apply and also a small coupling regime where $\varphi_N \cong \varphi_c$ and our previous analysis does apply. The large coupling regime corresponds to $g \gtrsim 2 \times 10^{-3}$ and again the cosmic string tension constraints are difficult to satisfy (see, however, [208]). We are therefore driven to consider only the small coupling regime, $g \lesssim 2 \times 10^{-3}$. For couplings $3 \times 10^{-7} \ll g \lesssim 2 \times 10^{-3}$ it can be shown that the quadratic term φ^4/M_p^4 in the potential (6.81) can be neglected and the dynamics is identical to D-term inflation, which we have already considered. Thus we consider only the F-term model for $g \ll 3 \times 10^{-7}$ since this is the only region of parameter space for which the model differs significantly from the D-term model. ¹³

 $^{^{12}}$ See [206] for further discussion of preheating in D-term inflation.

¹³We have neglected the intermediate regime $0.06 \lesssim g \lesssim 0.15$ which will not yield significant nongaussianity or spectral distortion.

For $g \lesssim 3 \times 10^{-7}$, so the *f*-term is dominating the potential, the slow roll parameter $\epsilon = \xi^3/(8M_p^6)$, and the COBE normalization fixes $\xi \cong 6.7 \times 10^6 g^2 M_p^2$ and the cosmic string tension is within observational bounds for $g \lesssim 2 \times 10^{-7}$. Again applying the general hybrid inflation constraints, we find the excluded region

$$-13.0 \lesssim \log_{10} g \lesssim -9.5 \tag{6.87}$$

which, as in the case of D-term inflation, comes from the k^3 spectral distortion effect rather than nongaussianity.

For more general P-term models with 0 < f < 1 we expect the excluded regions to interpolate between (6.86) and (6.87). In deriving our constraints we have been driven to the small coupling regime by the requirement that the cosmic sting tension be within observational bounds. Our analysis does not give any significant constraint on the string theoretic D3/D7 model [209] since in this case the cosmic strings are not stable and there is no motivation to consider the small values of the coupling g in (6.86) and (6.87). Indeed, such small couplings are difficult to motivate from string theory [210].

6.9 The Case of a Complex Tachyon

In the preceeding sections we have applied results which were derived under the assumption that σ is a real field, to models in which the tachyon is actually complex. In doing so we have assumed that the generalization of the analysis of [5] to the case of a complex tachyon does not significantly modify the exclusion plot, figure 6.8. Here we verify this claim.

6.9.1 Cosmological Perturbation Theory for an O(M) Multiplet

Before restricting to the case of a complex tachyon we consider the somewhat more general case of an O(M) symmetric multiplet of tachyon fields σ_A with $A = 1, \ldots, M$. The matter sector is expanded in perturbation theory as

$$\varphi(\tau, \vec{x}) = \varphi^{(0)}(\tau) + \delta^{(1)}\varphi(\tau, \vec{x}) + \frac{1}{2}\delta^{(2)}\varphi(\tau, \vec{x})$$
(6.88)

$$\sigma_A(\tau, \vec{x}) = \delta^{(1)} \sigma_A(\tau, \vec{x}) + \frac{1}{2} \delta^{(2)} \sigma_A(\tau, \vec{x}).$$
(6.89)

As before the time-dependent vacuum expectation value (VEV) of the tachyon fields are set to zero $\langle \sigma_A \rangle \equiv \sigma^{(0)} = 0$ which is a consequence of the O(M) symmetry of the theory. Notice, however, that the tachyon field *does* develop an effective VEV for the radial component

$$\langle |\sigma| \rangle \equiv \langle \sqrt{\sigma_A \sigma^A} \rangle \neq 0$$

We also assume that

$$\frac{\partial V}{\partial \sigma_A} = \frac{\partial^2 V}{\partial \sigma_A \partial \varphi} = 0$$

but V is, for the time being, otherwise arbitrary. Here and elsewhere the potential and its derivatives are understood to be evaluated on background values of the fields so that $V = V(\varphi^{(0)}, \sigma^{(0)})$, for example.

We consider only the $\delta^{(2)}G_0^0 = \kappa^2 \delta^{(2)}T_0^0$, $\partial_i \delta^{(2)}G_0^i = \kappa^2 \partial_i \delta^{(2)}T_0^i$ and $\delta_j^i \delta^{(2)}G_i^j = \kappa^2 \delta_j^i \delta^{(2)}T_i^j$ equations since the second order vector and tensor fluctuations decouple from this system. In the case that $\sigma_A^{(0)} = 0$ the second order tachyon fluctuations $\delta^{(2)}\sigma_A$ decouple from the inflaton and gravitational fluctuations up to second order and hence we do not need to solve for $\delta^{(2)}\sigma_A$. Note also that the Klein-Gordon equation for the inflaton fluctuations is not necessary to close the system. In this section we sometimes insert the slow roll parameters ϵ and η explicitly though we do not yet assume that they are small. We also introduce the shorthand notation $m_{\varphi}^2 \equiv \partial^2 V/\partial \varphi^2$.

The second order (0,0) equation is

$$3\mathcal{H}\psi^{\prime(2)} + (3-\epsilon)\mathcal{H}^{2}\phi^{(2)} - \partial^{k}\partial_{k}\psi^{(2)}$$

= $-\frac{\kappa^{2}}{2}\left[\varphi_{0}^{\prime}\delta^{(2)}\varphi^{\prime} + a^{2}\frac{\partial V}{\partial\varphi}\delta^{(2)}\varphi\right] + \Upsilon_{1}$ (6.90)

where Υ_1 is constructed entirely from first order quantities. Dividing Υ_1 into inflaton and tachyon contributions we have

$$\Upsilon_1=\Upsilon_1^arphi+\Upsilon_1^\sigma$$

where

$$\Upsilon_{1}^{\varphi} = 4(3-\epsilon)\mathcal{H}^{2}(\phi^{(1)})^{2} + 2\kappa^{2}\varphi_{0}'\phi^{(1)}\delta^{(1)}\varphi'
- \frac{\kappa^{2}}{2}(\delta^{(1)}\varphi')^{2} - \frac{\kappa^{2}}{2}a^{2}m_{\varphi}^{2}(\delta^{(1)}\varphi)^{2} - \frac{\kappa^{2}}{2}(\vec{\nabla}\delta^{(1)}\varphi)^{2}
+ 8\phi^{(1)}\partial^{k}\partial_{k}\phi^{(1)} + 3(\phi'^{(1)})^{2} + 3(\vec{\nabla}\phi^{(1)})^{2}$$
(6.91)

and

$$\Upsilon_{1}^{\sigma} = -\frac{\kappa^{2}}{2} \left[\delta^{(1)} \sigma_{A}^{\prime} \delta^{(1)} \sigma^{\prime A} + \partial_{i} \delta^{(1)} \sigma_{A} \partial^{i} \delta^{(1)} \sigma^{A} + a^{2} \frac{\partial^{2} V}{\partial \sigma_{A} \partial \sigma_{B}} \delta^{(1)} \sigma_{A} \delta^{(1)} \sigma_{B} \right].$$
(6.92)

The divergence of the second order (0, i) equation is

$$\partial^k \partial_k \left[\psi^{\prime(2)} + \mathcal{H} \phi^{(2)} \right] = \frac{\kappa^2}{2} \varphi_0^\prime \partial^k \partial_k \delta^{(2)} \varphi + \Upsilon_2 \tag{6.93}$$

where $\Upsilon_2 = \Upsilon_2^{\varphi} + \Upsilon_2^{\sigma}$ is constructed entirely from first order quantities. The inflaton part is

$$\Upsilon_{2}^{\varphi} = 2\kappa^{2}\varphi_{0}^{\prime}\partial_{i}\left(\phi^{(1)}\partial^{i}\delta^{(1)}\varphi\right) + \kappa^{2}\partial_{i}\left(\delta^{(1)}\varphi^{\prime}\partial^{i}\delta^{(1)}\varphi\right) - 8\partial_{i}\left(\phi^{(1)}\partial^{i}\phi^{\prime(1)}\right) - 2\partial_{i}\left(\phi^{\prime(1)}\partial^{i}\phi^{(1)}\right)$$
(6.94)

and the tachyon part is

$$\Upsilon_2^{\sigma} = \kappa^2 \partial_i \left(\delta^{(1)} \sigma'_A \partial^i \delta^{(1)} \sigma^A \right).$$
(6.95)

The trace of the second order (i, j) equation is

$$3\psi''^{(2)} + \partial^{k}\partial_{k} \left[\phi^{(2)} - \psi^{(2)}\right] + 6\mathcal{H}\psi'^{(2)}$$

+
$$3\mathcal{H}\phi'^{(2)} + 3(3 - \epsilon)\mathcal{H}^{2}\phi^{(2)}$$

=
$$\frac{3\kappa^{2}}{2} \left[\varphi_{0}'\delta^{(2)}\varphi' - a^{2}\frac{\partial V}{\partial\varphi}\delta^{(2)}\varphi\right] + \Upsilon_{3}$$
(6.96)

where $\Upsilon_3 = \Upsilon_3^{\varphi} + \Upsilon_3^{\sigma}$ is constructed entirely from first order quantities. The inflaton part is

$$\Upsilon_{3}^{\varphi} = 12(3-\epsilon)\mathcal{H}^{2}(\phi^{(1)})^{2} - 6\kappa^{2}\varphi_{0}'\phi^{(1)}\delta^{(1)}\varphi' + \frac{3\kappa^{2}}{2}(\delta^{(1)}\varphi')^{2} - \frac{3\kappa^{2}}{2}a^{2}m_{\varphi}^{2}(\delta^{(1)}\varphi)^{2} - \frac{\kappa^{2}}{2}(\vec{\nabla}\delta^{(1)}\varphi)^{2} + 3(\phi'^{(1)})^{2} + 8\phi^{(1)}\partial^{k}\partial_{k}\phi^{(1)} + 24\mathcal{H}\phi^{(1)}\phi'^{(1)} + 7(\vec{\nabla}\phi^{(1)})^{2}$$
(6.97)

and the tachyon part is

$$\Upsilon_{3}^{\sigma} = \kappa^{2} \left[\frac{3}{2} \delta^{(1)} \sigma_{A}^{\prime} \delta^{(1)} \sigma^{\prime A} - \frac{1}{2} \partial_{i} \delta^{(1)} \sigma_{A} \partial^{i} \delta^{(1)} \sigma^{A} - \frac{3}{2} a^{2} \frac{\partial^{2} V}{\partial \sigma_{A} \partial \sigma_{B}} \delta^{(1)} \sigma_{A} \delta^{(1)} \sigma_{B} \right]$$

$$(6.98)$$

The derivation of the master equation which was presented in appendix D-2 follows here unmodified except for the new definitions of Υ_1^{σ} , Υ_2^{σ} and Υ_3^{σ} . The master equation is

$$\phi^{\prime\prime(2)} + 2(\eta - \epsilon)\mathcal{H}\phi^{\prime(2)} + \left[2(\eta - 2\epsilon)\mathcal{H}^2 - \partial^k\partial_k\right]\phi^{(2)} = J$$
(6.99)

where the source is

$$J = \Upsilon_1 - \Upsilon_3 + 4\Delta^{-1}\Upsilon'_2 + 2(1 - \epsilon + \eta)\mathcal{H}\Delta^{-1}\Upsilon_2 + \Delta^{-1}\gamma'' - (1 + 2\epsilon - 2\eta)\mathcal{H}\Delta^{-1}\gamma'.$$
(6.100)

and the quantity γ is defined as

$$\gamma = \Upsilon_3 - 3\Delta^{-1}\Upsilon_2' - 6\mathcal{H}\Delta^{-1}\Upsilon_2 \tag{6.101}$$

We can split the source into tachyon and inflaton contributions $J = J^{\varphi} + J^{\sigma}$ in the obvious manner, by taking the tachyon and inflaton parts of $\Upsilon_1, \Upsilon_2, \Upsilon_3, \gamma$.

In appendix D-6 we prove the identity (see eqn. D-41)

$$\gamma_{\sigma} = -\frac{\kappa^2}{2} \left(\partial_i \delta^{(1)} \sigma_A \partial^i \delta^{(1)} \sigma^A \right) - 3\kappa^2 \Delta^{-1} \partial_i \left(\partial^k \partial_k \delta^{(1)} \sigma_A \partial^i \delta^{(1)} \sigma^A \right)$$

which is analogous to the result for a real tachyon field.

We now proceed to derive the tachyon curvature perturbation. The derivation of $\zeta_{\sigma}^{(2)}$ presented previously follows unmodified except, of course, for the change in the definitions of Υ_{1}^{σ} , Υ_{2}^{σ} , Υ_{3}^{σ} and γ_{σ} . From this point onwards we assume that ϵ , $|\eta| \ll 1$. The leading contribution to the tachyon curvature perturbation is

$$\begin{aligned} \zeta_{\sigma}^{(2)} &\cong \frac{1}{\epsilon} \int_{\tau_{i}}^{\tau} d\tau' \left[-\frac{\Upsilon_{1}^{\sigma}}{\mathcal{H}(\tau')} + \frac{1}{3} \frac{\Upsilon_{3}^{\sigma}}{\mathcal{H}(\tau')} \right. \\ &\left. - \frac{2}{3} \frac{\mathcal{H}(\tau')^{2}}{\mathcal{H}(\tau)^{3}} \Upsilon_{3}^{\sigma} \right] \end{aligned}$$
Now, using equations (D-8) and (D-14) we can write this in terms of the tachyon fluctuation $\delta^{(1)}\sigma$ as

$$\begin{aligned} \zeta_{\sigma}^{(2)} &\cong \frac{\kappa^2}{\epsilon} \int_{-1/a_i H}^{\tau} d\tau' \left[\frac{\delta^{(1)} \sigma'_A \delta^{(1)} \sigma'^A}{\mathcal{H}(\tau')} \right. \\ &- \frac{\mathcal{H}(\tau')^2}{\mathcal{H}(\tau)^3} \left(\delta^{(1)} \sigma'_A \delta^{(1)} \sigma'^A \right. \\ &- \left. a^2 \frac{\partial^2 V}{\partial \sigma_A \partial \sigma_B} \delta^{(1)} \sigma_A \delta^{(1)} \sigma_B \right) \right] \end{aligned} \tag{6.102}$$

The corrections to (6.102) are either total gradients or are subleading in the slow roll expansion. In deriving (6.102) we have restricted ourselves to the preheating phase during which the fluctuations $\delta^{(1)}\sigma_A$ grow exponentially.

Using (6.102) the second order tachyon curvature perturbation can be computed once the fluctuations $\delta^{(1)}\sigma_A$ are determined. The first order tachyon fluctuations are described by the perturbed Klein-Gordon equation

$$\delta^{(1)}\sigma_A'' + 2\mathcal{H}\delta^{(1)}\sigma_A - \partial^k\partial_k\delta^{(1)}\sigma_A + a^2\frac{\partial^2 V}{\partial\sigma_A\partial\sigma_B}\delta^{(1)}\sigma_B = 0$$
(6.103)

6.9.2 Complex Tachyon Mode Functions

At this point we restrict our attention to the case with M = 2 and the potential

$$V = \frac{\lambda}{4} \left(\sigma_A \sigma^A - v^2\right)^2 + \frac{g^2}{2} \varphi^2 \sigma_A \sigma^A + \frac{m_{\varphi}^2}{2} \varphi^2 \qquad (6.104)$$

For $\sigma_A^{(0)} = 0$ the mass matrix is diagonal

$$\frac{\partial^2 V}{\partial \sigma_A \partial \sigma_B} = \left(-\lambda v^2 + g^2 \varphi_0^2 \right) \delta_{AB}$$
$$\equiv m_\sigma^2 \delta_{AB}$$

so that the tachyon fluctuations with A = 1 and A = 2 evolve independently (see eqn. 6.103).

As previously the quantum mechanical solutions $\delta^{(1)}\sigma_A$ are written in terms of annihilation and creation operators a_k^A , $a_k^{A\dagger}$ in the usual way

$$\delta^{(1)}\sigma_A(x) = \int \frac{d^3k}{(2\pi)^{3/2}} a_k^A \xi_k(t) e^{ikx} + \text{h.c.}$$
(6.105)

Both components A = 1 and A = 2 have the same time dependence owing to the fact that the mass matrix is diagonal. The ξ_k in (6.105) are thus identical to the solutions (6.28), which we have already studied.

6.9.3 The End of Symmetry Breaking

For a multi-component tachyon the condition defining N_{\star} must be modified as $\langle \delta^{(1)} \sigma_A \delta^{(1)} \sigma^A \rangle (N = N_{\star}) = v^2/4$ which, for the case M = 2, changes (6.31) to

$$\int \frac{d^3k}{(2\pi)^3} |\xi_k|^2 \bigg|_{N=N_\star} = \frac{v^2}{8}$$

6.9.4 Tachyon Curvature Perturbation

For the potential (6.104) the tachyon curvature perturbation $\zeta_{\sigma}^{(2)}$ decomposes into a sum of term

$$\zeta_{\sigma}^{(2)} = \sum_{A=1,2} \zeta_{A}^{(2)}$$

where $\zeta_A^{(2)}$ is the contribution to $\zeta_{\sigma}^{(2)}$ coming from σ_A . Consider, as an example, the spectrum of the tachyon curvature perturbation

$$\begin{aligned} \langle \zeta_{\sigma,k_1}^{(2)} \zeta_{\sigma,k_2}^{(2)} \rangle &= \langle \zeta_{1,k_1}^{(2)} \zeta_{1,k_2}^{(2)} \rangle + \langle \zeta_{2,k_1}^{(2)} \zeta_{2,k_2}^{(2)} \rangle \\ &+ \langle \zeta_{1,k_1}^{(2)} \zeta_{2,k_2}^{(2)} \rangle + \langle \zeta_{1,k_2}^{(2)} \zeta_{2,k_1}^{(2)} \rangle \end{aligned}$$

Because the annihilation/creation operators a_k^1 and a_k^2 are independent the crossterms on the last line do not contribute to the connected part of the correlation function. This means that

$$\langle \zeta_{\sigma,k_1}^{(2)} \zeta_{\sigma,k_2}^{(2)} \rangle = 2 \, \langle \zeta_{1,k_1}^{(2)} \zeta_{1,k_2}^{(2)} \rangle$$

The quantity $\langle \zeta_{1,k_1}^{(2)} \zeta_{1,k_2}^{(2)} \rangle = \langle \zeta_{2,k_1}^{(2)} \zeta_{2,k_2}^{(2)} \rangle$ will be identical to the $\langle \zeta_{\sigma,k_1}^{(2)} \zeta_{\sigma,k_2}^{(2)} \rangle$ which we have already computed for the real tachyon field. We see, then, that the effect of having a complex tachyon field (as opposed to a real field) is to multiply f_L and f_{NL} by a factor of 2 and also to slightly reduce N_{\star} . The net change in f_L , f_{NL} is order unity and the new exclusion plot is difficult to visually distinguish from figure 6.8. This justifies our previous claims that our constraints do not change significantly when one considers a complex tachyon field.

6.10 Conclusions

In this chapter we have studied the evolution of the second order curvature perturbation during tachyonic preheating at the end of hybrid inflation. We have found that, depending on the values of certain model parameters, two interesting effects are possible:

- Preheating generates a scale-invariant contribution to the curvature perturbation. In this case significant nongaussianity can be generated during preheating and the model is even constrained by producing too high a level of nongaussianity.
- Preheating generates a nonscale-invariant contribution to the curvature perturbation with spectral index n = 4. In this case the strongest constraint comes from the distortion of the power spectrum and no significant nongaussianity can be produced.

In both cases one typically requires fairly small values of the dimensionless couplings g, λ in order to obtain a strong effect. Note that a small coupling g does not require fine tuning in the technical sense, since g^2 is only multiplicatively renormalized: $\beta(g^2) \sim \mathcal{O}(g^2\lambda, g^4)/(16\pi^2)$. That is, if g is small at tree level then loop corrections do not change its effective value significantly.

We have applied our constraints on hybrid inflation to several popular models: brane inflation, D-term inflation and F-term inflation. In the case of brane inflation we have found that significant nongaussianity from preheating is possible for sufficiently small values of the warp factor. For both D- and F-term inflation we have shown that no nongaussianity is produced during preheating, however, we still put interesting constraints on the model due to the distortion of the spectrum by nonscale-invariant fluctuations. We have also generalized our results to the case of a complex tachyon field, confirming the claim that this modification does not significantly alter our exclusion plots.

We should note that the model of hybrid inflation considered here always gives a small blue tilt to the spectral index, n > 1, which is disfavoured by recent data [16]. One avenue for future study [211] is to generalize our results to the case of inverted hybrid inflation [212] which always gives n < 1.

Acknowledgments. We would like to thank A. Mazumdar and A. Jokinen for stimulating discussions which motivated us to start on [5]. We are grateful to M. Sakellariadou for helpful discussions and suggestions concering our results for the complex tachyon. We thank L. Leblond for alerting us to an error in the original manuscript of [6]. We also wish to thank R. Brandenberger, C. Burgess, C. Byrnes, K. Dasgupta, J. Garcia-Bellido, J. Lidsey, A. Linde, D. Lyth, J. Magueijo, K. Malik, G. D. Moore, A. Notari, G. Rigopoulos, T. Riotto, D. Seery, P. Shellard, G. Shiu, B. van Tent and F. Vernizzi for enlightening discussions and correspondence. This work was supported by NSERC of Canada and FQRNT of Québec.

Chapter 7

p-adic Inflation

Abstract

We construct approximate inflationary solutions rolling away from the unstable maximum of *p*-adic string theory, a nonlocal theory with derivatives of all orders. Novel features include the existence of slow-roll solutions even when the slow-roll parameters, as usually defined, are much greater than unity, as well as the need for the Hubble parameter to exceed the string mass scale m_s . We show that the theory can be compatible with CMB observations if $g_s/\sqrt{p} \sim 10^{-7}$, where g_s is the string coupling, and if $m_s < 10^{-6}M_p$. A red-tilted spectrum is predicted, and the scalar-to-tensor ratio is bounded from above as r < 0.006. The *p*-adic theory is shown to have identical inflationary predictions to a local theory with superPlanckian parameter values, but with the advantage that the *p*-adic theory is ultraviolet complete.

7.1 Introduction

Many string theorists and cosmologists have turned their attention to building and testing stringy models of inflation in recent years. The goals have been to find natural realizations of inflation within string theory, and novel features which would help to distinguish the string-based models from their more conventional field theory counterparts. The more popular categories include brane-antibrane [24, 25, 53], D3/D7 [209], modular [213], DBI [162] and tachyon-driven [214] inflation (see [22] for a review).

In most examples to date, string theory has been used to derive an effective 4D field theory operating at energies below the string scale. Since string theory provides a complete description of dynamics also at higher energies, it may be interesting to consider a model which takes advantage of this distinctive feature. This is usually daunting since the field theory description should be supplemented by an infinite number of higher dimensional operators at energies above the string scale, whose detailed form is not known. In the present work, we propose to take a small step in the direction of overcoming this barrier, by considering a simplified model of string theory invented in 1987 [95], in which the world-sheet coordinates of the string are

restricted to the field of p-adic numbers. Scattering amplitudes of open string theory can be related to those of the p-adic strings. A great advantage in p-adic string theory is that it is possible to compute *all* amplitudes of its lowest state and to determine a simple field-theoretic Lagrangian which exactly reproduces them. The result is a nonlocal field theory which is nevertheless sensible in the far ultraviolet.

The *p*-adic string resembles the bosonic string in that its ground state is a tachyon, whose unstable maximum presumably indicates the presence of a decaying brane, analogous to the unstable D25-brane of the open bosonic string theory [215]. Similarly to the bosonic string, the potential is asymmetric around the maximum, with one direction leading to a zero-energy vacuum, while in the other direction the potential is unbounded from below. We will consider whether it is possible to get successful inflation from rolling toward the bounded direction. This has been tried before in the context of the open string tachyon, and is difficult [216] because the tachyon potential is not flat enough to give a significant period of inflation, and there are no parameters within the theory which can tune the potential to be more flat. In contrast, we will show that the *p*-adic string tachyon *can* roll slowly enough to give many e-foldings of inflation. There are two distinct regions of parameter space which allow for successful inflation. There is a region with $p = \mathcal{O}(1)$ for which the p-adic field potential is flat and slow roll inflation proceeds in the usual manner. However, there is also a region of parameter space with $p \gg 1$ for which the potential is extremely steep $(|\eta| = M_p^2 |V''/V|$ may be as large as 10^{11}) but the p-adic scalar field nevertheless rolls slowly. This remarkable behaviour relies on the nonlocal nature of the theory: the effect of the higher derivative terms in the action is to slow down the field sufficiently, despite its steep potential. This new effect manifests itself only in the regime where the higher derivative interactions cannot be ignored. It is also interesting that in our model the kinetic energy is responsible for driving inflation for a significant number of e-foldings, unlike in conventional models of inflation.

One may worry about the presence of these higher derivative terms, because they

7.1 Introduction

are usually known to introduce $ghosts^1$ into the theory. In fact, it is easy to check that for a scalar field theory if one introduces only a finite number of higher derivatives, then the model invariably contains ghost degrees of freedom. The reason why the *p*adic string action can evade this problem is because it is intrinsically nonperturbative in nature, the propagator being modified in such a way as to not contain any poles. In other words there are no physical states, ghosts or otherwise, around the true vacuum. This novel way of curing the problem of ghosts in higher derivative theories, while retaining some nice properties such as improved UV behaviour, was already pointed out in [218] in the context of gravity. It was also pointed out in [218] that such theories can exibit interesting new cosmological features. For instance, one can obtain nonsingular bouncing solutions by making gravity weak at short distances. More recently, such models have also been shown to possess inflationary solutions [219]. However, the models of [218, 219] are phenomenological, while the *p*-adic action that we consider is an actual (albeit exotic) string theory and reproduces many nontrivial features of conventional string theories.

We start by reviewing the salient features of p-adic string theory in section 7.2. In section 7.3 we show that this theory does not give inflation if the higher-derivative terms in the action are ignored. However, near the top of the potential, the energy can be large enough to justify keeping all higher derivative terms. In section 7.4 we

¹One may also worry about classical instabilities which usually plague higher derivative theories; generically they go by the name of Ostrogradski instabilities (see [217] for a review). These are the classical manifestations of having ghosts in the theory: they can have arbitrarily large negative energy, which leads to classical instability. Since the nonlocal theory under consideration does not contain any ghosts we also do not expect to find such instabilities. One way to see how such theories may avoid the Ostrogradski instability argument, valid for finite higher derivative theories, is by noting that one cannot construct the usual Ostrogradski Hamiltonian because there is no highest derivative in such nonlocal actions. Also, in arriving at the Ostrogradski Hamiltonian, one assumes that all the derivatives of the field (except the maximal one) are independent canonical variables. This is no longer true for theories with derivatives of infinite order. For instance, it is not possible to independently choose an initial condition with arbitrarily specified values of all the derivatives of the field. See ref. [87] for a discussion of this point.

show how to resum their contributions and we construct approximate inflationary solutions valid near the top of the potential, by solving the coupled equations for the tachyon and scale factor of the universe. We give two different approximate methods in this section. In section 7.5 we solve for the fluctuations around this background to determine the power spectrum of scalar and tensor perturbations which can be probed by the cosmic microwave background (CMB). There we show that it is possible to choose parameters which are compatible with the measured amplitude and spectral index, and that the scalar-to-tensor ratio is bounded from above as r < 0.006 in this model. We also argue that it can be natural to have initial conditions compatible with inflation in the *p*-adic theory. We give conclusions in section 7.6. Appendix E-1 gives details about the *p*-adic stress tensor and the approximate inflationary solution of the Friedmann equation. Appendix E-2 gives mathematical details about the incomplete cylindrical functions of the Sonine-Schaefli form. Appendix E-3 explains a formal equivalence between the dynamics of the *p*-adic tachyon and those of a local field theory with a super-Planckian vacuum expectation value (VEV).

7.2 Review of *p*-adic string theory

The action of p-adic string theory is given by [95]

$$S = \frac{m_s^4}{g_p^2} \int d^4 x \left(-\frac{1}{2} \phi \, p^{-\frac{\alpha}{2m_s^2}} \phi + \frac{1}{p+1} \phi^{p+1} \right)$$

$$\equiv \frac{m_s^4}{g_p^2} \int d^4 x \left(-\frac{1}{2} \phi \, e^{-\frac{\alpha}{m_p^2}} \phi + \frac{1}{p+1} \phi^{p+1} \right)$$
(7.1)

where $\Box = -\partial_t^2 + \nabla^2$ in flat space and we have defined

$$\frac{1}{g_p^2} \equiv \frac{1}{g_s^2} \frac{p^2}{p-1} \text{ and } m_p^2 \equiv \frac{2m_s^2}{\ln p}$$
(7.2)

The dimensionless scalar field $\phi(x)$ describes the open string tachyon, m_s is the string mass scale and g_s is the open string coupling constant. Though the action (7.1) was originally derived for p a prime number, it appears that it can be continued to any postive integer and even makes sense in the limit $p \to 1$ [96]. Setting $\Box = 0$ in the action, the resulting potential takes the form $V = (m_s^4/g_p^2)(\frac{1}{2}\phi^2 - \frac{1}{p+1}\phi^{p+1})$. Its shape is shown in figure 7.1.



Figure 7.1: The potential of the *p*-adic tachyon for p = 19.

The action (7.1) is a simplified model of the bosonic string which only qualitatively reproduces some aspects of a more realistic theory. That being said, there are several nontrivial similarities between *p*-adic string theory and the full string theory. For example, near the true vacuum of the theory $\phi = 0$ the field naively has no particlelike excitations since its mass squared goes to infinity.² This is the *p*-adic version of the statement that there are no open string excitations at the tachyon vacuum. A second similarity is the existence of lump-like soliton solutions representing *p*-adic D-branes [65]. The theory of small fluctuations about these lump solutions has a spectrum of equally spaced masses squared for the modes [65],[97], as in the case of normal bosonic string theory.

It should also be noted that the connection between (7.1) and the DBI-type tachyon actions, which have been widely studied in the literature in the context of tachyon matter [43], is not entirely clear (see [220] for a discussion of the relation between *p*-adic and ordinary strings). In the case of tachyon matter, solutions which roll towards the vacuum $T \to \infty$ have late time asymptotics $T \sim t$ and hence the

²Reference [87] found anharmonic oscillations around the vacuum by numerically solving the full nonlinear equation of motion. However, these solutions do not correspond to conventional physical states.

tachyon never reaches this point [47], whereas in the case of the p-adic string the vacuum is at a finite point in the field configuration space and homogeneous solutions rolling towards the vacuum typically pass this point without difficulty [87] (at least in flat space). In fact, the numerical studies of [87] found *no* homogeneous solutions which appeared to correspond to tachyon matter (vanishing pressure at late times). This issue has been considered in the case of cubic string field theory in [221]. Related rolling tachyon solutions in cubic string field theory have been discussed in [222].

It is worth pointing out that one obtains very similar actions to (7.1) with exponential kinetic operators (and usually assumed to have a cubic or quartic potential) when quantizing strings on random lattice [223]. These field theories are also known to reproduce several features, such as the Regge behaviour [224], of their stringy duals. Although our analysis focuses on the specific *p*-adic action, it can easily be applied to such theories as well. The connection between *p*-adic string theory and ordinary string theory on a discrete lattice was explored in [225].

The field equation that results from (7.1) is

$$e^{-\Box/m_p^2}\phi = \phi^p \tag{7.3}$$

We are interested in perturbing around the solution $\phi = 1$, which is a critical point of the potential, representing the unstable tachyonic maximum. For odd p one also has another unstable point at $\phi = -1$, but we will restrict our attention to solutions that start to evolve from the omnipresent $\phi = 1$ maximum. In passing we also note that there is also the stable vacuum of the tachyon, at $\phi = 0$. For both even and odd p the potential is unbounded from below.

It is also worth commenting on the physical interpretation of the fact that the potential is unbounded from below. The instability associated with the decay of the "closed string vacuum" $\phi = 0$ to the "true vacuum" $\phi = \infty$ is thought to be associated with the closed string tachyon instability [215].

Notice that in the limit $p \to 1$ the equation of motion (7.3) becomes a local equation

$$\Box \phi = 2\phi \ln \phi$$

Therefore $p \to 1$ is the limit of local field theory. In the present work we will also consider a very different limit, $p \gg 1$, in which the nonlocal structure of (7.1) is playing an important role in the dynamics.

7.3 Absence of Naive Slow Roll Dynamics

One may wonder whether the field theory (7.1) naively allows for slow roll inflation in the conventional sense. Naively one might expect that for a slowly rolling field the higher powers of \Box in the kinetic term are irrelevant and one may approximate (7.1)by a local field theory. The action (7.1) can be rewritten as

$$S = \int d^4x \left[\frac{1}{2} \chi \Box \chi - V(\chi) + \cdots \right]$$
(7.4)

where we have defined the field χ as

$$\chi = \chi_0 \phi \tag{7.5}$$

$$\chi_0 = \frac{p}{g_s} \sqrt{\frac{\ln p}{2(p-1)}} m_s$$
(7.6)

and the potential is

$$V(\chi) = \frac{m_s^2}{\ln p} \chi^2 - \frac{m_s^4}{g_s^2} \frac{p^2}{p^2 - 1} \left(\frac{\chi}{\chi_0}\right)^{p+1}$$
(7.7)

In (7.4) the \cdots denotes terms with higher powers of \Box . The truncation (7.4) is quite analogous to what is usually performed in the literature when one studies inflation from the string theory tachyon [214]. In the case of the usual string theory tachyon corrections involving \Box^2 and higher are expected, but the full infinite series of higher derivative terms is not known explicitly (see, however, [94] for a calculation of the string theory tachyon action up to order ∂^6). Thus, the standard approach is to simply neglect such terms. We will show, however, that under certain circumstances the higher derivative corrections may play an extremely important role in the inflationary dynamics.

Working in the context of the action (7.4) let us consider the slow roll parameters describing the flatness of the potential (7.7) about the unstable maximum $\chi = \chi_0$. It

7.4 Approximate Solutions: Analytical techniques

is straightforward to show that

$$\frac{M_p^2}{2} \frac{1}{V(\chi_0)^2} \left(\frac{\partial V(\chi)}{\partial \chi} \right)^2 \bigg|_{\chi = \chi_0} = 0$$
(7.8)

$$M_{p}^{2} \frac{1}{V(\chi_{0})} \frac{\partial^{2} V(\chi)}{\partial \chi^{2}} \bigg|_{\chi = \chi_{0}} = -\frac{4g_{s}^{2}}{\ln p} \frac{p^{2} - 1}{p^{2}} \left(\frac{M_{p}}{m_{s}}\right)^{2}$$
(7.9)

Thus one would naively expect that inflation is only possible in the theory (7.1) for $g_s M_p/(\sqrt{\ln p} m_s) \ll 1$. For $p = \mathcal{O}(1)$ we will see that this expectation is correct, however, for $p \gg 1$ this intuition is incorrect and that successful inflation can occur even with $M_p^2 V^{-1} \partial^2 V/\partial \chi^2 \sim -10^{16}$! The reason for this surprising result is that the dynamics of the *p*-adic tachyon field is set by the mass scale which appears in the kinetic term, m_s , rather than the mass scale which is naively implied by the potential

$$m_{\chi}^{2}(\chi_{0}) \equiv \left. \frac{\partial^{2} V}{\partial \chi^{2}} \right|_{\chi=\chi_{0}} = -\frac{2(p-1)}{\ln p} m_{s}^{2}$$

$$(7.10)$$

Clearly for $p \gg 1$ we have $|m_{\chi}^2| \gg m_s^2$.

7.4 Approximate Solutions: Analytical techniques

In this section we construct the approximate solutions for the scalar field and the quasi-de Sitter expansion of the universe, in which ϕ starts near the unstable maximum ($\phi = 1$) of its potential and rolls slowly toward the minimum ($\phi = 0$). Cosmological solutions of similar nonlocal theories have also been considered in [226].

We use two different formalisms to construct inflationary solutions. We first devise a perturbative expansion in $e^{\lambda t}$ similar to what was carried out in [87] to study rolling solutions in flat spacetime. Our second formalism is the analogue of the usual slow roll approximation: we assume that the friction term in the \Box operator dominates over the acceleration term and also neglect the time variation of H.

We first discuss the perturbative expansion in powers of $e^{\lambda t}$. Our starting point is the ansatz

$$\phi(t) = 1 - \sum_{r=1}^{\infty} \phi_r e^{r\lambda t} \tag{7.11}$$

and

$$H(t) = H_0 - \sum_{r=1}^{\infty} H_r e^{r\lambda t}$$
 (7.12)

We have chosen the parameterisation such that at $t \to -\infty$, ϕ starts from the top of the hill where the universe is undergoing de Sitter expansion with Hubble constant H_0 . As $t \to -\infty$, $e^{\lambda t} \to 0$ and all the correction terms, which quantify the departure from the pure de Sitter phase, vanish. As t increases, the field rolls toward the true vacuum $\phi = 0$, in fact reaching it at a finite time. Classically, the model admits infinitely many e-foldings of inflation, although only the last 60 e-foldings before the end of inflation are relevant for observation. This idealized behaviour is an artifact of neglecting quantum fluctuations; quantum mechanically the field cannot sit at $\phi = 1$ for an infinite amount of time. We will return to this issue later and show that the inclusion of quantum fluctuations does not spoil inflation, which it would in ordinary local field theory if the η parameter is large.

Given the ansatz (7.11,7.12), we can expand the field equation for the *p*-adic scalar (7.3) and the Friedmann equation as a series in $e^{\lambda t}$ and then determine the coefficients $\{\phi_r, H_r\}$ systematically, order by order. We will show that this can be done consistently. The zeroth order Klein-Gordon equation is trivially satisfied, by virtue of the fact that we start from a maximum of the potential.

7.4.1 *p*-adic Scalar Field Evolution

Let us first find an approximate solution for the scalar field equation of motion (7.3). We note that to compute quantities such as $\Box^n e^{\lambda t}$ to the order of interest, it is sufficient to use a truncation of (7.11) and (7.12):

$$\phi = 1 - u - \phi_2 u^2, \qquad H = H_0 - H_1 u - H_2 u^2$$
(7.13)

where we have used the freedom to choose the origin of time to set $\phi_1 \equiv 1$, and for convenience we have defined the new variable

$$u \equiv e^{\lambda t} \tag{7.14}$$

in terms of which the \Box operator takes the form

$$\Box = -\lambda^2 \left[u^2 \frac{\partial^2}{\partial u^2} + \left(1 + \frac{3H}{\lambda} \right) u \frac{\partial}{\partial u} \right]$$
(7.15)

We wish to compute quantities such a $\Box^n e^{\lambda t}$ up to $\mathcal{O}(u^2)$. This can be done recursively. Writing $\Box^n \phi$ (where $n \ge 1$) as

$$(-\Box)^n \phi = A_n u + B_n u^2 + \dots$$
 (7.16)

and applying another \Box operator one finds the following recursion relations for the coefficients A_n and B_n :

$$A_{n+1} = A_n(\lambda^2 + 3H_0\lambda)$$
 (7.17)

and

$$B_{n+1} = B_n (4\lambda^2 + 6H_0\lambda) - 3H_1\lambda A_n$$
(7.18)

Equation (7.17) has the solution

$$A_n = a_1 \left(\lambda^2 + 3H_0\lambda\right)^n \tag{7.19}$$

with $a_1 = -1$ (from examination of $\Box \phi$) while a suitable ansatz for B_n is given by

$$B_n = b_1 (\lambda^2 + 3H_0\lambda)^n + b_2 (4\lambda^2 + 6H_0\lambda)^n$$
(7.20)

The coefficients b_1, b_2 can be deduced from (7.18) using the initial values

$$b_1 = -\frac{H_1}{H_0 + \lambda} \text{ and } b_2 = \frac{H_1}{H_0 + \lambda} - \phi_2$$
 (7.21)

which follow from explicitly computing $\Box \phi$ and $\Box^2 \phi$. Putting everything together we now have

$$(-\Box)^{n}\phi = \delta_{n,0} - (\lambda^{2} + 3H_{0}\lambda)^{n}u + \left[\left(\frac{H_{1}}{H_{0} + \lambda} - \phi_{2}\right)(4\lambda^{2} + 6H_{0}\lambda)^{n} - \frac{H_{1}}{H_{0} + \lambda}(\lambda^{2} + 3H_{0}\lambda)^{n}\right]u^{2} + \mathcal{O}(u^{3})$$
(7.22)

which works also for the case n = 0. Using (7.22) one can resum the contributions coming from all the powers of \Box in the exponential operator $e^{-\Box/m_p^2}$ to give

$$e^{-\Box/m_p^2}\phi = 1 - e^{\mu_1}u + \left[\left(\sigma - \phi_2\right)e^{\mu_2} - \sigma e^{\mu_1}\right]u^2 + \cdots$$
(7.23)

where we have introduced

$$\mu_1 \equiv \frac{\lambda^2 + 3H_0\lambda}{m_p^2} \quad , \quad \mu_2 \equiv \frac{4\lambda^2 + 6H_0\lambda}{m_p^2} \quad \text{and} \quad \sigma \equiv \frac{H_1}{H_0 + \lambda} \tag{7.24}$$

We conjecture that such resummations are possible for higher order terms as well. Notice that (7.23) reduces to ϕ in the limit $m_p \to \infty$, as it should.

To solve the equation of motion for the scalar field, we must equate (7.23) to the right-hand-side of (7.3):

$$\phi^{p} = [1 - u - \phi_{2}u^{2}]^{p} = 1 - pu - p\left(\phi_{2} - \frac{p-1}{2}\right)u^{2} + \cdots$$
 (7.25)

matching coefficients for each order in u. The zeroth order equation is identically satisfied, as promised earlier, while matching at first order gives

$$e^{\mu_1} = e^{(\lambda^2 + 3H_0)/m_p^2} = p \tag{7.26}$$

which, using (7.2) can be rewritten in the form

$$\lambda^2 + 3H_0\lambda = 2m_s^2 \tag{7.27}$$

which is independent of p. Later we will see that $H_0 \gg m_s$ is necessary for getting inflation, so the solution of (7.27) is approximately

$$\lambda \cong \frac{2m_s^2}{3H_0} \tag{7.28}$$

Finally, matching coefficients at second order gives

$$\sigma e^{\mu_1} + (\phi_2 - \sigma) e^{\mu_2} = p\left(\phi_2 - \frac{p-1}{2}\right)$$
(7.29)

In summary, by solving the scalar field equation of motion (7.3) to second order in the expansion in powers of $u = e^{\lambda t}$, we have obtained two relations, (7.27) and (7.29). The former determines the parameter λ while the latter determines ϕ_2 .

7.4.2 The Stress Energy Tensor and the Friedmann Equation

To complete our approximate solution for the classical background, we must solve the Friedmann equation

$$H^2 = \frac{1}{3M_p^2} \,\rho_\phi \tag{7.30}$$

to second order in u. To find the energy density ρ_{ϕ} , we turn to the stress energy tensor for the *p*-adic scalar field. A convenient expression for $T_{\mu\nu}$ was derived in [227] (see also [228])

$$T_{\mu\nu} = \frac{m_s^4}{2g_p^2} g_{\mu\nu} \left[\phi e^{-\frac{\Box}{m_p^2}} \phi - \frac{2}{p+1} \phi^{p+1} + \frac{1}{m_p^2} \int_0^1 d\tau \left(\Box e^{-\frac{\tau\Box}{m_p^2}} \phi \right) \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) \right.$$
(7.31)
$$\left. + \frac{1}{m_p^2} \int_0^1 d\tau \left(\partial_\alpha e^{-\frac{\tau\Box}{m_p^2}} \phi \right) \left(\partial^\alpha e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) \right] - \frac{m_s^4}{m_p^2 g_p^2} \int_0^1 d\tau \left(\partial_\mu e^{-\frac{\tau\Box}{m_p^2}} \phi \right) \left(\partial_\nu e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right)$$

One may verify that the $T_{\mu\nu}$ is symmetric by changing the dummy integration variable $\tau \to 1 - \tau$ in the last term. For homogeneous $\phi(t)$ the above expression simplifies, and for T_{00} we find

$$\rho_{\phi} = -T_{00} = \frac{m_s^4}{2g_p^2} \left[\phi e^{-\frac{\Box}{m_p^2}} \phi - \frac{2}{p+1} \phi^{p+1} + \frac{1}{m_p^2} \int_0^1 d\tau \, \left(\Box e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) + \frac{1}{m_p^2} \int_0^1 d\tau \, \partial_t \left(e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \partial_t \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) \right]$$
(7.32)

One can evaluate the above expression term by term, keeping up to $\mathcal{O}(e^{2\lambda t}) \sim u^2$. The final result reads

$$T_{00} = \frac{m_s^4}{2g_p^2} \left[1 - u(1 + e^{\mu_1}) - \frac{2[1 - (p+1)u]}{p+1} + u(e^{\mu_1} - 1) \right] + \mathcal{O}(u^2)$$

$$= \frac{m_s^4(p-1)}{2g_p^2(p+1)} + \mathcal{O}(u^2)$$
(7.33)

The $\mathcal{O}(u)$ terms cancel out and matching the coefficients in the Friedmann equation gives us the simple results

$$H_0^2 = \frac{m_s^4}{6M_p^2} \frac{p-1}{g_p^2(p+1)}$$
(7.34)

and

$$H_1 = 0$$
 (7.35)

for zeroth and first order respectively.

The $\mathcal{O}(u^2)$ contribution to T_{00} is quite complicated (see appendix E-1) but once we use (7.35) it simplifies greatly. Matching coefficient at order $\mathcal{O}(u^2)$ in the Friedmann equation gives

$$H_2 = \frac{\lambda m_s^4}{4g_p^2 m_p^2 M_p^2} e^{\mu_1} = \frac{1}{8g_s^2} \frac{p^3 \ln p}{p-1} \left(\frac{m_s}{M_p}\right)^2 \lambda$$
(7.36)

Because of our sign convention for H_r , the fact that $H_2 > 0$ means that the expansion is slowing as ϕ rolls from the unstable maximum, as one would expect in a conventional inflationary model.

Using (7.28), (7.34) and (7.36) to compute $H_2/H_0 \sim p \ln p$ it is clear that the perturbative expansion in u breaks down once $u \sim (p \ln p)^{-1/2}$ (recall that $H \cong$ $H_0 - H_2 u^2$). Thus one expects that once $u \sim p^{-1/2}$ then inflation ends. We verify this claim using an alternative formalism in the next subsection.

To summarize, we have determined the five parameters λ , ϕ_2 , H_0 , H_1 and H_2 which appear in the solutions for $\phi(t)$ and H(t) up to $\mathcal{O}(u^2)$ through the equations (7.27-7.29), (7.34-7.36). As a check of our result, we can take $H_0 \ll m_s$ and compare it to the the Minkowski background solution that was found in [87]. For $H_0 \ll m_s$ we see from (7.27) that

$$\lambda \cong \sqrt{2}m_s \left(1 - \frac{3H_0}{2\sqrt{2}m_s}\right) \tag{7.37}$$

the first term corresponding to the known Minkowski result. We can also compute the coefficient ϕ_2 in this limit from (7.29),

$$\phi_2 \cong -\frac{1}{2(p^2 + p + 1)} \tag{7.38}$$

This too coincides with the coefficient that was determined in [87] for p = 2.

7.4.3 The Friction-Dominated Approximation

In the previous subsections we have constructed an approximate solution for the padic scalar rolling down its potential by performing an expansion of ϕ , H in a power series in u. Furthermore, we have shown that once $u \sim p^{-1/2}$ then this solution breaks down (because the $\mathcal{O}(u^2)$ term in H(t) become larger than the zeroth order term, H_0). At this point inflation has ended. Because the equations of motion are complicated, we now verify this behaviour using an alternative formalism which does not rely on small u.

The method is the same as the slow-roll approximation in ordinary inflation, which assumes that $\dot{\phi} \ll H\phi$. To justify it within the *p*-adic theory, we will provisionally assume that

 $\lambda^2 \ll 3H_0\lambda$

so that the evolution is friction-dominated in the usual sense. The consistency of this approximation will be established when we match the theory to the observables from the inflationary power spectrum later, in eqs. (7.70), (7.71). It follows that it is a good approximation to take

$$-\Box \cong 3H_0\partial_t$$

Then the p-adic scalar field equation becomes

$$e^{\alpha \partial_t} \phi = \phi^p \tag{7.39}$$

where we have defined

$$\alpha \equiv \frac{3H_0}{m_n^2} \tag{7.40}$$

Our procedure is to treat H_0 as exactly constant, solve for $\phi(t)$ and compute the energy density ρ_{ϕ} . If ρ_{ϕ} is approximately constant then this series of approximations is self-consistent and the solution is reliable. Once ρ_{ϕ} begins to deviate significantly from a constant value then the solution breaks down and we conclude that inflation has ended.

We now proceed to solve (7.39) for constant H_0 . To this end we expand $\phi(t)$ in Fourier modes as

$$\phi(t) = \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{+\infty} dk e^{-ikt} \phi_k \tag{7.41}$$

so that

$$e^{\alpha \partial_t} \phi(t) = \frac{e^{\alpha \partial_t}}{\sqrt{2\pi}} \int_{-\infty}^{+\infty} dk e^{-ikt} \phi_k$$
(7.42)

$$= \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{+\infty} dk e^{-ik(t+\alpha)} \phi_k \tag{7.43}$$

$$= \phi(t+\alpha) \tag{7.44}$$

The *p*-adic scalar equation of motion (7.39) then takes the simple form

$$\phi(t+\alpha) = \phi(t)^p \tag{7.45}$$

It is quite remarkable that this equation admits dynamical solutions. It is straightforward to check that (7.45) has the solution

$$\phi(t) = e^{-u(t)} \tag{7.46}$$

where $u(t) = e^{\lambda t}$ and $\lambda = \alpha^{-1} \ln p = 2m_s^2/(3H_0)$, as before. One can easily check by acting on (7.46) with the full operator $-\Box = \partial_t^2 + 3H_0\partial_t$ that indeed the friction term dominates as long as $u \lesssim 1$.

Performing a Taylor expansion of (7.46) about u = 0 gives

$$\phi \cong 1 - u + \frac{1}{2}u^2 + \cdots$$

which reproduces our solution in the small-*u* expansion in the limit that $m_s \ll H_0^3$ (see equations 7.13 and 7.72). We also have

$$e^{-\Box/m_p^2}\phi = e^{-up} \cong 1 - pu + \frac{1}{2}p^2u^2$$

which again reproduces our previous results (which can be verified by inserting the solutions for μ_1 , μ_2 , σ and ϕ_2 into equation 7.23).

We now proceed to construct the energy density for $\phi(t)$ in this approximation. It is straightforward to show that

$$e^{-\tau \Box/m_p^2} \phi = \exp\left(-up^{\tau}\right)$$
$$e^{-(1-\tau)\Box/m_p^2} \phi = \exp\left(-up^{1-\tau}\right)$$
(7.47)

We write ρ_{ϕ} (7.32) as

$$\rho_{\phi} \equiv \frac{m_s^4}{2g_p^2} \left[T_1 + T_2 + T_3 + T_4 \right] \tag{7.48}$$

where

$$T_1 \equiv \phi e^{-\Box/m_p^2} \phi \tag{7.49}$$

$$T_2 \equiv -\frac{2}{p+1}\phi^{p+1} \tag{7.50}$$

$$T_3 \equiv \frac{1}{m_p^2} \int_0^1 d\tau \left(\Box e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right)$$
(7.51)

$$T_4 \equiv \frac{1}{m_p^2} \int_0^1 d\tau \partial_t \left(e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \partial_t \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right)$$
(7.52)

³We will see later that this is the same as taking the spectral index equal to unity $n_s \rightarrow 1$.

Using the scalar field equation (7.3) and (7.46) the first two terms, (7.49) and (7.50), are trivial

$$T_1 + T_2 = \frac{p-1}{p+1}e^{-up-u} \tag{7.53}$$

Since this term is proportional to ϕ^{p+1} we identify it with the potential energy of the *p*-adic scalar.

The next simplest term to evaluate is T_4 (7.52) which gives

$$T_4 = \frac{\lambda^2 u^2}{2m_s^2} p \ln p \int_0^1 d\tau \exp\left[-u\left(p^{\tau} + p^{1-\tau}\right)\right]$$

It is useful to change the variable of integration to $\omega = up^{\tau}$ and cast this result in the form

$$T_4 = \frac{\lambda^2}{2m_s^2} (pu^2) \int_u^{up} \frac{d\omega}{\omega} \exp\left[-\omega - \frac{pu^2}{\omega}\right]$$
(7.54)

The $d\omega$ integral can be performed exactly (though not in closed form) in terms of special functions. Since T_4 is subdominant to a contribution coming from T_3 we will not investigate the behaviour of (7.54) any further.

We now study T_3 (7.51). This can be written in the form

$$T_{3} = -\frac{\lambda^{2}}{2m_{s}^{2}} \int_{u}^{up} d\omega \omega \exp\left[-\omega - \frac{pu^{2}}{\omega}\right] + \frac{\lambda^{2}}{2m_{s}^{2}} \left(1 + \frac{3H_{0}}{\lambda}\right) \int_{u}^{up} d\omega \exp\left[-\omega - \frac{pu^{2}}{\omega}\right]$$
(7.55)

The dominant contribution to T_3 is the one proportional to H_0 since the evolution is friction-dominated. This term is also larger than T_4 . The leading contribution to $T_3 + T_4$ is then

$$T_3 + T_4 \cong \int_u^{up} d\omega \exp\left[-\omega - \frac{pu^2}{\omega}\right]$$
(7.56)

where we have used the fact that $3H_0\lambda/(2m_s^2) \cong 1$ which follows from (7.27) when $\lambda^2 \ll 3H_0\lambda$. Since T_3 and T_4 contain time derivatives acting on ϕ it is natural to identify (7.56) with the kinetic energy of the *p*-adic scalar.

Let us study the behaviour of the kinetic energy, equation (7.56), as a function of u. The integral in (7.56) can be performed in terms of the incomplete cylindrical functions of the Sonine-Schlaefli form [230] (see appendix E-2 for a review).

$$\int_{u}^{up} d\omega \exp\left[-\omega - \frac{pu^{2}}{\omega}\right] = 2\pi u p^{1/2} S_{-1}(-u, -up; 2iup^{1/2})$$

We now study the behaviour of this integral in various limits. We assume that $p \gg 1$ throughout since the previous method of expanding in u is valid in the case where $p \sim 1$. At very early times, u < 1/p, this integral goes to zero as

$$\int_{u}^{up} d\omega \exp\left[-\omega - \frac{pu^2}{\omega}\right] \cong up \quad \text{for} \quad u < 1/p$$

For intermediate times, $1/p \ll u \lesssim 1/p^{1/2}$, it is a good approximation to treat the upper and lower limits of integration as 0 and $+\infty$ respectively. In this approximation we can write the incomplete cylinder function in terms of a Hankel function of order $\nu = -1$ (see appendix E-2 for details). The small argument asymptotics of $H_{-1}^{(1)}$ gives

$$\int_{u}^{up} d\omega \exp\left[-\omega - \frac{pu^2}{\omega}\right] \cong 1 + 2u^2 p \ln(2up^{1/2}) \quad \text{for} \quad 1/p \ll u < 1/p^{1/2}$$

Finally we consider late times, $u > 1/p^{1/2}$. It is still reasonable to extend the integral as $\int_{u}^{up} d\omega \cong \int_{0}^{\infty} d\omega$ and hence the integral can still be written in terms of $H_{-1}^{(1)}$. This time the large-argument asymptotics of the Hankel function are appropriate and one has

$$\int_{u}^{up} d\omega \exp\left[-\omega - \frac{pu^2}{\omega}\right] \cong \sqrt{\pi u p^{1/2}} e^{-2up^{1/2}} \quad \text{for} \quad u > 1/p^{1/2}$$

We have verified these asymptotic expressions numerically.

We can now write the dominant contribution to ρ_{ϕ} in the friction-dominated approximation,

$$\rho_{\phi} = \frac{m_s^4}{2g_s^2} \frac{p^2}{p-1} \left(\frac{p-1}{p+1} e^{-up-u} + \int_u^{up} d\omega \exp\left[-\omega - \frac{pu^2}{\omega} \right] \right)$$
(7.57)

The first term, proportional to e^{-up} , represents the potential energy and the second term represents the kinetic energy. Using our previous analysis of the kinetic energy the behaviour of ρ_{ϕ} as a function of u (assuming $p \gg 1$) is clear. At early times u < 1/p the potential energy dominates and we have $\rho_{\phi} \cong m_s^4/(2g_p^2)$. At intermediate times $1/p \ll u \lesssim 1/p^{1/2}$ the potential energy goes to zero and the kinetic energy dominates and we have $\rho \cong m_s^4/(2g_p^2)$. At late times $u > 1/p^{1/2}$ and ρ_{ϕ} damps to zero as $e^{-2up^{1/2}}$. We have verified this behaviour numerically. Figure 7.2 shows the behaviour of ρ_{ϕ} as a function of t, verifying that ρ_{ϕ} is approximately constant for $u < p^{-1/2}$. The time evolution of both the potential energy and the kinetic energy contributions to ρ_{ϕ} are shown, demonstrating that the latter dominates in the interval $p^{-1} < u < p^{-1/2}$. In this figure we have taken $p = 10^5$ for illustrative purposes.



Figure 7.2: Plot of the energy density of the *p*-adic scalar as a function of *t*, for $p = 10^5$. This figure shows that ρ_{ϕ} is approximately constant for $u < p^{-1/2}$. The individual contributions from the potential and kinetic energy are also shown.

We conclude that for $u \leq 1/p^{1/2}$ the scalar field energy density ρ_{ϕ} is approximately constant and inflation proceeds. At $u = 1/p^{1/2}$ the energy density of the *p*-adic scalar begins to decrease quickly and the analysis of this subsection is no longer applicable. We conclude that inflation ends, roughly, when $u = 1/p^{1/2}$.

It is quite interesting that during the intermediate phase $p^{-1} < u < p^{-1/2}$ it is in fact the *kinetic* energy which is driving inflation, rather than the potential energy. This is quite different from what occurs in a local field theory.

7.5 Fluctuations and Inflationary Predictions

In this section we consider the spectrum of cosmological fluctuations produced during p-adic inflation. The full cosmological perturbation theory for the p-adic string model (7.1), which should include metric perturbations and also take into account the departure of the background expansion from pure de Sitter, is complicated and beyond the scope of the present chapter. We leave a detailed study of these matters to future investigation [235]. To simplify the analysis we will neglect scalar metric perturbations as well as deviations of the background metric from pure de Sitter space during horizon crossing. In standard cosmological perturbation theory these approximations reproduce the more exact results to reasonable accuracy and we therefore assume that the situation is similar for the action (7.1).

7.5.1 *p*-adic Tachyon Fluctuations

We are approximating the background dynamics as de Sitter which amounts to working in the limit $u \to 0$ so that

$$H^{2} \equiv H_{0}^{2} = \frac{m_{s}^{4}}{6M_{p}^{2}} \frac{p-1}{g_{p}^{2}(p+1)}$$
(7.58)

$$\phi \equiv \phi_0 \equiv 1 \tag{7.59}$$

We expand the *p*-adic tachyon field in perturbation theory as

$$\phi(t, \vec{x}) = \phi^{(0)}(t) + \delta \phi(t, \vec{x})$$
$$= 1 + \delta \phi(t, \vec{x})$$

The perturbed Klein-Gordon equation (7.3) takes the form

$$e^{-\Box/m_p^2}\,\delta\phi = p\,\delta\phi\tag{7.60}$$

(Inhomogeneous solutions in *p*-adic string theory have also been considered in [2].) One can construct solutions by taking $\delta\phi$ to be an eigenfunction of the \Box operator. If we choose $\delta\phi$ to satisfy

$$-\Box\delta\phi = +B\,\delta\phi\tag{7.61}$$

then this is also a solution to (7.60) if

$$B = m_p^2 \ln p = 2m_s^2 \tag{7.62}$$

where in the second equality we have used (7.2).

The solutions of (7.61) are well known. However, in order to make contact with the usual treatment of cosmological perturbations we need to define a field in terms of which the action appears canonical. This presents a serious difficulty because, in general, there is no local field redefinition which will bring the kinetic term $\phi\left(1-e^{-\Box/m_p^2}\right)\phi$ into the canonical form $\phi\Box\phi$. (One might imagine simply truncating the expansion in powers of \Box as we have described in section 7.3, however, the higher order terms are not negligible in general.) Fortunately, for fields which are on-shell (that is, when (7.61) is solved) the field obeys

$$\begin{pmatrix} 1 - e^{-\Box/m_p^2} \end{pmatrix} \delta\phi = \left(1 - e^{B/m_p^2} \right) \delta\phi$$

$$= \left(1 - e^{B/m_p^2} \right) \frac{1}{(-B)} (-B) \delta\phi$$

$$= \left(1 - e^{B/m_p^2} \right) \frac{1}{(-B)} \Box \delta\phi$$

$$= \frac{p - 1}{2m_s^2} \Box \delta\phi$$

Thus, for on-shell fields the kinetic term in the Lagragian can be written as

$$\mathcal{L}_{\text{on-shell}} = \frac{m_s^4}{g_p^2} \frac{1}{2} \phi (1 - e^{-\Box/m_p^2}) \phi + \cdots$$
$$= \frac{m_s^4}{g_p^2} \frac{p-1}{2m_s^2} \frac{1}{2} \phi \Box \phi + \cdots$$
$$= \frac{1}{2} \varphi \Box \varphi + \cdots$$
(7.63)

In (7.63) we have defined the "canonical" field

$$\varphi \equiv A\phi \tag{7.64}$$

where

$$A \equiv \frac{m_s p}{\sqrt{2} q_s} \tag{7.65}$$

The field φ has a canonical kinetic term in the action, at least while (7.61) is satisfied. Notice that φ is distinct from the field χ (see (7.5)) which we introduced in section 7.3. The field χ corresponds to the canonical field which one would naively define when neglecting terms $\mathcal{O}(\Box^2)$ and higher in the action (as is typical in studies of tachyonic inflation) while φ is the appropriate definition of the canonically normalized field when taking into account the infinite series of higher derivative corrections.

Now, let us return to the task of solving (7.61), bearing in mind that $\delta \varphi = A \delta \phi$ is the appropriate canonically normalized field. We write the quantum mechanical solution in term of annihilation/creation operators as

$$\delta\varphi(t,\vec{x}) = \int \frac{d^3k}{(2\pi)^{3/2}} \left[a_k \varphi_k(t) e^{ikx} + \text{h.c.} \right]$$

and the mode functions $\varphi_k(t)$ are given by

$$\varphi_k(t) = \frac{1}{2} \sqrt{\frac{\pi}{a^3 H_0}} e^{\frac{i\pi}{2}(\nu + 1/2)} H_{\nu}^{(1)} \left(\frac{k}{a H_0}\right)$$
(7.66)

where the order of the Hankel functions is

$$\nu = \sqrt{\frac{9}{4} + \frac{B}{H_0^2}} = \sqrt{\frac{9}{4} + \frac{2m_s^2}{H_0^2}}$$
(7.67)

and of course $a = e^{H_0 t}$. In the second equality in (7.67) we have used (7.62) and (7.2). In writing (7.66) we have used the usual Bunch-Davies vacuum normalization so that on small scales, $k \gg aH_0$, one has

$$|\varphi_k| \cong \frac{a^{-1}}{\sqrt{2k}}$$

which reproduces the standard Minkowski space fluctuations. This is the usual procedure in cosmological perturbation theory. However, we note that the quantization of the theory (7.1) is not transparent and it might turn out that the usual prescription is incorrect in the present context. We defer this and other subtleties to future investigation.⁴ On large scales, $k \ll aH_0$, the solutions (7.66) behave as

$$|\varphi_k| \cong \frac{H}{\sqrt{2k^3}} \left(\frac{k}{aH_0}\right)^{3/2-\nu}$$

which gives a large-scale power spectrum for the fluctuations

$$P_{\delta\varphi} = \left(\frac{H_0}{2\pi}\right)^2 \left(\frac{k}{aH_0}\right)^{n_s - 1}$$

with spectral index

$$n_s - 1 = 3 - 2\nu$$

⁴Our prescription for choosing the vacuum has the property that in the local limit $p \rightarrow 1$ the cosmological fluctuations are identical to the well-known solutions in local field theory.

From (7.67) it is clear that to get an almost scale-invariant spectrum we require $m_s \ll H_0$. In this limit we have

$$n_s - 1 \cong -\frac{4}{3} \left(\frac{m_s}{H_0}\right)^2 \tag{7.68}$$

which gives a red tilt to the spectrum, in agreement with the latest WMAP data [16]. For $n_s \cong 0.95$ one has $m_s \cong 0.2H_0$. Comparing (7.66) to the corresponding solution in a local field theory we see that the *p*-adic tachyon field fluctuations evolve as though the mass-squared of the field was $-2m_s^2$ which may be quite different from the mass scale which one would infer by truncating the infinite series of derivatives: $\partial^2 V/\partial \chi^2 (\chi = \chi_0)$ (see eq. (7.10)). The fact that $H_0 > m_s$ is an unusual feature; we will comment on it below.

It is worth pointing out that we have constructed solutions of a partial differential equation with infinitely many derivatives, for which we are free to specify two initial data; to obtain eq. 7.66 we fixed these using the Bunch-Davies prescription. Precisely the same result was obtained when inhomogeneous solutions were studied in [2]. Partial differential equations with infinitely many derivatives constitute a new class of equations in mathematical physics about which little is presently known. In particular, it is not clear how to pose the intial value problem for such equations. We conjecture that the most general solutions of (7.3) are specified by two initial data (for example $\phi(0, \vec{x})$ and $\dot{\phi}(0, \vec{x})$), just like equations containing only one power of \Box . See [229] for mathematical work on constructing solutions of equations with infinitely many derivatives.

7.5.2 Determining Parameters

We now want to fix the parameters of the model by comparing to the observed features of the CMB perturbation spectrum. There are three dimensionless parameters, g_s , p and the ratio m_s/M_p . The important question is whether there is a sensible parameter range which can account for CMB observations, *i.e.*, the spectral tilt and the amplitude of fluctuations. Using (7.34) in (7.68), we can relate the tilt to the model parameters via

$$|n_s - 1| = \frac{8(p+1)}{p^2} \left(\frac{M_p}{m_s}\right)^2 g_s^2 \quad \Longleftrightarrow \quad \left(\frac{m_s}{M_p}\right)^2 = \frac{8(p+1)}{p^2} \frac{g_s^2}{|n_s - 1|} \tag{7.69}$$

Thus one can have a small tilt while ensuring that the string scale is smaller than the Planck scale, provided that $g_s^2/p \ll 1$. Henceforth we will use (7.69) to determine m_s/M_p in terms of p, g_s , and $|n_s - 1| \approx 0.05$. All the dimensionless parameters in our solution, m_s/H_0 , λ/m_s , ϕ_2 and H_2/m_s , are likewise functions of $n_s - 1$, p and g_s . From (7.34) and (7.69) we see that for $p \gg 1$,

$$\frac{m_s}{H_0} \cong \sqrt{6} g_p \frac{M_p}{m_s} \cong g_s \sqrt{\frac{6}{p}} \frac{M_p}{m_s} \cong \frac{1}{2} \sqrt{3|n_s - 1|}$$
(7.70)

It may seem strange to have H exceeding m_s since that means the energy density exceeds the fundamental scale, but this is an inevitable property of the *p*-adic tachyon at its maximum, as shown in eq. (7.33). This is similar to other attempts to get tachyonic or brane-antibrane inflation from string theory, since the false vacuum energy is just the brane tension which goes like m_s^4/g_s .

Next we determine λ/m_s , where λ is the mass scale appearing in the power series in $e^{\lambda t}$ which provides the ansatz for the background solutions. Consider eq. (7.26) for λ in the $H_0 \gg m_s$ limit. The positive root for λ gives

$$\frac{\lambda}{m_s} \cong \sqrt{\frac{|n_s - 1|}{3}} \tag{7.71}$$

From (7.70) and (7.71) we find that $3H_0\lambda \gg \lambda^2$ which means that the evolution is friction-dominated in the usual sense. As for ϕ_2 , from eq. (7.29) it follows that

$$\phi_2 \cong -\frac{1}{2} \frac{p-1}{p^{1+\beta}-1} \tag{7.72}$$

with

$$\beta \cong |n_s - 1|/3 + \mathcal{O}(|n_s - 1|^2)$$

so that $\beta \ll 1$. Notice that for $p \gg 1$, we have $\phi_2 \simeq -0.5p^{-\beta}$. Finally we have H_2 which from eq. (7.36) is given by

$$\frac{H_2}{m_s} = \frac{p}{\sqrt{3}} \frac{p-1}{p+1} \frac{\ln p}{|n_s - 1|^{1/2}}$$
(7.73)

To go further, we must impose the COBE normalization on the amplitude of the density perturbations, to show that it is possible to satisfy all the experimental constraints while keeping $m_s/M_p < 1$. The latter requirement is usually needed for the validity of any 4D effective description of string theory. In a compactification of the d extra dimensions whose volume is of order $V_d \sim m_s^{-d}$, we would have $M_p \sim m_s$ whereas more generally $M_p^2 = m_s^{2+d}V_d$. The 4D effective theory would normally need to be supplemented by higher dimension operators if V_d was small compared to m_s^{-d} .

7.5.3 Curvature Perturbation and COBE normalization

In order to fix the amplitude of the density perturbations we consider the curvature perturbation ζ . We assume that

$$\zeta \sim -\frac{H}{\dot{\varphi}}\delta\varphi$$

as in conventional inflation models. To evaluate the prefactor $H/\dot{\varphi}$ we must work beyond zeroth order in the small u expansion. We take $\phi = 1 - u$ to evaluate the prefactor, even though the perturbation $\delta \chi$ is computed in the limit that $\phi = 1$. This should reproduce the full answer up to $\mathcal{O}(u)$ corrections. The prefactor is

$$\begin{aligned} -\frac{H}{\dot{\varphi}} &\cong \frac{H_0}{A\lambda u} \\ &\cong \frac{2^{3/2}g_s}{p} \frac{1}{|n_s - 1|} \frac{1}{u} m_s^{-1} \end{aligned}$$

We should evaluate u at the time of horizon crossing, t_{\star} , defined to be approximately 60 e-foldings before the end of inflation $t_{\rm end}$, assuming that the energy scale of inflation is high (near the GUT scale). We must therefore estimate $t_{\rm end}$. We have shown in the last subsection that inflation ends when $u \sim 1/p^{1/2}$. From eqs. (7.70-7.71) we see that $H_0/\lambda = 2/|n_s - 1|$; therefore we can write the scale factor $a(t) \cong e^{H_0 t}$ in the form

$$a(t) \cong u(t)^{2/|n_s - 1|} \tag{7.74}$$

so that $a_{\star} = e^{-60} a_{\text{end}}$ corresponds to

$$u_{\star} = e^{-30|n_s - 1|} u_{\text{end}} \equiv e^{-30|n_s - 1|} \frac{1}{p^{1/2}}$$
(7.75)

The power spectrum of the curvature perturbation is given by

$$P_{\zeta} = \left| \frac{H}{\dot{\varphi}} \right|^2 P_{\delta\varphi} \equiv A_{\zeta}^2 \left(\frac{k}{aH_0} \right)^{n_s - 1} \tag{7.76}$$

where the amplitude of fluctuations A_{ζ} can now be read off as

$$A_{\zeta}^{2} = \frac{8}{3\pi^{2}} \frac{g_{s}^{2}}{p} \frac{e^{60|n_{s}-1|}}{|n_{s}-1|^{3}}$$
(7.77)

As an example, taking $n_s \cong 0.95$ one can fix the amplitude of the density perturbations $A_\zeta^2 \cong 10^{-10}$ by choosing

$$\frac{g_s}{\sqrt{p}} \cong 0.48 \times 10^{-7}$$
 (7.78)

To get a more general idea of how the inflationary observables constrain the parameters of the model, we will allow n_s to vary away from the value 0.95, which is a fit to the WMAP data under the assumption that the tensor contribution to the spectrum is negligible. Setting $A_{\zeta}^2 = 10^{-10}$ and using (7.77) gives and expression for g_s in terms of p and $|n_s - 1|$

$$g_s = \sqrt{\frac{3\pi^2}{8}} \sqrt{p} \, e^{-30|n_s - 1|} |n_s - 1|^{3/2} \times 10^{-5} \tag{7.79}$$

Combining (7.79) with (7.69), we also obtain

$$\frac{m_s}{M_p} = \sqrt{3\pi^2} \sqrt{\frac{p+1}{p}} e^{-30|n_s-1|} |n_s-1| \times 10^{-5}$$
(7.80)

We graph the dependence of g_s and m_s/M_p on p for several values of the spectral index in figures 7.3 and 7.4. We see that the string scale is bounded from above as $m_s/M_p \lesssim 0.94 \times 10^{-6}$ and that for typical values of p, n_s it is close to $m_s/M_p \cong$ 0.61×10^{-6} . We also see from (7.79) that g_s is unconstrained and that g_s, p are not independent parameters. If we wish to take $g_s \sim \mathcal{O}(1)$ then we must choose extremely large values $p \sim 10^{14}$. If we restrict ourselves to the perturbative regime $g_s < 1$ then this places an upper bound on p:

$$p < 4.3 \times 10^{14}$$



Figure 7.3: Log of the string coupling g_s as a function of $\log_{10} p$ for several values of n_s .



Figure 7.4: Log of the ratio of the string mass scale to the Planck mass as a function of $\log_{10} p$ for several values of n_s .

7.5.4 Comments on Slow Roll and the Relation to Local Field Theory

It is quite remarkable that our predictions for inflationary observables and also the solutions for χ , H are essentially identical to the results from *local* field theory with potential $V(\psi) = g_s(\psi^2 - v^2)^2/4$ where $g_sv^2 \equiv 2m_s^2$ (see appendix E-3 for a detailed comparison). Indeed, the dynamics of the *p*-adic tachyon are fixed by the mass scale in the kinetic term, m_s , rather than by the naive mass scale $\sqrt{-\partial^2 V/\partial \chi^2}$ (see section 7.3), which may be much larger than m_s .

It is an interesting feature of this theory that the canonical *p*-adic tachyon, ϕ , can roll slowly despite the fact that, working in a derivative truncatation (as in section 7.3), one would conclude that the tachyon has an extremely steep potential. To see this we first define the Hubble slow roll parameters $\epsilon_{\mathcal{H}}$, $\eta_{\mathcal{H}}$ by

$$\epsilon_{\mathcal{H}} \equiv \frac{1}{2M_p^2} \frac{\dot{\varphi}^2}{H^2} \tag{7.81}$$

$$\epsilon_{\mathcal{H}} - \eta_{\mathcal{H}} \equiv \frac{\ddot{\varphi}}{H\dot{\varphi}} \tag{7.82}$$

These are the appropriate parameters to describe the rate of time variation of the inflaton as compared to the Hubble scale. Using the solution $\phi \approx 1 - u$ (recall that $\varphi = A\phi, A = m_s p/(\sqrt{2}g_s)$) we find that

$$\epsilon_{\mathcal{H}} \cong \frac{1}{2} \frac{p+1}{p} e^{-60|n-1|} |n_s - 1|$$
(7.83)

$$\eta_{\mathcal{H}} \cong -\frac{|n_s - 1|}{2} \tag{7.84}$$

We see that the Hubble slow-roll parameters, as defined above, are small. This means that the *p*-adic tachyon field rolls slowly in the conventional sense. One reaches the same conclusion if one defines the potential slow roll parameters using the correct canonical field, which is φ (7.64):

$$\frac{M_p^2}{2} \left(\frac{1}{V} \frac{\partial V}{\partial \varphi} \right)^2 \bigg|_{\varphi=A} = 0$$
(7.85)

$$M_p^2 \left. \frac{1}{V} \frac{\partial^2 V}{\partial \varphi^2} \right|_{\varphi=A} = -\frac{1}{2} |n_s - 1|$$
(7.86)

On the other hand, consider the potential slow roll parameter which one would naively define using the the derivative truncated action (7.4):

$$\frac{M_p^2}{2} \left. \left(\frac{1}{V} \frac{\partial V}{\partial \chi} \right)^2 \right|_{\chi = \chi_0} = 0 \tag{7.87}$$

$$M_p^2 \left. \frac{1}{V} \frac{\partial^2 V}{\partial \chi^2} \right|_{\chi = \chi_0} = -\left(\frac{p-1}{\ln p}\right) \left. \frac{1}{2} |n_s - 1|$$
(7.88)

where in (7.88) we have used equations (7.9) and (7.69). We see that (7.88) can be enormous, though the tachyon field rolls slowly. Taking the largest allowed value of $p, p \sim 10^{14}$, and $n_s \approx 0.95$ we have $M_p^2 V^{-1} |\partial^2 V / \partial \chi^2| \sim 10^{11}!$

Since large values of p are required if one wants to obtain $g_s \sim 1$, it follows that it is somewhat natural for p-adic inflation to operate in the regime where the higher derivative corrections play an important role in the dyanimes. However, in the regime where $p \sim 1$ (corresponding to very small coupling g_s) this novel features is not present. For example, with p = 3, $n_s = 0.95$ one has $g_s \sim 10^{-7}$ and $M_p^2 V^{-1} |\partial^2 V / \partial \chi^2| \sim 0.05$ and so the slow-roll dynamics are not surprising.

7.5.5 Tensor Modes

Since the p-adic stress tensor (7.32) does not contribute any anisotropic stresses up to first order in perturbation theory it follows that the first order tensor perturbations of the metric do not couple to the first order tachyon perturbation. In fact, the action for the tensor perturbations is given by

$$S_{\rm grav} = \frac{M_p^2}{2} \int dt \, d^3x \, a(t)^3 \, \frac{1}{2} \partial_\mu h_{ij} \, \partial^\mu h^{ij}$$

The standard procedure gives a power spectrum for the gravity waves

$$P_T = A_T^2 \left(\frac{k}{aH_0}\right)^{n_T}$$

with amplitude

$$A_T^2 = \frac{8}{M_p^2} \left(\frac{H_0}{2\pi}\right)^2$$
(7.89)

We would like to compare this to the power in scalar modes (7.77). Defining the tensor-to-scalar ratio in the usual way

$$r \equiv \frac{\frac{1}{100}A_T^2}{\frac{4}{25}A_\zeta^2}$$

we find that

$$r = \frac{1}{2M_p^2} \frac{\dot{\varphi}^2}{H_0^2} = \epsilon_{\mathcal{H}}$$
(7.90)

which reproduces the usual result from local field theory. Using (7.83), we can evaluate r as a function of p and $n_s - 1$

$$r = \frac{1}{2} \frac{p+1}{p} e^{-60|n_s - 1|} |n_s - 1|$$
(7.91)

It is easy to see that r is maximal when p = 2, $n_s = 1 - 1/60$ and hence it follows that the scalar-tensor ratio is bounded from above as r < 0.006, which is very small.

7.5.6 Comments on Initial Conditions

As we have previously noted, our classical solution $\phi(t)$ sits at the unstable maximum of the potential for an infinite amount of time, thus this model would seem to admit infinitely many e-foldings of inflation. Of course, this cannot be the case quantum mechanically and one expects quantum fluctuations to displace ϕ from the false vacuum and cause it to roll down the potential (we have assumed that this rolling takes place towards $\phi = 0$, rather than down the unbounded side of the potential). As a consistency check we note that

$$\frac{\langle (\delta\phi)^2 \rangle^{1/2}}{\phi_0} = \langle (\delta\phi)^2 \rangle^{1/2} \sim \frac{H_0}{A}$$
(7.92)

We should compare this to u_{\star} (see equation 7.75), the distance the field has rolled classically at horizon crossing. It is straightforward to show that

$$\frac{\langle (\delta \phi)^2 \rangle^{1/2}}{u_{\star}} \sim 10^{-5} |n_s - 1|$$

where we have used (7.78). Since $\langle (\delta \phi)^2 \rangle^{1/2} \ll u_*$ for all parameter values it follows that the de Sitter space fluctuations (which are present as $u \to 0$) will not displace the

field far enough from the maximum of the potential to have any significant effect on the number of observable e-foldings of inflation, although they prevent inflation from being past-eternal. We note, however, that if one incorporates thermal fluctuations (and initial momentum) then this model may suffer from problems related to fine tuning the intial conditions as in small field inflationary models [231]. However, it is not clear if these objections apply to our model for several reasons. The first reason is that the dynamics of this theory is peculiar and it is not clear how (or if) the phase space arguments of [231] apply. The second reason is that it is not clear what initial conditions for the field ϕ are most natural from a string theory perspective. Finally we note that since the field φ rolls a distance $A > M_p$ in field space, our model is not a "small field" model in the conventional sense.⁵

If *p*-adic superstrings exist, it might also be possible to justify the initial conditions for inflation by having topological inflation [232], if the tachyon potential is symmetric about the unstable maximum. This distinction exists between the tachyon of the open bosonic string [215] (describing the instability of D25 branes), and the tachyon of unstable branes in superstring theory [233]. Any realistic extension of the model should have a potential which is bounded from below, and if it is supersymmetric, the minima should be at zero, hence the additional minima will be degenerate with the one at $\phi = 0$. The existence of domains of the universe in the different minima ensures that there will be regions in between where inflation from the maximum of the potential is taking place, so long as the minima are discrete and not connected to each other by a continuous symmetry.

As we have noted previously the fact that the potential $V(\phi)$ is unbounded from below is thought to be a reflection of the closed string tachyonic instability of bosonic string theory. If this conjecture is correct then the addition of supersymmetry should indeed lead to a symmetric potential for the *p*-adic tachyon which is bounded from below, as we have suggested above.

⁵The skeptical reader might have reservations about the validity of our analysis since $A > M_p$. We note that the action (7.1) is not a low energy effective field theory and hence we believe that we are justified in using this action even for super-Planckian symmetry breaking scale.

7.6 Conclusions

In this chapter we have constructed for the first time approximate solutions of the fully nonlocal *p*-adic string theory coupled to gravity, in which the *p*-adic tachyon drives a sufficiently long period of inflation while rolling away from the maximum of its potential. In our solution, the nonlocal nature of the theory played an essential role in obtaining slow-roll, since with a conventional kinetic term the potential would have been too steep to give inflation. One of the novel features of this construction is that the Hubble parameter is larger than the string scale during inflation, a condition which would usually invalidate an effective field theory description, but which is consistent in the present context because of the ultraviolet-complete nature of the theory.

We found that the experimental constraints on the amplitude of the spectrum of scalar perturbations produced by inflation require a small value of the string coupling g_s , and can be consistent with a large range of values of the parameter $p, 1 \leq p \leq 10^{14}$. The regime $p \gg 1$ is interesting because it exhibits qualitatively different behavior relative to conventional inflationary models: slow roll despite the potential being steep, and inflation being driven by the kinetic as well as potential energy of the field. This regime is also interesting because it corresponds to $g_s = \mathcal{O}(1)$ and hence appears more natural from a string theory perspective. Since the *p*-adic string is not construed to be a realistic model by itself, it may not be very meaningful to question how natural such values might be. However, it may not be unreasonable to think of real strings as being composed of constituent *p*-adic strings because the Veneziano amplitude of the *p*-adic theory is related to that of the full bosonic string by $\mathcal{A}^{-1} = \prod_p \mathcal{A}_p$ where the product is over all prime numbers *p*. Thus it may not be unreasonable to expect similar behavior to the large-*p* results from a more realistic model.

The model predicts a red spectrum, in agreement with the latest WMAP data, whose tilt is related to the ratio of the string scale to the Hubble rate during inflation via $H/m_s = 2/\sqrt{3|n_s - 1|}$. For $n_s = 0.95$ this gives $H/m_s \approx 5$. This is in contrast to
most stringy models of inflation which require $H < m_s$ in order for the effective field theory to be valid. We find the bound r < 0.006 on the tensor modes. It has been estimated that future experiments could eventually have a sensitivity of $r \sim 6 \times 10^{-5}$ [234] and hence the tensor components may in fact be observable.

We noted that the p-adic model succeeds with inflation where the real string theory tachyon fails. But our analysis makes it clear that this could be due to the failure to keep terms with arbitrary numbers of derivatives in the action. The effective tachyon action of Sen [43] is a truncation which keeps arbitrary powers of first derivatives but ignores higher order derivatives, which were essential for obtaining our solution. Thus the new features for inflation which we find in p-adic string theory could also be present in realistic string theories, if we knew how to include the whole tower of higher dimensional kinetic terms.

Acknowledgments

This work was supported in part by NSERC and FQRNT. We would like to thank R. Brandenberger, G. Calcagni, K. Dasgupta, A. Mazumdar and W. Siegel for interesting discussions, and D. Ghoshal for valuable comments on the manuscript.

Note Added

Upon completing this analysis a related work appeared [236] in which an alternative normalization for the fluctuations of the *p*-adic scalar was proposed. Motivated by this work, we reconsidered our choice of normalization and concluded that (7.64,7.65) is the most appropriate definition of a "canonical" field. Notice, however, that our field φ differs from the definition of a canonical field which was proposed in [236]. In [236] a field redefinition is advocated which puts the stress tensor $T_{\mu\nu}$ into canonical form, though this definition does not have canonical kinetic term in the action. Though we believe that our definition is more natural, we stress that in the case of current interest, *p*-adic inflation, the distinction does not generate any significant quantitative difference because our definition differs from that of [236] by a factor proportional to $\sqrt{\ln p}$ (which is less than an order of magnitude for the values of p which we consider). We are grateful to J. Lidsey for sending us a draft of his manuscript prior to publication and also for interesting and enlightening discussions.

Chapter 8

Conclusions and Discussion

All I know is what the words know, and dead things, and that makes a handsome little sum, with a beginning and a middle and an end, as in the well-built phrase and the long sonata of the dead. -Samuel Beckett

Inflation offers an elegant resolution to the conceptual problems associated with the simple big bang model, however, this resolution brings with it a new list of conceptual problems to be addressed. Lest the reader fear that we have simply traded one set of conceptual problems for another (neither representing an actual conflict with observation) we remind the reader that inflation is a predictive framework and that, to date, these predictions have recieved spectacular confirmation by observation. (Indeed, the predictions of inflation are so well confirmed by observation that it is difficult, at this point, to imagine any measurement which could falsify inflation.) We have argued that a proper resolution to the outstanding conceptual problems of inflation will require embedding into (or at least input from) a complete theory of particle physics which incorporates quantum gravity. Since the only viable candidate is string theory, we have argued that the construction of string theoretic inflationary models is well motivated and provides a rare (perhaps the only) potential window into stringy physics. It is quite a remarkable feature of the interaction between cosmology and particle physics that it may be possible to glean information about extremely high energy physics (as high as 10^{16} GeV) through observations which will be made

within our lifetimes.

In this thesis we have explored the cosmological consequences of tachyonic instabilities, both within the context of string theory and also in more conventional field theory models. Motivated by the KKLMMT model of brane inflation, we have considered the cosmological consequences of D-brane decay in chapters 2, 3, 4 and 5. We have studied in detail post-inflationary dynamics like reheating and cosmic string production in such models. We have shown that the dynamics of these processes may be quite different from the dynamics of analogous field theory models. We have also examined the cosmological consequences of these peculiar dynamics.

In chapter 6 we have studied tachyonic preheating at the end of hybrid inflation, showing that this phase of violent, nonperturbative particle production can have highly nontrivial implications for the observed spectrum of cosmological perturbations. We have shown that, depending on the values of certain model parameters, preheating may leave its imprint on the CMB either through n = 4 distortions of the spectrum or else through large nongaussian signatures. If large nongaussianity is observed in future missions it will certainly demonstrate the existence of some novel dynamics during (or shortly after) the inflationary epoch. There are very few inflationary models which can give rise to observably large nongaussianity and many of the models which do are somewhat contrived or are difficult to motivate from a particle physics perspective. Thus it is quite significant that large nongaussianity can be obtained in hybrid inflation, a simple and well-motivated model of inflation which appears to arise naturally in string theory and supergravity. We have used our analysis of nongaussianity from tachyonic preheating to show that current CMB observations constrain several popular string theory and supergravity models of inflation.

Finally, in chapter 7 we have explored a novel string theory model of inflation based on the tachyonic mode of the fully nonlocal p-adic string theory. The nonlocal structure of this theory leads to extremely peculiar dynamics, including the possibility of slow roll with a steep inflaton potential. This result opens up the possibility of addressing the issue of fine tuning of the inflaton potential in a new and novel way.

We feel that the work presented in this thesis represents some small progress in the

direction of producing a realistic and natural string theory model of inflation. Though we (as a community) are still far from this goal, it is encouraging that significant progress has been made in recent years. Furthermore, the interface between string theory and cosmology is one which has been mutually beneficial. Indeed, even if string theory turns out not to be the theory of quantum gravity which nature has chosen, we feel that the exercise has still been a fruitful one since it has revealed a wealth of previously unexpected possibilities for inflationary model building. We look forward to future surprises (from both theory and observation) and the exciting possibility that future cosmological measurements may provide a window into new physics at extraordinarily high energy scales.

Chapter A

Appendix to Chapter 3

A-1 The Legendre Transformation

In 1 + 1 dimensions equation (3.25) takes the form

$$\dot{T}^{2}\ddot{T} + T^{\prime 2}T^{\prime \prime} - 2\dot{T}T^{\prime}\dot{T}^{\prime} = 0$$
(A-1)

To construct general solutions of (A-1) we preform the Legendre transformation (see [101] for example) defined by the relations

$$\xi = T', \quad \eta = \dot{T}, \quad T(t, x) + \omega(\eta, \xi) = x\xi + t\eta \tag{A-2}$$

where $\omega(\eta, \xi)$ is a new dependent variable and $\{\eta, \xi\}$ are the new independent variables. It follows from (A-2) that

$$x = \frac{\partial \omega}{\partial \xi}, \quad t = \frac{\partial \omega}{\partial \eta}.$$
 (A-3)

It is straightforward to verify the relations

$$T'' = J \frac{\partial^2 \omega}{\partial \eta^2}, \quad \ddot{T} = J \frac{\partial^2 \omega}{\partial \xi^2}, \quad \dot{T}' = -J \frac{\partial^2 \omega}{\partial \xi \partial \eta}$$
 (A-4)

where the Jacobian of the transformation is

$$J = T''\ddot{T} - \left(\dot{T}'\right)^2$$

$$= \left[\frac{\partial^2 \omega}{\partial \xi^2} \frac{\partial^2 \omega}{\partial \eta^2} - \left(\frac{\partial^2 \omega}{\partial \xi \partial \eta}\right)^2\right]^{-1}$$
(A-5)

which we assume is nonzero. Note that this analysis excludes any solutions where J = 0. The nonlinear partial differential equation (A-1) is transformed to a linear partial differential equation in terms of the new variables

$$\xi^2 \frac{\partial^2 \omega}{\partial \eta^2} + \eta^2 \frac{\partial^2 \omega}{\partial \xi^2} + 2\eta \xi \frac{\partial^2 \omega}{\partial \xi \partial \eta} = 0.$$
 (A-6)

Given a solution $\omega(\eta,\xi)$ of (A-6) the relations (A-3) along with

$$T = \xi \frac{\partial \omega}{\partial \xi} + \eta \frac{\partial \omega}{\partial \eta} - \omega(\eta, \xi)$$
 (A-7)

define the solution of (A-1) parametrically.

We consider now solutions of the linear partial differential equation (A-6). To simplify the ensuing analysis we will restrict ourselves to solutions where $\partial_{\mu}T\partial^{\mu}T$ does not change sign. This is expected to be a reasonable restriction for the vacuum structure of the theory (3.24) since previous analysis of the actions (3.1,3.8,3.15) suggests that $\partial_{\mu}T\partial^{\mu}T$ is either increasing or decreasing as $T \to \pm \infty$. We only consider the case $\partial_{\mu}T\partial^{\mu}T \leq 0$, which we expect to most closely resemble the vacuum structure of the actions (3.1,3.8,3.15). In this regime we can define new coordinates $\{\rho, \sigma\}$ by

$$\eta = \rho \cosh \sigma, \quad \xi = \rho \sinh \sigma \tag{A-8}$$

such that

$$\rho^2 = -\xi^2 + \eta^2 = -\partial_\mu T \partial^\mu T \ge 0, \quad \tanh \sigma = \frac{\xi}{\eta}.$$

In terms of these new variables (A-6) takes the remarkably simple form

$$-\rho \frac{\partial \omega}{\partial \rho} + \frac{\partial^2 \omega}{\partial \sigma^2} = 0. \tag{A-9}$$

with particular solution

$$\omega_m(\eta,\xi) = \rho^{m^2} \left[\alpha_m e^{m\sigma} + \beta_m e^{-m\sigma} \right]$$

for arbitrary m, α_m , β_m . The most convenient way to fix the Cauchy data is to specify ω and $\partial \omega / \partial \rho$ at $\sigma = 0$. In this case we write the general solution of (A-9) as

$$\omega(\rho,\sigma) = \sum_{n=0}^{\infty} \rho^n \left(a_n \cosh\left(\sqrt{n}\,\sigma\right) + b_n \sinh\left(\sqrt{n}\,\sigma\right) \right) \tag{A-10}$$

where $\{a_n\}$ determine the coefficients in the taylor expansion of ω at $\sigma = 0$ and $\{b_n\}$ determine the coefficients in the taylor expansion of $\partial \omega / \partial \rho$ at $\sigma = 0$.

It is straightforward to compute $x(\eta, \xi)$, $t(\eta, \xi)$ from (A-3) and

$$T(\rho,\sigma) = \sum_{n=0}^{\infty} \rho^n (n-1) \left(a_n \cosh\left(\sqrt{n}\,\sigma\right) + b_n \sinh\left(\sqrt{n}\,\sigma\right) \right)$$

from (A-7). Because this solution is not in closed form and defined parametrically, it is difficult to get an intuitive feeling for the behaviour of T as a function of t and x. However, we believe that these general solutions do admit derivative singularities and have, for several choices of $\{a_n, b_n\}$, found points (ρ_c, σ_c) corresponding to finite x and t > 0 at which T'' is singular through T, T' and \dot{T} are regular. In this case the second derivative blows up because the Jacobian of the Legendre transformation (A-5) is singular.

Chapter B

Appendix to Chapter 4

B-1 The 4D View

In this appendix we compute the low-energy couplings amongst the bulk zero modes and brane modes in the effective 4D supergravity obtained after modulus stabilization $\dot{a} \ la$ KKLT [21]. Besides checking the scaling of the kinetic terms obtained by dimensionally reducing the Einstein-Hilbert action, this also allows the study of the couplings in the scalar potential which arise from modulus stabilization and so are more difficult to analyze from a semiclassical, higher-dimensional point of view.

To this end imagine integrating out all of the extra-dimensional physics to obtain the low-energy effective 4D supergravity for a Type IIB GKP vacuum having only the mandatory volume modulus (and its supersymmetric friends) plus various low-energy brane modes (such as those describing the motion of various D3 branes). The terms in this supergravity involving up to two derivatives are completely described once the Kähler function, K, superpotential, W, and gauge kinetic function, f_{ab} , are specified.

Denoting the bulk-modulus supermultiplet by T and the brane multiplets by ϕ^{I} , we use the Kähler potential [126, 21, 24, 127]

$$K = -3\log\left[r\right],\tag{B-1}$$

where $r = T + T^* + k(\phi, \phi^*)$. For instance, if ϕ^I denotes the position of single brane, then k is the Kähler potential for the underlying 6D manifold. This implies the scalar kinetic terms are governed by the following Kähler metric in field space

$$K_{TT^*} = \frac{3}{r^2}, \qquad K_{IT^*} = \frac{3k_I}{r^2} \qquad \text{and} \qquad K_{IJ^*} = \frac{3}{r^2} \left[k_I \, k_{J^*} - r \, k_{IJ^*} \right], \qquad (B-2)$$

with inverse

$$K^{T^*T} = \frac{r}{3} \left[r - k^{L^*N} k_{L^*} k_N \right], \quad K^{J^*T} = \frac{r k^{J^*L} k_L}{3} \quad \text{and} \quad K^{J^*I} = -\frac{r k^{J^*I}}{3}.$$
(B-3)

In the absence of modulus stabilization the superpotential of the effective theory is a constant [128], $W = w_0$, and the supergravity takes the usual no-scale form [129], with vanishing scalar potential. If, however, there are low-energy gauge multiplets associated with any of the D7 branes of the model then their gauge kinetic function is $f_{ab} = T \,\delta_{ab}$. For nonabelian multiplets of this type gaugino condensation [130, 131] can generate a nontrivial superpotential, of the form

$$W = w_0 + A \exp[-aT]$$
, (B-4)

where A and a are calculable constants.

With these choices the Kähler derivatives of the superpotential become

$$D_T W = W_T - \frac{3W}{r}$$
, and $D_I W = -\frac{3k_I W}{r}$, (B-5)

and so the supersymmetric scalar potential [132] becomes

$$V = \frac{1}{3r^2} \left[\left(r - k^{I^*J} k_{I*} k_J \right) |W_T|^2 - 3(W^* W_T + W W_T^*) \right] . \tag{B-6}$$

Notice that use of these expression implicitly requires that we work in the 4D Einstein frame, and so are using 4D Planck units for which $M_p = \mathcal{O}(1)$.

If we specialize to the case of several branes, for which $\{\phi^I\} = \{\phi_n^i\}$, with *i* labelling the fields on a given brane and n = 1, ..., N labelling which brane is involved, then we typically have

$$k(\phi^{I}, \phi^{I*}) = \sum_{n} k^{(n)}(\phi_{n}^{i}, \phi_{n}^{i*}).$$
 (B-7)

In this case the Kähler metric built from k is block diagonal, with $k_{i_n j_m} = k_{ij}^{(n)} \delta_{mn}$, and so $k^{I^*J} k_{I^*} k_J = \sum_n k_{(n)}^{i^*j} k_{i^*}^{(n)} k_j^{(n)}$ and so on.

We may now see how strongly the bulk KK zero modes, $g_{\mu\nu}$ and T, couple to one another and to the brane modes. Setting k = 0 in the above shows that the couplings of T and $g_{\mu\nu}$ to one another are order unity, and since our use of the standard 4D supergravity formalism requires us to be in the Einstein frame, this implies these are all of 4D Planck strength (in agreement with our higher-dimensional estimates).

Couplings to the branes are obtained by keeping k nonzero, and in the event that the branes are located in highly warped regions, we must take $k^{(n)} = \mathcal{O}(a_n^2)$ with $a_n \ll 1$ denoting the warp factor at the position of brane n.¹ In this case the combination $k_{(n)}^{i^*j}k_{i^*}^{(n)}k_j^{(n)}$ is also $\mathcal{O}(a_n^2)$.

¹For instance, this power of a_n reproduces the a_n -dependence of the factor $\sqrt{g}g^{\mu\nu}$ obtained by dimensionally reducing the higher-dimensional kinetic terms.

B-1 The 4D View

Suppose we now expand the functions $k^{(n)}$ in powers of ϕ and keep only the leading powers:

$$k^{(n)} \approx a_n^2 \sum_i \phi_n^{i*} \phi_n^i \,. \tag{B-8}$$

Then, since the ϕ_n kinetic terms are $\mathcal{O}(a_n^2)$, we see that the canonically-normalized fields are $\chi_n^i = a_n \phi_n^i$. Once this is done the leading couplings to T and $g_{\mu\nu}$ are those which involve those parts of $k^{(n)}$ that are quadratic in χ_n^i , and since these are also order unity, these couplings are also of Planck strength (again in agreement with our earlier estimates).

Alternatively, consider now those couplings which only involve the brane modes. Working to leading order in a_n^2 , we see that a term in $k^{(n)}$ of the form $(\chi_n^i)^k$ has a strength which is of order a_n^{2-k} . For instance the case k = 3 generates cubic couplings from the kinetic lagrangian of order $a_n^{-1}\chi\partial\chi\partial\chi$, whose coefficient is of order $(a_nM_p)^{-1} = M_{sm}^{-1}$. These are larger than Planck suppressed ones, as expected.

Chapter C

Appendix to Chapter 5

C-1 Initial Conditions for Defect Formation

Here we briefly discuss the initial conditions for defect formation at the end of brane inflation. During the slow roll inflationary phase the tachyon behaves as an ordinary massive Klein-Gordon scalar field (provided $T \ll l_s^{-1}$). We consider here for simplicity a standard field theory of a scalar χ in 3+1-dimensions whose mass squared parameter abruptly becomes negative. This type of theory has been considered in detail in [32, 33, 34].

During the de Sitter phase (before the mass parameter becomes tachyonic) vacuum fluctuations yield a blackbody spectrum of produced particles $\langle 0|N_k|0\rangle = (e^{\omega_k/T}-1)^{-1}$ with temperature $T = H/(2\pi)$ (see, for example, [154] for a review). However, in any region of de Sitter space which is small compared to the Hubble scale, the space is locally Minkowski, and even in the vacuum state there are quantum fluctuations quantified by the two point functions of the fields

$$\langle \chi^{\star}(\vec{k})\chi(\vec{k}')\rangle = \frac{1}{2|\vec{k}|} (2\pi)^{3} \delta^{3}(\vec{k}-\vec{k}') \langle \Pi^{\star}(\vec{k})\Pi(\vec{k}')\rangle = \frac{|\vec{k}|}{2} (2\pi)^{3} \delta^{3}(\vec{k}-\vec{k}')$$

where $\Pi = \dot{\chi}^{1}$ The initial stages of the string tachyon condensation are identical to the tachyonic preheating scenario described in [32, 33]. The tachyonic instability amplifies exponentially those modes with $|\vec{k}| < m$ where the χ field has mass squared parameter $-m^{2}$ and the variance of the fluctuations grows as [34]

$$\langle \chi^2(t) \rangle = \langle \chi^2(0) \rangle + \frac{1}{8\pi} \int_0^{m^2} dk^2 \frac{m^2}{m^2 - k^2} \sinh^2\left(t\sqrt{m^2 - k^2}\right)$$

where $\langle \chi^2(0) \rangle$ is a divergent vacuum contribution. The above result was derived in (3+1) dimensions, but the generalization to higher dimensions must have the form

$$\langle \chi^2(t) \rangle = \langle \chi^2(0) \rangle + c \sum_i \int_0^{m^2} dk^2 \frac{m^2}{m^2 - k^2 - m_i^2} \sinh^2 \left(t \sqrt{m^2 - k^2 - m_i^2} \right)$$

where m_i are the masses of the Kaluza-Klein excitations. These fluctuations grow to be of order the classical VEV, $\langle \chi^2(t) \rangle - \langle \chi^2(0) \rangle \sim m^2/\lambda$ before the linear treatment

¹Similar initial conditions are taken for defect formation at the end of inflation in [155].

breaks down. Notice that this growth occurs on a microscopic time scale. Since in the case of the string theory tachyon the potential has no minimum, we can conservatively take the initial exponential growth to be over a shorter time scale, $\langle \chi^2 \rangle \sim m^2 \sim M_s^2$. These fluctuations are much larger than the de Sitter fluctuations $\langle \chi^2 \rangle \sim H^2$ and are thus the dominant seeds for defect formation. Furthermore, these fluctuations have a minimum wavelength comparable to the string scale which is sufficient to initiate the formation of defects localized in the compact dimensions. This justifies our choice of random initial conditions for the tachyon field in the numerical studies of defect formation.

Chapter D

Appendices to Chapter 6

D-1 The Matching Time N_k

The matching time N_k which determines the boundary between large- and small-scale behaviour of the mode functions (6.28) is determined by the transcendental equation

$$|N_k|e^{2N_k} = \frac{\hat{k}^2}{c}$$
 (D-1)

The solutions may be written exactly in terms of the branches of the Lambert-W functions. In the region $\hat{k} < \sqrt{c/(2e)}$ the solution is triple-valued and may be written as

$$N_{k} = \begin{cases} \frac{1}{2}W_{-1}\left(-\frac{2\hat{k}^{2}}{c}\right) & \text{for the branch with } N_{k} < -1; \\ \frac{1}{2}W_{0}\left(-\frac{2\hat{k}^{2}}{c}\right) & \text{for the branch with } -1 < N_{k} < 0; \\ \frac{1}{2}W_{0}\left(+\frac{2\hat{k}^{2}}{c}\right) & \text{for the branch with } N_{k} > 0. \end{cases}$$
(D-2)

In the region $\hat{k} > \sqrt{c/(2e)}$ the solution is single valued and can be written as

$$N_k = \frac{1}{2} W_0 \left(+ \frac{2\hat{k}^2}{c} \right) \tag{D-3}$$

One may derive some asymptotic expressions for N_k in various regions of interest. When $|N_k| \gg 1$ we have

$$N_k \cong \ln\left(\frac{\hat{k}}{\sqrt{c}}\right) \tag{D-4}$$

which describes N_k at $\hat{k} \gg \sqrt{c/(2e)}$ and also the lower branch of N_k at $\hat{k} \ll \sqrt{c/(2e)}$. For $\hat{k} \lesssim \sqrt{c/(2e)}$ there are two more branches of the solution with approximate behaviour

$$N_k \cong \pm \frac{\hat{k}^2}{c} \tag{D-5}$$

In our analysis we have used the approximation that N_k is a single-valued function, described by

$$N_k^{\text{s.v.}} = \frac{1}{2} \Theta \left(\sqrt{c/(2e)} - \hat{k} \right) W_{-1} \left(-\frac{2\hat{k}^2}{c} \right)$$
$$+ \frac{1}{2} \Theta \left(\hat{k} - \sqrt{c/(2e)} \right) W_0 \left(+\frac{2\hat{k}^2}{c} \right)$$

where $\Theta(x)$ is the Heaviside step function. We have verified both numerically and analytically that this approximation does not alter our result

D-2 Perturbed Einstein Equations and the Master Equation

Using Maple we have carefully verified the results of [183] for the perturbed Einstein equations and the master equation. Here we briefly review those results relevent for the computation of $\zeta^{(2)}$. We present only the $\delta^{(2)}G_0^0 = \kappa^2 \delta^{(2)}T_0^0$, $\partial_i \delta^{(2)}G_0^i = \kappa^2 \partial_i \delta^{(2)}T_0^i$ and $\delta_j^i \delta^{(2)}G_i^j = \kappa^2 \delta_j^i \delta^{(2)}T_i^j$ equations since the second order vector and tensor fluctuations decouple from this system. In the case that $\sigma_0 = 0$ the second order tachyon fluctation $\delta^{(2)}\sigma$ decouples from the inflaton and gravitational fluctuations. Analogously to the first order fluctuations, the Klein-Gordon equation for the inflaton fluctation at second order $\delta^{(2)}\varphi$ is not necessary to close the system of equations. In the subsequent text we sometimes insert the slow roll parameters ϵ and η explicitly though we make no assumption that they are small. We also introduce the shorthand notation $m_{\varphi}^2 = \partial^2 V/\partial \varphi^2$ and $m_{\sigma}^2 = \partial^2 V/\partial \sigma^2$ and assume that $\partial^2 V/\partial \sigma \partial \varphi = 0$.

The second order (0,0) equation is

$$3\mathcal{H}\psi^{\prime(2)} + (3-\epsilon)\mathcal{H}^{2}\phi^{(2)} - \partial^{k}\partial_{k}\psi^{(2)}$$

= $-\frac{\kappa^{2}}{2}\left[\varphi_{0}^{\prime}\delta^{(2)}\varphi^{\prime} + a^{2}\frac{\partial V}{\partial\varphi}\delta^{(2)}\varphi\right] + \Upsilon_{1}$ (D-6)

where Υ_1 is constructed entirely from first order quantities. Dividing Υ_1 into inflaton and tachyon contributions we have

$$\Upsilon_1 = \Upsilon_1^{arphi} + \Upsilon_1^{\sigma}$$

where

$$\Upsilon_{1}^{\varphi} = 4(3-\epsilon)\mathcal{H}^{2}(\phi^{(1)})^{2} + 2\kappa^{2}\varphi_{0}'\phi^{(1)}\delta^{(1)}\varphi' - \frac{\kappa^{2}}{2}(\delta^{(1)}\varphi')^{2} - \frac{\kappa^{2}}{2}a^{2}m_{\varphi}^{2}(\delta^{(1)}\varphi)^{2} - \frac{\kappa^{2}}{2}(\vec{\nabla}\delta^{(1)}\varphi)^{2} + 8\phi^{(1)}\partial^{k}\partial_{k}\phi^{(1)} + 3(\phi'^{(1)})^{2} + 3(\vec{\nabla}\phi^{(1)})^{2}$$
(D-7)

and

$$\Upsilon_{1}^{\sigma} = -\frac{\kappa^{2}}{2} \left[\left(\delta^{(1)} \sigma' \right)^{2} + \left(\vec{\nabla} \delta^{(1)} \sigma \right)^{2} + a^{2} m_{\sigma}^{2} \left(\delta^{(1)} \sigma \right)^{2} \right].$$
(D-8)

The divergence of the second order (0, i) equation is

$$\partial^k \partial_k \left[\psi^{\prime(2)} + \mathcal{H} \phi^{(2)} \right] = \frac{\kappa^2}{2} \varphi_0^\prime \partial^k \partial_k \delta^{(2)} \varphi + \Upsilon_2 \tag{D-9}$$

where $\Upsilon_2 = \Upsilon_2^{\varphi} + \Upsilon_2^{\sigma}$ is constructed entirely from first order quantities. The inflaton part is

$$\Upsilon_{2}^{\varphi} = 2\kappa^{2}\varphi_{0}^{\prime}\partial_{i}\left(\phi^{(1)}\partial^{i}\delta^{(1)}\varphi\right) + \kappa^{2}\partial_{i}\left(\delta^{(1)}\varphi^{\prime}\partial^{i}\delta^{(1)}\varphi\right) - 8\partial_{i}\left(\phi^{(1)}\partial^{i}\phi^{\prime(1)}\right) - 2\partial_{i}\left(\phi^{\prime(1)}\partial^{i}\phi^{(1)}\right)$$
(D-10)

and the tachyon part is

$$\Upsilon_2^{\sigma} = \kappa^2 \partial_i \left(\delta^{(1)} \sigma' \partial^i \delta^{(1)} \sigma \right). \tag{D-11}$$

The trace of the second order (i, j) equation is

$$3\psi^{\prime\prime(2)} + \partial^{k}\partial_{k} \left[\phi^{(2)} - \psi^{(2)}\right] + 6\mathcal{H}\psi^{\prime(2)}$$

+
$$3\mathcal{H}\phi^{\prime(2)} + 3(3-\epsilon)\mathcal{H}^{2}\phi^{(2)}$$

=
$$\frac{3\kappa^{2}}{2} \left[\varphi_{0}^{\prime}\delta^{(2)}\varphi^{\prime} - a^{2}\frac{\partial V}{\partial\varphi}\delta^{(2)}\varphi\right] + \Upsilon_{3}$$
(D-12)

where $\Upsilon_3 = \Upsilon_3^{\varphi} + \Upsilon_3^{\sigma}$ is constructed entirely from first order quantities. The inflaton part is

$$\begin{split} \Upsilon_{3}^{\varphi} &= 12(3-\epsilon)\mathcal{H}^{2}\left(\phi^{(1)}\right)^{2} - 6\kappa^{2}\varphi_{0}^{\prime}\phi^{(1)}\delta^{(1)}\varphi^{\prime} \\ &+ \frac{3\kappa^{2}}{2}\left(\delta^{(1)}\varphi^{\prime}\right)^{2} - \frac{3\kappa^{2}}{2}a^{2}m_{\varphi}^{2}\left(\delta^{(1)}\varphi\right)^{2} - \frac{\kappa^{2}}{2}\left(\vec{\nabla}\delta^{(1)}\varphi\right)^{2} \\ &+ 3\left(\phi^{\prime(1)}\right)^{2} + 8\phi^{(1)}\partial^{k}\partial_{k}\phi^{(1)} + 24\mathcal{H}\phi^{(1)}\phi^{\prime(1)} \\ &+ 7\left(\vec{\nabla}\phi^{(1)}\right)^{2} \end{split}$$
(D-13)

and the tachyon part is

$$\Upsilon_{3}^{\sigma} = \kappa^{2} \left[\frac{3}{2} \left(\delta^{(1)} \sigma' \right)^{2} - \frac{1}{2} \left(\vec{\nabla} \delta^{(1)} \sigma \right)^{2} - \frac{3}{2} a^{2} m_{\sigma}^{2} \left(\delta^{(1)} \sigma \right)^{2} \right]$$

$$(D-14)$$

We now proceed to derive the master equation. Adding (D-6) to the inverse laplacian of the time derivative of (D-9) and then using (D-9) to eliminate $\delta^{(2)}\varphi$

yields

$$\psi^{\prime\prime(2)} - (1 + 2\epsilon - 2\eta)\mathcal{H}\psi^{\prime(2)} + \mathcal{H}\phi^{\prime(2)}$$

- $2(2\epsilon - \eta)\mathcal{H}^2\phi^{(2)} - \partial^k\partial_k\psi^{(2)} = \Upsilon_1 + \Delta^{-1}\Upsilon_2'$
- $2(2 + \epsilon - \eta)\mathcal{H}\Delta^{-1}\Upsilon_2.$ (D-15)

Notice that we have decoupled $\psi^{(2)}$ and $\phi^{(2)}$ from the inflaton perturbation $\delta^{(2)}\varphi$. It remains now to express $\psi^{(2)}$ in terms of $\phi^{(2)}$. To this end we subtract the inverse laplacian of (D-9) from (D-12) and again use (D-9) to eliminate $\delta^{(2)}\varphi$ which gives

$$\partial^k \partial_k \left[\phi^{(2)} - \psi^{(2)} \right] = \Upsilon_3 - 3 \triangle^{-1} \Upsilon_2' - 6 \mathcal{H} \triangle^{-1} \Upsilon_2$$

or, equivalently

$$\psi^{(2)} = \phi^{(2)} - \triangle^{-1}\gamma.$$
 (D-16)

Following the notation of [183] we have defined

$$\gamma = \Upsilon_3 - 3\triangle^{-1}\Upsilon_2' - 6\mathcal{H}\triangle^{-1}\Upsilon_2 \tag{D-17}$$

which can be split into inflaton and tachyon components $\gamma = \gamma_{\varphi} + \gamma_{\sigma}$ in an obvious fashion. Our Υ_2 is related to the quantities α, β defined in [183] by

$$\Upsilon_2 = \kappa^2 \beta - \alpha.$$

Now, using (D-16) to eliminate $\psi^{(2)}$ from (D-15) gives the master equation

$$\phi^{\prime\prime(2)} + 2(\eta - \epsilon)\mathcal{H}\phi^{\prime(2)} + \left[2(\eta - 2\epsilon)\mathcal{H}^2 - \partial^k\partial_k\right]\phi^{(2)}$$

= $\Upsilon_1 + \Delta^{-1}\Upsilon_2' - 2(2 + \epsilon - \eta)\mathcal{H}\Delta^{-1}\Upsilon_2 - \gamma$
- $(1 + 2\epsilon - 2\eta)\mathcal{H}\Delta^{-1}\gamma' + \Delta^{-1}\gamma''$

Inserting explicitly the expression for γ (D-17) this can be written as

$$\phi^{\prime\prime(2)} + 2(\eta - \epsilon)\mathcal{H}\phi^{\prime(2)} + \left[2(\eta - 2\epsilon)\mathcal{H}^2 - \partial^k\partial_k\right]\phi^{(2)} = J$$

where the source is

$$J = \Upsilon_1 - \Upsilon_3 + 4\Delta^{-1}\Upsilon'_2 + 2(1 - \epsilon + \eta)\mathcal{H}\Delta^{-1}\Upsilon_2 + \Delta^{-1}\gamma'' - (1 + 2\epsilon - 2\eta)\mathcal{H}\Delta^{-1}\gamma'.$$
(D-18)

We can split the source into tachyon and inflaton contributions $J = J^{\varphi} + J^{\sigma}$ in the obvious manner, by taking the tachyon and inflaton parts of $\Upsilon_1, \Upsilon_2, \Upsilon_3, \gamma$.

We now derive some results concering the tachyon source terms which will be useful in the text. First we consider the tachyon contribution to γ (D-17):

$$\gamma_{\sigma} = \Upsilon_3^{\sigma} - 3\Delta^{-1}\partial_{\tau}\Upsilon_2^{\sigma} - 6\mathcal{H}\Delta^{-1}\Upsilon_2^{\sigma}.$$

Using equations (D-11) and (D-14) we can write this as

$$\gamma_{\sigma} = \kappa^{2} \left[\frac{3}{2} (\delta^{(1)} \sigma')^{2} - \frac{1}{2} \partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma - \frac{3}{2} a^{2} m_{\sigma}^{2} (\delta^{(1)} \sigma)^{2} \right] - 3 \kappa^{2} \Delta^{-1} \partial_{\tau} \partial_{i} (\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma) - 6 \kappa^{2} \mathcal{H} \Delta^{-1} \partial_{i} (\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma)$$
(D-19)

We now write γ_{σ} as

$$\begin{split} \gamma_{\sigma} &= \kappa^{2} \Delta^{-1} \left[\frac{3}{2} \partial^{k} \partial_{k} (\delta^{(1)} \sigma')^{2} - \frac{1}{2} \partial^{k} \partial_{k} (\partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma) \right. \\ &- \frac{3}{2} a^{2} m_{\sigma}^{2} \partial^{k} \partial_{k} (\delta^{(1)} \sigma)^{2} - 3 \partial_{\tau} \partial_{i} (\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma) \\ &- \left. 6 \mathcal{H} \partial_{i} (\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma) \right] \end{split}$$

and, after some algebra, we have

$$\begin{split} \gamma_{\sigma} &= \kappa^{2} \triangle^{-1} \left[-\frac{1}{2} \partial^{k} \partial_{k} (\partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma) \right. \\ &- 3 \partial_{i} \left(\delta^{(1)} \sigma'' + 2 \mathcal{H} \delta^{(1)} \sigma' + a^{2} m_{\sigma}^{2} \delta^{(1)} \sigma \right) \partial^{i} \delta^{(1)} \sigma \\ &- 3 \left(\delta^{(1)} \sigma'' + 2 \mathcal{H} \delta^{(1)} \sigma' + a^{2} m_{\sigma}^{2} \delta^{(1)} \sigma \right) \partial^{k} \partial_{k} \delta^{(1)} \sigma \right] \end{split}$$

The last two lines can be simplified using the equation of motion for the tachyon fluctuation

$$\delta^{(1)}\sigma'' + 2\mathcal{H}\delta^{(1)}\sigma + a^2m_{\sigma}^2\delta^{(1)}\sigma = \partial^k\partial_k\delta^{(1)}\sigma$$

which gives

$$\gamma_{\sigma} = \kappa^{2} \Delta^{-1} \left[-\frac{1}{2} \partial^{k} \partial_{k} \left(\partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) \right. \\ \left. - 3 \partial_{i} \left(\partial^{k} \partial_{k} \delta^{(1)} \sigma \right) \partial^{i} \delta^{(1)} \sigma - 3 \partial_{i} \partial^{i} \delta^{(1)} \sigma \partial^{k} \partial_{k} \delta^{(1)} \sigma \right] \\ \left. = \kappa^{2} \Delta^{-1} \left[-\frac{1}{2} \partial^{k} \partial_{k} \left(\partial_{i} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) \right. \\ \left. - 3 \partial_{i} \left(\partial^{k} \partial_{k} \delta^{(1)} \sigma \partial^{i} \delta^{(1)} \sigma \right) \right]$$
(D-20)

This result has also been derived in [183]. We now consider the tachyon contribution to the source. We take

$$J^{\sigma} = \Upsilon_{1}^{\sigma} - \Upsilon_{3}^{\sigma} + 4\Delta^{-1}\partial_{\tau}\Upsilon_{2}^{\sigma} + 2(1-\epsilon+\eta)\mathcal{H}\Delta^{-1}\Upsilon_{2}^{\sigma} + \Delta^{-1}\gamma_{\sigma}'' - (1+2\epsilon-2\eta)\mathcal{H}\Delta^{-1}\gamma_{\sigma}'$$

and, using (D-8), (D-11) and (D-14), we have

$$J^{\sigma}(\tau, \vec{x}) = a^{2} \kappa^{2} m_{\sigma}^{2} \left(\delta^{(1)} \sigma\right)^{2} - 2\kappa^{2} \left(\delta^{(1)} \sigma'\right)^{2} + 2\kappa^{2} \mathcal{H}(1 + \eta - \epsilon) \Delta^{-1} \partial_{i} \left(\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma\right) + 4\kappa^{2} \Delta^{-1} \partial_{\tau} \partial_{i} \left(\delta^{(1)} \sigma' \partial^{i} \delta^{(1)} \sigma\right) - \mathcal{H}(1 + 2\epsilon - 2\eta) \Delta^{-1} \gamma_{\sigma}' + \Delta^{-1} \gamma_{\sigma}''.$$
(D-21)

D-3 The Inflaton Contribution to $\zeta^{(2)}$

In this appendix we consider the calculation of the inflaton part of the second order curvature perturbation $\zeta_{\varphi}^{(2)}$ using results from appendix D-2 and following closely the calculation of the tachyon part of the second order curvature perturbation.

The last four lines of (6.41) are relatively simple to evaluate in the large scale and slow roll limit. On the other hand, the first line of (6.41) is somewhat more complicated and its evaluation requires the second order Einstein equations. We first focus first on the following contribution to $\zeta^{(2)}$

$$\zeta^{(2)} \ni \frac{1}{3-\epsilon} \frac{1}{(\varphi_0')^2} \left[\varphi_0' Q_{\varphi}'^{(2)} + a^2 \frac{\partial V}{\partial \varphi} Q_{\varphi}^{(2)} \right]. \tag{D-22}$$

From the definition of $Q_{\varphi}^{(2)}$ it is clear that this contains a contribution of the form $\varphi'_0 \delta^{(2)} \varphi' + a^2 \partial V / \partial \varphi \, \delta^{(2)} \varphi$ which also appears in the second order (0,0) Einstein equation (D-6). We therefore use (D-6) to eliminate $\delta^{(2)} \varphi$ from (D-22). We also eliminate $\psi^{(2)}$ in favour of $\phi^{(2)}$ using (D-16). The result is that (D-22) takes the form

$$\begin{aligned} \zeta^{(2)} & \ni -\frac{\phi^{\prime(2)}}{\epsilon \mathcal{H}} - \left(\frac{1}{\epsilon} + 1\right) \phi^{(2)} + \frac{1}{3-\epsilon} \frac{\partial^k \partial_k \phi^{(2)}}{\epsilon \mathcal{H}^2} \\ &+ \frac{1}{\epsilon \mathcal{H}} \Delta^{-1} \gamma' + \Delta^{-1} \gamma - \frac{1}{3-\epsilon} \frac{1}{\epsilon \mathcal{H}^2} \gamma \\ &+ \frac{1}{3-\epsilon} \frac{\Upsilon_1}{\epsilon \mathcal{H}^2} + \frac{1}{3-\epsilon} \left(\frac{\Upsilon_4'}{\varphi_0'} + \frac{a^2}{(\varphi_0')^2} \frac{\partial V}{\partial \varphi} \Upsilon_4\right) \end{aligned}$$
(D-23)

where the quantities Υ_1, γ are constructed entirely from first order fluctuations and are defined explicitly in appendix D-2 in equations (D-7, D-8, D-10, D-11, D-13, D-14). The quantity Υ_4 is also contructed from first order fluctuations and is defined as the last two lines of (6.44). That is,

$$\Upsilon_{4} = (2 + 2\epsilon - \eta) \frac{\varphi_{0}'}{\mathcal{H}} (\phi^{(1)})^{2} + 2 \frac{\varphi_{0}'}{\mathcal{H}^{2}} \phi^{(1)} \phi'^{(1)} + \frac{2}{\mathcal{H}} \phi^{(1)} \delta^{(1)} \varphi'.$$
(D-24)

The first line of (D-23) can be solved for using the Green function for the master equation as:

$$- \frac{\phi_{k}^{\prime(2)}}{\epsilon \mathcal{H}} - \frac{\phi_{k}^{(2)}}{\epsilon} - \frac{k^{2}\phi_{k}^{(2)}}{3\epsilon \mathcal{H}^{2}} = \frac{1}{\epsilon} \int_{\tau_{i}}^{0} d\tau' \Theta(\tau - \tau') J_{k}^{\varphi}(\tau') \times \left[\tau' + \left(-\frac{1}{6}\tau'^{3} + \frac{5}{6}\tau^{2}\tau' - \frac{2}{3}\tau^{3}\right)k^{2}\right].$$
(D-25)

This equation allows us to remove all of the explicit dependence of $\zeta^{(2)}$ on the second order metric perturbation $\phi^{(2)}$.

Having removed the explicit dependence of $\zeta^{(2)}$ on the second order metric perturbation $\phi^{(2)}$ we next eliminate $\phi^{(1)}, \delta^{(1)}\varphi$ in favour of $\zeta^{(1)}$. Using the solution (1.47) and the first order constraint equation (1.40) one may verify that on large scales

$$\zeta^{(1)} \cong -\phi^{(1)} - \frac{\varphi_0'}{\mathcal{H}} \delta^{(1)} \varphi \cong -\frac{1}{\epsilon} \phi^{(1)}$$

and

$$\delta^{(1)} \varphi \cong rac{\varphi'_0}{\mathcal{H}} rac{1}{\epsilon} \phi^{(1)} \cong -rac{\varphi'_0}{\mathcal{H}} \zeta^{(1)}.$$

On large scales the first order curvature perturbation is approximately constant since [190]

$$\zeta'^{(1)} = \frac{\partial^k \partial_k}{\epsilon \mathcal{H}} \phi^{(1)}$$

using the fact that $\sigma_0 = 0$ (so that there are no anisotropic stresses, whose absence guarantees the conservation of $\zeta^{(1)}$ on super-Hubble scales). Thus on large scales we have

$$\delta^{(1)}\varphi' \cong -(2\epsilon - \eta)\varphi_0'\zeta^{(1)}$$

It is straightforward to compute the last three lines of (6.41) using the fact that $Q_{\varphi}^{(1)} \approx -\varphi_0' \zeta^{(1)}/\mathcal{H}$ and $Q_{\varphi}'^{(1)} \approx -(2\epsilon - \eta)\varphi_0' \zeta^{(1)}$ on large scales. The result is

$$\zeta^{(2)} \ni \left[\frac{1}{3} \left(\frac{a \, m_{\varphi}}{\mathcal{H}}\right)^2 + 2 + 2\epsilon - 2\eta\right] (\zeta^{(1)})^2 \tag{D-26}$$

where we have dropped terms which are higher order in slow roll parameters or which contain gradients. Notice that the quantity $(a m_{\varphi}/\mathcal{H})^2$ is first order in the slow roll expansion because

$$\eta \approx \frac{1}{\kappa^2} \frac{1}{V} \frac{\partial^2 V}{\partial \varphi^2} = \frac{1}{\kappa^2} \frac{m_{\varphi}^2}{V} \approx \frac{1}{3} \left(\frac{a \, m_{\varphi}}{\mathcal{H}}\right)^2$$

To (D-26) we must add the contribution coming from the fist line of (6.41) which can be written explicitly in terms of first order quantities using the Green function for the master equation. Here we consider only the particular solution for $\phi^{(2)}$ due to the inflaton source J^{φ} .

In order to compute J^{φ} we first study the quantities $\Upsilon_1, \Upsilon_2, \Upsilon_3, \Upsilon_4, \gamma$ which are defined in appendix D-2. On large scales and to leading order in slow roll we have

$$\Upsilon_1^{\varphi} \approx \left[12\epsilon^2 - \epsilon \left(\frac{a \, m_{\varphi}}{\mathcal{H}} \right)^2 \right] \mathcal{H}^2(\zeta^{(1)})^2.$$

The quantity $\triangle^{-1}\Upsilon_2^{\varphi}$ can be written as (see equation 39 of [183])

$$\Delta^{-1}\Upsilon_{2}^{\varphi} = \frac{\kappa^{2}}{2}(\epsilon - \eta)\mathcal{H}(\delta^{(1)}\varphi)^{2} + 3\mathcal{H}(\phi^{(1)})^{2} - 2\phi^{(1)}\phi'^{(1)}$$
$$+ \frac{2}{\varphi_{0}'}\Delta^{-1}\left(\partial_{i}\partial^{k}\partial_{k}\phi^{(1)}\partial^{i}\delta^{(1)}\varphi\right)$$
$$+ \partial^{k}\partial_{k}\phi^{(1)}\partial_{i}\partial^{i}\delta^{(1)}\varphi\right).$$

The inverse laplacians on the last two lines contribute only to the momentum dependence of f_{NL}^{φ} which we neglect. On large scales and in the slow roll limit we have

$$\Delta^{-1}\Upsilon_2^{\varphi} \approx (4\epsilon^2 - \epsilon\eta)\mathcal{H}(\zeta^{(1)})^2 + \cdots$$

where the \cdots denotes momentum dependent terms. On large scales and in the slow roll limit we also have

$$\Upsilon_3^{\varphi} \approx \left[36\epsilon^2 - 3\epsilon \left(\frac{a \, m_{\varphi}}{\mathcal{H}} \right)^2 \right] \mathcal{H}^2(\zeta^{(1)})^2.$$

The quantity Υ_4 defined in (D-24) depends only on the inflaton fluctuation and is given by

$$\Upsilon_4 = \Upsilon_4^{\varphi} \approx (6\epsilon^2 - 2\epsilon\eta) \frac{\varphi_0'}{\mathcal{H}} (\zeta^{(1)})^2$$

on large scales and in the slow roll limit. Using these results one may readily verify that $\gamma_{\varphi}/\mathcal{H}^2$ is third order in slow roll parameters

$$\gamma_{\varphi} \approx \mathcal{O}(\epsilon^3) \mathcal{H}^2(\zeta^{(1)})^2.$$

Using these results one may verify that the only term on the first line of (6.41) which contributes at lowest order in slow roll parameters is the term proportional to $\Upsilon_1^{\varphi}/\epsilon \mathcal{H}^2$. Thus, the contribution to the second order curvature perturbation due to the first line of (6.41) is

$$\zeta_{\varphi}^{(2)} \ni \left[4\epsilon - \frac{1}{3} \left(\frac{a \, m_{\varphi}}{\mathcal{H}} \right)^2 \right] (\zeta^{(1)})^2.$$

Adding all the contributions together we find

$$\zeta_{\varphi}^{(2)} \approx (2 - 2\eta + 6\epsilon)(\zeta^{(1)})^2.$$

The contribution $2(\zeta^{(1)})^2$ stems from using the Malik and Wands [189] definition of the second order curvature perturbation. It can be related to the definition of Lyth and Rodriguez [187] (which also agrees with Maldacena [18]) using

$$\zeta^{(2)} = \zeta_{LR}^{(2)} + 2(\zeta^{(1)})^2.$$

The Lyth-Rodriguez curvature perturbation, due to the inflaton up to second order, can thus be written as

$$\zeta_{LR}^{\varphi} = \zeta^{(1)} - \frac{3}{5} f_{NL}^{\varphi} (\zeta^{(1)})^2$$

where

$$f_{NL}^{\varphi} = \frac{5}{6}(2\eta - 6\epsilon).$$

In writing ζ_{LR}^{φ} above we have suppressed the homogeneous k = 0 mode of ζ which should be subtracted to ensure that $\langle \zeta \rangle = 0$. This result differs from previous studies [18, 158] by a factor of two. The calculation of [18, 158] takes into account the effect of nonlinear evolution as the effect of computing the bispectrum in the vacuum of the interacting theory, as opposed to the vacuum of the free theory. Our calculation does not consider the effect of the change in vacuum which is the same order of magnitude. Thus we should not expect to reproduce exactly the results of [18, 158]. The change in vacuum will not change the calculation of the tachyon part of the curvature perturbation $\zeta_{\sigma}^{(2)}$ since $\zeta^{(1)}$ does not depend on $\delta^{(1)}\sigma$ and hence contributions to ζ_{σ} due to the change in vacuum will be higher than second order in perturbation theory.

D-4 Fourier Transforms, Mode Functions and Inverse Laplacians

We define the Fourier transform of some first order quantity $\delta f(t, \vec{x})$ by

$$\delta f(t, \vec{x}) = \int \frac{d^{3}k}{(2\pi)^{3/2}} e^{i\vec{k}\cdot\vec{x}}\delta \tilde{f}_{k}(t)$$

$$\delta \tilde{f}_{\vec{k}}(t) = \int \frac{d^{3}x}{(2\pi)^{3/2}} e^{-i\vec{k}\cdot\vec{x}}\delta f(t, \vec{x})$$
(D-27)

where the Hermiticity of $\delta f(t, \vec{x})$ implies that $\delta \tilde{f}_k(t) = \delta \tilde{f}_{-k}(t)^{\dagger}$ so that we can define

$$\delta \tilde{f}_k(t) = a_k \xi_k(t) + a_{-k}^{\dagger} \xi_{-k}(t)^{\star}$$
(D-28)

where a_k is an operator and $\delta f_k(t)$ is a c-number valued mode function. We then re-write the Fourier transform as

$$\delta f(t,x) = \int \frac{d^3k}{(2\pi)^{3/2}} \left[a_k \xi_k(t) e^{ikx} + a^{\dagger}_{-k} \xi_{-k}(t)^* e^{ikx} \right]$$

=
$$\int \frac{d^3k}{(2\pi)^{3/2}} \left[a_k \xi_k(t) e^{ikx} + a^{\dagger}_k \xi_k(t)^* e^{-ikx} \right].$$

In this form it is clear that a_k is the usual creation operator satisfying

$$\left[a_k, a_{k'}^{\dagger}\right] = \delta^{(3)}(\vec{k} - \vec{k'})$$

and one should expect

$$\xi_k(\tau) \approx \frac{1}{\sqrt{2k}} e^{-ik\tau}$$

on small scales, which corresponds to the Bunch-Davies vacuum $a_k|0\rangle = 0$. This is consistent with the usual definition of the power spectrum in terms of the two-point function

$$\langle 0|\delta f^2(t,\vec{x})|0\rangle = \int \frac{d^3k}{(2\pi)^3} |\xi_k(t)|^2$$
$$= \int \frac{dk}{k} \mathcal{P}_f(k)$$

so that

$$\mathcal{P}_f(k) = \frac{k^3}{2\pi^2} |\xi_k(t)|^2.$$

We now discuss the Fourier transform of the tachyon source. Typical terms in the source have the form

$$J^{(1)}(t, \vec{x}) = b(t)\delta f(t, \vec{x})\delta g(t, \vec{x}),$$

$$J^{(2)}(t, \vec{x}) = b(t)\Delta^{-1} [\delta f(t, \vec{x})\delta g(t, \vec{x})],$$

$$J^{(3)}(t, \vec{x}) = b(t)\Delta^{-2} [\delta f(t, \vec{x})\delta g(t, \vec{x})]$$

where δf , δg are some first order quantities and b is constructed from zeroth order quantities. The Green's function for the laplacian is defined as appropriate in the absence of boundary surfaces

$$(\triangle^{-1}f)(t,\vec{x}) = \int d^3x' G(\vec{x}-\vec{x}')f(t,\vec{x}')$$

where

$$G(\vec{x} - \vec{x}') = -\frac{1}{4\pi} \frac{1}{|\vec{x} - \vec{x}'|}.$$

In Fourier space we have

$$G(x, x') = \int \frac{d^3k}{(2\pi)^{3/2}} e^{ikx} G_k(x')$$
$$G_k(x') = -\frac{1}{(2\pi)^{3/2}} \frac{e^{-ikx'}}{k^2}.$$

In Fourier space the souce terms contain convolutions

$$J_k^{(1)}(t) = b(t)(\delta \tilde{f} \star \delta \tilde{g})_k(t),$$

$$J_k^{(2)}(t) = -b(t)\frac{1}{k^2}(\delta \tilde{f} \star \delta \tilde{g})_k(t),$$

$$J_k^{(3)}(t) = b(t)\frac{1}{k^4}(\delta \tilde{f} \star \delta \tilde{g})_k(t).$$

where \tilde{f}_k , \tilde{g}_k are operator valued Fourier transforms defined as in (D-27). These may be related to the mode functions as (D-28). Finally, we have defined convolution by

$$(\delta \tilde{f} \star \delta \tilde{g})_{\vec{k}}(t) = \int \frac{d^3 k'}{(2\pi)^{3/2}} \delta \tilde{f}_{\vec{k}'}(t) \delta \tilde{g}_{\vec{k}-\vec{k}'}(t)$$

D-5 Construction of The Tachyon Curvature Perturbation

In this appendix we include technical details about the contruction of $\zeta_{\sigma}^{(2)}$ using the Green function for the master equation. For clarity we repeat some details which are included in the text. We consider only contributions to the equations which depend on the tachyon field and we remind the reader that $\zeta^{(1)}$ is independent of σ . The second order curvature perturbation is (see (6.41-6.43))

$$\begin{aligned} \zeta^{(2)} &= \frac{1}{3-\epsilon} \frac{1}{(\varphi_0')^2} \left[\varphi_0' Q_{\varphi}'^{(2)} + a^2 \frac{\partial V}{\partial \varphi} Q_{\varphi}^{(2)} \right] \\ &+ \frac{1}{3-\epsilon} \frac{1}{(\varphi_0')^2} \left[\left(\delta^{(1)} \sigma' \right)^2 + a^2 m_{\sigma}^2 \left(\delta^{(1)} \sigma \right)^2 \right] \\ &+ \text{ inflaton contributions} \end{aligned}$$
(D-29)

where the second order Sasaki-Mukhanov variable is

$$Q_{\varphi}^{(2)} = \delta^{(2)}\varphi + \frac{\varphi_0'}{\mathcal{H}}\psi^{(2)} + \text{ inflaton contributions}$$
(D-30)

Inserting (D-30) into (D-29) and using the (0,0) Einstein equation (D-6) to eliminate the contribution $\varphi'_0 \delta^{(2)} \varphi' + a^2 \partial V / \partial \varphi \delta^{(2)} \varphi$ gives

$$\begin{aligned} \zeta_{\sigma}^{(2)} &= -\frac{\phi^{\prime(2)}}{\epsilon \mathcal{H}} - \left(\frac{1}{\epsilon} + 1\right) \phi^{(2)} + \frac{1}{3-\epsilon} \frac{\partial^{k} \partial_{k} \phi^{(2)}}{\epsilon \mathcal{H}^{2}} \\ &+ \frac{1}{\epsilon \mathcal{H}} \Delta^{-1} \gamma_{\sigma}^{\prime} + \Delta^{-1} \gamma_{\sigma} - \frac{1}{3-\epsilon} \frac{1}{\epsilon \mathcal{H}^{2}} \gamma_{\sigma} \end{aligned} \tag{D-31}$$

where we have also eliminated $\psi^{(2)}$ in favour of $\phi^{(2)}$ using (D-16). Notice that using (D-6) introduces a term proportional to Υ_1^{σ} to the curvature perturbation which cancels the contribution proportional to $(\delta^{(1)}\sigma')^2 + a^2m_{\sigma}^2(\delta^{(1)}\sigma)^2$ on the second line

of (D-29) up to a gradient term $(\vec{\nabla}\delta^{(1)}\sigma)^2$ which can be neglected on large scales. The result (D-31) is valid only on large scales but we have not yet assumed slow roll. In (D-31) it is understood that $\phi^{(2)}$ denotes only the particular solution due to the tachyon source, J^{σ} . We now use the Green function (6.36) to solve for $\phi^{(2)}$ in terms of the tachyon source (D-21). We work only to leading order in slow roll parameters. We also work in the large scale limit. To lowest order in ϵ , η and up to order k^2 we can write

$$- \frac{\phi_k^{\prime(2)}}{\epsilon \mathcal{H}} - \frac{\phi_k^{(2)}}{\epsilon} - \frac{k^2 \phi_k^{(2)}}{3\epsilon \mathcal{H}^2} = \frac{1}{\epsilon} \int_{\tau_i}^0 d\tau' \Theta(\tau - \tau') J_k^{\sigma}(\tau') \times \left[\tau' + \left(-\frac{1}{6} \tau'^3 + \frac{5}{6} \tau^2 \tau' - \frac{2}{3} \tau^3 \right) k^2 \right]$$
(D-32)

using the Green function (6.36) and the relation (6.37). Equation (D-32) allows us to eliminate the dependence of $\zeta_{\sigma}^{(2)}$ on $\phi^{(2)}$. At leading order in slow roll parameters the tachyon source is (see (D-18))

$$J^{\sigma} = \Upsilon_{1}^{\sigma} - \Upsilon_{3}^{\sigma} + 4 \Delta^{-1} \partial_{\tau} \Upsilon_{2}^{\sigma} + 2\mathcal{H} \Delta^{-1} \Upsilon_{2}^{\sigma} + \Delta^{-1} \partial_{\tau}^{2} \gamma_{\sigma} - \mathcal{H} \Delta^{-1} \partial_{\tau} \gamma_{\sigma}.$$
(D-33)

We must now insert (D-33) into (D-32), perform numerous integrations by parts, and then insert this result into (D-31). We evaluate term-by-term the last line of (D-32). The first contribution (the term proportional to k^0) is

$$-\frac{\phi_k^{\prime(2)}}{\epsilon \mathcal{H}} - \frac{\phi_k^{(2)}}{\epsilon} - \frac{k^2 \phi_k^{(2)}}{3\epsilon \mathcal{H}^2} \ni -\frac{1}{\epsilon} \int_{\tau_i}^{\tau} d\tau' \frac{J^{\sigma}(\tau')}{\mathcal{H}(\tau')}$$

Inserting (D-33), noting that $\mathcal{H}(\tau') = -1/\tau'$ at leading order in slow roll and integrating by parts gives

$$-\frac{\phi_{k}^{\prime(2)}}{\epsilon\mathcal{H}} - \frac{\phi_{k}^{(2)}}{\epsilon} - \frac{k^{2}\phi_{k}^{(2)}}{3\epsilon\mathcal{H}^{2}} \ni -\frac{1}{\epsilon}\int_{\tau_{i}}^{\tau} d\tau' \frac{J^{\sigma}(\tau')}{\mathcal{H}(\tau')}$$

$$\cong \frac{1}{\epsilon}\int_{\tau_{i}}^{\tau} d\tau' \left[-\frac{\Upsilon_{1}^{\sigma}}{\mathcal{H}(\tau')} + \frac{\Upsilon_{3}^{\sigma}}{\mathcal{H}(\tau')} - 6\Delta^{-1}\Upsilon_{2}^{\sigma} \right]$$

$$+ \frac{1}{\epsilon} \left[-4\frac{\Delta^{-1}\Upsilon_{2}^{\sigma}}{\mathcal{H}} - \frac{\Delta^{-1}\gamma_{\sigma}'}{\mathcal{H}} \right] + \cdots$$
(D-34)

where the terms under the $d\tau'$ integral are evaluated at τ' while the terms in the square braces on the second line are evaluated at τ . The \cdots denotes constant terms evaluated at $\tau = \tau_i$ which arise from the integration by parts. Since our interest is in the preheating phase during which the fluctuations $\delta^{(1)}\sigma$ are amplified exponentially we can safely drop these constant terms.

The second contribution to the last line of (D-32) has the form

$$-\frac{\phi_k^{\prime(2)}}{\epsilon\mathcal{H}} - \frac{\phi_k^{(2)}}{\epsilon} - \frac{k^2\phi_k^{(2)}}{3\epsilon\mathcal{H}^2} \ni \frac{1}{6\epsilon}\int_{\tau_i}^{\tau} d\tau' \frac{k^2 J_k^{\sigma}}{\mathcal{H}(\tau')^3}.$$

In evaluating this we need only consider terms in $k^2 J_k^{\sigma}$ which are not suppressed on large scales. Using (D-33) one may verify that

$$k^2 J_k^{\sigma} \cong -\gamma_{\sigma,k}^{\prime\prime} + \mathcal{H} \gamma_{\sigma,k}^{\prime} \tag{D-35}$$

on large scales (recall that Υ_2^{σ} is a gradient, see (D-11)) so that we have

$$- \frac{\phi_{k}^{\prime(2)}}{\epsilon \mathcal{H}} - \frac{\phi_{k}^{(2)}}{\epsilon} - \frac{k^{2}\phi_{k}^{(2)}}{3\epsilon \mathcal{H}^{2}} \ni \frac{1}{6\epsilon} \int_{\tau_{i}}^{\tau} d\tau' \frac{k^{2}J_{k}^{\sigma}}{\mathcal{H}(\tau')^{3}}$$
$$\cong \frac{1}{\epsilon} \int_{\tau_{i}}^{\tau} d\tau' \left[-\frac{2}{3}\frac{\Upsilon_{3}^{\sigma}}{\mathcal{H}} + 6\Delta^{-1}\Upsilon_{2}^{\sigma} \right]$$
$$+ \left[-\frac{1}{6\epsilon}\frac{\gamma_{\sigma}'}{\mathcal{H}^{3}} - \frac{1}{3\epsilon}\frac{\gamma_{\sigma}}{\mathcal{H}^{2}} + 2\frac{\Delta^{-1}\Upsilon_{2}^{\sigma}}{\mathcal{H}} \right] + \cdots$$
(D-36)

Equation (D-35) shows why it was necessary to include the k^2 terms in the large scale expansion of the Green function (D-32), noting that γ_{σ} may be written as (D-17) and comparing to (D-33) we see that $k^2 J_k$ contains terms which are of the same size as those in J_k , on large scales. Thus a consistent large scale expansion of $\zeta_{\sigma}^{(2)}$ requires that we work up to order k^2 in the expansion of the Green function.

The third contribution to the last line of (D-32) is

$$-\frac{\phi_k^{\prime(2)}}{\epsilon\mathcal{H}} - \frac{\phi_k^{(2)}}{\epsilon} - \frac{k^2\phi_k^{(2)}}{3\epsilon\mathcal{H}^2} \ni -\frac{5}{6\epsilon}\int_{\tau_i}^{\tau} d\tau' \frac{k^2 J_k^{\sigma}}{\mathcal{H}(\tau')\mathcal{H}(\tau)^2}$$
$$\cong \frac{5}{6\epsilon} \left[\frac{\gamma_{\sigma}'}{\mathcal{H}^3}\right] + \cdots$$
(D-37)

using the same procedure as above.

Finally, the fourth contribution to the last line of (D-32) is

$$-\frac{\phi_{k}^{\prime(2)}}{\epsilon\mathcal{H}} - \frac{\phi_{k}^{(2)}}{\epsilon} - \frac{k^{2}\phi_{k}^{(2)}}{3\epsilon\mathcal{H}^{2}} \ni \frac{2}{3\epsilon} \int_{\tau_{i}}^{\tau} d\tau' \frac{k^{2}J_{k}^{\sigma}}{\mathcal{H}(\tau)^{3}}$$
$$\cong -\frac{2}{3\epsilon} \frac{1}{\mathcal{H}(\tau)^{3}} \int_{\tau_{i}}^{\tau} d\tau' \mathcal{H}(\tau')^{2} \Upsilon_{3}^{\sigma}$$
$$+ \frac{1}{\epsilon} \left[2\frac{\Delta^{-1}\Upsilon_{2}^{\sigma}}{\mathcal{H}} + \frac{2}{3}\frac{\gamma_{\sigma}}{\mathcal{H}^{2}} - \frac{2}{3}\frac{\gamma_{\sigma}'}{\mathcal{H}^{3}} \right] + \cdots$$
(D-38)

Summing up (D-34), (D-36), (D-37) and (D-38) and inserting the result into (D-31) gives

$$\begin{aligned} \zeta_{\sigma}^{(2)} &\cong \frac{1}{\epsilon} \int_{\tau_{i}}^{\tau} d\tau' \left[-\frac{\Upsilon_{1}^{\sigma}}{\mathcal{H}(\tau')} + \frac{1}{3} \frac{\Upsilon_{3}^{\sigma}}{\mathcal{H}(\tau')} \right. \\ &\left. - \frac{2}{3} \frac{\mathcal{H}(\tau')^{2}}{\mathcal{H}(\tau)^{3}} \Upsilon_{3}^{\sigma} \right] \end{aligned}$$

Now, using equations (D-8) and (D-14) we can write this in terms of the tachyon fluctuation $\delta^{(1)}\sigma$ as

$$\begin{aligned} \zeta_{\sigma}^{(2)} &\cong \frac{\kappa^2}{\epsilon} \int_{-1/a_i H}^{\tau} d\tau' \left[\frac{\left(\delta^{(1)} \sigma' \right)^2}{\mathcal{H}(\tau')} - \frac{\mathcal{H}(\tau')^2}{\mathcal{H}(\tau)^3} \left(\left(\delta^{(1)} \sigma' \right)^2 - a^2 m_{\sigma}^2 \left(\delta^{(1)} \sigma \right)^2 \right) \right] \end{aligned} \tag{D-39}$$

D-6 An Identity Concerning γ_{σ}

In this appendix we derive an identity concering the tachyon source term γ_{σ} (6.101):

$$\gamma_{\sigma} = \Upsilon_3^{\sigma} - 3 \triangle^{-1} \partial_{\tau} \Upsilon_2^{\sigma} - 6 \mathcal{H} \triangle^{-1} \Upsilon_2^{\sigma}.$$

Using equations (6.95) and (6.98) we can write this

$$\begin{split} \gamma_{\sigma} &= \kappa^{2} \Delta^{-1} \left[\frac{3}{2} \partial^{k} \partial_{k} (\delta^{(1)} \sigma'_{A} \delta^{(1)} \sigma'^{A}) \right. \\ &- \frac{1}{2} \partial^{k} \partial_{k} (\partial_{i} \delta^{(1)} \sigma_{A} \partial^{i} \delta^{(1)} \sigma^{A}) \\ &- \frac{3}{2} a^{2} \frac{\partial^{2} V}{\partial \sigma_{A} \partial \sigma_{B}} \partial^{k} \partial_{k} (\delta^{(1)} \sigma_{A} \delta^{(1)} \sigma_{B}) \\ &- 3 \partial_{\tau} \partial_{i} (\delta^{(1)} \sigma'_{A} \partial^{i} \delta^{(1)} \sigma^{A}) \\ &- 6 \mathcal{H} \partial_{i} (\delta^{(1)} \sigma'_{A} \partial^{i} \delta^{(1)} \sigma^{A}) \right] \end{split}$$

and, after some algebra, we have

$$\gamma_{\sigma} = \kappa^{2} \Delta^{-1} \left[-\frac{1}{2} \partial^{k} \partial_{k} (\partial_{i} \delta^{(1)} \sigma_{A} \partial^{i} \delta^{(1)} \sigma^{A}) \right]$$
(D-40)
$$- 3 \partial_{i} \left(\delta^{(1)} \sigma_{A}'' + 2 \mathcal{H} \delta^{(1)} \sigma_{A}' + a^{2} \frac{\partial^{2} V}{\partial \sigma_{A} \partial \sigma_{B}} \delta^{(1)} \sigma_{B} \right) \partial^{i} \delta^{(1)} \sigma^{A}$$
$$- 3 \left(\delta^{(1)} \sigma_{A}'' + 2 \mathcal{H} \delta^{(1)} \sigma_{A}' + a^{2} \frac{\partial^{2} V}{\partial \sigma_{A} \partial \sigma_{B}} \delta^{(1)} \sigma_{B} \right) \partial^{k} \partial_{k} \delta^{(1)} \sigma^{A}$$

In deriving this equation we have used that fact that

$$\frac{\partial^2 V}{\partial \sigma_A \partial \sigma_B} = \frac{\partial^2 V}{\partial \sigma_B \partial \sigma_A}$$

which follows from the O(M) symmetry of the theory. The last two lines of (D-40) can be simplified using the equation of motion for the tachyon fluctuation (6.103) which gives

$$\gamma_{\sigma} = -\frac{\kappa^2}{2} \left(\partial_i \delta^{(1)} \sigma_A \partial^i \delta^{(1)} \sigma^A \right) - 3\kappa^2 \Delta^{-1} \partial_i \left(\partial^k \partial_k \delta^{(1)} \sigma_A \partial^i \delta^{(1)} \sigma^A \right)$$
(D-41)

Chapter E

Appendices to Chapter 7

E-1 The Stress Energy Tensor and the Friedmann Equation

Here we compute the $O(u^2)$ term of the approximate solutions

$$\phi = 1 - u - \phi_2 u^2$$
$$H = H_0 - H_2 u^2$$

We write the different terms appearing in the energy density as

$$\rho_{\phi} = -T_{00} = \frac{m_s^4}{2g_p^2} \left[\phi e^{-\frac{\Box}{m_p^2}} \phi - \frac{2}{p+1} \phi^{p+1} + \frac{1}{m_p^2} \int_0^1 d\tau \ \left(\Box e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) + \frac{1}{m_p^2} \int_0^1 d\tau \ \partial_t \left(e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \partial_t \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) \right] \equiv \frac{m_s^4}{2g_p^2} (T_1 + T_2 + T_3 + T_4)$$
(E-1)

where T_1 , T_2 , T_3 and T_4 are defined as in (7.49-7.52). We now evaluate this expression term-by-term.

We first consider $T_1 + T_2$, which can be written as

$$T_1 + T_2 = \frac{p-1}{p+1}\phi^{p+1}$$

using (7.3). Using (E-1) we have

$$T_1 + T_2 = \frac{p-1}{p+1} + (-p+1)u + (p-1)\left[-\phi_2 + \frac{p}{2}\right]u^2 + \mathcal{O}(u^3)$$
(E-2)

To evaluate T_3 , T_4 we note that

$$e^{-\tau \Box/m_p^2} \phi = 1 - e^{\tau \mu_1} u - \phi_2 e^{\tau \mu_2} u^2 + \cdots$$
$$e^{-(1-\tau)\Box/m_p^2} \phi = 1 - e^{(1-\tau)\mu_1} u - \phi_2 e^{(1-\tau)\tau \mu_2} u^2 + \cdots$$
(E-3)

(see equation 7.23 with $\sigma = 0$). It is straightforward to show that

$$\Box e^{-\tau \Box/m_p^2} \phi = e^{\tau \mu_1} (\lambda^2 + 3H_0 \lambda) u + \phi_2 e^{\tau \mu_2} (4\lambda^2 + 3H_0 \lambda) u^2 + \cdots$$
 (E-4)

We are now in a position to compute T_3 . Using (E-3) and (E-4) we have

$$T_{3} = \frac{1}{m_{p}^{2}} \int_{0}^{1} d\tau \left[(\lambda^{2} + 3H_{0}\lambda)e^{\tau\mu_{1}}u + \left[-(\lambda^{2} + 3H_{0}\lambda)e^{\mu_{1}} + \phi_{2}(4\lambda^{2} + 6H_{0}\lambda)e^{\tau\mu_{2}} \right] u^{2} + \cdots \right]$$

The $d\tau$ integrals are trivial to perform using the identity

$$\int_0^1 d\tau e^{\alpha\tau} = \frac{e^\alpha - 1}{\alpha} \tag{E-5}$$

We find that

$$T_3 = (p-1)u + \left[-p\mu_1 + (e^{\mu_2} - 1)\phi_2\right]u^2 + \mathcal{O}(u^3)$$
(E-6)

We now consider the integrand of T_4 :

$$\partial_t \left(e^{-\frac{\tau \Box}{m_p^2}} \phi \right) \partial_t \left(e^{-\frac{(1-\tau)\Box}{m_p^2}} \phi \right) \cong \lambda^2 u^2 \partial_u \left(1 - e^{\tau \mu_1} u \right) \partial_u \left(1 - e^{(1-\tau)\mu_1} u \right) = \lambda^2 u^2 e^{\mu_1} \quad (\text{E-7})$$

The $d\tau$ integral is trivial and gives

$$T_4 = u^2 \frac{\lambda^2}{m_p^2} e^{\mu_1}$$
 (E-8)

It is straightforward to sum up the various contributions to T_{00} . We find that

$$\rho_{\phi} = \frac{m_s^4}{2g_p^2} \left[\frac{p-1}{p+1} + \left[(-p+e^{\mu_2})\phi_2 + \frac{p}{2}(p-1) - p\ln p + p\frac{\lambda^2}{m_p^2} \right] u^2 + \cdots \right]$$
(E-9)

The fact that the coefficient of the $\mathcal{O}(u)$ term is zero verifies that $H_1 = 0$. We now solve the Friedmann equation

$$3H^2 = \frac{1}{M_p^2} \rho_\phi$$

noting that

$$H^2 = H_0^2 - 2H_0H_2u^2 + \mathcal{O}(u^3)$$

Matching the coefficients at order u^0 gives

$$H_0^2 = \frac{1}{6g_p^2} \frac{m_s^4}{M_p^2} \frac{p-1}{p+1}$$

as before. Matching the coefficients at order u^2 gives

$$-6H_0H_2 = \frac{m_s^4}{2g_p^2 M_p^2} \left[(e^{\mu_2} - p)\phi_2 + \frac{p}{2}(p-1) - p\ln p + p\frac{\lambda^2}{m_p^2} \right]$$
$$= \frac{m_s^4}{M_p^2} \frac{p}{2g_p^2} \left[\frac{\lambda^2}{m_p^2} - \ln p \right]$$
$$= \frac{m_s^4}{M_p^2} \frac{p}{2g_p^2} \left[\frac{-3H_0\lambda}{m_p^2} \right]$$
In the first equality we have used that fact that

$$\phi_2 e^{\mu_2} = p\phi_2 - \frac{p}{2}(p-1)$$

which follows from the second order Klein-Gordon equation and in the second equality we have used

$$\frac{\lambda^2}{m_p^2} + 3\frac{H_0\lambda}{m_p^2} = \mu_1 = \ln p$$

which follows from the first order Klein-Gordon equation. Finally we arise at the result for H_2 :

$$H_2 = \frac{m_s^4}{4g_p^2 M_p^2} \frac{p\lambda}{m_p^2} = \frac{1}{8g_s^2} \frac{p^3 \ln p}{p-1} \left(\frac{m_s}{M_p}\right)^2 \lambda$$
(E-10)

E-2 Incomplete Cylindrical Functions of the Sonine-Schaefli Form

The Bessel function can be represented by a contour integral of the Sonine-Schaefli form

$$J_{\nu}(z) = \frac{1}{2\pi i} \left(\frac{z}{2}\right)^{\nu} \int_{c-i\infty}^{c+i\infty} \omega^{-\nu-1} \exp\left[\omega - \frac{z^2}{4\omega}\right] d\omega$$
(E-11)

as long as $\operatorname{Re}(\nu) > -1$. In (E-11) c is an arbitrary positive constant. The incomplete cylindrical function, $S_{\nu}(r, s; z)$ generalizes (E-11) to arbitrary limits of integration

$$S_{\nu}(r,s;z) = \frac{1}{2\pi i} \left(\frac{z}{2}\right)^{\nu} \int_{r}^{s} \omega^{-\nu-1} \exp\left[\omega - \frac{z^{2}}{4\omega}\right] d\omega$$
 (E-12)

It follows that

$$\int_{r}^{s} \omega^{-\nu-1} \exp\left[-\omega - \frac{z^2}{4\omega}\right] d\omega = 2\pi i e^{i\nu\pi/2} \left(\frac{z}{2}\right)^{-\nu} S(-r, -s; iz)$$
(E-13)

Taking the limits $r \to 0$ and $s \to +\infty$ this integral can be written in terms of Hankel functions of imaginary argument, $K_{\nu}(z)$, as

$$\int_0^\infty \omega^{-\nu-1} \exp\left[-\omega - \frac{z^2}{4\omega}\right] d\omega = 2\left(\frac{z}{2}\right)^{-\nu} K_\nu(z)$$
 (E-14)

The function $K_{\nu}(z)$ is real-valued for real z and is related to the usual Hankel function as $K_{\nu}(z) = (i\pi/2)e^{i\pi\nu/2}H_{\nu}^{(1)}(iz)$. Using the well known large-argument asymptotics of the Hankel functions one may show that

$$\int_0^\infty \omega^{-\nu-1} \exp\left[-\omega - \frac{z^2}{4\omega}\right] d\omega \cong \sqrt{\frac{2\pi}{z}} \left(\frac{z}{2}\right)^{-\nu} e^{-z}$$
(E-15)

for $z \gg 1$.

E-3 Comparison to Local Field Theory

In this appendix we perform a detailed comparison of our results for the action (7.1) to the theory

$$S = -\int d^4x \left[\frac{1}{2} \partial_\mu \psi \partial^\mu \psi + V(\psi) \right]$$
(E-16)

with potential

$$V(\psi) = \frac{g_s}{4} (\psi^2 - v^2)^2$$

$$\equiv \frac{gv^4}{4} - m_s^2 \psi^2 + \frac{g}{4} \psi^4$$
(E-17)

where we have defined $m_s^2 \equiv 2g_s v^2$. We are interested in obtaining inflation near the unstable maximum $\psi = 0$. The flatness of the potential is parameterized by the dimensionless slow roll parameters

$$\epsilon = \frac{M_p^2}{2} \left(\frac{V''(\psi)}{V(\psi)}\right)^2 = \frac{8\psi^2 M_p^2}{(\psi^2 - v^2)^2}$$
(E-18)

$$\eta = M_p^2 \frac{V''(\psi)}{V(\psi)} = \frac{4M_p^2}{(\psi^2 - v^2)^2} \left[3\psi^2 - v^2\right]$$
(E-19)

Unlike in the *p*-adic theory we do not need to distinguish between $\epsilon_{\mathcal{H}}$, $\eta_{\mathcal{H}}$ and ϵ_{V} , η_{V} (see equations 7.81-7.88) and hence we drop the subscripts on the slow roll parameters. For $\psi \ll v$ the ϵ parameter (E-18) is automatically small while

$$|\eta| \cong 8\left(\frac{M_p}{v}\right)^2$$

which is small compared to unity for $v \gg M_p$. The fact that the symmetry breaking scale is large compared to the Planck scale may be reason to doubt the validity of the field theory (E-16). However, for our purposes this is irrelevant. It is instructive to consider solving the equations of motion for this theory using the formalism of section 7.4. We begin by speculating solutions of the form

$$\psi = v e^{\lambda t} + \psi_2 v e^{2\lambda t} + \psi_3 v e^{3\lambda t} + \mathcal{O}(e^{3\lambda t})$$
(E-20)

$$H = H_0 - H_1 e^{\lambda t} - H_2 e^{2\lambda t} + \mathcal{O}(e^{3\lambda t})$$
 (E-21)

and solving order by order in $e^{\lambda t}$. The ansatz (E-20,E-21) is analogous to (7.11,7.12) since in both cases the (classical) field spends an infinite amount of time at the unstable maximum, driving a past-eternal de Sitter phase, before rolling towards the true minimum of the potential. We suppose that $e^{\lambda t} \ll 1$ so that $\psi \ll v$ initially.

The Klein-Gordon equation is

$$\frac{\ddot{\psi}}{v}+3H\frac{\dot{\psi}}{v}=2m_{s}^{2}\frac{\psi}{v}+2m_{s}^{2}\left(\frac{\psi}{v}\right)^{3}$$

Plugging in the ansatz (E-20) we obtain

$$\lambda^2 + 3H_0\lambda = 2m_s^2 \tag{E-22}$$

at first order in $e^{\lambda t}$. This result is identical to (7.27). At second and third order respectively we find

$$\psi_2 = 0$$

$$\psi_3 = \frac{2m_s^2 + 3H_2\lambda}{9\lambda^2 - 2m_s^2}$$

The Friedmann equation is

$$3H^{2} = \frac{1}{M_{p}^{2}} \left[\frac{gv^{4}}{4} - m_{s}^{2}v^{2} \left(\frac{\psi}{v}\right)^{2} + \frac{gv^{4}}{4} \left(\frac{\psi}{v}\right)^{4} + \frac{v^{2}}{2} \left(\frac{\dot{\psi}}{v}\right)^{2} \right]$$

At zeroth order in $e^{\lambda t}$ we obtain the familiar result

$$H_0 = \frac{g^{1/2}}{2\sqrt{3}} \frac{v^2}{M_p} \tag{E-23}$$

At first and second order we find that

$$H_1 = 0 \tag{E-24}$$

$$H_2 = \frac{2}{\sqrt{3g}} \frac{2m_s^2 - \lambda^2}{M_p} = \frac{1}{4} \left(\frac{v}{M_p}\right)^2 \lambda$$
 (E-25)

In writing the second equality in (E-25) we have used $\lambda^2 + 3H_0\lambda = 2m_s^2$.

Though a complete treatment of inhomogeneities including metric perturbations and nonzero slow roll parameters is straightforward in this context we choose to analyze this theory using the same approximations as we used in section 7.5 in order to make more explicit the comparison between the two theories. The perturbed Klein-Gordon equation (neglecting metric fluctuations and in the limit $e^{\lambda t} \rightarrow 0$) is

$$\left(\partial_t^2 + 3H_0\partial_t - \frac{\partial_i\partial^i}{a^2} - 2m_s^2\right)\delta\psi = 0$$

where $a = e^{H_0 t}$. The large scale solution is

$$\left|\delta\psi_{k}\right| \cong \frac{H}{\sqrt{2k^{3}}} \left(\frac{k}{aH_{0}}\right)^{3/2-k}$$

where

$$\nu = \sqrt{\frac{9}{4} + \frac{2m_s^2}{H_0^2}}$$

Near scale-invariance of the spectrum requires $m_s \ll H_0$. In this limit we have

$$n_s - 1 \cong -\frac{4}{3} \frac{m_s^2}{H_0^2} = -8 \left(\frac{M_p}{v}\right)^2$$
 (E-26)

which is identical to (7.68). This result also reproduces the full calculation incorporating metric perturbations: $n_s - 1 \approx 2\eta - 6\epsilon \approx -8M_p^2/v^2$.

We can use (E-26) to write the dimensionless quantities H_0/m_s , H_2/m_s and λ/m_s in terms of $|n_s - 1|$. The solution of (E-22) can be written as

$$\frac{\lambda}{m_s} \cong \sqrt{\frac{|n_s - 1|}{3}}$$

which is identical to (7.71). For this solution we of course have $3\lambda H_0 \gg \lambda^2$ so that the evolution is friction dominated. It is also straightforward to show that

$$\frac{H_0}{m_s} = \frac{2}{\sqrt{3|n_s - 1|}} \\ H_2 = H_0$$

We see that the solutions H, χ for the theory (E-16) are identical to those of the theory (7.1) up to order $e^{\lambda t}$. At order $e^{2\lambda t}$ and higher, however, the dynamics of the two theories differs.

From (E-18) one can check that $\epsilon \cong 1$ at $\psi \lesssim v$ so that it is a good approximation to suppose that inflation ends at $e^{\lambda t} \cong 1$. It is straightforward to impose the COBE normalization $V/(\epsilon M_p^4) = 6 \times 10^{-7}$ for this model (which will impose $g_s \ll |n_s - 1|^2$), however, this is unnecessary for our purposes. It is clear that the inflationary dynamics and predictions predictions of the theory (E-16) are identical to those of the theory (7.1).

Bibliography

- N. Barnaby and J. M. Cline, "Creating the universe from brane-antibrane annihilation," Phys. Rev. D 70, 023506 (2004) [arXiv:hep-th/0403223].
- [2] N. Barnaby, "Caustic formation in tachyon effective field theories," JHEP 0407, 025 (2004) [arXiv:hep-th/0406120].
- [3] N. Barnaby, C. P. Burgess and J. M. Cline, "Warped reheating in braneantibrane inflation," JCAP 0504, 007 (2005) [arXiv:hep-th/0412040].
- [4] N. Barnaby, A. Berndsen, J. M. Cline and H. Stoica, "Overproduction of cosmic superstrings," JHEP 0506, 075 (2005) [arXiv:hep-th/0412095].
- [5] N. Barnaby and J. M. Cline, "Nongaussian and nonscale-invariant perturbations from tachyonic preheating in hybrid inflation," Phys. Rev. D 73, 106012 (2006)
 [arXiv:astro-ph/0601481].
- [6] N. Barnaby and J. M. Cline, "Nongaussianity from tachyonic preheating in hybrid inflation," arXiv:astro-ph/0611750.
- [7] N. Barnaby, T. Biswas and J. M. Cline, "p-adic inflation," arXiv:hepth/0612230.
- [8] A. H. Guth, "The Inflationary Universe: A Possible Solution To The Horizon And Flatness Problems," Phys. Rev. D 23, 347 (1981).
- [9] A. Riotto, "Inflation and the theory of cosmological perturbations," arXiv:hepph/0210162.

 [10] R. H. Brandenberger, "Inflationary cosmology: Progress and problems," arXiv:hep-ph/9910410.

R. H. Brandenberger, "Lectures on the theory of cosmological perturbations," Lect. Notes Phys. **646**, 127 (2004) [arXiv:hep-th/0306071].

- [11] D. Langlois, "Inflation, quantum fluctuations and cosmological perturbations," arXiv:hep-th/0405053.
- [12] A. Vilenkin and L. H. Ford, "Gravitational Effects Upon Cosmological Phase Transitions," Phys. Rev. D 26, 1231 (1982).
- [13] J. C. R. Magueijo, "Local gauge invariant cosmological perturbations," Phys. Rev. D 46, 3360 (1992).
- [14] L. Kofman, A. D. Linde and A. A. Starobinsky, "Towards the theory of reheating after inflation," Phys. Rev. D 56, 3258 (1997) [arXiv:hep-ph/9704452].
- [15] J. C. Mather *et al.*, "Measurement of the Cosmic Microwave Background spectrum by the COBE FIRAS instrument," Astrophys. J. **420**, 439 (1994).
- [16] D. N. Spergel et al. [WMAP Collaboration], "Wilkinson Microwave Anisotropy Probe (WMAP) three year results: Implications for cosmology," arXiv:astroph/0603449.
- [17] N. Bartolo, E. Komatsu, S. Matarrese and A. Riotto, "Non-Gaussianity from inflation: Theory and observations," Phys. Rept. 402, 103 (2004) [arXiv:astroph/0406398].
- [18] J. M. Maldacena, "Non-Gaussian features of primordial fluctuations in single field inflationary models," JHEP 0305, 013 (2003) [arXiv:astro-ph/0210603].
- [19] A. D. Linde, "Chaotic Inflation," Phys. Lett. B 129, 177 (1983).
- [20] S. B. Giddings, S. Kachru and J. Polchinski, "Hierarchies from fluxes in string compactifications," Phys. Rev. D 66, 106006 (2002) [arXiv:hep-th/0105097].

- [21] S. Kachru, R. Kallosh, A. Linde and S. P. Trivedi, "De Sitter vacua in string theory," Phys. Rev. D 68, 046005 (2003) [arXiv:hep-th/0301240].
- [22] J. M. Cline, "String cosmology," arXiv:hep-th/0612129.
- [23] J. M. Cline and J. Vinet, "Problems with time-varying extra dimensions or 'Cardassian expansion' as alternatives to dark energy," Phys. Rev. D 68, 025015 (2003) [arXiv:hep-ph/0211284].
- [24] S. Kachru, R. Kallosh, A. Linde, J. M. Maldacena, L. McAllister and S. P. Trivedi, "Towards inflation in string theory," JCAP 0310, 013 (2003) [arXiv:hep-th/0308055].
- [25] C. P. Burgess, J. M. Cline, H. Stoica and F. Quevedo, "Inflation in realistic D-brane models," JHEP 0409, 033 (2004) [arXiv:hep-th/0403119].
- [26] L. Kofman and P. Yi, "Reheating the universe after string theory inflation," Phys. Rev. D 72, 106001 (2005) [arXiv:hep-th/0507257].

A. R. Frey, A. Mazumdar and R. Myers, "Stringy effects during inflation and reheating," Phys. Rev. D 73, 026003 (2006) [arXiv:hep-th/0508139].

D. Chialva, G. Shiu and B. Underwood, "Warped reheating in multi-throat brane inflation," JHEP 0601, 014 (2006) [arXiv:hep-th/0508229].

H. Firouzjahi and S. H. Tye, "The shape of gravity in a warped deformed conifold," JHEP 0601, 136 (2006) [arXiv:hep-th/0512076].

X. Chen and S. H. Tye, "Heating in brane inflation and hidden dark matter," JCAP 0606, 011 (2006) [arXiv:hep-th/0602136].

P. Langfelder, "On tunnelling In two-throat warped reheating," JHEP 0606, 063 (2006) [arXiv:hep-th/0602296].

[27] C. P. Burgess, J. M. Cline, K. Dasgupta and H. Firouzjahi, "Uplifting and inflation with D3 branes," arXiv:hep-th/0610320.

- [28] C. P. Burgess, "Strings, branes and cosmology: What can we hope to learn?," arXiv:hep-th/0606020.
- [29] R. Brandenberger, private communication.
- [30] J. M. Cline and L. Hoi, "Inflationary Potential Reconstruction for a WMAP Running Power Spectrum," JCAP 0606, 007 (2006) [arXiv:astro-ph/0603403].
- [31] A. D. Linde, "Hybrid inflation," Phys. Rev. D 49, 748 (1994) [arXiv:astroph/9307002].
- [32] G. N. Felder, J. Garcia-Bellido, P. B. Greene, L. Kofman, A. D. Linde and I. Tkachev, "Dynamics of symmetry breaking and tachyonic preheating," Phys. Rev. Lett. 87, 011601 (2001) [arXiv:hep-ph/0012142].
- [33] G. N. Felder, L. Kofman and A. D. Linde, "Tachyonic instability and dynamics of spontaneous symmetry breaking," Phys. Rev. D 64, 123517 (2001) [arXiv:hep-th/0106179].
- [34] E. J. Copeland, S. Pascoli and A. Rajantie, "Dynamics of tachyonic preheating after hybrid inflation," Phys. Rev. D 65, 103517 (2002) [arXiv:hep-ph/0202031].
- [35] T. W. B. Kibble, "Topology Of Cosmic Domains And Strings," J. Phys. A 9, 1387 (1976).
 T. W. B. Kibble, "Some Implications Of A Cosmological Phase Transition," Phys. Rept. 67, 183 (1980).
- [36] J. Magueijo and R. H. Brandenberger, "Cosmic defects and cosmology," arXiv:astro-ph/0002030.
- [37] T. Garagounis and M. Hindmarsh, "Scaling in numerical simulations of domain walls," Phys. Rev. D 68, 103506 (2003) [arXiv:hep-ph/0212359].
- [38] J. C. R. Oliveira, C. J. A. Martins and P. P. Avelino, "The cosmological evolution of domain wall networks," Phys. Rev. D 71, 083509 (2005) [arXiv:hepph/0410356].

- [39] W. Taylor, "Lectures on D-branes, tachyon condensation, and string field theory," arXiv:hep-th/0301094.
- [40] A. Sen, "Tachyon dynamics in open string theory," Int. J. Mod. Phys. A 20, 5513 (2005) [arXiv:hep-th/0410103].
- [41] C. V. Johnson, "Etudes on D-branes," arXiv:hep-th/9812196.
- [42] M. R. Garousi, "Tachyon couplings on non-BPS D-branes and Dirac-Born-Infeld action," Nucl. Phys. B 584, 284 (2000) [arXiv:hep-th/0003122].
 E. A. Bergshoeff, M. de Roo, T. C. de Wit, E. Eyras and S. Panda, "T-duality and actions for non-BPS D-branes," JHEP 0005, 009 (2000) [arXiv:hep-th/0003221].
 - J. Kluson, "Proposal for non-BPS D-brane action," Phys. Rev. D 62, 126003 (2000) [arXiv:hep-th/0004106].
- [43] A. Sen, "Tachyon matter," JHEP 0207, 065 (2002) [arXiv:hep-th/0203265].
- [44] A. Sen, "Field theory of tachyon matter," Mod. Phys. Lett. A 17, 1797 (2002)
 [arXiv:hep-th/0204143].
- [45] N. Lambert, H. Liu and J. M. Maldacena, "Closed strings from decaying Dbranes," arXiv:hep-th/0303139.
- [46] D. Kutasov and V. Niarchos, "Tachyon effective actions in open string theory," Nucl. Phys. B 666, 56 (2003) [arXiv:hep-th/0304045].
 M. Smedback, "On effective actions for the bosonic tachyon," JHEP 0311, 067 (2003) [arXiv:hep-th/0310138].
- [47] A. Sen, "Rolling tachyon," JHEP **0204**, 048 (2002) [arXiv:hep-th/0203211].
- [48] A. Sen, "Time evolution in open string theory," JHEP 0210, 003 (2002)[arXiv:hep-th/0207105].

- [49] F. Larsen, A. Naqvi and S. Terashima, "Rolling tachyons and decaying branes," JHEP 0302, 039 (2003) [arXiv:hep-th/0212248].
- [50] A. Sen, "Dirac-Born-Infeld action on the tachyon kink and vortex," Phys. Rev. D 68, 066008 (2003) [arXiv:hep-th/0303057].
- [51] J. M. Cline and H. Firouzjahi, "Real-time D-brane condensation," Phys. Lett. B 564, 255 (2003) [arXiv:hep-th/0301101].
- [52] A. Sen, "Open and closed strings from unstable D-branes," Phys. Rev. D 68, 106003 (2003) [arXiv:hep-th/0305011].

A. Sen, "Open-closed duality at tree level," Phys. Rev. Lett. **91**, 181601 (2003) [arXiv:hep-th/0306137].

H. U. Yee and P. Yi, "Open / closed duality, unstable D-branes, and coarsegrained closed strings," Nucl. Phys. B 686, 31 (2004) [arXiv:hep-th/0402027].

[53] G. R. Dvali and S. H. H. Tye, "Brane inflation," Phys. Lett. B 450, 72 (1999)
 [arXiv:hep-ph/9812483].

S. H. S. Alexander, "Inflation from D- anti-D brane annihilation," Phys. Rev. D 65, 023507 (2002) [arXiv:hep-th/0105032].

A. Mazumdar, S. Panda and A. Perez-Lorenzana, "Assisted inflation via tachyon condensation," Nucl. Phys. B **614**, 101 (2001) [arXiv:hep-ph/0107058].

C. P. Burgess, M. Majumdar, D. Nolte, F. Quevedo, G. Rajesh and
R. J. Zhang, "The inflationary brane-antibrane universe," JHEP 0107, 047
(2001) [arXiv:hep-th/0105204].

G. R. Dvali, Q. Shafi and S. Solganik, "D-brane inflation," arXiv:hep-th/0105203.

B. s. Kyae and Q. Shafi, "Branes and inflationary cosmology," Phys. Lett. B
526, 379 (2002) [arXiv:hep-ph/0111101].

C. P. Burgess, P. Martineau, F. Quevedo, G. Rajesh and R. J. Zhang, "Brane antibrane inflation in orbifold and orientifold models," JHEP **0203**, 052 (2002) [arXiv:hep-th/0111025].

J. Garcia-Bellido, R. Rabadan and F. Zamora, "Inflationary scenarios from branes at angles," JHEP **0201**, 036 (2002) [arXiv:hep-th/0112147].

R. Blumenhagen, B. Kors, D. Lust and T. Ott, "Hybrid inflation in intersecting brane worlds," Nucl. Phys. B 641, 235 (2002) [arXiv:hep-th/0202124].

K. Dasgupta, C. Herdeiro, S. Hirano and R. Kallosh, "D3/D7 inflationary model and M-theory," Phys. Rev. D 65, 126002 (2002) [arXiv:hep-th/0203019].

- [54] N. Jones, H. Stoica and S. H. H. Tye, "Brane interaction as the origin of inflation," JHEP 0207, 051 (2002) [arXiv:hep-th/0203163].
- [55] J. P. Hsu, R. Kallosh and S. Prokushkin, "On brane inflation with volume stabilization," JCAP 0312, 009 (2003) [arXiv:hep-th/0311077].
 H. Firouzjahi and S. H. H. Tye, "Closer towards inflation in string theory," arXiv:hep-th/0312020.
 E. Halyo, "Inflation on fractional branes: D-brane inflation as D-term inflation," arXiv:hep-th/0312042.

E. Halyo, "D-brane inflation on conifolds," arXiv:hep-th/0402155.

- [56] S. Sarangi and S. H. H. Tye, "Cosmic string production towards the end of brane inflation," Phys. Lett. B 536, 185 (2002) [arXiv:hep-th/0204074].
- [57] N. T. Jones, H. Stoica and S. H. H. Tye, "The production, spectrum and evolution of cosmic strings in brane inflation," Phys. Lett. B 563, 6 (2003) [arXiv:hep-th/0303269].
- [58] M. Majumdar and A. Christine-Davis, "Cosmological creation of D-branes and anti-D-branes," JHEP 0203, 056 (2002) [arXiv:hep-th/0202148].

L. Pogosian, S. H. H. Tye, I. Wasserman and M. Wyman, "Observational constraints on cosmic string production during brane inflation," Phys. Rev. D 68, 023506 (2003) [Erratum-ibid. D 73, 089904 (2006)] [arXiv:hep-th/0304188].

G. Dvali and A. Vilenkin, "Formation and evolution of cosmic D-strings," arXiv:hep-th/0312007.

E. J. Copeland, R. C. Myers and J. Polchinski, "Cosmic F- and D-strings," arXiv:hep-th/0312067.

E. Halyo, "Cosmic D-term strings as wrapped D3 branes," JHEP 0403, 047
(2004) [arXiv:hep-th/0312268].

L. Leblond and S. H. H. Tye, "Stability of D1-strings inside a D3-brane," arXiv:hep-th/0402072.

T. Matsuda, "String production after angled brane inflation," arXiv:hep-ph/0403092.

- [59] J. M. Cline, H. Firouzjahi and P. Martineau, "Reheating from tachyon condensation," JHEP 0211, 041 (2002) [arXiv:hep-th/0207156].
- [60] S. M. Carroll and M. M. Guica, "Sidestepping the cosmological constant with football-shaped extra arXiv:hep-th/0302067.

I. Navarro, "Codimension two compactifications and the cosmological constant problem," JCAP **0309**, 004 (2003) [arXiv:hep-th/0302129].

Y. Aghababaie, C. P. Burgess, S. L. Parameswaran and F. Quevedo, "Towards a naturally small cosmological constant from branes in 6D supergravity," Nucl. Phys. B 680, 389 (2004) [arXiv:hep-th/0304256].

Y. Aghababaie *et al.*, "Warped brane worlds in six dimensional supergravity," JHEP **0309**, 037 (2003) [arXiv:hep-th/0308064].

P. Bostock, R. Gregory, I. Navarro and J. Santiago, "Einstein gravity on the codimension 2 brane?," arXiv:hep-th/0311074.

- [61] J. M. Cline, J. Descheneau, M. Giovannini and J. Vinet, "Cosmology of codimension-two braneworlds," JHEP 0306, 048 (2003) [arXiv:hepth/0304147].
- [62] L. Kofman and A. Linde, "Problems with tachyon inflation," JHEP 0207, 004 (2002) [arXiv:hep-th/0205121].
 G. Shiu, S. H. H. Tye and I. Wasserman, "Rolling tachyon in brane world cosmology from superstring field theory," Phys. Rev. D 67, 083517 (2003) [arXiv:hep-th/0207119].
- [63] A. Strominger, "Open string creation by S-branes," arXiv:hep-th/0209090.
 G. Dvali and A. Vilenkin, "Solitonic D-branes and brane annihilation," Phys. Rev. D 67, 046002 (2003) [arXiv:hep-th/0209217].
- [64] A. Sen, "SO(32) spinors of type I and other solitons on brane-antibrane pair," JHEP 9809, 023 (1998) [arXiv:hep-th/9808141].
 A. Sen, "Descent relations among bosonic D-branes," Int. J. Mod. Phys. A 14, 4061 (1999) [arXiv:hep-th/9902105].
- [65] D. Ghoshal and A. Sen, "Tachyon condensation and brane descent relations in p-adic string theory," Nucl. Phys. B 584, 300 (2000) [arXiv:hep-th/0003278].
- [66] P. J. E. Peebles and A. Vilenkin, "Quintessential inflation," Phys. Rev. D 59, 063505 (1999) [arXiv:astro-ph/9810509].
- [67] P. Brax, J. Mourad and D. A. Steer, "Tachyon kinks on non BPS D-branes," Phys. Lett. B 575, 115 (2003) [arXiv:hep-th/0304197].
 C. Kim, Y. Kim, O. K. Kwon and C. O. Lee, "Tachyon kinks on unstable Dp-branes," JHEP 0311, 034 (2003) [arXiv:hep-th/0305092].
 E. J. Copeland, P. M. Saffin and D. A. Steer, "Singular tachyon kinks from regular profiles," Phys. Rev. D 68, 065013 (2003) [arXiv:hep-th/0306294].
 P. Brax, J. Mourad and D. A. Steer, "On tachyon kinks from the DBI action," arXiv:hep-th/0310079.

- [68] G. N. Felder, L. Kofman and A. Starobinsky, "Caustics in tachyon matter and other Born-Infeld scalars," JHEP 0209, 026 (2002) [arXiv:hep-th/0208019].
- [69] G. N. Felder and L. Kofman, "Inhomogeneous fragmentation of the rolling tachyon," arXiv:hep-th/0403073.
- [70] S. Mukohyama, "Inhomogeneous tachyon decay, light-cone structure and Dbrane network problem in tachyon cosmology," Phys. Rev. D 66, 123512 (2002) [arXiv:hep-th/0208094].
- [71] A. Ishida and S. Uehara, "Rolling down to D-brane and tachyon matter," JHEP 0302, 050 (2003) [arXiv:hep-th/0301179].
- [72] O. K. Kwon and P. Yi, "String fluid, tachyon matter, and domain walls," JHEP 0309, 003 (2003) [arXiv:hep-th/0305229].
- [73] G. W. Gibbons, K. Hori and P. Yi, "String fluid from unstable D-branes," Nucl. Phys. B 596, 136 (2001) [arXiv:hep-th/0009061].
 G. Gibbons, K. Hashimoto and P. Yi, "Tachyon condensates, Carrollian contraction of Lorentz group, and fundamental strings," JHEP 0209, 061 (2002) [arXiv:hep-th/0209034].
- [74] L. Leblond and S. H. H. Tye, "Stability of D1-strings inside a D3-brane," arXiv:hep-th/0402072.
- [75] D. P. Jatkar, G. Mandal and S. R. Wadia, "Nielsen-Olesen vortices in noncommutative Abelian Higgs model," JHEP 0009, 018 (2000) [arXiv:hepth/0007078].

Y. Hikida, M. Nozaki and T. Takayanagi, "Tachyon condensation on fuzzy sphere and noncommutative solitons," Nucl. Phys. B **595**, 319 (2001) [arXiv:hep-th/0008023].

P. Kraus and F. Larsen, "Boundary string field theory of the DD-bar system,"Phys. Rev. D 63, 106004 (2001) [arXiv:hep-th/0012198].

T. Takayanagi, S. Terashima and T. Uesugi, "Brane-antibrane action from boundary string field theory," JHEP 0103, 019 (2001) [arXiv:hep-th/0012210].
N. T. Jones and S. H. H. Tye, "An improved brane anti-brane action from boundary superstring field theory and multi-vortex solutions," JHEP 0301, 012 (2003) [arXiv:hep-th/0211180].

- [76] K. Hashimoto and S. Nagaoka, "Realization of brane descent relations in effective theories," Phys. Rev. D 66, 026001 (2002) [arXiv:hep-th/0202079].
- [77] F. Epple and D. Lust, "Tachyon condensation for intersecting branes at small and large angles," arXiv:hep-th/0311182.
- [78] J. Polchinski, "String Theory. Vol. 2: Superstring Theory And Beyond," p. 150.
- [79] J. A. Minahan and B. Zwiebach, "Field theory models for tachyon and gauge field string dynamics," JHEP 0009, 029 (2000) [arXiv:hep-th/0008231].
 J. A. Minahan and B. Zwiebach, "Effective tachyon dynamics in superstring theory," JHEP 0103, 038 (2001) [arXiv:hep-th/0009246].
 J. A. Minahan and B. Zwiebach, "Gauge fields and fermions in tachyon effective field theories," JHEP 0102, 034 (2001) [arXiv:hep-th/0011226].
- [80] G. Gibbons, K. Hashimoto and P. Yi, "Tachyon condensates, Carrollian contraction of Lorentz group, and fundamental strings," JHEP 0209, 061 (2002) [arXiv:hep-th/0209034].
- [81] F. Leblond, R. C. Myers and D. J. Winters, "Consistency conditions for brane worlds in arbitrary dimensions," JHEP 0107, 031 (2001) [arXiv:hepth/0106140];

F. Leblond, R. C. Myers and D. J. Winters, "Brane world sum rules and the AdS soliton," arXiv:hep-th/0107034.

[82] C. P. Burgess, J. M. Cline, N. R. Constable and H. Firouzjahi, "Dynamical stability of six-dimensional warped brane-worlds," JHEP 0201, 014 (2002) [arXiv:hep-th/0112047]. [83] M. Fairbairn and M. H. Tytgat, "Inflation from a tachyon fluid?," arXiv:hepth/0204070.

S. Mukohyama, "Brane cosmology driven by the rolling tachyon," Phys. Rev.
D 66, 024009 (2002) [arXiv:hep-th/0204084].

A. Feinstein, "Power-law inflation from the rolling tachyon," arXiv:hep-th/0204140.

T. Padmanabhan, "Accelerated expansion of the universe driven by tachyonic matter," Phys. Rev. D 66, 021301 (2002) [arXiv:hep-th/0204150].

D. Choudhury, D. Ghoshal, D. P. Jatkar and S. Panda, "On the cosmological relevance of the tachyon," arXiv:hep-th/0204204.

X. z. Li, J. g. Hao and D. j. Liu, "Can quintessence be the rolling tachyon?," arXiv:hep-th/0204252.

H. B. Benaoum, "Accelerated universe from modified Chaplygin gas and tachyonic fluid," arXiv:hep-th/0205140.

M. Sami, "Implementing power law inflation with rolling tachyon on the brane," arXiv:hep-th/0205146.

M. Sami, P. Chingangbam and T. Qureshi, "Aspects of tachyonic inflation with exponential potential," arXiv:hep-th/0205179.

J. c. Hwang and H. Noh, "Cosmological perturbations in a generalized gravity including tachyonic condensation," arXiv:hep-th/0206100.

M. Sami, "Implementing power law inflation with rolling tachyon on the brane," Mod. Phys. Lett. A 18, 691 (2003) [arXiv:hep-th/0205146].

M. Sami, P. Chingangbam and T. Qureshi, "Aspects of tachyonic inflation with exponential potential," Phys. Rev. D 66, 043530 (2002) [arXiv:hep-th/0205179].

C. j. Kim, H. B. Kim, Y. b. Kim and O. K. Kwon, "Cosmology of rolling tachyon," arXiv:hep-th/0301142.

A. Das and A. DeBenedictis, "Inhomogeneous cosmologies with tachyonic dust as dark matter," arXiv:gr-qc/0304017.

V. Gorini, A. Y. Kamenshchik, U. Moschella and V. Pasquier, "Tachyons, scalar fields and cosmology," arXiv:hep-th/0311111.

B. C. Paul and M. Sami, "A note on inflation with tachyon rolling on the Gauss-Bonnet brane," arXiv:hep-th/0312081.

C. Kim, H. B. Kim, Y. Kim, O. K. Kwon and C. O. Lee, "Unstable D-branes and string cosmology," arXiv:hep-th/0404242.

 [84] G. W. Gibbons, "Thoughts on tachyon cosmology," Class. Quant. Grav. 20, S321 (2003) [arXiv:hep-th/0301117].

A. Sen, "Remarks on tachyon driven cosmology," arXiv:hep-th/0312153.

- [85] A. Frolov, L. Kofman and A. A. Starobinsky, "Prospects and problems of tachyon matter cosmology," arXiv:hep-th/0204187.
 G. Shiu and I. Wasserman, "Cosmological constraints on tachyon matter," arXiv:hep-th/0205003.
 L. Kofman and A. Linde, "Problems with tachyon inflation," JHEP 0207, 004 (2002) [arXiv:hep-th/0205121].
- [86] J. M. Cline, H. Firouzjahi and P. Martineau, "Reheating from tachyon condensation," JHEP 0211, 041 (2002) [arXiv:hep-th/0207156].
- [87] N. Moeller and B. Zwiebach, "Dynamics with infinitely many time derivatives and rolling tachyons," JHEP 0210, 034 (2002) [arXiv:hep-th/0207107].
- [88] A. Buchel, P. Langfelder and J. Walcher, "Does the tachyon matter?," Annals Phys. 302, 78 (2002) [arXiv:hep-th/0207235].
 A. Buchel and J. Walcher, "Comments on supergravity description of S-branes," JHEP 0305, 069 (2003) [arXiv:hep-th/0305055].

[89] S. Mukohyama, "Inhomogeneous tachyon decay, light-cone structure and Dbrane network problem in tachyon cosmology," Phys. Rev. D 66, 123512 (2002) [arXiv:hep-th/0208094].

P. Brax, J. Mourad and D. A. Steer, "Tachyon kinks on non BPS D-branes," Phys. Lett. B 575, 115 (2003) [arXiv:hep-th/0304197].

C. Kim, Y. Kim, O. K. Kwon and C. O. Lee, "Tachyon kinks on unstable Dp-branes," JHEP **0311**, 034 (2003) [arXiv:hep-th/0305092].

E. J. Copeland, P. M. Saffin and D. A. Steer, "Singular tachyon kinks from regular profiles," Phys. Rev. D 68, 065013 (2003) [arXiv:hep-th/0306294].

P. Brax, J. Mourad and D. A. Steer, "On tachyon kinks from the DBI action," arXiv:hep-th/0310079.

 [90] G. W. Gibbons, K. Hori and P. Yi, "String fluid from unstable D-branes," Nucl. Phys. B 596, 136 (2001) [arXiv:hep-th/0009061].

G. Gibbons, K. Hashimoto and P. Yi, "Tachyon condensates, Carrollian contraction of Lorentz group, and fundamental strings," JHEP **0209**, 061 (2002) [arXiv:hep-th/0209034].

O. K. Kwon and P. Yi, "String fluid, tachyon matter, and domain walls," JHEP 0309, 003 (2003) [arXiv:hep-th/0305229].

[91] D. Kutasov, M. Marino and G. W. Moore, "Some exact results on tachyon condensation in string field theory," JHEP 0010, 045 (2000) [arXiv:hepth/0009148].

D. Kutasov, M. Marino and G. W. Moore, "Remarks on tachyon condensation in superstring field theory," arXiv:hep-th/0010108.

[92] S. Sugimoto and S. Terashima, "Tachyon matter in boundary string field theory," JHEP 0207, 025 (2002) [arXiv:hep-th/0205085].

J. A. Minahan, "Rolling the tachyon in super BSFT," JHEP **0207**, 030 (2002) [arXiv:hep-th/0205098].

- [93] N. D. Lambert and I. Sachs, "On higher derivative terms in tachyon effective actions," JHEP 0106, 060 (2001) [arXiv:hep-th/0104218].
 N. D. Lambert and I. Sachs, "Tachyon dynamics and the effective action approximation," Phys. Rev. D 67, 026005 (2003) [arXiv:hep-th/0208217].
- [94] M. Laidlaw and M. Nishimura, "Comments on higher derivative terms for the tachyon action," arXiv:hep-th/0403269.
- [95] P. G. O. Freund and M. Olson, "Nonarchimedean strings," Phys. Lett. B 199, 186 (1987);

P. G. O. Freund and E. Witten, "Adelic string amplitudes," Phys. Lett. B 199, 191 (1987).

L. Brekke, P.G. Freud, M. Olson and E. Witten, "Nonarchimedean String Dynamics," Nucl. Phys. B **302**, 365 (1998).

- [96] A. A. Gerasimov and S. L. Shatashvili, "On exact tachyon potential in open string field theory," JHEP 0010, 034 (2000) [arXiv:hep-th/0009103].
- [97] J. A. Minahan, "Mode interactions of the tachyon condensate in p-adic string theory," JHEP 0103, 028 (2001) [arXiv:hep-th/0102071].
 N. Moeller and M. Schnabl, "Tachyon condensation in open-closed p-adic string theory," JHEP 0401, 011 (2004) [arXiv:hep-th/0304213].
- [98] V. S. Vladimirov and Y. I. Volovich, "On the nonlinear dynamical equation in the p-adic string theory," Theor. Math. Phys. 138, 297 (2004) [Teor. Mat. Fiz. 138, 355 (2004)] [arXiv:math-ph/0306018].
- [99] R. Courant and D. Hilbert, "Methods of mathematical physics", Vol. 1, Wiley-Interscience Publ., New York (1989).
- [100] N. N. Lebedev, "Special functions and their applications", Prentice-Hall, INC., Englewood Cliffs, N.J. (1965).

L. C. Andrews, "Special functions for engineers and applied mathematicians", Macmillan publishing company, New York (1985).

- [101] R. Courant and D. Hilbert, "Methods of mathematical physics", Vol. 2, Wiley-Interscience Publ., New York (1989).
- [102] D.V. Nanopoulos, K.A. Olive, M. Srednicki and K. Tamvakis, Phys. Lett. B123 (1983) 41; A.B. Goncharov and A.D. Linde, Phys. Lett. B139 (1984) 27; B. Gato, J. Leon, M. Quiros and M. Ramon-Medrano, Z. Phys. C22 (1984) 345; G. Gelmini, C. Kounnas and D.V. Nanopoulos, Nucl. Phys. B250 (1985) 177; L.G. Jensen and K.A. Olive, Nucl. Phys. B263 (1986) 731; P. Binetruy and M. K. Gaillard, Phys. Rev. D34 (1986) 3069; G.L. Cardoso and B.A. Ovrut, Phys. Lett. B298 (1993) 292-298 [hep-th/9210114]; R. Brustein and P. J. Steinhardt, Phys. Lett. B302, 196 (1993) [hep-th/9212049]; J.A. Adams, G.G. Ross and S. Sarkar, Phys. Lett. B391 (1997) 271-280 [hep-ph/9608336]; E. Halyo, Phys. Lett. B388 (1996) 241 [hep-ph/9606342]; A.D. Linde and A. Riotto, Phys. Rev. D56 (1997) 1841-1844 [hep-ph/9703209].
- [103] G. R. Dvali and S. H. H. Tye, Phys. Lett. B450 (1999) 72 [hep-ph/9812483].
- [104] C. P. Burgess, M. Majumdar, D. Nolte, F. Quevedo, G. Rajesh and R. J. Zhang, "The inflationary brane-antibrane universe," JHEP 0107 (2001) 047 [hepth/0105204].
- [105] S. H. Alexander, "Inflation from D anti-D brane annihilation," Phys. Rev. D65 (2002) 023507 [hep-th/0105032]; G. R. Dvali, Q. Shafi and S. Solganik, "D-brane inflation," [hep-th/0105203].
- [106] J. Garcia-Bellido, R. Rabadan and F. Zamora, "Inflationary scenarios from branes at angles," JHEP 0201 (2002) 036 [hep-th/0112147]; N. Jones, H. Stoica and S. H. Tye, "Brane interaction as the origin of inflation," JHEP 0207 (2002) 051 [hep-th/0203163]; M. Gomez-Reino and I. Zavala, "Recombination

of intersecting D-branes and cosmological inflation," JHEP **0209** (2002) 020 [hep-th/0207278].

- [107] A. Mazumdar, S. Panda and A. Pérez-Lorenzana, "Assisted Inflation via Tachyon Condensation," Nucl. Phys. B614 (2001) 101-116 [hep-ph/0107058];
 C. P. Burgess, P. Martineau, F. Quevedo, G. Rajesh and R. J. Zhang, "Brane antibrane inflation in orbifold and orientifold models," JHEP 0203 (2002) 052 [hep-th/0111025]; C. Herdeiro, S. Hirano and R. Kallosh, "String theory and hybrid inflation / acceleration," JHEP 0112 (2001) 027 [hep-th/0110271]; K. Dasgupta, C. Herdeiro, S. Hirano and R. Kallosh, "D3/D7 inflationary model and M-theory," Phys. Rev. D 65 (2002) 126002 [hep-th/0203019]; L. Pilo, A. Riotto and A. Zaffaroni, "Old inflation in string theory," JHEP 0407, 052 (2004) [hepth/0401004]; J. P. Hsu, R. Kallosh and S. Prokushkin, "On brane inflation with volume stabilization," JCAP 0312 (2003) 009 [hep-th/0311077]; F. Koyama, Y. Tachikawa and T. Watari, "Supergravity analysis of hybrid inflation model from D3-D7 system", [hep-th/0311191];
- S. Sethi, C. Vafa and E. Witten, "Constraints on low-dimensional string compactifications," Nucl. Phys. B 480 (1996) 213 [hep-th/9606122]; K. Dasgupta, G. Rajesh and S. Sethi, "M theory, orientifolds and G-flux," JHEP 9908 (1999) 023 [hep-th/9908088].
- [109] J.J. Blanco-Pillado, C.P. Burgess, J.M. Cline, C. Escoda, M. Gómez-Reino,
 R. Kallosh, A. Linde, and F. Quevedo, "Racetrack Inflation," JHEP 0411, 063
 (2004) [arXiv:hep-th/0406230].
- [110] A.R. Frey, M. Lippert and B. Williams, "The fall of stringy de Sitter", Phys. Rev. D68 (2003) 046008 [hep-th/0305018]; S. Shandera, B. Shlaer, H. Stoica and S. H. H. Tye, "Inter-brane interactions in compact spaces and brane inflation," JCAP 0402, 013 (2004) [arXiv:hep-th/0311207]. C. Escoda, M. Gomez-Reino and F. Quevedo, "Saltatory de Sitter string vacua," JHEP 0311 (2003) 065, [hep-th/0307160]; C. P. Burgess, R. Kallosh and F. Quevedo, "de Sit-

ter string vacua from supersymmetric D-terms," JHEP 0310 (2003) 056 [hep-th/0309187] A. Buchel and R. Roiban, "Inflation in warped geometries," Phys. Lett. B 590, 284 (2004) hep-th/0311154; H. Firouzjahi and S. H. H. Tye, "Closer towards inflation in string theory," Phys. Lett. B 584, 147 (2004) [hep-th/0312020]; A. Saltman and E. Silverstein, "The Scaling of the No Scale Potential and de Sitter Model Building," JHEP 0411, 066 (2004) [hep-th/0402135]; E. Halyo, "D-brane inflation on conifolds," hep-th/0402155; R. Kallosh and S. Prokushkin, "SuperCosmology," [hep-th/0403060]; M. Becker, G. Curio and A. Krause, Nucl. Phys. B693 (2004) 223-260 [hep-th/0403027]; M. Alishahiha, E. Silverstein and D. Tong, "DBI in the sky," Phys. Rev. D 70, 123505 (2004) [arXiv:hep-th/0404084]. J. P. Hsu and R. Kallosh, "Volume stabilization and the origin of the inflaton shift symmetry in string theory," JHEP 0404 (2004) 042 [hep-th/0402047]; O. DeWolfe, S. Kachru and H. Verlinde, "The giant inflaton," JHEP 0405 (2004) 017 [hep-th/0403123].

- [111] F. Quevedo, Class. Quant. Grav. 19 (2002) 5721, [hep-th/0210292]; A. Linde, "Prospects of inflation," [hep-th/0402051]; C.P. Burgess, "Inflatable string theory?," Pramana 63, 1269 (2004) [arXiv:hep-th/0408037].
- [112] N. Iizuka and S. P. Trivedi, "An inflationary model in string theory," Phys. Rev. D 70, 043519 (2004) hep-th/0403203. X. Chen, "Multi-throat Brane Inflation,"
 [hep-th/040084].
- [113] J. F. G. Cascales, M. P. Garcia del Moral, F. Quevedo and A. M. Uranga, "Realistic D-brane models on warped throats: Fluxes, hierarchies and moduli stabilization," hep-th/0312051.
- [114] R. Kallosh and A. Linde, "Landscape, the scale of SUSY breaking, and inflation," JHEP 0412, 004 (2004) [arXiv:hep-th/0411011].
- [115] Y. Himemoto and T. Tanaka, Phys. Rev. D67 (2003) 084014 [gr-qc/0212114];
 R. Allahverdi, A. Mazumdar and A. Perez-Lorenzana, Phys. Lett. B516 (2001)
 431-438 [hep-ph/0105125]; J.H. Brodie and D.A. Easson, JCAP 0312 (2003)

004 [hep-th/0301138]; Y. Takamizu and K. Maeda, "Collision of Domain Walls and Reheating of the Brane Universe," [hep-th/0406235]; E. Papantonopoulos and V. Zamarias, JHEP 0410 (2004) 051 [hep-th/0408227]; C.J. Chang and I.G. Moss, "Brane Inflation With Dark Reheating," [hep-ph/0411021].

- [116] L. Randall and R. Sundrum, "A large mass hierarchy from a small extra dimension," Phys. Rev. Lett. 83 (1999) 3370 [hep-ph9905221]; L. Randall and R. Sundrum, "An alternative to compactification," Phys. Rev. Lett. 83 (1999) 4690 [hep-th9906064].
- [117] K. Benakli, Phenomenology of low quantum gravity scale models, Phys. Rev. D 60, 104002 (1999) [hep-ph/9809582]; C. P. Burgess, L. E. Ibanez and F. Quevedo, Strings at the intermediate scale or is the Fermi scale dual to the Planck scale? Phys. Lett. B 447, 257 (1999) [hep-ph/9810535].
- [118] A. Frey, R. Myers and A. Mazumdar, in preparation.
- [119] O. De Wolfe and S.B. Giddings, "Scales and Hierarchies in Warped Compactifications and Brane Worlds," Phys. Rev. D67 (2003) 066008 [hep-th/0208123].
- [120] S. Dimopoulos, S. Kachru, N. Kaloper, A. E. Lawrence and E. Silverstein, "Generating small numbers by tunneling in multi-throat compactifications," Int. J. Mod. Phys. A 19, 2657 (2004) [arXiv:hep-th/0106128]; S. Dimopoulos, S. Kachru, N. Kaloper, A. E. Lawrence and E. Silverstein, "Small numbers from tunneling between brane throats," Phys. Rev. D 64, 121702 (2001) [arXiv:hepth/0104239].
- [121] W. D. Goldberger and M. B. Wise, "Bulk fields in the Randall-Sundrum compactification scenario," Phys. Rev. D 60, 107505 (1999) [hep-ph/9907218];
 H. Davoudiasl, J. L. Hewett and T. G. Rizzo, "Phenomenology of the Randall-Sundrum gauge hierarchy model," Phys. Rev. Lett. 84, 2080 (2000) [hepph/9909255].

- [122] L. McAllister and I. Mitra, "Relativistic D-Brane Scattering is Extremely Inelastic," [hep-th/0408085].
- [123] R. Iengo and J.G. Russo, JHEP 0211 (2002) 045 [hep-th/0210245]; R. Iengo and J.G. Russo, JHEP 0303 (2003) 030 [hep-th/0301109]; D. Chialva, R. Iengo and J.G. Russo, JHEP 0312 (2003) 014 [hep-th/0310283]; D. Chialva, R. Iengo and J.G. Russo, [hep-th/0410152].
- [124] J. Lykken, R. Myers and J. Wang, "Gravity in a Box," JHEP 0009 (2000)009 [hep-th/0006191].
- [125] L. F. Abbott, E. Farhi and M. B. Wise, "Particle Production In The New Inflationary Cosmology," Phys. Lett. B 117, 29 (1982); A. D. Linde, "Particle Physics And Inflationary Cosmology," Chur, Switzerland: Harwood (1990) 362 p. (Contemporary concepts in physics, 5).
- [126] E. Witten, "Dimensional Reduction Of Superstring Models," Phys. Lett. B
 155 (1985) 151; C. P. Burgess, A. Font and F. Quevedo, "Low-Energy Effective Action For The Superstring," Nucl. Phys. B 272 (1986) 661.
- [127] M. Graã, T.W. Grimm, H. Jockers and J. Louis, Nucl. Phys. B690 (2004) 21-61
 [hep-th/0312232]; T.W. Grimm and J. Louis, Nucl. Phys. B699 (2004) 387-426
 [hep-th/0403067].
- [128] S. Gukov, C. Vafa and E. Witten, "CFTs from Calabi-Yau Fourfolds," Nucl. Phys. B584, 69 (2000).
- [129] E. Cremmer, S. Ferrara, C. Kounnas and D.V. Nanonpoulos, "Naturally vanishing cosmological constant in N = 1 supergravity," Phys. Lett. B133, 61 (1983); J. Ellis, A.B. Lahanas, D.V. Nanopoulos and K. Tamvakis, "No-scale Supersymmetric Standard Model," Phys. Lett. B134, 429 (1984).
- [130] J. P. Derendinger, L. E. Ibanez and H. P. Nilles, "On The Low-Energy D = 4, N=1 Supergravity Theory Extracted From The D = 10, N=1 Superstring,"

Phys. Lett. B **155** (1985) 65; M. Dine, R. Rohm, N. Seiberg and E. Witten, "Gluino Condensation In Superstring Models," Phys. Lett. B **156** (1985) 55.

- [131] C.P. Burgess, J.-P. Derendinger, F. Quevedo and M. Quirós, "Gaugino Condensates and Chiral-Linear Duality: An Effective-Lagrangian Analysis", Phys. Lett. B 348 (1995) 428–442; "On Gaugino Condensation with Field-Dependent Gauge Couplings", Ann. Phys. 250 (1996) 193-233.
- [132] E. Cremmer, B. Julia, J. Scherk, S. Ferrara, L. Girardello and P. van Nieuwenhuizen, "Spontaneous Symmetry Breaking And Higgs Effect In Supergravity Without Cosmological Constant," Nucl. Phys. B 147, 105 (1979).
- [133] M. Kawasaki, K. Kohri and T. Moroi, "Hadronic decay of the gravitino in the early universe and its implications to inflation," arXiv:hep-ph/0410287.
- [134] N. Arkani-Hamed and S. Dimopoulos, "Supersymmetric unification without low energy supersymmetry and signatures for fine-tuning at the LHC," arXiv:hepth/0405159;

G. F. Giudice and A. Romanino, "Split supersymmetry," Nucl. Phys. B 699, 65 (2004) [arXiv:hep-ph/0406088].

- [135] T. Damour and A. Vilenkin, "Gravitational radiation from cosmic (super)strings: Bursts, stochastic background, and observational windows," arXiv:hep-th/0410222.
- [136] G. D. Starkman and D. B. Stojkovic, "How Frustrated Strings Would Pull the Black Holes from the Centers of Galaxies," Phys. Rev. D 63, 045008 (2001) [arXiv:astro-ph/0010563].
- [137] M. G. Jackson, N. T. Jones and J. Polchinski, "Collisions of cosmic F- and D-strings," arXiv:hep-th/0405229.
- [138] G. Gibbons, K. Hashimoto and P. Yi, "Tachyon condensates, Carrollian contraction of Lorentz group, and fundamental strings," JHEP 0209, 061 (2002) [arXiv:hep-th/0209034].

K. Hashimoto and S. Terashima, "Brane decay and death of open strings," JHEP 0406, 048 (2004) [arXiv:hep-th/0404237].

- [139] N. T. Jones and S. H. H. Tye, "An improved brane anti-brane action from boundary superstring field theory and multi-vortex solutions," JHEP 0301, 012 (2003) [arXiv:hep-th/0211180].
- [140] D. Kutasov, M. Marino and G. W. Moore, "Remarks on tachyon condensation in superstring field theory," arXiv:hep-th/0010108.

P. Kraus and F. Larsen, "Boundary string field theory of the DD-bar system," Phys. Rev. D 63, 106004 (2001) [arXiv:hep-th/0012198].

T. Takayanagi, S. Terashima and T. Uesugi, "Brane-antibrane action from boundary string field theory," JHEP 0103, 019 (2001) [arXiv:hep-th/0012210].

- [141] M. R. Garousi, "D-brane anti-D-brane effective action and brane interaction in open string channel," arXiv:hep-th/0411222.
- [142] A. Avgoustidis and E. P. S. Shellard, "Cosmic string evolution in higher dimensions," arXiv:hep-ph/0410349.
- [143] M. Sakellariadou, "A note on the evolution of cosmic string / superstring networks," arXiv:hep-th/0410234.
- [144] T. Damour and A. Vilenkin, "Gravitational radiation from cosmic (super)strings: Bursts, stochastic background, and observational windows," arXiv:hep-th/0410222;
- [145] C. J. A. Martins and E. P. S. Shellard, "Quantitative String Evolution," Phys. Rev. D 54, 2535 (1996) [arXiv:hep-ph/9602271].
 C. J. A. Martins and E. P. S. Shellard, "Extending the velocity-dependent one-scale string evolution model," Phys. Rev. D 65, 043514 (2002) [arXiv:hepph/0003298].

- [146] "Gravitational wave bursts from cusps and kinks on cosmic strings," Phys. Rev. D 64, 064008 (2001) [arXiv:gr-qc/0104026]; T. Damour and A. Vilenkin, "Gravitational wave bursts from cosmic strings," Phys. Rev. Lett. 85, 3761 (2000) [arXiv:gr-qc/0004075].
- [147] M. Majumdar and A. C. Davis, "D-brane anti-brane annihilation in an expanding universe," JHEP 0312, 012 (2003) [arXiv:hep-th/0304153].
- [148] T. Garagounis and M. Hindmarsh, "Scaling in numerical simulations of domain walls," Phys. Rev. D 68, 103506 (2003) [arXiv:hep-ph/0212359].
- [149] J. C. R. Oliveira, C. J. A. Martins and P. P. Avelino, "The cosmological evolution of domain wall networks," arXiv:hep-ph/0410356.
- [150] A. G. Smith and A. Vilenkin, "Numerical Simulation Of Cosmic String Evolution In Flat Space-Time," Phys. Rev. D 36, 990 (1987).
 M. Sakellariadou and A. Vilenkin, "Cosmic-String Evolution In Flat Space-Time," Phys. Rev. D 42, 349 (1990).
- [151] D. Stojkovic and K. Freese, "A black hole solution to the cosmological monopole problem," arXiv:hep-ph/0403248.
- [152] C. J. A. Martins, "Scaling laws for non-intercommuting cosmic string networks," Phys. Rev. D 70, 107302 (2004) [arXiv:hep-ph/0410326].
- [153] J. Polchinski, "Cosmic superstrings revisited," arXiv:hep-th/0410082.
- [154] R. H. Brandenberger, "Quantum Field Theory Methods And Inflationary Universe Models," Rev. Mod. Phys. 57, 1 (1985).
- [155] S. Kasuya and M. Kawasaki, "Can topological defects be formed during preheating?," Phys. Rev. D 56, 7597 (1997) [arXiv:hep-ph/9703354].
- [156] N. Bartolo, S. Matarrese and A. Riotto, "Non-Gaussianity from inflation," Phys. Rev. D 65, 103505 (2002) [arXiv:hep-ph/0112261].

- [157] V. Acquaviva, N. Bartolo, S. Matarrese and A. Riotto, "Second-order cosmological perturbations from inflation," Nucl. Phys. B 667, 119 (2003) [arXiv:astroph/0209156].
- [158] D. Seery and J. E. Lidsey, "Primordial non-gaussianities in single field inflation," JCAP 0506, 003 (2005) [arXiv:astro-ph/0503692].
- [159] D. Seery, J. E. Lidsey and M. S. Sloth, "The inflationary trispectrum," arXiv:astro-ph/0610210.

M. x. Huang and G. Shiu, "The inflationary trispectrum for models with large non-Gaussianities," arXiv:hep-th/0610235.

D. Seery and J. E. Lidsey, "Non-gaussianity from the inflationary trispectrum," arXiv:astro-ph/0611034.

C. T. Byrnes, M. Sasaki and D. Wands, "The primordial trispectrum from inflation," arXiv:astro-ph/0611075.

[160] N. Bartolo, S. Matarrese and A. Riotto, "Enhancement of non-Gaussianity after inflation," JHEP 0404, 006 (2004) [arXiv:astro-ph/0308088].

N. Bartolo, S. Matarrese and A. Riotto, "Evolution of second-order cosmological perturbations and non-Gaussianity," JCAP **0401**, 003 (2004) [arXiv:astroph/0309692].

N. Bartolo, S. Matarrese and A. Riotto, "Gauge-invariant temperature anisotropies and primordial non-Gaussianity," Phys. Rev. Lett. **93**, 231301 (2004) [arXiv:astro-ph/0407505].

G. I. Rigopoulos, E. P. S. Shellard and B. J. W. van Tent, "A simple route to non-Gaussianity in inflation," Phys. Rev. D 72, 083507 (2005) [arXiv:astro-ph/0410486].

S. Koh, S. P. Kim and D. J. Song, "Nonlinear evolutions and non-Gaussianity in generalized gravity," Phys. Rev. D **71**, 123511 (2005) [arXiv:astro-ph/0501401].

N. Bartolo, S. Matarrese and A. Riotto, "Signatures of Primordial Non-Gaussianity in the Large-Scale Structure of the Universe," arXiv:astro-ph/0501614.

D. H. Lyth and Y. Rodriguez, "Non-gaussianity from the second-order cosmological perturbation," Phys. Rev. D **71**, 123508 (2005) [arXiv:astro-ph/0502578].

D. H. Lyth and Y. Rodriguez, "The inflationary prediction for primordial non-gaussianity," arXiv:astro-ph/0504045.

L. Boubekeur and D. H. Lyth, "Detecting a small perturbation through its non-Gaussianity," Phys. Rev. D 73, 021301(R) (2006) [arXiv:astro-ph/0504046].

N. Bartolo, S. Matarrese and A. Riotto, "Non-Gaussianity of Large-Scale CMB Anisotropies beyond Perturbation Theory," JCAP **0508**, 010 (2005) [arXiv:astro-ph/0506410].

D. H. Lyth and I. Zaballa, "A Bound Concerning Primordial Non-Gaussianity," arXiv:astro-ph/0507608.

M. Liguori, F. K. Hansen, E. Komatsu, S. Matarrese and A. Riotto, "Testing Primordial Non-Gaussianity in CMB Anisotropies," arXiv:astro-ph/0509098.

M. P. Salem, "On the generation of density perturbations at the end of inflation," Phys. Rev. D 72, 123516 (2005) [arXiv:astro-ph/0511146].

E. W. Kolb, A. Riotto and A. Vallinotto, "Non-Gaussianity from broken symmetries," Phys. Rev. D 73, 023522 (2006) [arXiv:astro-ph/0511198].

I. Zaballa, Y. Rodriguez and D. H. Lyth, "Higher order contributions to the primordial non-gaussianity," arXiv:astro-ph/0603534.

D. Seery and J. E. Lidsey, "Non-Gaussian inflationary perturbations from the dS/CFT correspondence," JCAP 0606, 001 (2006) [arXiv:astro-ph/0604209].

L. Alabidi and D. Lyth, "Curvature perturbation from symmetry breaking the end of inflation," arXiv:astro-ph/0604569.

L. Alabidi, "Non-gaussianity for a two component hybrid model of inflation," JCAP 0610, 015 (2006) [arXiv:astro-ph/0604611].

T. Matsuda, "Generating the curvature perturbation with instant preheating," arXiv:hep-th/0610232.

- [161] X. Chen, M. x. Huang, S. Kachru and G. Shiu, "Observational signatures and non-Gaussianities of general single field inflation," arXiv:hep-th/0605045.
- [162] E. Silverstein and D. Tong, "Scalar speed limits and cosmology: Acceleration from D-cceleration," Phys. Rev. D 70, 103505 (2004) [arXiv:hep-th/0310221].
 M. Alishahiha, E. Silverstein and D. Tong, "DBI in the sky," Phys. Rev. D 70, 123505 (2004) [arXiv:hep-th/0404084].

S. E. Shandera and S. H. Tye, "Observing brane inflation," JCAP 0605, 007 (2006) [arXiv:hep-th/0601099].

S. Kecskemeti, J. Maiden, G. Shiu and B. Underwood, "DBI inflation in the tip region of a warped throat," JHEP **0609**, 076 (2006) [arXiv:hep-th/0605189].

J. E. Lidsey and D. Seery, "Primordial non-Gaussianity and gravitational waves: Observational tests of brane inflation in string theory," arXiv:astro-ph/0610398.

G. Shiu and B. Underwood, "Observing the geometry of warped compactification via cosmic inflation," arXiv:hep-th/0610151.

T. Matsuda, "Brane inflation without slow-roll," arXiv:astro-ph/0610402.

- [163] L. Leblond and S. Shandera, "Cosmology of the tachyon in brane inflation," arXiv:hep-th/0610321.
- [164] X. Chen, R. Easther and E. A. Lim, "Large Non-Gaussianities in Single Field Inflation," arXiv:astro-ph/0611645.
- [165] D. H. Lyth and D. Wands, "Generating the curvature perturbation without an inflaton," Phys. Lett. B 524, 5 (2002) [arXiv:hep-ph/0110002].

[166] D. H. Lyth, C. Ungarelli and D. Wands, "The primordial density perturbation in the curvaton scenario," Phys. Rev. D 67, 023503 (2003) [arXiv:astroph/0208055].

N. Bartolo, S. Matarrese and A. Riotto, "On non-Gaussianity in the curvaton scenario," Phys. Rev. D 69, 043503 (2004) [arXiv:hep-ph/0309033].

- [167] K. Enqvist and S. Nurmi, "Non-Gaussianity in Curvaton Models with Nearly Quadratic Potential," arXiv:astro-ph/0508573.
- [168] D. Seery and J. E. Lidsey, "Primordial non-gaussianities from multiple-field inflation," JCAP 0509, 011 (2005) [arXiv:astro-ph/0506056].

K. A. Malik, "Gauge-invariant perturbations at second order: Multiple scalar fields on large scales," JCAP **0511**, 005 (2005) [arXiv:astro-ph/0506532].

G. I. Rigopoulos, E. P. S. Shellard and B. J. W. van Tent, "Non-linear perturbations in multiple-field inflation," Phys. Rev. D **73**, 083521 (2006) [arXiv:astroph/0504508].

G. I. Rigopoulos, E. P. S. Shellard and B. J. W. van Tent, "Large non-Gaussianity in multiple-field inflation," arXiv:astro-ph/0506704.

C. Y. Sun and D. H. Zhang, "Non Gaussianity of general multiple-field inflationary models," arXiv:astro-ph/0510709.

L. E. Allen, S. Gupta and D. Wands, "Non-Gaussian perturbations from multi-field inflation," JCAP **0601**, 006 (2006) [arXiv:astro-ph/0509719].

G. I. Rigopoulos, E. P. S. Shellard and B. J. W. van Tent, "Quantitative bispectra from multifield inflation," arXiv:astro-ph/0511041.

F. Vernizzi and D. Wands, "Non-Gaussianities in two-field inflation," JCAP 0605, 019 (2006) [arXiv:astro-ph/0603799].

S. A. Kim and A. R. Liddle, "Nflation: Non-gaussianity in the horizon-crossing approximation," Phys. Rev. D 74, 063522 (2006) [arXiv:astro-ph/0608186].

T. Battefeld and R. Easther, "Non-gaussianities in multi-field inflation," arXiv:astro-ph/0610296.

K. A. Malik, "A not so short note on the Klein-Gordon equation at second order," arXiv:astro-ph/0610864.

[169] E. J. Copeland, A. R. Liddle, D. H. Lyth, E. D. Stewart and D. Wands, "False vacuum inflation with Einstein gravity," Phys. Rev. D 49, 6410 (1994) [arXiv:astro-ph/9401011].

J. Garcia-Bellido and D. Wands, "The spectrum of curvature perturbations from hybrid inflation," Phys. Rev. D 54, 7181 (1996) [arXiv:astro-ph/9606047].

- [170] D. Parkinson, S. Tsujikawa, B. A. Bassett and L. Amendola, "Testing for double inflation with WMAP," Phys. Rev. D 71, 063524 (2005) [arXiv:astroph/0409071].
- [171] F. Finelli and S. Khlebnikov, "Metric perturbations at reheating: The use of spherical symmetry," Phys. Rev. D 65, 043505 (2002) [arXiv:hep-ph/0107143].
- [172] K. Enqvist, A. Jokinen, A. Mazumdar, T. Multamaki and A. Vaihkonen, "Non-Gaussianity from Preheating," Phys. Rev. Lett. 94, 161301 (2005) [arXiv:astro-ph/0411394].
- [173] K. Enqvist, A. Jokinen, A. Mazumdar, T. Multamaki and A. Vaihkonen, "Nongaussianity from instant and tachyonic preheating," JCAP 0503, 010 (2005) [arXiv:hep-ph/0501076].
- [174] K. Enqvist, A. Jokinen, A. Mazumdar, T. Multamaki and A. Vaihkonen, "Cosmological constraints on string scale and coupling arising from tachyonic instability," JHEP 0508, 084 (2005) [arXiv:hep-th/0502185].
- [175] Q. G. Huang and K. Ke, "Non-Gaussianity in KKLMMT model," arXiv:hepth/0504137.

- [176] E. W. Kolb, A. Riotto and A. Vallinotto, "Non-Gaussianity from broken symmetries," arXiv:astro-ph/0511198.
- [177] A. Jokinen and A. Mazumdar, "Very Large Primordial Non-Gaussianity from multi-field: Application to Massless Preheating," arXiv:astro-ph/0512368.
- [178] H. Kodama and T. Hamazaki, "Evolution of cosmological perturbations in a stage dominated by an oscillatory scalar field," Prog. Theor. Phys. 96, 949 (1996) [arXiv:gr-qc/9608022].

Y. Nambu and A. Taruya, "Evolution of cosmological perturbation in reheating phase of the universe," Prog. Theor. Phys. **97**, 83 (1997) [arXiv:gr-qc/9609029].

T. Hamazaki and H. Kodama, "Evolution of cosmological perturbations during reheating," Prog. Theor. Phys. 96, 1123 (1996) [arXiv:gr-qc/9609036].
B. A. Bassett, D. I. Kaiser and R. Maartens, "General relativistic preheating after inflation," Phys. Lett. B 455, 84 (1999) [arXiv:hep-ph/9808404].

B. A. Bassett, F. Tamburini, D. I. Kaiser and R. Maartens, "Metric preheating and limitations of linearized gravity. II," Nucl. Phys. B 561, 188 (1999) [arXiv:hep-ph/9901319].

- [179] F. Finelli and R. H. Brandenberger, "Parametric amplification of gravitational fluctuations during reheating," Phys. Rev. Lett. 82, 1362 (1999) [arXiv:hepph/9809490].
- [180] A. R. Liddle, D. H. Lyth, K. A. Malik and D. Wands, "Super-horizon perturbations and preheating," Phys. Rev. D 61, 103509 (2000) [arXiv:hep-ph/9912473].
- [181] F. Finelli and R. H. Brandenberger, "Parametric amplification of metric fluctuations during reheating in two field models," Phys. Rev. D 62, 083502 (2000) [arXiv:hep-ph/0003172].
- [182] R. Kallosh and A. Linde, "P-term, D-term and F-term inflation," JCAP 0310, 008 (2003) [arXiv:hep-th/0306058].

- [183] K. Enqvist and A. Vaihkonen, "Non-Gaussian perturbations in hybrid inflation," JCAP 0409, 006 (2004) [arXiv:hep-ph/0405103].
- [184] T. Asaka, W. Buchmuller and L. Covi, "False vacuum decay after inflation," Phys. Lett. B 510, 271 (2001) [arXiv:hep-ph/0104037].
- [185] J. Garcia-Bellido, M. Garcia Perez and A. Gonzalez-Arroyo, "Symmetry breaking and false vacuum decay after hybrid inflation," Phys. Rev. D 67, 103501 (2003) [arXiv:hep-ph/0208228].
- [186] A. Arrizabalaga, J. Smit and A. Tranberg, "Tachyonic preheating using 2PI 1/N dynamics and the classical approximation," JHEP 0410 (2004) 017 [arXiv:hep-ph/0409177].

B. van Tent, J. Smit and A. Tranberg, "Electroweak-scale inflation, inflaton-Higgs mixing and the scalar spectral index," JCAP **0407** (2004) 003 [arXiv:hepph/0404128].

A. Tranberg and J. Smit, "Baryon asymmetry from electroweak tachyonic preheating," JHEP **0311** (2003) 016 [arXiv:hep-ph/0310342].

J. I. Skullerud, J. Smit and A. Tranberg, "W and Higgs particle distributions during electroweak tachyonic preheating," JHEP **0308** (2003) 045 [arXiv:hep-ph/0307094].

J. Smit and A. Tranberg, "Chern-Simons number asymmetry from CP violation at electroweak tachyonic preheating," JHEP **0212** (2002) 020 [arXiv:hepph/0211243].

- [187] D. H. Lyth and Y. Rodriguez, "Non-gaussianity from the second-order cosmological perturbation," Phys. Rev. D 71, 123508 (2005) [arXiv:astroph/0502578].
- [188] K. A. Malik, "Gauge-invariant perturbations at second order: Multiple scalar fields on large scales," arXiv:astro-ph/0506532.

- [189] K. A. Malik and D. Wands, "Evolution of second order cosmological perturbations," Class. Quant. Grav. 21, L65 (2004) [arXiv:astro-ph/0307055].
- [190] J. Garcia-Bellido and D. Wands, "Metric perturbations in two-field inflation," Phys. Rev. D 53, 5437 (1996) [arXiv:astro-ph/9511029].
- [191] D. H. Lyth, K. A. Malik and M. Sasaki, "A general proof of the conservation of the curvature perturbation," JCAP 0505, 004 (2005) [arXiv:astro-ph/0411220].
- [192] J. Garcia-Bellido and D. Wands, "Metric perturbations in two-field inflation," Phys. Rev. D 53, 5437 (1996) [arXiv:astro-ph/9511029].
 D. Langlois and F. Vernizzi, "Conserved non-linear quantities in cosmology," arXiv:astro-ph/0509078.
- [193] F. Vernizzi, "On the conservation of second-order cosmological perturbations in a scalar field dominated universe," Phys. Rev. D 71, 061301 (2005) [arXiv:astroph/0411463].
- [194] D. Langlois and F. Vernizzi, "Nonlinear perturbations of cosmological scalar fields," arXiv:astro-ph/0610064.
- [195] F. Bernardeau, T. Brunier and J. P. Uzan, "High order correlation functions for self interacting scalar field in de Sitter space," Phys. Rev. D 69, 063520 (2004) [arXiv:astro-ph/0311422].
- [196] J. H. Traschen, "Causal Cosmological Perturbations And Implications For The Sachs-Wolfe Effect," Phys. Rev. D 29, 1563 (1984).
 J. H. Traschen, "Constraints On Stress Energy Perturbations In General Relativity," Phys. Rev. D 31, 283 (1985).
 L. F. Abbott and J. H. Traschen, "Causality Constraints On Cosmological Perturbations," BRX-TH-174
- [197] I. R. Klebanov and M. J. Strassler, "Supergravity and a confining gauge theory: Duality cascades and χ SB-resolution of naked singularities," JHEP **0008**, 052 (2000) [arXiv:hep-th/0007191].
- [198] G. Calcagni, "Non-Gaussianity in braneworld and tachyon inflation," arXiv:astro-ph/0411773.

C. Y. Sun and D. H. Zhang, "The non-Gaussianity of racetrack inflation models," arXiv:astro-ph/0604298.

 [199] S. E. Shandera and S. H. Tye, "Observing brane inflation," JCAP 0605, 007 (2006) [arXiv:hep-th/0601099].

S. H. Tye, "Brane inflation: String theory viewed from the cosmos," arXiv:hep-th/0610221.

- [200] P. Binetruy and G. R. Dvali, "D-term inflation," Phys. Lett. B 388, 241 (1996)
 [arXiv:hep-ph/9606342].
- [201] E. J. Copeland, A. R. Liddle, D. H. Lyth, E. D. Stewart and D. Wands, "False vacuum inflation with Einstein gravity," Phys. Rev. D 49, 6410 (1994)
 [arXiv:astro-ph/9401011].

G. Dvali, Q. Shafi and R. Schaefer, "Large scale structure and supersymmetric inflation without fine tuning," Phys. Rev. Lett. **73**, 1886 (1994) [arXiv:hep-ph/9406319].

M. Bastero-Gil and S. F. King, "F-term hybrid inflation in effective supergravity theories," Nucl. Phys. B 549, 391 (1999) [arXiv:hep-ph/9806477].

- [202] P. Brax, C. van de Bruck, A. C. Davis and S. C. Davis, "Coupling hybrid inflation to moduli," JCAP 0609, 012 (2006) [arXiv:hep-th/0606140].
 Ph. Brax, C. . v. de Bruck, A. C. Davis, S. C. Davis, R. Jeannerot and M. Postma, "Moduli corrections to D-term inflation," arXiv:hep-th/0610195.
- [203] D. H. Lyth and A. Riotto, "Comments on D-term inflation," Phys. Lett. B 412, 28 (1997) [arXiv:hep-ph/9707273].

J. R. Espinosa, A. Riotto and G. G. Ross, "D-term inflation in superstring theories," Nucl. Phys. B 531, 461 (1998) [arXiv:hep-ph/9804214].

C. F. Kolda and D. H. Lyth, "D-term inflation and M-theory," arXiv:hep-ph/9812234.

E. Halyo, "D-term inflation at the TeV scale and large internal dimensions," Phys. Lett. B 461, 109 (1999) [arXiv:hep-ph/9905244].

[204] J. Urrestilla, A. Achucarro and A. C. Davis, "D-term inflation without cosmic strings," Phys. Rev. Lett. 92, 251302 (2004) [arXiv:hep-th/0402032].

J. Rocher and M. Sakellariadou, "D-term inflation, cosmic strings, and consistency with cosmic microwave background measurement," Phys. Rev. Lett. **94**, 011303 (2005) [arXiv:hep-ph/0412143].

O. Seto and J. Yokoyama, "Hiding cosmic strings in supergravity D-term inflation," Phys. Rev. D 73, 023508 (2006) [arXiv:hep-ph/0508172].

C. M. Lin and J. McDonald, "Supergravity modification of D-term hybrid inflation: Solving the cosmic string and spectral index problems via a right-handed sneutrino," Phys. Rev. D 74, 063510 (2006) [arXiv:hep-ph/0604245].

Ph. Brax, C. van de Bruck, A. C. Davis and S. C. Davis, "Cosmic D-strings and vortons in supergravity," Phys. Lett. B 640, 7 (2006) [arXiv:hep-th/0606036].

J. Rocher and M. Sakellariadou, "D-term inflation in non-minimal supergravity," arXiv:hep-th/0607226.

- [205] R. A. Battye, B. Garbrecht and A. Moss, "Constraints on supersymmetric models of hybrid inflation," JCAP 0609, 007 (2006) [arXiv:astro-ph/0607339].
- [206] C. Kolda and J. March-Russell, "Supersymmetric D-term inflation, reheating and Affleck-Dine baryogenesis," Phys. Rev. D 60, 023504 (1999) [arXiv:hepph/9802358].

M. Bastero-Gil, S. F. King and J. Sanderson, "Preheating in supersymmetric hybrid inflation," Phys. Rev. D 60, 103517 (1999) [arXiv:hep-ph/9904315].

[207] J. A. Casas, G. B. Gelmini and A. Riotto, "F-term inflation in superstring theories," Phys. Lett. B 459, 91 (1999) [arXiv:hep-ph/9903492].

S. M. Harun-or-Rashid, T. Kobayashi and H. Shimabukuro, "F-term inflation in M-theory with five-branes," Phys. Lett. B **466**, 95 (1999) [arXiv:hep-ph/9908266].

J. McDonald, "F-term hybrid inflation, the eta-problem and extra dimensions," JHEP 0212, 029 (2002) [arXiv:hep-ph/0201016].

B. Kain, "Semi-realistic' F-term inflation model building in supergravity," arXiv:hep-ph/0608279.

- [208] R. Jeannerot and M. Postma, "Enlarging the parameter space of standard hybrid inflation," JCAP 0607, 012 (2006) [arXiv:hep-th/0604216].
- [209] K. Dasgupta, J. P. Hsu, R. Kallosh, A. Linde and M. Zagermann, "D3/D7 brane inflation and semilocal strings," JHEP 0408, 030 (2004) [arXiv:hepth/0405247].

P. Chen, K. Dasgupta, K. Narayan, M. Shmakova and M. Zagermann, "Brane inflation, solitons and cosmological solutions: I," JHEP **0509**, 009 (2005) [arXiv:hep-th/0501185].

- [210] K. Dasgupta, C. Herdeiro, S. Hirano and R. Kallosh, "D3/D7 inflationary model and M-theory," Phys. Rev. D 65, 126002 (2002) [arXiv:hep-th/0203019].
- [211] N. Barnaby and J. Cline, in preparation.
- [212] D. H. Lyth and E. D. Stewart, "More varieties of hybrid inflation," Phys. Rev. D 54, 7186 (1996) [arXiv:hep-ph/9606412].
- [213] J. J. Blanco-Pillado *et al.*, "Racetrack inflation," JHEP **0411**, 063 (2004)
 [arXiv:hep-th/0406230].

J. P. Conlon and F. Quevedo, "Kaehler moduli inflation," JHEP 0601, 146 (2006) [arXiv:hep-th/0509012].

J. J. Blanco-Pillado *et al.*, "Inflating in a better racetrack," JHEP **0609**, 002 (2006) [arXiv:hep-th/0603129].

J. R. Bond, L. Kofman, S. Prokushkin and P. M. Vaudrevange, "Roulette Inflation with Kahler Moduli and their Axions," arXiv:hep-th/0612197.

[214] M. Fairbairn and M. H. G. Tytgat, "Inflation from a tachyon fluid?," Phys. Lett. B 546, 1 (2002) [arXiv:hep-th/0204070].

A. Feinstein, "Power-law inflation from the rolling tachyon," Phys. Rev. D 66, 063511 (2002) [arXiv:hep-th/0204140].

M. Sami, "Implementing power law inflation with rolling tachyon on the brane," Mod. Phys. Lett. A 18, 691 (2003) [arXiv:hep-th/0205146].

M. Majumdar and A. C. Davis, "Inflation from tachyon condensation, large N effects," Phys. Rev. D 69, 103504 (2004) [arXiv:hep-th/0304226].

D. A. Steer and F. Vernizzi, "Tachyon inflation: Tests and comparison with single scalar field inflation," Phys. Rev. D 70, 043527 (2004) [arXiv:hep-th/0310139].

J. Raeymaekers, "Tachyonic inflation in a warped string background," JHEP 0410, 057 (2004) [arXiv:hep-th/0406195].

D. Cremades, F. Quevedo and A. Sinha, "Warped tachyonic inflation in type IIB flux compactifications and the open-string completeness conjecture," JHEP **0510**, 106 (2005) [arXiv:hep-th/0505252].

D. Cremades, "Electric tachyon inflation," Fortsch. Phys. 54, 357 (2006) [arXiv:hep-th/0512294].

L. Leblond and S. Shandera, "Cosmology of the tachyon in brane inflation," arXiv:hep-th/0610321.

[215] D. Kutasov, M. Marino and G. W. Moore, "Some exact results on tachyon condensation in string field theory," JHEP 0010, 045 (2000) [arXiv:hepth/0009148].

- [216] L. Kofman and A. Linde, "Problems with tachyon inflation," JHEP 0207, 004
 (2002) [arXiv:hep-th/0205121].
- [217] R. P. Woodard, "Avoiding dark energy with 1/R modifications of gravity," arXiv:astro-ph/0601672.
- [218] T. Biswas, A. Mazumdar and W. Siegel, "Bouncing universes in string-inspired gravity," JCAP 0603, 009 (2006) [arXiv:hep-th/0508194];
 T. Biswas, R. Brandenberger, A. Mazumdar and W. Siegel, "Non-perturbative
 - gravity, Hagedorn bounce and CMB," arXiv:hep-th/0610274.
- [219] J. Khoury, "Fading Gravity and Self-Inflation," arXiv:hep-th/0612052.
- [220] D. Ghoshal, "On the relation between p-adic and ordinary strings," arXiv:hepth/0606082.
- [221] E. Coletti, I. Sigalov and W. Taylor, "Taming the tachyon in cubic string field theory," JHEP 0508, 104 (2005) [arXiv:hep-th/0505031].
- [222] I. Y. Aref'eva, L. V. Joukovskaya and A. S. Koshelev, "Time evolution in superstring field theory on non-BPS brane. I: Rolling tachyon and energy-momentum conservation," JHEP 0309, 012 (2003) [arXiv:hep-th/0301137].

M. Fujita and H. Hata, "Rolling tachyon solution in vacuum string field theory," Phys. Rev. D **70**, 086010 (2004) [arXiv:hep-th/0403031].

T. G. Erler, "Level truncation and rolling the tachyon in the lightcone basis for open string field theory," arXiv:hep-th/0409179.

V. Forini, G. Grignani and G. Nardelli, "A new rolling tachyon solution of cubic string field theory," arXiv:hep-th/0502151.

V. Forini, G. Grignani and G. Nardelli, "A solution to the 4-tachyon off-shell amplitude in cubic string field theory," JHEP **0604**, 053 (2006) [arXiv:hep-th/0603206].

- [223] M. R. Douglas and S. H. Shenker, "Strings In Less Than One-Dimension," Nucl. Phys. B 335, 635 (1990); D. J. Gross and A. A. Migdal, "Nonperturbative Two-Dimensional Quantum Gravity," Phys. Rev. Lett. 64, 127 (1990); E. Brezin and V. A. Kazakov, "EXACTLY SOLVABLE FIELD THEORIES OF CLOSED STRINGS," Phys. Lett. B 236, 144 (1990).
- [224] T. Biswas, M. Grisaru and W. Siegel, "Linear Regge trajectories from worldsheet lattice parton field theory," Nucl. Phys. B 708, 317 (2005) [arXiv:hepth/0409089].
- [225] D. Ghoshal, "p-adic string theories provide lattice discretization to the ordinary string worldsheet," Phys. Rev. Lett. 97, 151601 (2006).
- [226] I. Y. Aref'eva, "Nonlocal string tachyon as a model for cosmological dark energy," AIP Conf. Proc. 826, 301 (2006) [arXiv:astro-ph/0410443].

I. Y. Aref'eva and L. V. Joukovskaya, "Time lumps in nonlocal stringy models and cosmological applications," JHEP **0510**, 087 (2005) [arXiv:hep-th/0504200].

I. Y. Aref'eva, A. S. Koshelev and S. Y. Vernov, "Crossing of the w = -1 barrier by D3-brane dark energy model," Phys. Rev. D 72, 064017 (2005) [arXiv:astro-ph/0507067].

G. Calcagni, "Cosmological tachyon from cubic string field theory," JHEP 0605, 012 (2006) [arXiv:hep-th/0512259].

I. Y. Aref'eva and A. S. Koshelev, "Cosmic acceleration and crossing of w = -1 barrier from cubic superstring field theory," arXiv:hep-th/0605085.

I. Y. Aref'eva and I. V. Volovich, "On the null energy condition and cosmology," arXiv:hep-th/0612098.

[227] H. t. Yang, "Stress tensors in *p*-adic string theory and truncated OSFT," JHEP 0211, 007 (2002) [arXiv:hep-th/0209197].

- [228] A. Sen, "Energy momentum tensor and marginal deformations in open string field theory," JHEP 0408, 034 (2004) [arXiv:hep-th/0403200].
- [229] Y. Volovich, "Numerical study of nonlinear equations with infinite number of derivatives," J. Phys. A 36, 8685 (2003) [arXiv:math-ph/0301028].
 V. S. Vladimirov and Y. I. Volovich, "On the nonlinear dynamical equation in the *p*-adic string theory," Theor. Math. Phys. 138, 297 (2004) [Teor. Mat. Fiz. 138, 355 (2004)] [arXiv:math-ph/0306018].
 V. S. Vladimirov, "On the equation of the *p*-adic open string for the scalar tachyon field," arXiv:math-ph/0507018.

D. V. Prokhorenko, "On Some Nonlinear Integral Equation in the (Super)String Theory," arXiv:math-ph/0611068.

- [230] M. M. Agrest, M. S. Maksimov, "Theory on Incomplete Cylindrical Functions and their Applications," Springer-Verlag Berlin Heidelberg New York, 1971.
- [231] D. S. Goldwirth, "Initial Conditions For New Inflation," Phys. Lett. B 243, 41 (1990).

D. S. Goldwirth and T. Piran, "Initial conditions for inflation," Phys. Rept. 214, 223 (1992).

R. Brandenberger, G. Geshnizjani and S. Watson, "On the initial conditions for brane inflation," Phys. Rev. D 67, 123510 (2003) [arXiv:hep-th/0302222].

- [232] A. D. Linde, "Monopoles as big as a universe," Phys. Lett. B 327, 208 (1994)
 [arXiv:astro-ph/9402031]; A. D. Linde and D. A. Linde, "Topological defects as seeds for eternal inflation," Phys. Rev. D 50, 2456 (1994) [arXiv:hep-th/9402115]. A. Vilenkin, "Topological inflation," Phys. Rev. Lett. 72, 3137 (1994) [arXiv:hep-th/9402085].
- [233] D. Kutasov, M. Marino and G. W. Moore, "Remarks on tachyon condensation in superstring field theory," arXiv:hep-th/0010108.

- [234] A. R. Cooray, "Polarized CMB: Reionization and primordial tensor modes," arXiv:astro-ph/0311059.
- [235] N. Barnaby, T. Biswas, J. Cline, work in progress.
- $[236]\,$ J. E. Lidsey, "Stretching the inflaton potential with kinetic energy," arXiv:hep-th/0703007.